

Signed contract document

This contract has been signed by the authorised signatory for the Education and Skills Funding Agency, acting on behalf of the Secretary of State, and has been digitally signed by all parties.

Document reference: 10032931_SBPD-1088_v1	
---	--

Signed by **Signed by Control** on 22 December 2023 as the provider's authorised signatory

User ID:



INTERTRAIN UK LTD. Carr Hill Doncaster South Yorkshire DN4 8DE

Company No: 04696164

December 2023

Contract for Services Funding for Skills Bootcamps DPS Second Competitions – ESFA-24998 ("the Contract").

Please find attached your Contract which details the requirements for Skills Bootcamps DPS Second Competitions provision.

The terms and conditions set out in the attached Contract for Services will apply to all funding received from the Secretary of State for Education for this provision.

Your total Funding for Skills Bootcamps DPS Second Competitions can be found immediately below in the Summary of Funding.

A copy of your completed Quality Tender Response is attached at Schedule 13: Contract Tender. This document forms part of the terms and conditions of the Contract.

If you have any questions about your Contract, please contact your Contract Manager.

SUMMARY OF FUNDING

Organisation Name:	INTERTRAIN UK LTD.				UKPRN:	10032931
Master Contract Number:	ESFA-24998	Start Da	ate:	01/02/2024	End Date:	31/01/2026
		Allocation				
		Contract Ref		t per Learner fully-funded)	Number of Learners	Total Funding
Skills Bootcamps						£302,225.00
of which Pathway to Accele	rated Apprenticeships - OLE	SBD-1198-452				£36,960.00
of which Pathway to Accele	rated Apprenticeship - Track	SBD-1198-453				£122,980.00
of which Pathway to Acceler	rated Apprenticeship - Signalling	SBD-1198-454				£142,285.00
		1	Total	Funding for th	nis contract:	£302,225.00
	Commence of Fo					

Summary of Funding – Page 1

Date: Friday, December 15, 2023 9:07 AM



DfE Skills Bootcamps Dynamic Purchasing System Skills Bootcamps Call Off Agreement:

DPS Second Competitions



Agreement Type	Contract for Services
Funding Period	1 st September 2023 to 31 st August 2024
Between	The Secretary of State for Education
And	INTERTRAIN UK LTD.
Funding for	Skills Bootcamps DPS Second Competitions
Master Contract Number	ESFA-24998

ACCEPTANCE BY THE CONTRACTOR

BY ACCEPTING THIS CONTRACT VIA THE MANAGE YOUR EDUCATION & SKILLS FUNDING SERVICE THE PERSON TAKING THIS ACTION ON BEHALF OF THE CONTRACTOR REPRESENTS AND WARRANTS THAT THE CONTRACTOR HAS READ AND UNDERSTOOD THIS CONTRACT, THE CONTRACTOR AGREES TO BE BOUND BY THIS CONTRACT AND THAT HE/SHE IS DULY AUTHORISED TO ACCEPT THIS CONTRACT AND LEGALLY BIND THE CONTRACTOR

SIGNED FOR AND ON BEHALF OF THE SECRETARY OF STATE FOR EDUCATION By **Control of Control of Control** This Contract is made on the date the Contract is digitally signed by the Contractor on the Manage Your Education & Skills Funding Service between:

INTERTRAIN UK LTD. Carr Hill Doncaster South Yorkshire DN4 8DE AND

THE SECRETARY OF STATE FOR EDUCATION DEPARTMENT FOR EDUCATION 20 GREAT SMITH STREET LONDON SW1P 3BT

Company No: 04696164

Hereinafter called the "Contractor"

Hereinafter called the "DFE"

each a "Party" and together the "Parties".

It is agreed that:

- 1. this contract, together with the attached schedules and annexes, collectively form the "Contract"; and
- 2. if there is a conflict between the provisions of the clauses of the Contract and the provisions of the schedules, contract clause 1.4 in the contract below shall apply.

Call off Contract Terms for Skills Bootcamp Services Dynamic Purchasing Agreement

1. DEFINITIONS AND INTERPRETATION	1
2. NOT USED	2
3. DUE DILIGENCE	2
4. REPRESENTATIONS AND WARRANTIES	2
5. CONTRACT PERIOD	4
6. IMPLEMENTATION	4
7. SERVICES	4
8. SERVICES	5
9. QUALITY STANDARDS	6
10. DISRUPTION	7
11. FINANCIAL DISTRESS	7
12. RECORDS, AUDIT ACCESS AND OPEN BOOK DATA	7
13. CHANGE	8
14. ILR DATA, CONTRACT CHARGES AND PAYMENT	9
15. PROMOTING TAX COMPLIANCE	12
16. SUPPLIER PERSONNEL	12
17. STAFF TRANSFER	13
18. SUPPLY CHAIN RIGHTS AND PROTECTION	14
19. NOT USED	17
20. NOT USED	17
21. NOT USED	17
22. INTELLECTUAL PROPERTY RIGHTS	17
23. SECURITY AND PROTECTION OF INFORMATION	20
24. PUBLICITY AND BRANDING	26
25. LIABILITY	26
26. INSURANCE	27
27. CUSTOMER REMEDIES FOR DEFAULT	28
28. OFSTED MONITORING VISITS	29
29. FORCE MAJEURE	31
30. CUSTOMER TERMINATION RIGHTS	32
31. SUPPLIER TERMINATION RIGHTS	34
32. TERMINATION BY EITHER PARTY	34
33. PARTIAL TERMINATION, SUSPENSION AND PARTIAL SUSPENSION	34
34. CONSEQUENCES OF EXPIRY OR TERMINATION	34
35. COMPLIANCE	36
36. ASSIGNMENT AND NOVATION	37
37. WAIVER AND CUMULATIVE REMEDIES	38
38. RELATIONSHIP OF THE PARTIES	38
39. PREVENTION OF FRAUD AND BRIBERY	38
40. SEVERANCE	39
41. MODERN SLAVERY	39
42. FURTHER ASSURANCES	40
43. ENTIRE AGREEMENT	40
44. THIRD PARTY RIGHTS	41
45. NOTICES	41
46. DISPUTE RESOLUTION	42
	42
47. GOVERNING LAW AND JURISDICTION CONTRACT SCHEDULE 1: DEFINITIONS	42
	43 60
CONTRACT SCHEDULE 2: SERVICES, PERFORMANCE MANAGEMENT AND CONTRACT MANANGEMENT	
CONTRACT SCHEDULE 3: CONTRACT CHARGES, PAYMENT AND INVOICING CONTRACT SCHEDULE 4: SECURITY	80 06
	96 104
CONTRACT SCHEDULE 5: STAFF TRANSFER	104

CONTRACT SCHEDULE 6: DISPUTE RESOLUTION PROCEDURE	131
CONTRACT SCHEDULE 7: PROCESSING PERSONAL DATA AND DATA SUBJECTS	134
CONTRACT SCHEDULE 8: JOINT CONTROLLER AGREEMENT	136
CONTRACT SCHEDULE 9: NOT USED	138
CONTRACT SCHEDULE 10: EXIT MANAGEMENT	139
CONTRACT SCHEDULE 11: VARIATION FORM	149
CONTRACT SCHEDULE 12: ALTERNATIVE AND/OR ADDITIONAL CLAUSES	150
CONTRACT SCHEDULE 13: CONTRACT TENDER	155
CONTRACT SCHEDULE 14: SUB-CONTRACTING	156
CONTRACT SCHEDULE 15: FINANCIAL DISTRESS	159

A. <u>BACKGROUND</u>

- 1. The Supplier is a participant on the Dynamic Purchasing System for Skills Bootcamp Services System and has entered into the Dynamic Purchasing Agreement, "the DPA".
- 2. The Customer has, through the DPA, selected the Supplier to provide services required as detailed in the Contract Order Form.
- 3. Both the Customer and the Supplier have agreed that these terms and conditions, together with the Contract Order Form, will govern their relationship.

B. <u>PRELIMINARIES</u>

1. DEFINITIONS AND INTERPRETATION

- 1.1. In this Contract, unless the context otherwise requires, capitalised expressions shall have the meanings set out in Contract Schedule 1 (Definitions) or the relevant Contract Schedule in which that capitalised expression appears.
- 1.2. If a capitalised expression does not have an interpretation in Contract Schedule 1 (Definitions) or relevant Contract Schedule, it shall have the meaning given to it in the DPA. If no meaning is given to it in the DPA, it shall, in the first instance, be interpreted in accordance with the common interpretation within the relevant market sector/industry where appropriate. Otherwise, it shall be interpreted in accordance with the dictionary meaning.
- 1.3. In this Contract, unless the context otherwise requires:
 - 1.3.1 the singular includes the plural and vice versa;
 - 1.3.2 reference to a gender includes the other gender and the neuter;
 - 1.3.3 references to a person include an individual, company, body corporate, corporation, unincorporated association, firm, partnership or other legal entity or Crown Body;
 - 1.3.4 a reference to any Law includes a reference to that Law as amended, extended, consolidated or re-enacted from time to time;
 - 1.3.5 the words "including", "other", "in particular", "for example" and similar words shall not limit the generality of the preceding words and shall be construed as if they were immediately followed by the words "without limitation";
 - 1.3.6 references to "writing" include typing, printing, lithography, photography, display on a screen, electronic and facsimile transmission and other modes of representing or reproducing words in a visible form, and expressions referring to writing shall be construed accordingly;
 - 1.3.7 references to "**representations**" shall be construed as references to present facts, to "**warranties**" as references to present and future facts and to "**undertakings**" as references to obligations under this Contract;
 - 1.3.8 references to "Clauses" and "Contract Schedules" are, unless otherwise provided, references to the clauses and schedules of this Contract and references in any Contract Schedule to parts, paragraphs, annexes and tables are, unless otherwise provided, references to the parts, paragraphs, annexes and tables of the Contract Schedule in which these references appear; and
 - 1.3.9 the headings in this Contract are for ease of reference only and shall not affect the interpretation or construction of this Contract.

- 1.4. In the event of and only to the extent of any conflict between the Contract Order Form, the Contract Terms and the provisions of the DPA, the conflict shall be resolved in accordance with the following order of precedence:
 - 1.4.1 The DPA;
 - 1.4.2 the Contract Order Form; and
 - 1.4.3 the Contract Terms.
- 1.5. Any permitted changes by the Customer to the Template Contract Terms and the Template Contract Order Form under Clause 5 (Call for Competition Procedure) of the DPA and DPA Schedule 5 (Call for Competition Procedure) prior to them becoming the Contract Terms and the Contract Order Form which comprise this Contract shall prevail over the DPA.

2. NOT USED

3. DUE DILIGENCE

- 3.1. The Supplier acknowledges that:
 - 3.1.1 the Customer has delivered or made available to the Supplier all of the information and documents that the Supplier considers necessary or relevant for the performance of its obligations under this Contract;
 - 3.1.2 it has made its own enquiries to satisfy itself as to the accuracy and adequacy of the Due Diligence Information;
 - 3.1.3 it has raised all relevant due diligence questions with the Customer before the Contract Commencement Date;
 - 3.1.4 it has undertaken all necessary due diligence and has entered into this Contract in reliance on its own due diligence alone; and
 - 3.1.5 it shall not be excused from the performance of any of its obligations under this Contract on the grounds of, nor shall the Supplier be entitled to recover any additional costs or Contract Charges, arising as a result of any:
 - (a) misinterpretation of the requirements of the Customer in the Contract Order Form or elsewhere in this Contract;
 - (b) failure by the Supplier to satisfy itself as to the accuracy and/or adequacy of the Due Diligence Information; and/or
 - (c) failure by the Supplier to undertake its own due diligence.

4. **REPRESENTATIONS AND WARRANTIES**

- 4.1. Each Party represents and warrants that:
 - 4.1.1 it has full capacity and authority to enter into and to perform this Contract;
 - 4.1.2 this Contract is executed by its duly authorised representative;
 - 4.1.3 there are no actions, suits or proceedings or regulatory investigations before any court or administrative body or arbitration tribunal pending or, to its knowledge, threatened against it (or, in the case of the Supplier, any of its Affiliates) that might affect its ability to perform its obligations under this Contract; and
 - 4.1.4 its obligations under this Contract constitute its legal, valid and binding obligations, enforceable in accordance with their respective terms subject to applicable (as the case may be for each Party) bankruptcy, reorganisation, insolvency, moratorium or similar Laws affecting creditors' rights generally and subject, as to enforceability, to equitable principles

of general application (regardless of whether enforcement is sought in a proceeding in equity or Law).

- 4.2. The Supplier represents and warrants that:
 - 4.2.1 it is validly incorporated, organised and subsisting in accordance with the Laws of its place of incorporation;
 - 4.2.2 it has all necessary consents (including, where its procedures so require, the consent of its Parent Company) and regulatory approvals to enter into this Contract;
 - 4.2.3 its execution, delivery and performance of its obligations under this Contract does not and will not constitute a breach of any Law or obligation applicable to it and does not and will not cause or result in a Default under any agreement by which it is bound;
 - 4.2.4 as at the Contract Commencement Date, all written statements and representations in any written submissions made by the Supplier as part of the procurement process, and any other documents submitted remain true and accurate except to the extent that such statements and representations have been superseded or varied by this Contract;
 - 4.2.5 if the Contract Charges payable under this Contract exceed or are likely to exceed five (5) million pounds, as at the Contract Commencement Date it has notified the Customer in writing of any Occasions of Tax Non-Compliance or any litigation that it is involved in connection with any Occasions of Tax Non Compliance;
 - 4.2.6 it has and shall continue to have all necessary rights in and to the Third Party IPR, the Supplier Background IPRs and any other materials made available by the Supplier (and/or any Sub-Contractor) to the Customer which are necessary for the performance of the Supplier's obligations under this Contract including the receipt of the Services by the Customer;
 - 4.2.7 it shall take all steps, in accordance with Good Industry Practice, to prevent the introduction, creation or propagation of any disruptive elements (including any virus, worms and/or trojans, spyware or other malware) into systems, data, software or the Customer's Confidential Information (held in electronic form) owned by or under the control of, or used by, the Customer;
 - 4.2.8 it is not subject to any contractual obligation, compliance with which is likely to have a material adverse effect on its ability to perform its obligations under this Contract;
 - 4.2.9 it is not affected by an Insolvency Event and no proceedings or other steps have been taken and not discharged (nor, to the best of its knowledge, are threatened) for the winding up of the Supplier or for its dissolution or for the appointment of a receiver, administrative receiver, liquidator, manager, administrator or similar officer in relation to any of the Supplier's assets or revenue; and
 - 4.2.10 for the Contract Period and for a period of twelve (12) Months after the termination or expiry of this Contract, the Supplier shall not employ or offer employment to any staff of the Customer which have been associated with the provision of the Services without Approval or the prior written consent of the Customer which shall not be unreasonably withheld.
- 4.3. Each of the representations and warranties set out in Clauses 4.1 and 4.2 shall be construed as a separate representation and warranty and shall not be limited or restricted by reference to, or inference from, the terms of any other representation, warranty or any undertaking in this Contract.
- 4.4. If at any time a Party becomes aware that a representation or warranty given by it under Clauses 4.1 and 4.2 has been breached, is untrue or is misleading, it shall immediately notify the other Party of the relevant occurrence in sufficient detail to enable the other Party to make an accurate assessment of the situation.
- 4.5. For the avoidance of doubt, the fact that any provision within this Contract is expressed as a warranty shall not preclude any right of termination the Customer may have in respect of breach of that provision by the Supplier which constitutes a material Default.

C. DURATION OF CONTRACT

5. CONTRACT PERIOD

- 5.1. This Contract shall take effect on the Contract Commencement Date and will expire on the Contract Expiry Date unless terminated earlier in accordance with the provisions of this Contract.
- 5.2. The term of this Contract shall be the Contract Period.
- 5.3. The Customer may extend the Contract Period on one or more occasions for a minimum of 6 months per extension subject to the maximum aggregate extension that is permissible being 24 months.
- 5.4. The contract contains extension provisions which enable the Authority to exercise discretion and extend for up to 24 months on 4 occasions up to a maximum total term (including the initial term) of 6 years to align with the DPS lifespan.

D. <u>CONTRACT PERFORMANCE</u>

6. IMPLEMENTATION

- 6.1. The Supplier shall provide to the Customer the Services as identified in Schedule 2 Part 1 (Services) to this Contract and in the Contract Order Form.
- 6.2. When providing the Services, as required by Clause 6.1, the Supplier shall comply with the Performance Measures detailed in Schedule 2 Part 2 (Performance Measures).

7. SERVICES

- 7.1. Provision of the Services
- 7.2. The Supplier acknowledges and agrees that the Customer relies on the skill and judgment of the Supplier in the provision of the Services and the performance of its obligations under this Contract.
- 7.3. The Supplier shall ensure that the Services:
 - 7.3.1 comply in all respects with the description of the Services in Contract Schedule 2 Part 1 (Services) or elsewhere in this Contract; and
 - 7.3.2 are supplied in accordance with the provisions of this Contract.
- 7.4. The Supplier shall perform its obligations under this Contract in accordance with:
 - 7.4.1 All applicable Law;
 - 7.4.2 Good Industry Practice;
 - 7.4.3 the Quality Standards;
 - 7.4.4 the Skills Bootcamp Guidance; and
 - 7.4.5 the Supplier's own established procedures and practices to the extent the same do not conflict with the requirements of Clauses 7.4.1 to 7.4.4.
- 7.5. The Supplier shall:
 - 7.5.1 at all times allocate sufficient resources with the appropriate technical expertise to deliver the Services in accordance with this Contract;
 - 7.5.2 obtain, and maintain throughout the Contract Period, all the consents, approvals, licences and permissions (statutory, regulatory contractual or otherwise) it may require which are necessary for the provision of the Services;
 - 7.5.3 ensure that any Services recommended or otherwise specified by the Supplier for use by the Customer in conjunction with the Services shall meet the requirements of the Customer;
 - 7.5.4 ensure that the Supplier Assets will be free of all encumbrances (except as agreed in writing with the Customer);
 - 7.5.5 minimise any disruption to the Sites and/or the Customer's operations when providing the Services;

- 7.5.6 ensure that any Documentation and training provided by the Supplier to the Customer are comprehensive, accurate and prepared in accordance with Good Industry Practice;
- 7.5.7 co-operate with any other Supplier's and provide reasonable information (including any Documentation), advice and assistance in connection with the Services to any other supplier and, on the Contract Expiry Date for any reason, to enable the timely transition of the supply of the Services (or any of them) to the Customer and/or to any Replacement Supplier;
- 7.5.8 assign to the Customer, or if it is unable to do so, shall (to the extent it is legally able to do so) hold on trust for the sole benefit of the Customer, all warranties and indemnities provided by third parties or any Sub-Contractor in respect of Services. Where any such warranties are held on trust, the Supplier shall enforce such warranties in accordance with any reasonable directions that the Customer may notify from time to time to the Supplier;
- 7.5.9 provide the Customer with such assistance as the Customer may reasonably require during the Contract Period in respect of the supply of the Services;
- 7.5.10 deliver the Services in a proportionate and efficient manner;
- 7.5.11 gather, collate and provide such information and co-operation as the Customer may reasonably request for the purposes of ascertaining the Supplier's compliance with its obligations under this Contract; and
- 7.5.12 provide the Customer with access to all information and data relating to the Services (including to Learner files) for the purposes for evaluation as detailed in the Service Requirements and co-operate fully and promptly with the requests made by the Customer including those relating to the retention of information and data (including the Learner files) and evaluation.
- 7.6. An obligation on the Supplier to do, or to refrain from doing, any act or thing shall include an obligation upon the Supplier to procure that all Sub-Contractors and Supplier Personnel also do, or refrain from doing, such act or thing.

Drafting Note: clause 7.7 will be marked "Not Used" where a Parent Company Guarantee is not required.

7.7. On or before the date of this Contract the Supplier shall procure the execution of a Guarantor by the Guarantor in favour of the Customer in the form annexed at Schedule 13 to the DPA.

8. SERVICES

General application

8.1. The Services to be delivered have been included in Schedule 2 Part 1 (Services).

Time of Delivery of the Services

8.2. The Supplier shall provide the Services on the date(s) specified in the Contract Order Form (or elsewhere in this Contract).

Location and Manner of Delivery of the Services

- 8.3. Except where otherwise provided in this Contract, the Supplier shall provide the Services to the Customer through the Supplier Personnel at the Sites.
- 8.4. The Customer may inspect and examine the manner in which the Supplier provides the Services at the Sites and if the Sites are not the Customer Premises, the Customer may carry out such inspection and examination on reasonable notice during normal business hours.

Undelivered Services

8.5. In the event that any of the Services are not delivered in accordance with Clauses 7 (Provision of the Services), 8.2 (Time of Delivery of the Services) and 8.3 and 8.4 (Location and Manner of Delivery of the Services) ("Undelivered Services"), the Customer, without prejudice to any other rights and remedies

of the Customer howsoever arising, shall be entitled to withhold payment of any applicable Contract Charges for the Services that were not so delivered until such time as the undelivered Services are delivered.

8.6. The Customer may, at its discretion and without prejudice to any other rights and remedies of the Customer howsoever arising, deem the failure to comply with Clauses 7 (Provision of the Services), 8.2 (Time of Delivery of the Services) and 8.3 and 8.4 (Location and Manner of Delivery of the Services) to be a material Default.

Obligation to Remedy of Default in the Supply of the Services

- 8.7. Subject to Clauses 22.9 and 22.20 (IPR Indemnity) and without prejudice to any other rights and remedies of the Customer howsoever arising (including under Clauses 8.6 (Undelivered Services) and 27 (Customer Remedies for Default)), the Supplier shall, where practicable:
 - 8.7.1 remedy any breach of its obligations in Clauses 8 and 9 within three (3) Working Days of becoming aware of the relevant Default or being notified of the Default by the Customer or within such other time period as may be agreed with the Customer (taking into account the nature of the breach that has occurred);
 - 8.7.2 meet all the costs of, and incidental to, the performance of such remedial work

Continuing Obligation to Provide the Services

- 8.8. The Supplier shall continue to perform all of its obligations under this Contract and shall not suspend the provision of the Services, notwithstanding:
 - 8.8.1 any withholding or deduction by the Customer of any Contract Charges or other sum due to the Supplier pursuant to the exercise of a right of the Customer to such withholding or deduction under this Contract;
 - 8.8.2 the existence of an unresolved Dispute; and/or
 - 8.8.3 any failure by the Customer to pay any Contract Charges, unless the Supplier is entitled to terminate this Contract under Clauses 31.1 to 31.4 (Termination on Customer Failure to Pay) for failure by the Customer to pay undisputed Contract Charges.

9. QUALITY STANDARDS

- 9.1. The Supplier shall at all times during the Contract Period comply with the Quality Standards and maintain, where applicable, accreditation with the relevant Quality Standards authorisation body.
- 9.2. Throughout the Contract Period, the Parties shall notify each other of any new or emergent Quality Standards which could affect the Supplier's provision, or the receipt by the Customer, of the Services. A proposed adoption of any such new or emergent Quality standard, or changes to existing Quality Standards (including any specified in the Contract Order Form), shall be made by the Supplier by adhering to the Variation procedure set out at Clauses 13.3 to 13.5.
- 9.3. Where a new or emergent Quality Standard is to be developed or introduced by the Customer, the Supplier shall be responsible for ensuring that the potential impact on the Supplier's provision, or the Customer's receipt of the Services is explained to the Customer (within a reasonable timeframe), prior to the implementation of the new or emergent Quality Standard.
- 9.4. Where Quality Standards conflict with each other or with Good Industry Practice, then the later Quality Standard or best practice shall be adopted by the Supplier. The proposal to use the later Quality Standard or best practice shall require Approval (and the written consent of the Customer where the relevant Standard or Standards is/are included in DPA Schedule 2 (Services, Performance Measures and Contract Performance) and shall be implemented within an agreed timescale.
- 9.5. Where a Quality Standard, policy or document is referred to by reference to a hyperlink, then if the hyperlink is changed or no longer provides access to the relevant standard, policy or document, the Supplier shall notify the Customer and provide access to an updated hyperlink.

10. DISRUPTION

- 10.1. The Supplier shall take reasonable care to ensure that in the performance of its obligations under this Contract it does not disrupt the operations of the Customer, its employees or any other contractor engaged by the Customer.
- 10.2. The Supplier shall immediately inform the Customer of any actual or potential industrial action, whether such action be by the Supplier Personnel or others, which affects or might affect the Supplier's ability at any time to perform its obligations under this Contract.
- 10.3. In the event of industrial action by the Supplier Personnel, the Supplier shall seek Approval to its proposals for the continuance of the supply of the Services in accordance with its obligations under this Contract.
- 10.4. If the Supplier's proposals referred to in Clause 10.3 are considered insufficient or unacceptable by the Customer acting reasonably then the Customer may terminate this Contract for material Default.

11. FINANCIAL DISTRESS

11.1. The Parties shall comply with the provisions of Contract Schedule 15 (Financial Distress) in relation to the assessment of the financial standing of the Supplier and the consequences of a change to that financial standing.

E. <u>CONTRACT GOVERNANCE</u>

12. RECORDS, AUDIT ACCESS AND OPEN BOOK DATA

- 12.1. The Supplier shall keep and maintain for seven (7) years after the Contract Expiry Date or Termination Date, whichever is the earlier, full and accurate records and accounts of the operation of this Contract including the Services provided under it, any Sub-Contracts and the amounts paid by the Customer.
- 12.2. The Supplier shall:
 - 12.2.1 keep the records and accounts referred to in Clause 12.1 in accordance with Good Industry Practice and Law; and
 - 12.2.2 afford any Auditor access to the records and accounts referred to in Clause 12.1 at the Supplier's premises and/or provide records and accounts (including copies of the Supplier's

published accounts) or copies of the same, as may be required by any of the Auditors from time to time during the Contract Period and the period specified in Clause 12.1, in order that the Auditor(s) may carry out an inspection to assess compliance by the Supplier and/or its Sub-Contractors of any of the Supplier's obligations under this Contract including in order to:

- (a) verify the accuracy of the Contract Charges and any other amounts payable by the Customer under this Contract (and proposed or actual variations to them in accordance with this Contract);
- (b) verify the costs of the Supplier (including the costs of all Sub Contractors and any third party Supplier's) in connection with the provision of the Services;
- (c) verify the Open Book Data;
- (d) verify the Supplier's and each Sub-Contractor's compliance with the applicable Law;
- (e) identify or investigate an actual or suspected Prohibited Act, impropriety or accounting mistakes or any breach or threatened breach of security and in these circumstances the Customer shall have no obligation to inform the Supplier of the purpose or objective of its investigations;
- (f) identify or investigate any circumstances which may impact upon the financial stability of the Supplier, and/or any Sub-Contractors or their ability to perform the Services;

- (g) obtain such information as is necessary to fulfil the Customer's obligations to supply information for parliamentary, ministerial, judicial or administrative purposes including the supply of information to the Comptroller and Auditor General;
- (h) review any books of account and the internal contract management accounts kept by the Supplier in connection with this Contract;
- (i) carry out the Customer's internal and statutory audits and to prepare, examine and/or certify the Customer's annual and interim reports and accounts;
- (j) enable the National Audit Office to carry out an examination pursuant to Section 6(1) of the National Audit Act 1983 of the economy, efficiency and effectiveness with which the Customer has used its resources;
- (k) verify the accuracy and completeness of any information delivered or required by this Contract;
- (I) review the Supplier's quality management systems (including any quality manuals and procedures);
- (m) review the Supplier's compliance with the Standards;
- (n) inspect the Customer Assets, including the Customer's IPRs, equipment and facilities, for the purposes of ensuring that the Customer Assets are secure and that any register of assets is up to date; and/or
- (0) review the integrity, confidentiality and security of the Customer Data.
- 12.3. The Customer shall use reasonable endeavours to ensure that the conduct of each audit does not unreasonably disrupt the Supplier or delay the provision of the Services save insofar as the Supplier accepts and acknowledges that control over the conduct of audits carried out by the Auditor(s) is outside of the control of the Customer.
- 12.4. Subject to the Supplier's rights in respect of Confidential Information, the Supplier shall on demand provide the Auditor(s) with all reasonable co-operation and assistance in:
 - 12.4.1 all reasonable information requested by the Customer within the scope of the audit;
 - 12.4.2 reasonable access to sites controlled by the Supplier and to any Supplier Equipment used in the provision of the Services; and
 - 12.4.3 access to the Supplier Personnel.
- 12.5. The Parties agree that they shall bear their own respective costs and expenses incurred in respect of compliance with their obligations under this Clause 12, unless the audit reveals a Default by the Supplier in which case the Supplier shall reimburse the Customer for the Customer's reasonable costs incurred in relation to the audit.

13. CHANGE

Legislative Change

- 13.1. The Supplier shall neither be relieved of its obligations under this Contract nor be entitled to an increase in the Contract Charges as the result of a:
 - 13.1.1 General Change in Law;
 - 13.1.2 Specific Change in Law where the effect of that Specific Change in Law on the Services is reasonably foreseeable at the Contract Commencement Date.
- 13.2. If a Specific Change in Law occurs or will occur during the Contract Period (other than as referred to in Clause 13.1.2, the Supplier shall:
 - 13.2.1 notify the Customer as soon as reasonably practicable of the likely effects of that change including:

- (a) whether, in their opinion a contract variation is required to the provision of the Services, the Contract Charges or this Contract; and
- (b) whether any relief from compliance with the Supplier's obligations is required; and
- 13.2.2 provide to the Customer with evidence:
 - (a) that the Supplier has minimised any increase in costs or maximised any reduction in costs, including in respect of the costs of its Sub-Contractors;
 - (b) as to how the Specific Change in Law has affected the cost of providing the Services; and
 - (C) demonstrating that any expenditure that has been avoided, has been taken into account in amending the Contract Charges.

Variation Process

- 13.3. Subject to the provisions of this Clause 13.3 and Call Off Schedule 3 (Call Off Contract Charges, Payment and Invoicing), either Party may request a variation to this Call Off Contract provided that such variation does not amount to a material change of this Call Off Contract within the meaning of the Regulations and the Law. Such a change once implemented is hereinafter called a "Variation"
- 13.4. A Party may request a Variation by completing, signing and sending the Variation Form to the other Party giving sufficient information for the receiving Party to assess the extent of the proposed Variation and any additional cost that may be incurred.
- 13.5. If the Parties agree the Variation, the Supplier shall implement such Variation and be bound by the same provisions so far as is applicable, as though such Variation was stated in this Call Off Contract.

F. PAYMENT, TAXATION AND VALUE FOR MONEY PROVISIONS

14. ILR DATA, CONTRACT CHARGES AND PAYMENT

The Submission of ILR Data

- 14.1. The Provider must register with UKRLP (UK Register of Learning Providers (ukrlp.co.uk)) and maintain contact details on an on-going basis.
- 14.2. The Supplier must submit promptly to the Customer an accurate Individualised Learning Record ("ILR") in relation to each individual Learner in accordance with this Contract and the ILR technical documents, guidance and requirements as amended and updated from time to time.
- 14.3. The Supplier can view the ILR technical documents, guidance and requirements by visiting the website https://guidance.submit-learner-data.service.gov.uk/. Suppliers should view the "Skills Bootcamps Funding and Performance Management Guidance", which includes the latest instructions on how to complete the ILR for Skills Bootcamp courses.
- 14.4. The Supplier must notify the Customer through the ILR of any new Learners within the timescale specified in the Skills Bootcamp Guidance.
- 14.5. ILR Data collected by the Supplier must be transmitted to the Customer through the Submit Learner Data service .
- 14.6. The Supplier shall ensure that information about Learners (including new Learners) is recorded on the ILR promptly (as defined in the Skills Bootcamp Guidance) and regularly reviewed and updated within a month when there has been a change.
- 14.7. Further to clause 14.6 the Supplier must also and in any event check the accuracy of the ILR Data submitted via the Submit Learner Data service and the Supplier must correct any errors immediately.
- 14.8. All submissions to the ILR must be supported by evidence including but not limited to the information specified in Annex 1 to Schedule 2 Part 3 (Contract Management), and the Skills Bootcamp Guidance.

- 14.9. The Customer may publish monitoring reports via View your education data. If these monitoring reports are published, the Supplier must access the monitoring reports every month and amend any errors for the next submission of the ILR by taking the actions specified in the report user guide. Failure to amend the ILR data as required will constitute a breach of this Contract and in addition the provisions of Clause 14.15 will apply.
- 14.10. Where the Customer has grounds to be concerned about the ILR Data submitted by the Supplier, including but not limited to the completeness or accuracy of the ILR Data provided by the Supplier under this Contract or any other agreement between the Supplier and the Customer, the Customer may take one or more of the following actions:
 - 14.10.1 require the Supplier, at its own cost, to carry out such work as the Customer deems necessary to improve the quality of the ILR Data;
 - 14.10.2 require the Supplier at its own cost to carry out an audit in accordance with Clause 14.15;
 - 14.10.3 where ILR Data submitted by the Supplier on an ILR is missing, incomplete or incorrect in the final ILR submission of the Academic Year, the Customer will recover the Contract Charges that correspond with the missing, incomplete or incorrect ILR Data;
 - 14.10.4 require the Supplier to supply ILR Data at such intervals and for such a period as is specified by the Customer;
 - 14.10.5 suspend the payment of the Contract Charges for a specified period in accordance; and / or
 - 14.10.6 terminate the Contract in accordance with Clauses 30 or 32.
- 14.11. If served with notice of a breach of the submission requirements as referred to in this Clause, the Supplier must correct the ILR Data or supply the Customer with evidence in support of its ILR submission within the period specified in the notice. This evidence can include digital copies of documents such as Learner files, commitment statements or employer declarations. The Supplier should note that annotated spreadsheets or notes are not acceptable to the Customer as evidence.
- 14.12. Where the Supplier is delivering Skills Bootcamp Courses to one or more Learners claiming state benefits, it must provide data to the Secretary of State with responsibility for unemployment or their nominated representative in accordance with the requirements notified to the Supplier.
- 14.13. The Customer will pay the Contract Charge due to the Supplier in accordance with the payment profile set out in the table at Annex 2 of Contract Schedule 3 (Contract Charges, Payment & Invoices).
- 14.14. The payment of Contract Charges by the Customer shall be without prejudice to any claims or rights, which the Customer may have against the Supplier and shall not be taken as an acknowledgement that the Supplier has fulfilled its obligations under this Contract. Prior to payment of Contract Charges, the Customer shall be entitled to make deductions or deferments in respect of any disputes or claims whatsoever with or against the Supplier, arising from this Contract or any other Contract between the Supplier and the Customer.
- 14.15. Where the Customer identifies errors which it deems to be material in the ILR Data, the Customer reserves the right at its absolute discretion to require the Supplier at the Supplier's cost to carry out a 100% audit of all or part of the Skills Bootcamp Course by a deadline specified by the Customer and / or to recover from the Supplier Contract Charges equivalent to an amount based on the error rate identified and the total value of the Contract Charges paid to the Supplier under this Contract.
- 14.16. Without prejudice to any other provisions of this Contract, such amounts may be recovered by making adjustments to ILR Data submitted by the Supplier under the Contract, or by raising an invoice for payment by the Supplier, or by making deductions from future payments (including Contract Charges) due to the Supplier under the Contract. Failure by the Supplier to settle such amounts will constitute a material Default of this Contract. The decision of the Customer as to the amount of recovery under this Clause is final.
- 14.17. All payments by the Customer of the Contract Charges will be made via BACS.

Contract Charges

- 14.18. In consideration of the Supplier carrying out its obligations under this Contract the Customer shall pay the Contract Charges in accordance with Clause 14.13.
- 14.19. Except as otherwise provided, each Party shall bear its own costs and expenses incurred in respect of compliance with its obligations under Clauses 12 (Records, Audit Access and Open Book Data), 23.20 to 23.21 (Freedom of Information) and 23.22 to 23.36 (Protection of Personal Data).
- 14.20. If the Customer fails to pay the Contract Charges the Supplier shall have the right to charge interest on the overdue amount at the applicable rate under the Late Payment of Commercial Debts (Interest) Act 1998, accruing on a daily basis from the due date up to the date of actual payment, whether before or after judgment.
- 14.21. The Customer is not responsible for any shortfall in fees or any lost revenue suffered by the Supplier, in the event that a Learner does not complete their Skills Bootcamp course and withdraws early from it.

VAT

- 14.22. The Contract Charges are stated exclusive of VAT, which shall be added at the prevailing rate as applicable and paid by the Customer in accordance with Clause 14.13.
- 14.23. Where VAT is applicable, the Supplier shall indemnify the Customer on a continuing basis against any liability, including any interest, penalties or costs incurred, which is levied, demanded or assessed on

the Customer at any time (whether before or after the making of a demand pursuant to the indemnity hereunder) in respect of the Supplier's failure to account for or to pay any VAT relating to payments made to the Supplier under this Contract

14.24. Any amounts due under Clause 14.23 (VAT) shall be paid in cleared funds by the Supplier to the Customer not less than five (5) Working Days before the date upon which the tax or other liability is payable by the Customer.

Retention and Set Off

- 14.25. The Customer may retain or set off any amount owed to it by the Supplier against any amount due to the Supplier under this Contract or under any other agreement between the Supplier and the Customer.
- 14.26. If the Customer wishes to exercise its right pursuant to Clause 14.25 it shall give notice to the Supplier setting out the Customer's reasons for retaining or setting off the relevant Contract Charges.
- 14.27. The Supplier shall make any payments due to the Customer without any deduction whether by way of set-off, counterclaim, discount, abatement or otherwise unless the Supplier has obtained a sealed court order requiring an amount equal to such deduction to be paid by the Customer to the Supplier.

Foreign Currency

- 14.28. Any requirement of Law to account for the Services in any currency other than Sterling, (or to prepare for such accounting) instead of and/or in addition to Sterling, shall be implemented by the Supplier free of charge to the Customer.
- 14.29. The Customer shall provide all reasonable assistance to facilitate compliance with Clause 14.28 by the Supplier.

Income Tax and National Insurance Contributions

- 14.30. Where the Supplier or any Supplier Personnel are liable to be taxed in the UK or to pay national insurance contributions in respect of consideration received under this Contract, the Supplier shall:
 - 14.30.1 at all times comply with the Income Tax (Earnings and Pensions) Act 2003 and all other statutes and regulations relating to income tax, and the Social Security Contributions and Benefits Act 1992 and all other statutes and regulations relating to national insurance contributions, in respect of that consideration; and
 - 14.30.2 indemnify the Customer against any income tax, national insurance and social security contributions and any other liability, deduction, contribution, assessment or claim arising

from or made (whether before or after the making of a demand pursuant to the indemnity hereunder) in connection with the provision of the Services by the Supplier or any Supplier Personnel.

- 14.31. In the event that any one of the Supplier Personnel is a Worker as defined in Contract Schedule 1 (Definitions) who receives consideration relating to the Services, then, in addition to its obligations under Clause 14.30, the Supplier shall ensure that its contract with the Worker contains the following requirements:
 - 14.31.1 that the Customer may, at any time during the Contract Period, request that the Worker provides information which demonstrates how the Worker complies with the requirements of Clause 14.30, or why those requirements do not apply to it. In such case, the Customer may specify the information which the Worker must provide and the period within which that information must be provided;
 - 14.31.2 that the Worker's contract may be terminated at the Customer's request if:
 - (a) the Worker fails to provide the information requested by the Customer within the time specified by the Customer under Clause 14.31.1; and/or
 - (b) the Worker provides information which the Customer considers is inadequate to demonstrate how the Worker complies with Clause 14.30 or confirms that the Worker is not complying with those requirements; and
 - (C) that the Customer may supply any information it receives from the Worker to HMRC for the purpose of the collection and management of revenue for which they are responsible

15. PROMOTING TAX COMPLIANCE

- 15.1. This Clause 15 shall apply if the Contract Charges payable under this Contract exceed or are likely to exceed five (5) million pounds during the Contract Period.
- 15.2. If, at any point during the Contract Period, an Occasion of Tax Non-Compliance occurs, the Supplier shall:
 - 15.2.1 notify the Customer in writing of such fact within five (5) Working Days of its occurrence; and
 - 15.2.2 promptly provide to the Customer:
 - (a) details of the steps that the Supplier is taking to address the Occasion of Tax Non-Compliance and to prevent the same from recurring, together with any mitigating factors that it considers relevant; and
 - (b) such other information in relation to the Occasion of Tax Noncompliance as the Customer may reasonably require.

In the event that the Supplier fails to comply with this Clause 15 and/or does not provide details of proposed mitigating factors which in the reasonable opinion of the Customer are acceptable, then the Customer reserves the right to terminate this Contract for material Default.

G. <u>SUPPLIER PERSONNEL AND SUPPLY CHAIN MATTERS</u>

16. SUPPLIER PERSONNEL

Supplier Personnel

16.1. The Supplier shall:

16.1.1 provide a list of the names of all Supplier Personnel requiring admission to Customer Premises, specifying the capacity in which they require admission and giving such other particulars as the Customer may reasonably require; 16.1.2 ensure that all Supplier Personnel:

- (a) are appropriately qualified, trained and experienced to provide the Services with all reasonable skill, care and diligence;
- (b) are vetted in accordance with Good Industry Practice and, where applicable, the Security Policy and the Standards;
- (c) obey all lawful instructions and reasonable directions of the Customer (including, if so required by the Customer, the ICT Policy) and provide the Services to the reasonable satisfaction of the Customer; and
- (d) comply with all reasonable requirements of the Customer concerning conduct at the Customer Premises, including the security requirements set out in Contract Schedule 4 (Security);
- 16.1.3 subject to Contract Schedule 5 (Staff Transfer), retain overall control of the Supplier Personnel at all times so that the Supplier Personnel shall not be deemed to be employees, agents or contractors of the Customer;
- 16.1.4 be liable at all times for all acts or omissions of Supplier Personnel, so that any act or omission of a member of any Supplier Personnel which results in a Default under this Contract shall be a Default by the Supplier;
- 16.1.5 use all reasonable endeavours to minimise the number of changes in Supplier Personnel;
- 16.1.6 replace (temporarily or permanently, as appropriate) any Supplier Personnel as soon as practicable if any Supplier Personnel have been removed or are unavailable for any reason whatsoever;
- 16.1.7 bear the programme familiarisation and other costs associated with any replacement of any Supplier Personnel; and
- 16.1.8 procure that the Supplier Personnel shall vacate the Customer Premises immediately upon the Contract Expiry Date.
- 16.2. If the Customer reasonably believes that any of the Supplier Personnel are unsuitable to undertake work in respect of this Contract, it may:
 - 16.2.1 refuse admission to the relevant person(s) to the Customer Premises; and/or
 - 16.2.2 direct the Supplier to end the involvement in the provision of the Services of the relevant person(s).
- 16.3. The decision of the Customer as to whether any person is to be refused access to the Customer Premises shall be final and conclusive.

Relevant Convictions

- 16.4. For each member of Supplier Personnel who, in providing the Services, has, will have or is likely to have access to children, vulnerable persons or other members of the public to whom the Customer owes a special duty of care, the Supplier shall (and shall procure that the relevant Sub-Contractor shall):
 - 16.4.1 carry out a check with the records held by the Department for Education (DfE);
 - 16.4.2 conduct thorough questioning regarding any Relevant Convictions; and
 - 16.4.3 ensure a police check is completed and such other checks as may be carried out through the Disclosure and Barring Service (DBS), and the Supplier shall not (and shall ensure that any Sub-Contractor shall not) engage or continue to employ in the provision of the Services any person who has a Relevant Conviction or an inappropriate record.

17. STAFF TRANSFER

17.1. Where the commencement of the provision of the Services or any part of the Services results in one or more Relevant Transfers, Contract Schedule 5 (Staff Transfer) shall apply as follows:

- 17.1.1 where the Relevant Transfer involves the transfer of Transferring Customer Employees, Part A of Contract Schedule 5 (Staff Transfer) shall apply;
- 17.1.2 where the Relevant Transfer involves the transfer of Transferring Former Supplier Employees, Part B of Contract Schedule 5 (Staff Transfer) shall apply;
- 17.1.3 where the Relevant Transfer involves the transfer of Customer Employees and Transferring Former Supplier Employees, Parts A and B of Contract Schedule 5 (Staff Transfer) shall apply; and
- 17.1.4 Part C of Contract Schedule 5 (Staff Transfer) shall not apply;
- 17.2. Where commencement of the provision of the Services or a part of the Services does not result in a Relevant Transfer, Part C of Contract Schedule 5 (Staff Transfer) shall apply and Parts A and B of Contract Schedule 5 (Staff Transfer) shall not apply.
- 17.3. Part D of Contract Schedule 5 (Staff Transfer) shall apply on the expiry or termination of the Services or any part of the Services.
- 17.4. The Supplier shall both during and after the Contract Period indemnify the Customer against all Employee Liabilities that may arise as a result of any claims brought against the Customer by any person where such claim arises from any act or omission of the Supplier or any Supplier Personnel.

18. SUPPLY CHAIN RIGHTS AND PROTECTION

Appointment of Sub-Contractors

- 18.1. The Supplier may enter into one or more Sub-Contracts provided that it does so in accordance with this Clause 18, Contract Schedule 1 (Service Requirements) and Contract Schedule 14 (Sub-Contracting). Clauses 18.9 and 18.11 will apply to Key Sub-Contractors.
- 18.2. Failure to comply with any of the requirements under this Clause 18 may result in the Customer taking such actions as it deems appropriate, which may include but is not limited to suspending payment of the Contract Charges and/or termination of this Contract under Clause 30.1 (Termination).
- 18.3. The Supplier shall exercise due skill and care in the selection of any Sub-Contractors to ensure that the Supplier is able to:
 - 18.3.1 manage any Sub-Contractors in accordance with Good Industry Practice;
 - 18.3.2 comply with its obligations under this Contract; and
 - 18.3.3 assign, novate or otherwise transfer to the Customer or any Replacement Supplier any of its rights and/or obligations under each Sub-Contract that relate exclusively to this Contract.
- 18.4. Prior to sub-contacting any of its obligations under this Contract, the Supplier shall notify the Customer and provide the Customer with:
 - 18.4.1 the proposed Sub-Contractor's name, registered office and company registration number;
 - 18.4.2 the scope of any Services to be provided by the proposed Sub-Contractor; and
 - 18.4.3 where the proposed Sub-Contractor is an Affiliate of the Supplier, evidence that demonstrates to the reasonable satisfaction of the Customer that the proposed SubContract has been agreed on "arm's-length" terms.
- 18.5. If requested by the Customer within ten (10) Working Days of receipt of the Supplier's notice issued pursuant to Clause 18.4, the Supplier shall also provide:
 - 18.5.1 a copy of the proposed Sub-Contract; and
 - 18.5.2 any further information reasonably requested by the Customer.
- 18.6. The Customer may, within ten (10) Working Days of receipt of the Supplier's notice issued pursuant to Clause 18.4 (or, if later, receipt of any further information requested pursuant to Clause 18.5), object to the appointment of the relevant Sub-Contractor if they consider that:

- 18.6.1 the appointment of a proposed Sub-Contractor may prejudice the provision of the Services or may be contrary to the interests respectively of the Customer under this Contract
- 18.6.2 the proposed Sub-Contractor is unreliable and/or has not provided reliable and/or reasonable services to its other customers; and/or
- 18.6.3 the proposed Sub-Contractor employs unfit persons,

in which case, the Supplier shall not proceed with the proposed appointment.

- 18.7. The Supplier may proceed with the proposed appointment if
 - 18.7.1 the Customer has not notified the Supplier that it objects to the proposed Sub-Contractor's appointment by the later of ten (10) Working Days of receipt of:
 - (a) the Supplier's notice issued pursuant to Clause18.4; and
 - (b) any further information requested by the Customer pursuant to Clause 18.5; and
 - 18.7.2 the proposed Sub-Contract is not a Key Sub-Contract which shall require the written consent of the Authority and the Customer in accordance with Clause 18.10 to 18.11 (Appointment of Key Sub-Contractors).
- 18.8. The Supplier shall ensure that each Sub-Contract shall include a provision restricting the ability of the Sub-Contractor to Sub-Contract all or any part of the provision of the Services to the Supplier under the Sub-Contract without first seeking the written consent of the Customer.

Appointment of Key Sub-Contractors

- 18.9. Any Key Sub-Contractors that have been appointed by the Supplier at the Contract Commencement Date are set out at Paragraph 2.1 of Schedule 14 (Sub-Contracting).
- 18.10. Where the Supplier wishes to enter into a new Key Sub-Contract or replace a Key Sub-Contractor, it must obtain the prior written consent of the Customer (the decision to consent or otherwise not to be unreasonably withheld or delayed). The Customer may reasonably withhold its consent to the appointment of a Key Sub-Contractor if any of them considers that:
 - 18.10.1 the appointment of a proposed Key Sub-Contractor may prejudice the provision of the Services or may be contrary to its interests;

18.10.2 the proposed Key Sub-Contractor is unreliable and/or has not provided reliable and/or reasonable services to its other customers; and/or 18.10.3 the proposed Key Sub-Contractor employs unfit persons.

- 18.11. The Supplier shall ensure that each Key Sub-Contract shall include:
 - 18.11.1 provisions which will enable the Supplier to discharge its obligations under this Contract;
 - 18.11.2 a right under CRTPA for the Customer to enforce any provisions under the Key SubContract which confer a benefit upon the Customer;
 - 18.11.3 a provision enabling the Customer to enforce the Key Sub-Contract as if it were the Supplier;
 - 18.11.4 a provision enabling the Supplier to assign, novate or otherwise transfer any of its rights and/or obligations under the Key Sub-Contract to the Customer or any Replacement Supplier;
 - 18.11.5 obligations no less onerous on the Key Sub-Contractor than those imposed on the Supplier under this Contract in respect of:
 - (a) data protection requirements set out in Clauses 23.1 (Security Requirements),
 23.2 to 23.9 (Protection of Customer Data) and 23.22 to 23.36 (Data Protection);
 - (b) FOIA requirements set out in Clause 23.20 to 23.21 (Freedom of Information);

- (c) the keeping of records in respect of the Services being provided under the Key Sub-Contract, including the maintenance of Open Book Data;
- (d) the conduct of audits set out in Clause 12 (Records, Audit Access & Open Book Data);
- 18.11.6 provisions enabling the Supplier to terminate the Key Sub-Contract on notice on terms no more onerous on the Supplier than those imposed on the Customer under Clauses 30 (Customer Termination Rights), 32 (Termination by Either Party) and 34 (Consequences of Expiry or Termination) of this Contract;
- 18.11.7 the provision set out in clause 18.8 above; and
- 18.11.8 a provision, where a provision in Contract Schedule 5 (Staff Transfer) imposes an obligation on the Supplier to provide an indemnity, undertaking or warranty, requiring the Key SubContractor to provide such indemnity, undertaking or warranty to the Customer, Former Supplier or the Replacement Supplier as the case may be.

Supply Chain Protection

- 18.12. The Supplier shall ensure that all Sub-Contracts contain a provision:
 - 18.12.1 requiring the Supplier to pay any undisputed sums which are due from it to the SubContractor within a specified period not exceeding thirty (30) days from the receipt of a Valid Invoice;
 - 18.12.2 requiring that any invoices submitted by a Sub-Contractor shall be considered and verified by the Supplier in a timely fashion and that undue delay in doing so shall not be sufficient justification for failing to regard an invoice as valid and undisputed;
 - 18.12.3 requiring the Sub-Contractor to include in any Sub-Contract which it in turn awards suitable provisions to impose, as between the parties to that Sub-Contract, requirements to the same effect as those required by sub-clauses 18.12.1 and 18.12.2 directly above; and
 - 18.12.4 conferring a right to the Customer to publish the Supplier's compliance with its obligation to pay undisputed invoices within the specified payment period.
- 18.13. The Supplier shall:
 - 18.13.1 pay undisputed sums which are due from it to a Sub-Contractor within thirty (30) days from the receipt of a Valid Invoice;
 - 18.13.2 Provide a summary of its compliance with this Clause 18.13.1, such data to be certified each quarter by a director of the Supplier as being accurate and not misleading.
- 18.14. Any invoices submitted by a Sub-Contractor to the Supplier shall be considered and verified by the Supplier in a timely fashion. Undue delay in doing so shall not be sufficient justification for the Supplier failing to regard an invoice as valid and undisputed.
- 18.15. Notwithstanding any provision of Clauses 23.10 to 23.18 (Confidentiality) and 24 (Publicity and Branding) if the Supplier notifies the Customer that the Supplier has failed to pay an undisputed SubContractor's invoice within thirty (30) days of receipt, or the Customer otherwise discovers the same, the Customer shall be entitled to publish the details of the late or non-payment (including on government websites and in the press).

Termination of Sub-Contracts

- 18.16. The Customer may require the Supplier to terminate:
 - 18.16.1 a Sub-Contract where:
 - (a) the acts or omissions of the relevant Sub-Contractor have caused or materially contributed to the Customer's right of termination pursuant to any of the termination events in Clause 30 (Customer Termination Rights) except Clause 30.10 (Termination Without Cause); and/or

- (b) relevant Sub-Contractor or its Affiliates embarrassed the Customer or otherwise brought the Customer into disrepute by engaging in any act or omission which is reasonably likely to diminish the trust that the public places in the Customer, regardless of whether or not such act or omission is related to the Sub-Contractor's obligations in relation to the Services or otherwise;
- 18.16.2 a Key Sub-Contract where there is a Change of Control of the relevant Key Sub-Contractor, unless:
 - (a) the Customer has given its prior written consent to the particular Change of Control, which subsequently takes place as proposed; or
 - (b) the Customer has not served its notice of objection within six (6) Months of the later of the date the Change of Control took place or the date on which the Customer was given notice of the Change of Control.

Retention of Legal Obligations

- 18.17. Notwithstanding the Supplier's right to Sub-Contract pursuant to Clause 18 (Supply Chain Rights and Protection), the Supplier shall remain responsible for all acts and omissions of its Sub-Contractors and the acts and omissions of those employed or engaged by the Sub-Contractors as if they were its own.
- H. <u>NOT USED</u>
- 19. NOT USED
- 20. NOT USED
- 21. NOT USED
- I. INTELLECTUAL PROPERTY AND INFORMATION

22. INTELLECTUAL PROPERTY RIGHTS

Allocation of title to IPR

- 22.1. Save as expressly granted elsewhere under this Contract:
 - 22.1.1 the Customer shall not acquire any right, title or interest in or to the Intellectual Property Rights of the Supplier or its licensors, including:
 - (a) the Supplier Background IPR;
 - (b) the Third Party IPR; and
 - (c) the Project Specific IPR.
 - 22.1.2 the Supplier shall not acquire any right, title or interest in or to the Intellectual Property Rights

of the Customer or its licensors, including the: (a) Customer Background IPR; and

- (b) Customer Data.
- 22.2. Where either Party acquires, by operation of Law, title to Intellectual Property Rights that is inconsistent with the allocation of title set out in Clause 22.1, 22.2 and 22.3, it shall assign in writing such Intellectual Property Rights as it has acquired to the other Party on the request of the other Party (whenever made).
- 22.3. Neither Party shall have any right to use any of the other Party's names, logos or trade marks on any of its products or services without the other Party's prior written consent.

Licence granted by the Supplier: Project Specific IPR

22.4. The Supplier if requested by the Customer will grant to the Customer or shall procure the direct grant to the Customer of, a 10 year, royalty-free, irrevocable, nonexclusive licence to use the Project Specific IPR including but not limited to the right to copy, adapt, publish and distribute such Project Specific IPR.

Licence granted by the Supplier: Supplier Background IPR

- 22.5. The Supplier if requested by the Customer will grant to the Customer a 10 year, royalty-free and nonexclusive licence to use the Supplier Background IPR for any purpose relating to the Services (or substantially equivalent services) or for any purpose relating to the exercise of the Customer's (or, if the Customer is a Central Government Body, any other Central Government Body's) business or function.
- 22.6. In the event the licence of the Supplier Background IPR is terminated for Default or expires, the Customer shall:
 - 22.6.1 immediately cease all use of the Supplier Background IPR;
 - 22.6.2 at the discretion of the Supplier, return or destroy documents and other tangible materials that contain any of the Supplier Background IPR, provided that if the Supplier has not made an election within six (6) Months of the termination of the licence, the Customer may destroy the documents and other tangible materials that contain any of the Supplier Background IPR; and
 - 22.6.3 ensure, so far as reasonably practicable, that any Supplier Background IPR that is held in electronic, digital or other machine-readable form ceases to be readily accessible (other than by the information technology staff of the Customer) from any computer, word processor, voicemail system or any other device containing such Supplier Background IPR.

Customer's right to sub-license

- 22.7. The Customer shall be freely entitled to sub-license the rights granted to it pursuant to Clause 22.4 (Licence granted by the Supplier: Project Specific IPR).
- 22.8. The Customer may sub-license:
 - 22.8.1 the rights granted under Clause 22.5 (Licence granted by the Supplier: Supplier Background IPR) to a third party (including for the avoidance of doubt, any Replacement Supplier) provided that:
 - (a) the sub-licence is on terms no broader than those granted to the Customer; and
 - (b) the sub-licence only authorises the third party to use the rights licensed in Clause 22.5 (Licence granted by the Supplier: Supplier Background IPR) for purposes relating to the Services (or substantially equivalent services) or for any purpose relating to the exercise of the Customer's (or, if the Customer is a Central Government Body, any other Central Government Body's) business or function; and
 - 22.8.2 the rights granted under Clause 22.5 (Licence granted by the Supplier: Supplier Background IPR) to any Approved Sub-Licensee to the extent necessary to use and/or obtain the benefit of the Project Specific IPR provided that the sub-licence is on terms no broader than those granted to the Customer.

Customer's right to assign/novate licences

- 22.9. The Customer shall be freely entitled to assign, novate or otherwise transfer its rights and obligations under the licence granted to it pursuant to Clause 22.4 (Licence granted by the Supplier: Project Specific IPR).
- 22.10. The Customer may assign, novate or otherwise transfer its rights and obligations under the licence granted pursuant to Clause 22.5 (Licence granted by the Supplier: Supplier Background IPR) to:
 - 22.10.1 a Central Government Body; or

- 22.10.2 to any body (including any private sector body) which performs or carries on any of the functions and/or activities that previously had been performed and/or carried on by the Customer.
- 22.11. Where the Customer is a Central Government Body, any change in the legal status of the Customer which means that it ceases to be a Central Government Body shall not affect the validity of any licence granted in Clause 22.4 (Licence granted by the Supplier: Project Specific IPR) and/or Clause 22.5 (Licences granted by the Supplier: Supplier Background IPR). If the Customer ceases to be a Central Government Body, the successor body to the Customer shall still be entitled to the benefit of the licences granted in Clause 22.4 (Licence granted by the Supplier: Project Specific IPR) and Clause 22.5 (Licence granted by the Supplier: Supplier: Supplier: Project Specific IPR) and Clause 22.5 (Licence granted by the Supplier: Supplier: Background IPR).
- 22.12. If a licence granted in Clause 22.4 (Licence granted by the Supplier: Project Specific IPR) and/or Clause 22.5 (Licence granted by the Supplier: Supplier Background IPR) is novated under Clauses 22.9 and/or 22.10 or there is a change of the Customer's status pursuant to Clause 22.11 (both such bodies being referred to as the **"Transferee"**), the rights acquired by the Transferee shall not extend beyond those previously enjoyed by the Customer.

Third Party IPR

- 22.13. The Supplier shall procure that the owners or the authorised licensors of any Third Party IPR grant a direct licence to the Customer on terms at least equivalent to those set out in Clause 22.5 (Licence granted by the Supplier: Supplier Background IPR) and Clause 22.10 (Customer's right to assign/novate licences). If the Supplier cannot obtain for the Customer a licence materially in accordance with the licence terms set out in Clause 22.5 (Licences granted by the Supplier: Supplier Background IPR) and Clause 22.10 (Customer's right to assign/novate licence terms set out in Clause 22.5 (Licences granted by the Supplier: Supplier Background IPR) and Clause 22.10 (Customer's right to assign/novate licences) in respect of any such Third Party IPR, the Supplier shall:
 - 22.13.1 notify the Customer in writing giving details of what licence terms can be obtained from the relevant third party and whether there are alternative providers which the Supplier could seek to use; and
 - 22.13.2 only use such Third Party IPR if the Customer Approves the terms of the licence from the relevant third party.

Licence granted by the Customer

- 22.14. The Customer hereby grants to the Supplier a royalty-free, nonexclusive, non-transferable licence during the Contract Period to use the Customer Background IPR and the Customer Data solely to the extent necessary for providing the Services in accordance with this Contract, including (but not limited to) the right to grant sub-licences to Sub-Contractors provided that:
 - 22.14.1 any relevant Sub-Contractor has entered into a confidentiality undertaking with the Supplier on the same terms as set out in Clause 23.10 to 23.18 (Confidentiality); and
 - 22.14.2 the Supplier shall not without Approval use the licensed materials for any other purpose or for the benefit of any person other than the Customer.

Termination of Licences

- 22.15. Subject to Clause 22.5 (Licence granted by the Supplier: Supplier Background IPR), all licences granted pursuant to Clause 22 (Intellectual Property Rights) (other than those granted pursuant to Clause 22.13 (Third Party IPR) and 22.14 (Licence granted by the Customer)) shall survive the Contract Expiry Date.
- 22.16. The Supplier shall, if requested by the Customer as a result of a contract termination in accordance with Clause 30, grant (or procure the grant) to the Replacement Supplier of a licence to use any Supplier Background IPR and/or Third Party IPR on terms equivalent to those set out in Clause 22.5 (Licence granted by the Supplier: Supplier Background IPR) subject to the Replacement Supplier entering into reasonable confidentiality undertakings with the Supplier.
- 22.17. The licence granted pursuant to Clause 22.14 (Licence granted by the Customer) and any sub-licence granted by the Supplier in accordance with Clause 22.14 (Licence granted by the Customer) shall terminate automatically on the Contract Expiry Date and the Supplier shall:

- 22.17.1 immediately cease all use of the Customer Background IPR and the Customer Data (as the case may be);
- 22.17.2 at the discretion of the Customer, return or destroy documents and other tangible materials that contain any of the Customer Background IPR and the Customer Data, provided that if the Customer has not made an election within six months of the termination of the licence, the Supplier may destroy the documents and other tangible materials that contain any of the Customer Background IPR and the Customer Data (as the case may be); and
- 22.17.3 ensure, so far as reasonably practicable, that any Customer Background IPR and Customer Data that are held in electronic, digital or other machine-readable form ceases to be readily accessible from any computer, word processor, voicemail system or any other device of the Supplier containing such Customer Background IPR and/or Customer Data.

IPR Indemnity

- 22.18. The Supplier shall, during and after the Contract Period, on written demand, indemnify the Customer against all Losses incurred by, awarded against, or agreed to be paid by the Customer (whether before or after the making of the demand pursuant to the indemnity hereunder) arising from an IPR Claim.
- 22.19. If an IPR Claim is made, or the Supplier anticipates that an IPR Claim might be made, the Supplier may, at its own expense and sole option, either:
 - 22.19.1 procure for the Customer the right to continue using the relevant item which is subject to the IPR Claim; or
 - 22.19.2 replace or modify the relevant item with non-infringing substitutes provided that:
 - (a) the performance and functionality of the replaced or modified item is at least equivalent to the performance and functionality of the original item;
 - (b) the replaced or modified item does not have an adverse effect on any other Services;
 - (c) there is no additional cost to the Customer; and
 - (d) the terms and conditions of this Contract shall apply to the replaced or modified Services.
- 22.20. If the Supplier elects to procure a licence in accordance with Clause 22.19.1 or to modify or replace an item pursuant to Clause 22.19.2, but this has not avoided or resolved the IPR Claim, then:
 - 22.20.1 the Customer may terminate this Contract by written notice with immediate effect; and
 - 22.20.2 without prejudice to the indemnity set out in Clause 22.18, the Supplier shall be liable for all reasonable and unavoidable costs of the services including the additional costs of procuring, implementing and maintaining the substitute items.
- 22.21. Within one month of the Contract Commencement Date, the Supplier will compile an IPR Register which lists the categories of IPR that are referred to in this Clause and the corresponding documents, software, materials or any other asset in which there is IPR that are used in and / or generated under this DPA.
- 22.22. The IPR Register must be kept up to date during the Contract Period.

23. SECURITY AND PROTECTION OF INFORMATION

Security Requirements

23.1. The Supplier shall comply with the requirements of Contract Schedule 4 (Security) including the Security Management Plan (if any).

Protection of Customer Data

23.2. The Supplier shall not delete or remove any proprietary notices contained within or relating to the Customer Data.

- 23.3. The Supplier shall not store, copy, disclose, or use the Customer Data except as necessary for the performance by the Supplier of its obligations under this Contract or as otherwise Approved by the Customer.
- 23.4. To the extent that the Customer Data is held and/or Processed by the Supplier, the Supplier shall supply that Customer Data to the Customer as requested by the Customer and in the format (if any) specified by the Customer in the Contract Order Form and, in any event, as specified by the Customer from time to time in writing.
- 23.5. The Supplier shall take responsibility for preserving the integrity of Customer Data and preventing the corruption or loss of Customer Data.
- 23.6. The Supplier shall perform secure back-ups of all Customer Data and shall ensure that up-to-date backups are stored off-site at an Approved location in accordance with any business continuity plan or otherwise. The Supplier shall ensure that such back-ups are available to the Customer (or to such other person as the Customer may direct) at all times upon request and are delivered to the Customer at no less than six (6) Monthly intervals (or such other intervals as may be agreed in writing between the Parties).
- 23.7. The Supplier shall ensure that any system on which the Supplier holds any Customer Data, including backup data, is a secure system that complies with the Security Policy and the Security Management Plan (if any)
- 23.8. If at any time the Supplier suspects or has reason to believe that the Customer Data is corrupted, lost or sufficiently degraded in any way for any reason, then the Supplier shall notify the Customer immediately and inform the Customer of the remedial action the Supplier proposes to take.
- 23.9. If the Customer Data is corrupted, lost or sufficiently degraded as a result of a Default so as to be unusable, the Customer may:
 - 23.9.1 require the Supplier (at the Supplier's expense) to restore or procure the restoration of Customer Data as required by the Customer, and the Supplier shall do so as soon as practicable but not later than five (5) Working Days from the date of receipt of the Customer's notice; and/or
 - 23.9.2 itself restore or procure the restoration of Customer Data, and shall be repaid by the Supplier any reasonable expenses incurred in doing so as required by the Customer.

Confidentiality

- 23.10. For the purposes of Clauses 23.11 to 23.18, the term **"Disclosing Party"** shall mean a Party which discloses or makes available directly or indirectly its Confidential Information and **"Recipient"** shall mean the Party which receives or obtains directly or indirectly Confidential Information.
- 23.11. Except to the extent set out in Clauses 23.11 to 23.18 or where disclosure is expressly permitted elsewhere in this Contract, the Recipient shall:
 - 23.11.1 treat the Disclosing Party's Confidential Information as confidential and keep it in secure custody (which is appropriate depending upon the form in which such materials are stored and the nature of the Confidential Information contained in those materials); and
 - 23.11.2 not disclose the Disclosing Party's Confidential Information to any other person except as expressly set out in this Contract or without obtaining the owner's prior written consent;
 - 23.11.3 not use or exploit the Disclosing Party's Confidential Information in any way except for the purposes anticipated under this Contract ; and
 - 23.11.4 immediately notify the Disclosing Party if it suspects or becomes aware of any unauthorised access, copying, use or disclosure in any form of any of the Disclosing Party's Confidential Information.
- 23.12. The Recipient shall be entitled to disclose the Confidential Information of the Disclosing Party where:
 - 23.12.1 the Recipient is required to disclose the Confidential Information by Law, provided that Clause 23.20 to 23.21 (Freedom of Information) shall apply to disclosures required under the FOIA or the EIRs;

- 23.12.2 the need for such disclosure arises out of or in connection with:
 - (a) any legal challenge or potential legal challenge against the Customer arising out of or in connection with this Contract;
 - (b) the examination and certification of the Customer's accounts (provided that the disclosure is made on a confidential basis) or for any examination pursuant to Section 6(1) of the National Audit Act 1983 of the economy, efficiency and effectiveness with which the Customer is making use of any Services provided under this Contract; or
 - (c) the conduct of a Central Government Body review in respect of this Contract; or
- 23.12.3 the Recipient has reasonable grounds to believe that the Disclosing Party is involved in activity that may constitute a criminal offence under the Bribery Act 2010 and the disclosure is being made to the Serious Fraud Office.
- 23.13. If the Recipient is required by Law to make a disclosure of Confidential Information, the Recipient shall as soon as reasonably practicable and to the extent permitted by Law notify the Disclosing Party of the full circumstances of the required disclosure including the relevant Law and/or regulatory body requiring such disclosure and the Confidential Information to which such disclosure would apply.
- 23.14. Subject to Clause 23.11, the Supplier may only disclose the Confidential Information of the Customer on a confidential basis to:
 - 23.14.1 Supplier Personnel who are directly involved in the provision of the Services and need to know the Confidential Information to enable performance of the Supplier's obligations under this Contract; and
 - 23.14.2 its professional advisers for the purposes of obtaining advice in relation to this Contract.
- 23.15. Where the Supplier discloses Confidential Information of the Customer pursuant to Clause 23.14, it shall remain responsible at all times for compliance with the confidentiality obligations set out in this Contract by the persons to whom disclosure has been made.
- 23.16. The Customer may disclose the Confidential Information of the Supplier:
 - 23.16.1 to any Central Government Body on the basis that the information may only be further disclosed to Central Government Bodies;
 - 23.16.2 to the British Parliament and any committees of the British Parliament or if required by any British Parliamentary reporting requirement;
 - 23.16.3 to the extent that the Customer (acting reasonably) deems disclosure necessary or appropriate in the course of carrying out its public functions;
 - 23.16.4 on a confidential basis to a professional adviser, consultant, supplier or other person engaged by any of the entities described in Clause 23.16.4 (including any benchmarking organisation) for any purpose relating to or connected with this Contract;
 - 23.16.5 on a confidential basis for the purpose of the exercise of its rights under this Contract; or
 - 23.16.6 to a proposed transferee, assignee or novatee of, or successor in title to the Customer, and for the purposes of the foregoing, references to disclosure on a confidential basis shall mean disclosure subject to a confidentiality agreement or arrangement containing terms no less stringent than those placed on the Customer under Clause 23.11 to 23.18.
- 23.17. Nothing in Clause 23.11 to 23.18 shall prevent a Recipient from using any techniques, ideas or KnowHow gained during the performance of this Contract in the course of its normal business to the extent that this use does not result in a disclosure of the Disclosing Party's Confidential Information or an infringement of Intellectual Property Rights.
- 23.18. In the event that the Supplier fails to comply with Clauses 23.11 to 23.12, the Customer reserves the right to terminate this Contract for material Default.

Transparency

23.19. The Supplier recognises that the Customer is subject to PPN 01/17 (Updates to transparency principles v1.1 <u>https://www.gov.uk/government/publications/procurement-policy-note-0117-updateto-transparency-principles</u> and shall provide the Customer with such information as it reasonably requests in this regard.

Freedom of Information

- 23.20. The Supplier acknowledges that the Customer is subject to the requirements of the FOIA and the EIRs. The Supplier shall:
 - 23.20.1 provide all necessary assistance and cooperation as reasonably requested by the Customer to enable the Customer to comply with its Information disclosure obligations under the FOIA and EIRs;
 - 23.20.2 transfer to the Customer all Requests for Information relating to this Contract that it receives as soon as practicable and in any event within two (2) Working Days of receipt;
 - 23.20.3 provide the Customer with a copy of all Information belonging to the Customer requested in the Request for Information which is in its possession or control in the form that the Customer requires within five (5) Working Days (or such other period as the Customer may reasonably specify) of the Customer's request for such Information; and
 - 23.20.4 not respond directly to a Request for Information unless authorised in writing to do so by the Customer.
- 23.21. The Supplier acknowledges that the Customer may be required under the FOIA and EIRs to disclose Information (including Commercially Sensitive Information) without consulting or obtaining consent from the Supplier. The Customer shall take reasonable steps to notify the Supplier of a Request for Information (in accordance with the Secretary of State's Section 45 Code of Practice on the Discharge of the Functions of Public Authorities under Part 1 of the FOIA) to the extent that it is permissible and reasonably practical for it to do so but (notwithstanding any other provision in this Contract) the Customer shall be responsible for determining in its absolute discretion whether any Commercially Sensitive Information and/or any other information is exempt from disclosure in accordance with the FOIA and/or the EIRs.

Data Protection

- 23.22. The Parties acknowledge that for the purposes of the Data Protection Legislation, the Customer is the Controller and the Supplier is the Processor unless otherwise specified in Contract Schedule 7. The only processing that the Processor is authorised to do is listed in Contract Schedule 7 by the Controller and may not be determined by the Processor.
- 23.23. The Processor shall notify the Controller immediately if it considers that any of the Controller's instructions infringe the Data Protection Legislation.
- 23.24. The Processor shall provide all reasonable assistance to the Controller in the preparation of any Data Protection Impact Assessment prior to commencing any processing. Such assistance may, at the discretion of the Controller, include:
 - 23.24.1 a systematic description of the envisaged processing operations and the purpose of the processing;
 - 23.24.2 an assessment of the necessity and proportionality of the processing operations in relation to the Services;
 - 23.24.3 an assessment of the risks to the rights and freedoms of Data Subjects; and
 - 23.24.4 the measures envisaged to address the risks, including safeguards, security measures and mechanisms to ensure the protection of Personal Data.
- 23.25. The Processor shall, in relation to any Personal Data processed in connection with its obligations under this Contract:

- 23.25.1 process that Personal Data only in accordance with Contract Schedule 7, unless the Processor is required to do otherwise by Law. If it is so required the Processor shall promptly notify the Controller before processing the Personal Data unless prohibited by Law;
- 23.25.2 ensure that it has in place Protective Measures, which are appropriate to protect against a Data Loss Event, which the Controller may reasonably reject (but failure to reject shall not amount to approval by the Controller of the adequacy of the Protective Measures), having taken account of the:
 - (a) nature of the data to be protected;
 - (b) harm that might result from a Data Loss Event;
 - (c) state of technological development; and
 - (d) cost of implementing any measures; 23.25.3

ensure that:

- (a) the Processor Personnel do not process Personal Data except in accordance with this Contract (and in particular Schedule 7);
- (b) it takes all reasonable steps to ensure the reliability and integrity of any Processor Personnel who have access to the Personal Data and ensure that they:
 - (i) are aware of and comply with the Processor's duties under this clause;
- (ii) are subject to appropriate confidentiality undertakings with the Processor or any Sub-processor;
- (iii) are informed of the confidential nature of the Personal Data and do not publish, disclose or divulge any of the Personal Data to any third Party unless directed in writing to do so by the Controller or as otherwise permitted by this Contract; and
- (iv) have undergone adequate training in the use, care, protection and handling of Personal Data; and
- 23.25.4 not transfer Personal Data outside of the EU unless the prior written consent of the Controller has been obtained and the following conditions are fulfilled:
 - (a) the Controller or the Processor has provided appropriate safeguards in relation to the transfer (whether in accordance with GDPR Article 46 or LED Article 37) as determined by the Controller;
 - (b) the Data Subject has enforceable rights and effective legal remedies;
 - (c) the Processor complies with its obligations under the Data Protection Legislation by providing an adequate level of protection to any Personal Data that is transferred (or, if it is not so bound, uses its best endeavours to assist the Controller in meeting its obligations); and
 - (d) the Processor complies with any reasonable instructions notified to it in advance by the Controller with respect to the processing of the Personal Data;
- 23.25.5 at the written direction of the Controller, delete or return Personal Data (and any copies of it) to the Controller on termination of the Contract unless the Processor is required by Law to retain the Personal Data.
- 23.26. Subject to Clause 23.27, the Processor shall notify the Controller immediately if it:
 - 23.26.1 receives a Data Subject Request (or purported Data Subject Request);
 - 23.26.2 receives a request to rectify, block or erase any Personal Data;

- 23.26.3 receives any other request, complaint or communication relating to either Party's obligations under the Data Protection Legislation;
- 23.26.4 receives any communication from the Information Commissioner or any other regulatory authority in connection with Personal Data processed under this Contract;
- 23.26.5 receives a request from any third Party for disclosure of Personal Data where compliance with such request is required or purported to be required by Law; or
- 23.26.6 becomes aware of a Data Loss Event.
- 23.27. The Processor's obligation to notify under Clause 23.26 shall include the provision of further information to the Controller in phases, as details become available.
- 23.28. Taking into account the nature of the processing, the Processor shall provide the Controller with full assistance in relation to either Party's obligations under Data Protection Legislation and any complaint, communication or request made under Clause 23.26 (and insofar as possible within the timescales reasonably required by the Controller) including by promptly providing:
 - 23.28.1 the Controller with full details and copies of the complaint, communication or request;
 - 23.28.2 such assistance as is reasonably requested by the Controller to enable the Controller to comply with a Data Subject Request within the relevant timescales set out in the Data Protection Legislation;
 - 23.28.3 the Controller, at its request, with any Personal Data it holds in relation to a Data Subject;
 - 23.28.4 assistance as requested by the Controller following any Data Loss Event;
 - 23.28.5 assistance as requested by the Controller with respect to any request from the Information Commissioner's Office, or any consultation by the Controller with the Information Commissioner's Office.
- 23.29. The Processor shall maintain complete and accurate records and information to demonstrate its compliance with this clause. This requirement does not apply where the Processor employs fewer than 250 staff, unless:
 - 23.29.1 the Controller determines that the processing is not occasional;
 - 23.29.2 the Controller determines the processing includes special categories of data as referred to in Article 9(1) of the GDPR or Personal Data relating to criminal convictions and offences referred to in Article 10 of the GDPR; or
 - 23.29.3 the Controller determines that the processing is likely to result in a risk to the rights and freedoms of Data Subjects.
- 23.30. The Processor shall allow for audits of its Data Processing activity by the Controller or the Controller's designated auditor.
- 23.31. Each Party shall designate its own data protection officer if required by the Data Protection Legislation.
- 23.32. Before allowing any Sub-processor to process any Personal Data related to this Contract, the Processor must:
 - 23.32.1 notify the Controller in writing of the intended Sub-processor and processing;
 - 23.32.2 obtain the written consent of the Controller;
 - 23.32.3 enter into a written agreement with the Sub-processor which give effect to the terms set out in clauses 23.22 to 23.36 (Data Protection) such that they apply to the Sub-processor; and
 - 23.32.4 provide the Controller with such information regarding the Sub-processor as the Controller may reasonably require.
- 23.33. The Processor shall remain fully liable for all acts or omissions of any of its Sub-processors.

- 23.34. The Controller may, at any time on not less than 30 Working Days' notice, revise this clause by replacing it with any applicable controller to processor standard clauses or similar terms forming part of an applicable certification scheme (which shall apply when incorporated by attachment to this Contract).
- 23.35. The Parties agree to take account of any guidance issued by the Information Commissioner's Office. The Controller may on not less than 30 Working Days' notice to the Processor amend this agreement to ensure that it complies with any guidance issued by the Information Commissioner's Office.
- 23.36. Where the Parties include two or more Joint Controllers as identified in Contract Schedule 7 in accordance with GDPR Article 26, those Parties shall enter into a Joint Controller Agreement based on the terms outlined in Contract Schedule 8 in replacement of Clauses 23.22 to 23.36 for the Personal Data under Joint Control.

24. PUBLICITY AND BRANDING

- 24.1. The Supplier shall not, without Approval (the decision of the Customer to Approve or not shall not be unreasonably withheld or delayed):
 - 24.1.1 make any press announcements or publicise this Contract in any way; or
 - 24.1.2 use the Customer's name or brand in any promotion or marketing or announcement of orders,
- 24.2. Each Party acknowledges to the other that nothing in this Contract either expressly or by implication constitutes an endorsement of any products or services of the other Party (including the Services and Supplier Equipment) and each Party agrees not to conduct itself in such a way as to imply or express any such approval or endorsement.
- 24.3. The Supplier must adhere to the Communications Guidelines provided by the Customer, "Skills Bootcamps Communications Guidance". The Customer reserves the right to update the said document from time to time.

J. LIABILITY AND INSURANCE

25. LIABILITY

Unlimited Liability

- 25.1. Neither Party excludes or limits it liability for:
 - 25.1.1 death or personal injury caused by its negligence, or that of its employees, agents or SubContractors (as applicable);
 - 25.1.2 bribery or Fraud by it or its employees; or
 - 25.1.3 any liability to the extent it cannot be excluded or limited by Law.
- 25.2. The Supplier does not exclude or limit its liability in respect of the indemnity in Clauses 22.18 to 22.20 (IPR Indemnity) and in each case whether before or after the making of a demand pursuant to the indemnity therein.

Financial Limits

- 25.3. The Supplier's total aggregate liability: in respect of all Losses incurred by the Customer under or in connection with this Contract as a result of Defaults shall in no event exceed: the higher of ten million pounds (£10,000,000) or a sum equal to one hundred and fifty per cent (150%) of the Contract Charges unless such Losses are in whole or in part incurred under Clauses 25.1 and / or 25.2 in which case the Supplier has unlimited liability.
- 25.4. Without prejudice to its obligation to pay the undisputed Contract Charges as and when they fall due for payment, the Customer's total aggregate liability in respect of all Losses shall be limited to a sum equal

to Contract Charges unless such Losses are in whole or in part incurred under Clauses 25.1 and / or 25.2 in which case the Customer has unlimited liability.

Non-recoverable Losses

- 25.5. Subject to Clause 25.1 and 25.2 (Unlimited Liability) neither Party shall be liable to the other Party for any:
 - 25.5.1 indirect, special or consequential Loss;
 - 25.5.2 loss of profits, turnover, savings, business opportunities or damage to goodwill (in each case whether direct or indirect).

Recoverable Losses

- 25.6. Subject to Clause 25.3 and 25.4 (Financial Limits), and notwithstanding Clause 25.5 (Non-recoverable Losses), the Supplier acknowledges that the Customer may, amongst other things, recover from the Supplier the following Losses incurred by the Customer to the extent that they arise as a result of a Default by the Supplier:
 - 25.6.1 any additional operational and/or administrative costs and expenses incurred by the Customer, including costs relating to time spent by or on behalf of the Customer in dealing with the consequences of the Default;
 - 25.6.2 any wasted expenditure or Contract Charges;

25.6.3 the additional cost of procuring Replacement Services for the remainder of the Contract Period, which shall include any incremental costs associated with such Replacement Services above those which would have been payable under this Contract; 25.6.4 any compensation or interest paid to a third party by the Customer; and

25.6.5 any fine, penalty or costs incurred by the Customer pursuant to Law.

Miscellaneous

- 25.7. Each Party shall use all reasonable endeavours to mitigate any loss or damage suffered arising out of or in connection with this Contract.
- 25.8. Any Deductions shall not be taken into consideration when calculating the Supplier's liability under Clause 25.3 to 25.4 (Financial Limits).
- 25.9. Subject to any rights of the Customer under this Contract (including in respect of an IPR Claim), any claims by a third party where an indemnity is sought by that third party from a Party to this Contract shall be dealt with in accordance with the provisions of DPA Schedule 20 (Conduct of Claims).

26. INSURANCE

- 26.1. This Clause 26 will only apply where specified in the Contract Order Form or elsewhere in this Contract.
- 26.2. Notwithstanding any benefit to the Customer of the policy or policies of insurance referred to in Clause 31 (Insurance) of the DPA, the Supplier shall effect and maintain such further policy or policies of insurance or extensions to such existing policy or policies of insurance procured under the DPA in respect of all risks which may be incurred by the Supplier arising out of its performance of its obligations under this Contract.
- 26.3. Without limitation to the generality of Clause 26.2 the Supplier shall ensure that it maintains the following insurance policies at the level set out below in respect of each and every claim:
 - 26.3.1 Employers' Liability £10 million;
 - 26.3.2 Public Liability £10 million; and
 - 26.3.3 Professional Indemnity £5 million
- 26.4. The Supplier shall effect and maintain the policy or policies of insurance referred to in Clause 26.3 for the Contract Period and for six (6) years after the Contract Expiry Date.

- 26.5. The Supplier shall give the Customer, on request, copies of all insurance policies referred to in Clause 26 or a broker's verification of insurance to demonstrate that the appropriate cover is in place, together with receipts or other evidence of payment of the latest premiums due under those policies.
- 26.6. If, for whatever reason, the Supplier fails to give effect to and maintain the insurance policies required under Clause 26 the Customer may make alternative arrangements to protect its interests and may recover the premium and other costs of such arrangements as a debt due from the Supplier.
- 26.7. The provisions of any insurance or the amount of cover shall not relieve the Supplier of any liability under this Contract. It shall be the responsibility of the Supplier to determine the amount of insurance cover that will be adequate to enable the Supplier to satisfy any liability in relation to the performance of its obligations under this Contract.
- 26.8. The Supplier shall ensure that nothing is done which would entitle the relevant insurer to cancel, rescind or suspend any insurance or cover, or to treat any insurance, cover or claim as voided in whole or part. The Supplier shall use all reasonable endeavours to notify the Customer (subject to third party confidentiality obligations) as soon as practicable when it becomes aware of any relevant fact, circumstance or matter which has caused, or is reasonably likely to provide grounds to, the relevant insurer to give notice to cancel, rescind, suspend or void any insurance, or any cover or claim under any insurance in whole or in part.

K. <u>REMEDIES AND RELIEF</u>

27. CUSTOMER REMEDIES FOR DEFAULT

Remedies

- 27.1. Without prejudice to any other right or remedy of the Customer howsoever arising, if the Supplier commits any Default of this Contract then the Customer may (whether or not any part of the Services have been Delivered) do any of the following:
 - 27.1.1 at the Customer's option, give the Supplier the opportunity (at the Supplier's expense) to remedy the Default together with any damage resulting from such Default (where such Default is capable of remedy) or to supply Replacement Services and carry out any other necessary work to ensure that the terms of this Contract are fulfilled, in accordance with the Customer's instructions;
 - 27.1.2 carry out, at the Supplier's expense, any work necessary to make the provision of the Services comply with this Contract;
 - 27.1.3 if the Default is a material Default that is capable of remedy (and for these purposes a material Default may be a single material Default or a number of Defaults or repeated Defaults whether of the same or different obligations and regardless of whether such Defaults are remedied which taken together constitute a material Default):
 - (a) instruct the Supplier to comply with the Rectification Plan Process;
 - (b) suspend this Contract (whereupon the relevant provisions of Clause 33 (Partial Termination, Suspension and Partial Suspension) shall apply) and step-in to itself supply or procure a third party to supply (in whole or in part) the Services;
 - (c) without terminating or suspending the whole of this Contract, terminate or suspend this Contract in respect of part of the provision of the Services only (whereupon the relevant provisions of Clause 33 (Partial Termination, Suspension and Partial Suspension) shall apply) and step-in to itself supply or procure a third party to supply (in whole or in part) such part of the Services;
- 27.2. Where the Customer exercises any of its step-in rights under Clauses 27.1.3(b) or 27.1.3(c), the Customer shall have the right to charge the Supplier for and the Supplier shall on demand pay any costs reasonably incurred by the Customer (including any reasonable administration costs) in respect of the supply of any part of the Services by the Customer or a third party and provided that the Customer uses its reasonable endeavours to mitigate any additional expenditure in obtaining Replacement Services.

Rectification Plan Process

- 27.3. Where the Customer has instructed the Supplier to comply with the Rectification Plan Process pursuant to Clause 27.1.3(a):
 - 27.3.1 the Supplier shall submit a draft Rectification Plan to the Customer for it to review as soon as possible and in any event within 10 (ten) Working Days (or such other period as may be agreed between the Parties) from the date of Customer's instructions. The Supplier shall submit a draft Rectification Plan even if the Supplier disputes that it is responsible for the Default giving rise to the Customer's request for a draft Rectification Plan.
 - 27.3.2 the draft Rectification Plan shall set out:
 - (a) full details of the Default that has occurred, including a cause analysis;
 - (b) the actual or anticipated effect of the Default; and
 - (c) the steps which the Supplier proposes to take to rectify the Default (if applicable) and to prevent such Default from recurring, including timescales for such steps and for the rectification of the Default (where applicable).
- 27.4. The Supplier shall promptly provide to the Customer any further documentation that the Customer requires to assess the Supplier's root cause analysis. If the Parties do not agree on the root cause set out in the draft Rectification Plan, either Party may refer the matter to be determined by an expert in accordance with paragraph 5 of Contract Schedule 6 (Dispute Resolution Procedure).
- 27.5. The Customer may reject the draft Rectification Plan by notice to the Supplier if, acting reasonably, it considers that the draft Rectification Plan is inadequate, for example because the draft Rectification Plan:
 - 27.5.1 is insufficiently detailed to be capable of proper evaluation;
 - 27.5.2 will take too long to complete;
 - 27.5.3 will not prevent reoccurrence of the Default; and/or
 - 27.5.4 will rectify the Default but in a manner which is unacceptable to the Customer.
- 27.6. The Customer shall notify the Supplier whether it consents to the draft Rectification Plan as soon as reasonably practicable. If the Customer rejects the draft Rectification Plan, the Customer shall give reasons for its decision and the Supplier shall take the reasons into account in the preparation of a revised Rectification Plan. The Supplier shall submit the revised draft of the Rectification Plan to the Customer for review within five (5) Working Days (or such other period as agreed between the Parties) of the Customer's notice rejecting the first draft.
- 27.7. If the Customer consents to the Rectification Plan, the Supplier shall immediately start work on the actions set out in the Rectification Plan.

L. <u>OFSTED</u>

28. OFSTED MONITORING VISITS

- 28.1. Where the Customer is made aware that OFSTED has assessed the Supplier as having made "insufficient progress" after a monitoring visit, the Customer may, at its absolute discretion take one or more of the following actions:
 - 28.1.1 require the Supplier to accept and comply with additional contractual obligations relating to the improvement of the Services assessed as insufficient progress;
 - 28.1.2 reduce and / or recover from the Supplier Contract Charges paid by the Customer in respect of that part of the Services that has been assessed as having made insufficient progress;
 - 28.1.3 suspend the payment of Contract Charges for current Learners for a specified period;
 - 28.1.4 not consider any applications or pay any Contract Charges for new Learners for a specified period;

- 28.1.5 require the Supplier to inform all of its existing Employers and Subcontractors as relevant, about the outcome of the OFSTED visit;
- 28.1.6 require the Supplier to enter into a subcontracting arrangement with another provider in order to ensure that there is minimal disruption to the Supplier's current Learners;
- 28.1.7 terminate this Contract for material Default in accordance with Clause 30.1.3 (Termination) where Learners may be at immediate risk on the grounds of safeguarding issues and / or the quality of leadership and / or training provision is such that one or more Learner has no reasonable prospect of achieving his or her training objective; and / or
- 28.1.8 terminate this Contract for material Default in accordance with Clause 30.1.3 (Termination) if the Supplier has two consecutive monitoring visits each resulting in one or more "insufficient progress" judgements in relation to one or more themes
- 28.2. Where the Customer is made aware that OFSTED has assessed a Subcontractor to the Supplier as having made "insufficient progress" during OFSTED's Supplier monitoring visits, if requested in writing by the Customer, the Supplier will suspend the delivery of the Services under the Subcontract and shall not subcontract any further Services to the Subcontractor until permitted to do so by the Customer.

OFSTED Inspection

- 28.3. When the Supplier receives notification from OFSTED that the Services are going to be inspected, the Supplier shall on request provide the Customer with details of its quality improvement activity, and any other relevant information at the same time as it submits it to OFSTED in accordance with the timescale specified by OFSTED. The Supplier must notify the Customer in writing of the date of the meeting at which OFSTED will give feedback on the inspection and allow the Customer's nominated representative to attend the meeting. The Supplier must confirm to the Customer in writing the outcome of the inspection within 5 Working Days of receiving OFSTED's feedback.
- 28.4. OFSTED may, at any time during the Contract Period, undertake an inspection of the Supplier. The Customer will consider the outcome of any such inspection as follows

Inadequate in Part

- 28.5. Where OFSTED has assessed the Services to be inadequate in part, the Customer may, at its absolute discretion, take one or more of the following actions;
 - 28.5.1 require the Supplier to accept and comply with additional contractual obligations relating to the improvement of the Services assessed as inadequate;
 - 28.5.2 suspend the payment of Contract Charges for current Learners for a specified period;
 - 28.5.3 not consider any applications or pay any Contract Charges for new Learners for a specified period;
 - 28.5.4 reduce and / or recover from the Supplier Contract Charges paid by the Customer in respect of that part of the Services that has been assessed as inadequate;
 - 28.5.5 require the Supplier to inform all of its existing Employers, Subcontractors and EmployerProviders, as relevant, about the outcome of the OFSTED visit; and / or
 - 28.5.6 terminate this Contract for material Default in accordance with Clause 30.1.3 (Termination) where Learners may be at immediate risk on the grounds of safeguarding issues and / or the quality of leadership and / or training provision within the 'adult learning programmes' judgment is judged inadequate.

Inadequate Overall

- 28.6. Where OFSTED has assessed the Services as inadequate overall, the Customer may, at its absolute discretion take one or more of the following actions:
 - 28.6.1 require the Supplier to accept and comply with additional conditions relating to the improvement of the overall Services;

- 28.6.2 suspend the payment of Contract Charges to the Supplier in relation to Learners for a specified period;
- 28.6.3 not consider any applications or pay any Contract Charges for new Learners for a specified period;
- 28.6.4 give consideration to the assessment of inadequate when considering access to Contract Charges in any subsequent agreements and / or contracts between the Parties;
- 28.6.5 reduce and / or recover the Contract Charges paid to the Supplier;
- 28.6.6 require the Supplier to inform all of its existing Employers, Subcontractors and EmployerProviders, as relevant, about the outcome of the OFSTED visit; and / or
- 28.6.7 terminate this Contract for material Default in accordance with Clause 30.1 (Termination) if Learners may be at immediate risk on the grounds of safeguarding issues; and / or the quality of leadership is inadequate; and / or training provision within the 'adult learning programmes' is judged to be inadequate.
- 28.7. If a Sub-Contractor is assessed by OFSTED as having made "insufficient progress" in relation to one or more themes during an OFSTED monitoring visit, or judged as 'inadequate' in one or more theses in an Ofsted full inspection, unless permitted to continue by the Customer, the Supplier shall suspend delivery of the Services under the Sub-Contract and shall not subcontract any further Services to that Sub-Contractor.
- 28.8. The failure of the Supplier, as assessed by the Customer, to comply with any requirements of Clauses 28.1, 28.5 and 28.6 within such time as the Customer may deem reasonable may lead to the Customer taking such action as it deems appropriate which may include, but is not limited to, terminating this Contract in accordance with Clause 30.1.5 (Termination).
- 28.9. The Customer will take action based on OFSTED's provisional and confirmed outcomes. Where the Customer is made aware that the Supplier has made a complaint about the outcome of the OFSTED assessment, the Customer will continue to consider what action it needs to take under this Contract but will be mindful of the implications arising from the outcome of a complaint. The Customer will review any decisions made at such time when the outcome of any complaint is made known.

29. FORCE MAJEURE

- 29.1. Subject to the remainder of Clause L, a Party may claim relief under Clause L from liability for failure to meet its obligations under this Contract for as long as and only to the extent that the performance of those obligations is directly affected by a Force Majeure Event. Any failure or delay by the Supplier in performing its obligations under this Contract which results from a failure or delay by an agent, SubContractor or supplier shall be regarded as due to a Force Majeure Event only if that agent, SubContractor or supplier is itself impeded by a Force Majeure Event from complying with an obligation to the Supplier.
- 29.2. The Affected Party shall as soon as reasonably practicable issue a Force Majeure Notice, which shall include details of the Force Majeure Event, its effect on the obligations of the Affected Party and any action the Affected Party proposes to take to mitigate its effect.
- 29.3. If the Supplier is the Affected Party, it shall not be entitled to claim relief under Clause L to the extent that consequences of the relevant Force Majeure Event:
 - 29.3.1 are capable of being mitigated by any of the provision of any Services, but the Supplier has failed to do so; and/or
 - 29.3.2 should have been foreseen and prevented or avoided by a prudent provider of services similar to the Services, operating to the standards required by this Contract.
- 29.4. Subject to Clause 29.5, as soon as practicable after the Affected Party issues the Force Majeure Notice, and at regular intervals thereafter, the Parties shall consult in good faith and use reasonable endeavours to agree any steps to be taken and an appropriate timetable in which those steps should be taken, to enable continued provision of the Services affected by the Force Majeure Event.

- 29.5. The Parties shall at all times following the occurrence of a Force Majeure Event and during its subsistence use their respective reasonable endeavours to prevent and mitigate the effects of the Force Majeure Event. Where the Supplier is the Affected Party, it shall take all steps in accordance with Good Industry Practice to overcome or minimise the consequences of the Force Majeure Event.
- 29.6. Where, as a result of a Force Majeure Event, an Affected Party fails to perform its obligations in accordance with this Contract, then during the continuance of the Force Majeure Event:
 - 29.6.1 the other Party shall not be entitled to exercise any rights to terminate this Contract in whole or in part as a result of such failure unless the provision of the Services is materially impacted by a Force Majeure Event which endures for a continuous period of more than ninety (90) days; and
 - 29.6.2 the Supplier shall not be liable for any Default arising as a result of such failure;
- 29.7. Where, as a result of a Force Majeure Event the Supplier fails to perform its obligations in accordance with this Contract:
 - 29.7.1 the Customer shall not be entitled: during the continuance of the Force Majeure Event to exercise its step-in rights under Clause 27.1.2 and 27.1.3 (Customer Remedies for Default) as a result of such failure
 - 29.7.2 the Supplier shall be entitled to receive payment of the Contract Charges (or a proportional payment of them) only to the extent that the Services (or part of the Services) continue to be provided in accordance with the terms of this Contract during the occurrence of the Force Majeure Event.
- 29.8. The Affected Party shall notify the other Party as soon as practicable after the Force Majeure Event ceases or no longer causes the Affected Party to be unable to comply with its obligations under this Contract.
- 29.9. Relief from liability for the Affected Party under Clause L shall end as soon as the Force Majeure Event no longer causes the Affected Party to be unable to comply with its obligations under this Contract and shall not be dependent on the serving of notice under Clause 29.8.

M. <u>TERMINATION AND EXIT MANAGEMENT</u>

30. CUSTOMER TERMINATION RIGHTS

Termination on Material Default

- 30.1. The Customer may terminate this Contract for material Default by issuing a Termination Notice to the Supplier where:
 - 30.1.1 the representation and warranty given by the Supplier pursuant to Clause 4.2.5 (Representations and Warranties) is materially untrue or misleading, and the Supplier fails to provide details of proposed mitigating factors which in the reasonable opinion of the Customer are acceptable;
 - 30.1.2 as a result of any Defaults, the Customer incurs Losses in any Contract Year which exceed 80% (unless stated differently in the Contract Order Form) of the value of the Supplier's aggregate annual liability limit for that Contract Year as set out in Clauses 25.3 and 25.4 (Liability);
 - 30.1.3 the Customer expressly reserves the right to terminate this Contract for material Default, including pursuant to any of the following Clauses: 10.4 (Disruption), 12.5 (Records, Audit Access and Open Book Data), 14.10.6 (ILR Data, Contract Charges and Payment) 15 (Promoting Tax Compliance), 23.18 (Confidentiality), 28.1.7 (Ofsted Monitoring Visits), 28.1.8 (Ofsted Monitoring Visits), 28.6.7 (Ofsted Inadequate Overall) 39.6.2 (Prevention of Fraud and Bribery), Paragraph 1.2.4 of the Annex to Part A and Paragraph 1.2.4 of the Annex to Part B of Contract Schedule 5 (Staff Transfer);

- 30.1.4 the Supplier commits any material Default of this Contract which is not, in the reasonable opinion of the Customer, capable of remedy; and/or
- 30.1.5 the Supplier commits a Default, including a material Default, which in the opinion of the Customer is remediable but has not remedied such Default to the satisfaction of the Customer in accordance with the Rectification Plan Process.
- 30.2. For the purpose of Clause 30.1, a material Default may be a single material Default or a number of Defaults or repeated Defaults (whether of the same or different obligations and regardless of whether such Defaults are remedied) which taken together constitute a material Default.

Termination in Relation to Financial Standing

- 30.3. The Customer may terminate this Contract by issuing a Termination Notice to the Supplier:
 - 30.3.1 in the circumstances set out in Paragraph 6 of Schedule 15 (Financial Distress); or
 - 30.3.2 where in the reasonable opinion of the Customer there is a material detrimental change in the financial standing and/or the credit rating of the Supplier which adversely impacts on the Supplier's ability to supply the Services under this Contract or which could reasonably be expected to have an adverse impact on the Supplier's ability to supply the Services under this Contract.

Termination on Insolvency

30.4. The Customer may terminate this Contract by issuing a Termination Notice to the Supplier where an Insolvency Event affecting the Supplier occurs.

Termination on Change of Control

- 30.5. The Supplier shall notify the Customer immediately in writing and as soon as the Supplier is aware (or ought reasonably to be aware) that it is anticipating, undergoing, undergoes or has undergone a Change of Control and provided such notification does not contravene any Law.
- 30.6. The Supplier shall ensure that any notification made pursuant to Clause 30.5 shall set out full details of the Change of Control including the circumstances suggesting and/or explaining the Change of Control.
- 30.7. The Customer may terminate this Contract by issuing a Termination Notice to the Supplier within six(6) Months of:
 - 30.7.1 being notified in writing that a Change of Control is anticipated or in contemplation or has occurred; or
 - 30.7.2 where no notification has been made, the date that the Customer becomes aware that a Change of Control is anticipated or is in contemplation or has occurred, but shall not be permitted to terminate where an Approval was granted prior to the Change of Control

Termination in relation to IPR Indemnity

30.8. The Customer may terminate this Contract with immediate effect by issuing a Termination Notice to the Supplier where the circumstances in Clause 22.20.1 (IPR Indemnity) arise.

Termination for breach of Regulations

30.9. The Customer may terminate this Contract by issuing a Termination Notice to the Supplier on the occurrence of any of the statutory provisos contained in Regulation 73 (1) (a) to (c) or any applicable relevant successor legislation.

Termination Without Cause

30.10. The Customer shall have the right to terminate this Contract at any time by issuing a Termination Notice to the Supplier giving at least ninety (90) Working Days written notice (unless stated differently in the Contract Order Form).

Termination in Relation to Dynamic Purchasing Agreement

30.11. The Customer may terminate this Contract by issuing a Termination Notice to the Supplier if a Call Off Contract entered under this Contract is terminated for any reason whatsoever.

31. SUPPLIER TERMINATION RIGHTS

Termination on Customer Failure to Pay

- 31.1. The Supplier may, by issuing a Termination Notice to the Customer, terminate this Contract if the Customer fails to pay an undisputed sum due to the Supplier under this Contract which in aggregate exceeds an amount equal to one month's average Contract Charges (unless a different amount has been specified in the Contract Order Form), for the purposes of this Clause 31.1 (the **"Undisputed Sums Limit"**), and the said undisputed sum due remains outstanding for forty (40) Working Days (the **"Undisputed Sums Time Period**") after the receipt by the Customer of a written notice of non-payment from the Supplier specifying:
 - 31.1.1 the Customer's failure to pay; and
 - 31.1.2 the correct overdue and undisputed sum; and
 - 31.1.3 the reasons why the undisputed sum is due; and
 - 31.1.4 the requirement on the Customer to remedy the failure to pay;
- 31.2. If a Termination Notice is issued in accordance with Clause 31.1 this Contract will terminate on the date specified in the Termination Notice (which shall not be less than twenty (20) Working Days from the date of the issue of the Termination Notice).
- 31.3. Termination rights under Clause 31.1 shall not apply where the failure to pay is due to the Customer exercising its right(s) of retention and/or set off.
- 31.4. The Supplier shall not suspend the supply of the Services for failure of the Customer to pay undisputed sums of money (whether in whole or in part).

32. TERMINATION BY EITHER PARTY

32.1. Either Party may, by issuing a Termination Notice to the other Party, terminate this Contract in accordance with Clause 29.6.1 (Force Majeure).

33. PARTIAL TERMINATION, SUSPENSION AND PARTIAL SUSPENSION

- 33.1. Where the Customer has the right to terminate this Contract, the Customer shall be entitled to terminate or suspend all or part of this Contract provided always that, if the Customer elects to terminate or suspend this Contract in part, the parts of this Contract not terminated or suspended can, in the Customer's reasonable opinion, operate effectively to deliver the intended purpose of the surviving parts of this Contract.
- 33.2. Any suspension of this Contract under Clause 33.1 shall be for such period as the Customer may specify and without prejudice to any right of termination which has already accrued, or subsequently accrues, to the Customer.

34. CONSEQUENCES OF EXPIRY OR TERMINATION

Consequences of termination under Clauses 30.1 and 30.2 (Termination on Material Default), 30.3 (Termination in Relation to Financial Standing), 30.11 (Termination in Relation to DPA)

- 34.1. Where the Customer:
 - 34.1.1 terminates (in whole or in part) this Contract under any of Clauses 30.1, 30.2, 30.3 and 30.11; and
 - 34.1.2 then makes other arrangements for the supply of the Services,

the Customer may recover from the Supplier the cost reasonably incurred of making those other arrangements and any additional expenditure incurred by the Customer throughout the remainder of the Contract Period provided that Customer shall take all reasonable steps to mitigate such additional expenditure. No further payments shall be payable by the Customer to the Supplier until the Customer has established the final cost of making those other arrangements

Consequences of termination under Clause 30.10 (Termination without Cause) and 31.1 (Termination on Customer Failure to Pay)

34.2. Where:

- 34.2.1 the Customer terminates (in whole or in part) this Contract under Clause 30.10 (Termination without Cause); or
- 34.2.2 the Supplier terminates this Contract pursuant to Clause 31.1 (Termination on Customer Failure to Pay),

the Customer shall indemnify the Supplier against any reasonable and proven Losses which would otherwise represent an unavoidable loss by the Supplier by reason of the termination of this Contract.

- 34.3. The Supplier shall take all reasonable steps to mitigate Losses identified in accordance with Clause 34.2.
- 34.4. The Supplier shall submit a fully itemised and costed list of such Losses identified in accordance with supporting evidence including such further evidence as the Customer may require, reasonably and actually incurred by the Supplier.
- 34.5. The Customer shall not be liable under Clause 34.2 to pay any sum which:
 - 34.5.1 was claimable under insurance held by the Supplier, and the Supplier has failed to make a claim on its insurance, or has failed to make a claim in accordance with the procedural requirements of the insurance policy; or
 - 34.5.2 when added to any sums paid or due to the Supplier under this Contract, exceeds the total sum that would have been payable to the Supplier if this Contract had not been terminated.

Consequences of termination under Clause 32.1 (Termination for Continuing Force Majeure Event)

34.6. The costs of termination incurred by the Parties shall lie where they fall if either Party terminates or partially terminates this Contract for a continuing Force Majeure Event pursuant to Clause 32.1 (Termination for Continuing Force Majeure Event).

Consequences of Termination for any reason

- 34.7. Save as otherwise expressly provided in this Contract,
 - 34.7.1 termination or expiry of this Contract shall be without prejudice to any rights, remedies or obligations accrued under this Contract prior to termination or expiration and nothing in this Contract shall prejudice the right of either Party to recover any amount outstanding at the time of such termination or expiry; and
 - 34.7.2 termination of this Contract shall not affect the continuing rights, remedies or obligations of the Customer or the Supplier under Clauses 12 (Records, Audit Access & Open Book Data), 22 (Intellectual Property Rights), 23.10 to 23.18 (Confidentiality), (Freedom of Information) 23.22 to 23.36 (Data Protection), 25 (Liability), 34 (Consequences of Expiry or Termination), 40 (Severance), 43 (Entire Agreement), 44 (Third Party Rights) 46 (Dispute Resolution) and 47 (Governing Law and Jurisdiction), and the provisions of Contract Schedule 1 (Definitions), Contract Schedule 3 (Contract Charges, Payment and Invoicing), Contract Schedule 5 (Staff Transfer), Contract Schedule 6 (Dispute Resolution Procedure) and, without limitation to the foregoing, any other provision of this Contract which expressly or by implication is to be performed or observed notwithstanding termination or

expiry shall survive the Contract Expiry Date.

- 34.7.3 On or before the Contract Expiry Date or Termination Date, the Supplier must ensure that all documents or computer records in its possession, custody or control including but not limited to e-portfolios, which contain information relating to the Services including any documents in the possession, custody or control of a Sub-Contractor are made available upon request to the Customer.
- 34.7.4 For the avoidance of doubt, after notice of termination and/or an Insolvency Event affecting the Supplier, the Supplier must not share any information about Learners, including but not limited to Department Data, with another organisation unless the Department provides written authorisation for the Contractor to do so.
- 34.7.5 The Supplier hereby grants the Customer a non-exclusive licence to access the Supplier's Premises from the date of a notice of termination for such periods as may be reasonably necessary to enable the Customer to retrieve the information referred to in Clause 34.7.3. The Customer will exercise the rights provided under this clause where the Supplier has failed to comply with Clause 34.7.3.
- 34.7.6 The Customer reserves the right to retain Contract Charges that would otherwise be paid to the Supplier prior to the Expiry Date or Termination Date and/or to demand repayment of Contract Charges, as relevant, in order to reconcile what has already been paid to the Supplier under this Contract.
- 34.7.7 Where this Contract is terminated or expires, the Customer may elect to take the role of Data Controller to secure and protect Learner Files, Learner Records and/or Evidence Packs, including e-portfolios until the Learner information can be transferred to a new provider, the Learner, or destroyed in accordance with defined retention periods. If the Customer elects to assume this role, the Supplier shall co-operate fully to facilitate this.
- 34.7.8 On expiry or termination of this Contract for any reason, the Supplier shall do its utmost to minimise any disruption to Learners and shall co-operate fully with any reasonable requests made by the Customer relating to this. For the avoidance of doubt the

Department will be entitled to request that where the Supplier cannot complete Learners that it will co-operate in transferring the Learners to a new provider even if this is prior to the Termination Date or Expiry Date of this Contract and the Supplier's Exit Plan should reflect this. The Customer will not be liable for any costs prior to or after the Termination Date or Expiry Date incurred by the Supplier in complying with this Clause 34.7.8.

34.7.9 If the Supplier does not co-operate with the Customer in relation to exit and the Customer incurs additional expenditure of any description as a result, the Customer reserves the right to require the Supplier to reimburse the Customer for this additional expenditure.

N. MISCELLANEOUS AND GOVERNING LAW

35. COMPLIANCE

Health and Safety

- 35.1. The Supplier shall perform its obligations under this Contract (including those in relation to the Services) in accordance with:
 - 35.1.1 all applicable Law regarding health and safety; and
 - 35.1.2 the Customer's health and safety policy (as provided to the Supplier from time to time) whilst at the Customer Premises.
- 35.2. Each Party shall promptly notify the other of as soon as possible of any health and safety incidents or material health and safety hazards at the Customer Premises of which it becomes aware and which relate to or arise in connection with the performance of this Contract

35.3. While on the Customer Premises, the Supplier shall comply with any health and safety measures implemented by the Customer in respect of Supplier Personnel and other persons working there and any instructions from the Customer on any necessary associated safety measures.

Equality and Diversity

- 35.4. The Supplier shall:
 - 35.4.1 perform its obligations under this Contract (including those in relation to the Services) in accordance with:
 - (a) all applicable equality Law (whether in relation to race, sex, gender reassignment, age, disability, sexual orientation, religion or belief, pregnancy, maternity or otherwise);
 - (b) the Authority's equality and diversity policy as provided to the Supplier from time to time; and
 - (c) any other requirements and instructions which the Authority reasonably imposes in connection with any equality obligations imposed on the Authority at any time under applicable equality Law; and
 - 35.4.2 take all necessary steps, and inform the Authority of the steps taken, to prevent unlawful discrimination designated as such by any court or tribunal, or the Equality and Human Rights Commission or (any successor organisation).

Finance Act

35.5. The Supplier shall comply with the provisions of section 182 of the Finance Act 1989.

Environmental Requirements

- 35.6. The Supplier shall, when working on the Sites, perform its obligations under this Contract in accordance with the Environmental Policy of the Customer.
- 35.7. The Customer shall provide a copy of its written Environmental Policy (if any) to the Supplier upon the Supplier's written request.

36. ASSIGNMENT AND NOVATION

- 36.1. The Supplier shall not assign, novate, Sub-Contract or otherwise dispose of or create any trust in relation to any or all of its rights, obligations or liabilities under this Contract or any part of it without Approval.
- 36.2. The Customer may assign, novate or otherwise dispose of any or all of its rights, liabilities and obligations under this Contract or any part thereof to:
 - 36.2.1 any other Contracting Authority; or
 - 36.2.2 any other body established by the Crown or under statute in order substantially to perform any of the functions that had previously been performed by the Customer; or

36.2.3 any private sector body which substantially performs the functions of the Customer, and the Supplier shall, at the Customer's request, enter into a novation agreement in such form as the Customer shall reasonably specify in order to enable the Customer to exercise its rights pursuant to this Clause 36.2.

- 36.3. A change in the legal status of the Customer shall not, subject to Clause 36.4 affect the validity of this Contract and this Contract shall be binding on any successor body to the Customer.
- 36.4. If the Customer assigns, novates or otherwise disposes of any of its rights, obligations or liabilities under this Contract to a private sector body in accordance with Clause 36.2.3 (the "Transferee" in the rest of this Clause 36.4 the right of termination of the Customer in Clause 30.4 (Termination on Insolvency) shall be available to the Supplier in the event of insolvency of the Transferee (as if the references to Supplier in Clause 30.4 (Termination on Insolvency) and to Supplier or DPA Guarantor or Contract Guarantor in the definition of Insolvency Event were references to the Transferee).

37. WAIVER AND CUMULATIVE REMEDIES

- 37.1. The rights and remedies under this Contract may be waived only by notice and in a manner that expressly states that a waiver is intended. A failure or delay by a Party in ascertaining or exercising a right or remedy provided under this Contract or by law shall not constitute a waiver of that right or remedy, nor shall it prevent or restrict the further exercise of that or any other right or remedy. No single or partial exercise of any right or remedy shall prevent or restrict the further exercise of that or any other right or any other right or remedy.
- 37.2. Unless otherwise provided in this Contract, rights and remedies under this Contract are cumulative and do not exclude any rights or remedies provided by law, in equity or otherwise.

38. RELATIONSHIP OF THE PARTIES

38.1. Except as expressly provided otherwise in this Contract, nothing in this Contract, nor any actions taken by the Parties pursuant to this Contract, shall create a partnership, joint venture or relationship of employer and employee or principal and agent between the Parties, or authorise either Party to make representations or enter into any commitments for or on behalf of any other Party.

39. PREVENTION OF FRAUD AND BRIBERY

- 39.1. The Supplier represents and warrants that neither it, nor to the best of its knowledge any Supplier Personnel, have at any time prior to the Contract Commencement Date:
 - 39.1.1 committed a Prohibited Act or been formally notified that it is subject to an investigation or prosecution which relates to an alleged Prohibited Act; and/or
 - 39.1.2 been listed by any government department or agency as being debarred, suspended, proposed for suspension or debarment, or otherwise ineligible for participation in government procurement programmes or contracts on the grounds of a Prohibited Act.
- **39.2.** The Supplier shall not during the Contract Period:
 - 39.2.1 commit a Prohibited Act; and/or
 - 39.2.2 do or suffer anything to be done which would cause the Customer or any of the Customer's employees, consultants, contractors, sub-contractors or agents to contravene any of the Relevant Requirements or otherwise incur any liability in relation to the Relevant Requirements.
- 39.3. The Supplier shall during the Contract Period:
 - 39.3.1 establish, maintain and enforce, and require that its Sub-Contractors establish, maintain and enforce, policies and procedures which are adequate to ensure compliance with the Relevant Requirements and prevent the occurrence of a Prohibited Act;
 - 39.3.2 keep appropriate records of its compliance with its obligations under Clause 39.3.1 and make such records available to the Customer on request;
 - 39.3.3 have in place reasonable prevention measures (as defined in sections 45(3) and 46(4) of the Criminal Finance Act 2017) to ensure that Associated Persons of the Supplier do not commit tax evasion facilitation offences as defined under that Act;
 - 39.3.4 if so required by the Customer, within twenty (20) Working Days of the Contract Commencement Date, and annually thereafter, certify to the Customer in writing that the Supplier and all persons associated with it or its Sub-Contractors or other persons who are supplying the Services in connection with this Contract are compliant with the Relevant Requirements. The Supplier shall provide such supporting evidence of compliance as the Customer may reasonably request;
 - 39.3.5 take account of any guidance about preventing facilitation of tax evasion offences which may be published and updated in accordance with Section 47 of the Criminal Finances Act 2017 and

- 39.3.6 have, maintain and where appropriate enforce an anti-bribery policy (which shall be disclosed to the Customer on request) to prevent it and any Supplier Personnel or any person acting on the Supplier's behalf from committing a Prohibited Act.
- 39.4. The Supplier shall immediately notify the Customer in writing if it becomes aware of any breach of Clause 39.1 and/or Clause 39.2, or has reason to believe that it has or any of the Supplier Personnel have:
 - 39.4.1 been subject to an investigation or prosecution which relates to an alleged Prohibited Act;
 - 39.4.2 been listed by any government department or agency as being debarred, suspended, proposed for suspension or debarment, or otherwise ineligible for participation in government procurement programmes or contracts on the grounds of a Prohibited Act; and/or
 - 39.4.3 received a request or demand for any undue financial or other advantage of any kind in connection with the performance of this Contract or otherwise suspects that any person

or Party directly or indirectly connected with this Contract has committed or attempted to commit a Prohibited Act.

- 39.5. If the Supplier makes a notification to the Customer pursuant to Clause 39.4, the Supplier shall respond promptly to the Customer's enquiries, co-operate with any investigation, and allow the Customer to audit any books, records and/or any other relevant documentation in accordance with Clause 12 (Records, Audit Access and Open Book Data).
- 39.6. If the Supplier is in Default under Clauses 39.1 and/or 39.2, the Customer may by notice:
 - 39.6.1 require the Supplier to remove from performance of this Contract any Supplier Personnel whose acts or omissions have caused the Default; or
 - 39.6.2 immediately terminate this Contract.
- 39.7. Any notice served by the Customer under Clause 39.4 shall specify the nature of the Prohibited Act, the identity of the Party who the Customer believes has committed the Prohibited Act and the action that the Customer has elected to take (including, where relevant, the date on which this Contract shall terminate).

40. SEVERANCE

- 40.1. If any provision of this Contract (or part of any provision) is held to be void or otherwise unenforceable by any court of competent jurisdiction, such provision (or part) shall to the extent necessary to ensure that the remaining provisions of this Contract are not void or unenforceable be deemed to be deleted and the validity and/or enforceability of the remaining provisions of this Contract shall not be affected.
- 40.2. In the event that any deemed deletion under Clause 40.1 is so fundamental as to prevent the accomplishment of the purpose of this Contract or materially alters the balance of risks and rewards in this Contract, either Party may give notice to the other Party requiring the Parties to commence good faith negotiations to amend this Contract so that, as amended, it is valid and enforceable, preserves the balance of risks and rewards in this Contract and, to the extent that is reasonably practicable, achieves the Parties' original commercial intention.
- 40.3. If the Parties are unable to resolve the Dispute arising under Clause 40 within twenty (20) Working Days of the date of the notice given pursuant to Clause 40.2, this Contract shall automatically terminate with immediate effect. The costs of termination incurred by the Parties shall lie where they fall if this Contract is terminated pursuant to Clause 40.

41. MODERN SLAVERY

41.1. The Supplier:

- 41.1.1 shall not use, nor allow its Sub-Contractors to use forced, bonded or involuntary prison labour;
- 41.1.2 shall not require any Supplier Personnel or the personnel of any Sub-Contractors to lodge deposits or identity papers with their employer and shall be free to leave their employer after reasonable notice;

- 41.1.3 warrants and represents that it has not been convicted of any slavery or human trafficking offences anywhere around the world;
- 41.1.4 warrants that to the best of its knowledge it is not currently under investigation, inquiry or enforcement proceedings in relation to any allegation of slavery or human trafficking offences anywhere around the world;
- 41.1.5 shall make reasonable enquires to ensure that its officers, employees and sub-contractors have not been convicted of slavery or human trafficking offences anywhere around the world;
- 41.1.6 shall have and maintain throughout the Contract Period its own policies and procedures to ensure its compliance with the Modern Slavery Act 2015 and include in its contracts with its Sub-Contractors anti-slavery and human trafficking provisions;
- 41.1.7 shall implement due diligence procedures to ensure that there is no slavery or human trafficking in any part of its supply chain performing obligations under the Contract;
- 41.1.8 shall prepare and deliver to the Customer, an annual slavery and human trafficking report setting out the steps it has taken to ensure that slavery and human trafficking is not taking place in any of its supply chains or in any part of its business;
- 41.1.9 shall not use, nor allow its employees or Sub-Contractors to use physical abuse or discipline, the threat of physical abuse, sexual or other harassment and verbal abuse or other forms of intimidation of its employees or sub-contractors;
- 41.1.10 shall not use or allow child or slave labour to be used by its Sub-contractors; and
- 41.1.11 shall report the discovery or suspicion of any slavery or trafficking by it or its subcontractors to the Customer and the Modern Slavery Helpline (modernslaveryhelpline.org).
- 41.2. In addition to any other rights under the Contract, the Customer may instruct the Supplier to carry out such an audit of any Sub-Contractor by an independent third party and, if so instructed, the Supplier shall deliver a report to the Customer within ninety (90) days of such instruction.
- 41.3. If the Supplier notifies the Customer pursuant to Clause 41.5 it shall respond promptly to the Customer's enquiries, co-operate with any investigation, and allow the Customer to audit any books, records and/or any other relevant documentation in accordance with the Contract.
- 41.4. If the Supplier is in Default under Clause 41.1 the Customer may by notice:
 - 41.4.1 require the Supplier to remove from performance of the Contract any Sub-Contractor, Supplier Personnel or other persons associated with it whose acts or omissions have caused the Default; or
 - 41.4.2 immediately terminate the Contract.

Whistleblowing

- 41.5. As soon as it is aware of it the Supplier and Supplier Personnel must report to the Customer any actual or suspected breach of;
 - 41.5.1 Law;
 - 41.5.2 Clauses 41.1
- 41.6. The Supplier must not retaliate against any of the Supplier Personnel who in good faith reports a breach listed in this Clause to the Customer or a Prescribed Person.

42. FURTHER ASSURANCES

42.1. Each Party undertakes at the request of the other, and at the cost of the requesting Party to do all acts and execute all documents which may be necessary to give effect to the meaning of this Contract.

43. ENTIRE AGREEMENT

- 43.1. This Contract and the documents referred to in it constitute the entire agreement between the Parties in respect of the matter and supersede and extinguish all prior negotiations, course of dealings or agreements made between the Parties in relation to its subject matter, whether written or oral.
- 43.2. Neither Party has been given, nor entered into this Contract in reliance on, any warranty, statement, promise or representation other than those expressly set out in this Contract.
- 43.3. Nothing in Clause 43 shall exclude any liability in respect of misrepresentations made fraudulently.

44. THIRD PARTY RIGHTS

- 44.1. The provisions of paragraphs 2.1 and 2.6 of Part A, paragraphs 2.1, 2.6, 3.1 and 3.3 of Part B, paragraphs 2.1 and 2.3 of Part C and paragraphs and 1.4, 2.3 and 2.8 of Part D of Contract Schedule 5 (Staff Transfer) (together "Third Party Provisions") confer benefits on persons named in such provisions other than the Parties (each such person a "Third Party Beneficiary") and are intended to be enforceable by Third Parties Beneficiaries by virtue of the CRTPA.
- 44.2. Subject to Clause 44.1, a person who is not a Party to this Contract has no right under the CTRPA to enforce any term of this Contract but this does not affect any right or remedy of any person which exists or is available otherwise than pursuant to that Act.
- 44.3. No Third Party Beneficiary may enforce, or take any step to enforce, any Third Party Provision without the prior written consent of the Customer, which may, if given, be given on and subject to such terms as the Customer may determine.
- 44.4. Any amendments or modifications to this Contract may be made, and any rights created under Clause 44.1 may be altered or extinguished, by the Parties without the consent of any Third Party Beneficiary.

45. NOTICES

- 45.1. Except as otherwise expressly provided within this Contract, any notices sent under this Contract must be in writing. For the purpose of Clause 45, an e-mail is accepted as being "in writing".
- 45.2. Subject to Clause 45.3, the following table sets out the method by which notices may be served under this Contract and the respective deemed time and proof of service:

Manner of delivery	Deemed time of delivery	Proof of Service
Email (Subject to Clauses 45.3 and 45.4)	9.00am on the first Working Day after sending	Dispatched as a pdf attachment to an e-mail to the correct e-mail address without any error message
Personal delivery	On delivery, provided delivery is between 9.00am and 5.00pm on a Working Day. Otherwise, delivery will occur at 9.00am on the next Working Day	Properly addressed and delivered as evidenced by signature of a delivery receipt

Royal Mail Signed For™ 1 st Class or other prepaid, next Working Day service providing proof of delivery	At the time recorded by the delivery service, provided that delivery is between 9.00am and 5.00pm on a Working Day. Otherwise, delivery will occur at 9.00am on the same Working Day (if delivery before 9.00am) or on the next Working Day (if after 5.00pm)	Properly addressed prepaid and delivered as evidenced by signature of a delivery receipt
--	--	--

45.3. The following notices may only be served as an attachment to an email if the original notice is then sent to the recipient by personal delivery or Royal Mail Signed For[™] 1st Class or other prepaid in the manner set out in the table in Clause 45.2:

45.3.1 any Termination Notice (Clause 30 (Customer Termination Rights)), 45.3.2

any notice in respect of:

- (a) partial termination, suspension or partial suspension (Clause 33 (Partial Termination, Suspension and Partial Suspension))
- (b) waiver (Clause 37 (Waiver and Cumulative Remedies))
- (c) Default ; and
- (d) Any Dispute Notice.
- 45.4. Failure to send any original notice by personal delivery or recorded delivery in accordance with Clause 45.3 shall invalidate the service of the related e-mail transmission. The deemed time of delivery of such notice shall be the deemed time of delivery of the original notice sent by personal delivery or Royal Mail Signed For[™] 1st Class delivery (as set out in the table in) or, if earlier, the time of response or acknowledgement by the other Party to the email attaching the notice.
- 45.5. Clause 45 does not apply to the service of any proceedings or other documents in any legal action or, where applicable, any arbitration or other method of dispute resolution (other than the service of a Dispute Notice under the Dispute Resolution Procedure).
- 45.6. For the purposes of Clause 45, the address and email address of each Party shall be as specified in the Contract Order Form.

46. DISPUTE RESOLUTION

- 46.1. The Parties shall resolve Disputes arising out of or in connection with this Contract in accordance with the Dispute Resolution Procedure.
- 46.2. The Supplier shall continue to provide the Services in accordance with the terms of this Contract until a Dispute has been resolved.

47. GOVERNING LAW AND JURISDICTION

- 47.1. This Contract and any issues, Disputes or claims (whether contractual or non-contractual) arising out of or in connection with it or its subject matter or formation shall be governed by and construed in accordance with the laws of England and Wales.
- 47.2. Subject to Clause 46 (Dispute Resolution) and Contract Schedule 6 (Dispute Resolution Procedure) (including the Customer's right to refer the Dispute to arbitration), the Parties agree that the courts of England and Wales (unless stated differently in the Contract Order Form) shall have exclusive jurisdiction to settle any Dispute or claim (whether contractual or non-contractual) that arises out of or in connection with this Contract or its subject matter or formation.

CONTRACT SCHEDULE 1: DEFINITIONS

1. In accordance with Clause 1 (Definitions and Interpretations), in this Dynamic Purchasing System Agreement including its Recitals the following expressions shall have the following meanings:

including its Recitals the followin "Academic Year"	g expressions shall have the following meanings: means 1 August to 31 July;		
"Achieve"	means I August to 31 July; means in respect of a Test, to successfully pass such Test without any Test Issues in accordance with the Test Strategy Plan and in respect of a Milestone, the issue of a Satisfaction Certificate in respect of that Milestone and " Achieved ", " Achieving " and " Achievement " shall be construed accordingly;		
"Acquired Rights Directive"	the European Council Directive 2001/23/EC on the approximation of the laws of the member states of the European Union relating to the safeguarding of employees' rights in the event of transfers of undertakings, businesses or parts of undertakings or businesses, as amended or re-enacted from time to time;		
"Additional Clauses"	means the additional Clauses in Contract Schedule 12 (Alternative and/or Additional Clauses) and any other additional Clauses set out in the Contract Order Form or elsewhere in this Contract;		
"Affected Party"	means the party seeking to claim relief in respect of a Force Majeure;		
"Affiliates"	has the meaning given to it in DPA Schedule 1 (Definitions);		
"Alternative Clauses"	means the alternative Clauses in Contract Schedule 12 (Alternative and/or Additional Clauses) and any other alternative Clauses set out in the Contract Order Form or elsewhere in this Contract;		
"Approval"	means the prior written consent of the Customer and "Approve" and "Approved" and "Approved" shall be construed accordingly;		
"Approved Sub-Licensee"	means any of the following:		
	a) a Central Government Body;		
	 any third party providing services to a Central Government Body; and/or 		
	 any body (including any private sector body) which performs or carries on any of the functions and/or activities that previously had been performed and/or carried on by the Customer; 		
"Associated Persons"	has the meaning as set out in Section 44(4) of the Criminal Finance Act 2017;		
"Auditor"	means:		
	a) the Customer's internal and external auditors;		
	b) the Customer's statutory or regulatory auditors;		
	 c) the Comptroller and Auditor General, their staff and/or any appointed representatives of the National Audit Office; 		
	d) HM Treasury or the Cabinet Office;		
	 e) any party formally appointed by the Customer to carry out audit or similar review functions; and 		
	f) successors or assigns of any of the above.		
"Authority"	has the meaning given to it in DPA Schedule 1 (Definitions);		

"BACS"	means the Bankers' Automated Clearing Services, which is a scheme for the electronic processing of financial transactions within the United Kingdom;		
"Call for Competition Procedure"	means the competition procedure described in paragraph 2 of DPA Schedule 5 (Call for Competition Procedure);		
"Category"	means the Institute for Apprenticeships and Technical Education occupational routes as referred to at <u>Occupational Maps / Institute</u> for Apprenticeships and Technical Education;		
"Comptroller and Auditor General"	means the head of the National Audit Office;		
"Contract Commencement Date"	means the date of commencement of this Contract set out in the Contract Order Form;		
"Contract "	means this contract between the Customer and the Supplier (entered into pursuant to the provisions of the DPA), which consists of the terms set out in the Contract Order Form and the Contract Terms;		
"Contract Charges"	means the prices (inclusive of any Milestone Payments and exclusive of any applicable VAT), payable to the Supplier by the Customer under this Contract, as set out in Annex 1 of Contract Schedule 3 (Contract Charges, Payment and Invoicing), for the full and proper performance by the Supplier of its obligations under this Contract less any Deductions;		
"Contract Finder"	means the online portal administered by the UK Government that lets economic operators Contracts Finder lets you search for information about contracts worth over £10,000 with the UK Government and its agencies.		
"Contract Period"	means the term of this Contract from the Contract Commencement Date until the Contract Expiry Date;		
"Contract Year"	means a consecutive period of twelve (12) Months commencing on the Contract Commencement Date or each anniversary thereof;		
"Contract Expiry Date"	 means: a) the end date of the Contract Initial Period or any Contract Extension Period; or b) if this Contract is terminated before the date specified in (a) above, the earlier date of termination of this Contract; 		
"Contract Extension Period"	means such period or periods up to a maximum of the number of years in total as may be specified by the Customer, pursuant to Clause 5.2 and in the Contract Order Form;		
"Contract Initial Period"	means the initial term of this Contract from the Contract Commencement Date to the end date of the initial term stated in the Contract Order Form;		
"Contract Order Form"	means the order form applicable to this Contract containing details of the parties and the service to be provided that is generated at the end of the DPA process or otherwise provided;		
"Call for Competition Procedure"	has the meaning given to it in DPA Schedule 1 (Definitions);		
"Contract Schedule"	means a schedule to this Contract;		

"Contract Tender"	means the tender submitted by the Supplier in response to the Customer's Statement of Requirements following a Call for Competition Procedure and set out at Contract Schedule 5 (Contract Tender);
"Contract Terms"	means the terms applicable to and set out in this Contract;
"Central Government Body"	has the meaning given to it in DPA Schedule 1 (Definitions);
"Change in Law"	means any change in Law which impacts on the supply of the Services and performance of the Contract which comes into force after the Contract Commencement Date;
"Change of Control"	has the meaning given to it in DPA Schedule 1 (Definitions);
"Charging Structure"	means the structure to be used in the establishment of the charging model which is applicable to the Contract, which is set out in Contract Schedule 3 (Contract Prices and Charging Structure);
"Commercially Sensitive Information"	means the Confidential Information listed in the Contract Order Form (if any) comprising of commercially sensitive information relating to the Supplier, its IPR or its business or which the Supplier has indicated to the Customer that, if disclosed by the Customer, would cause the Supplier significant commercial disadvantage or material financial loss;
"Comparable Supply"	means the supply of Services to another customer of the Supplier that are the same or similar to the Services;
"Confidential Information"	means the Customer's Confidential Information and/or the Supplier's Confidential Information, as the context specifies;
"Call Off Contract" or "Contract"	means this contract;
"Contract Charges"	means the charges raised under or in connection with this Contract from time to time, which shall be calculated in a manner that is consistent with the Charging Structure;
"Contract Commencement Date	means the date of commencement of the Contract as written at the top of the Contract;
"Contract Monitoring Plan" or "CMP" and "Learner Datasheet" or "LD"	means a submission by the Supplier that incorporates the information required in the form of contract monitoring plan set out at Schedule 2 Part 3 (Contract Management), Annex 1;
"Contracting Authority"	means the Authority, the Customer and any other bodies listed in the OJEU Notice;
"Control"	has the meaning given to it in DPA Schedule 1 (Definitions);
"Controller"	take the meaning given in the GDPR
"Conviction"	means other than for minor road traffic offences, any previous or pending prosecutions, convictions, cautions and binding over orders (including any spent convictions as contemplated by section 1(1) of the Rehabilitation of Offenders Act 1974 by virtue of the exemptions specified in Part II of Schedule 1 of the Rehabilitation of Offenders Act 1974 (Exemptions) Order 1975 (SI 1975/1023) or any replacement or amendment to that Order, or being placed on a list kept pursuant to section 1 of the Protection of Children Act 1999 or being placed on a list kept pursuant to the Safeguarding Vulnerable Groups Act 2006;

the following costs (without double recovery) to the extent that they are reasonably and properly incurred by the Supplier in providing the Services:
a) the cost to the Supplier or the Key Sub-Contractor (as the context requires), calculated per Man Day, of engaging the Supplier Personnel, including:

"Costs"

- i) base salary paid to the Supplier Personnel; ii)
- employer's national insurance contributions; iii)

pension contributions; iv) car

allowances;

- v) any other contractual employment benefits; vi) staff training; vii) work place accommodation;
 viii) work place IT equipment and tools reasonably necessary to provide the Services (but not including items included within limb (b) below); and
- ix) reasonable recruitment costs, as agreed with the Customer;
- a) costs incurred in respect of those Supplier Assets which are detailed on the Registers and which would be treated as capital costs according to generally accepted accounting principles within the UK, which shall include the cost to be charged in respect of Supplier Assets by the Supplier to the Customer or (to the extent that risk and title in any Supplier Asset is not held by the Supplier) any cost actually incurred by the Supplier in respect of those Supplier Assets;
- b) operational costs which are not included within (a) or (b) above, to the extent that such costs are necessary and properly incurred by the Supplier in the provision of the Services;

SubContractors for the provision of the Services (or any of them);

"Crown"	has the meaning given to it in DPA Schedule 1 (Definitions);
"Crown Body"	has the meaning given to it in DPA Schedule 1 (Definitions);
"CRTPA"	has the meaning given to it in DPA Schedule 1 (Definitions);

"Customer" means the customer(s) identified in the Contract Order Form;

"Customer Assets" means the Customer's infrastructure, data, software, materials, assets, equipment or other property owned by and/or licensed or leased to the Customer and which is or may be used in connection with the provision of the Services; "Customer Background IPR" means:

	a)	IPRs the	owned Contrac	,	the	Customer	before
			r's Know-	-	-	tion, software, pr	
	b)	IPRs creat; and/or	ted by the	e Custom	er indep	endently of this Co	ontract
	c)	Crown Co otherwise				ble to the Supplier	
"Customer Data"	means	s data that	refers to	or belon	gs to the	Customer;	
"Customer Premises"		•				cupied by the Cu y the Supplier	

46

"Customer Property"	equip	s the property, other than real property and IPR, including any ment issued or made available to the Supplier by the Customer inection with this Contract ;
"Customer Representative"		s the representative appointed by the Customer from time to n relation to this Contract;
"Customer Responsibilities"	Order	s the responsibilities of the Customer set out in the Contract Form or agreed in writing between the Parties from time to n connection with this Contract ;
"Customer's Confidential Information"	tł ri (i (b) ar	s: II Personal Data and any information, however it is conveyed, nat relates to the business, affairs, developments, property ghts, trade secrets, Know-How and IPR of the Customer ncluding all Customer Background IPR and Project Specific IPR); ny other information clearly designated as being confidential whether or not it is marked "confidential") or which ought
	re	easonably be considered confidential which comes (or has ome) to the Customer's attention or into the Customer's ossession in connection with this Contract ; and
	inforn	nation derived from any of the above;
"Data Loss Event", Data Protection Impact Assessment", "Data Protection Officer", "Data Subject" "Data Subject Request", "Personal Data", "Processor" and "Processor Personnel"		have the meaning given to the relevant term in the Data ction Act 2018;
"Data Protection Legislation"		
	mean (i)	s: The General Data Protection Regulation 2016/679 ("GDPR"), the law enforcement directive 2016/680 ("LED") and any applicable national implementing laws as amended from time to time;
	(ii)	The DPA 2018 to the extent that it relates to processing of personal data and privacy;
	(iii)	all applicable Law about the processing of personal data and privacy;
"Deductions"		s any form of Service Credits, Delay Payments or any other ction which the Customer is paid or is payable under this act;
"Default"	limite terms omiss SubCo conne	s any breach of the obligations of the Supplier (including but not d to including abandonment of this Contract in breach of its) or any other default (including material Default), act, ion, negligence or statement of the Supplier, of its patractors or any Supplier Personnel howsoever arising in ection with or in relation to the subject-matter of this Contract or respect of which the Supplier is liable to the Customer;
"Deliverable"	to be	s an item or feature in the supply of the Services delivered or delivered by the Supplier at any other stage during the rmance of this Contract;

"Delivery"	means delivery in accordance with the terms of this Contract as confirmed by the issue by the Customer of a Satisfaction Certificate in respect of the relevant Milestone thereof (if any) or otherwise in accordance with this Contract and accepted by the Customer and " Deliver " and " Delivered " shall be construed accordingly;		
"Delivery Plan"	means the plan setting out key timings and trajectories for the Supplier's delivery during the Contract Period, as agreed by the Customer and including but not limited to		
	(a) the Delivery Plan Trajectory Points; and		
	(b) how many Learners will start or Complete at each of those Delivery Plan Trajectory Points.		
	as is set out further in Schedule 2 Part 2 (Performance Management and Part 3 (Contract Management);;		
"Disclosing Party"	has the meaning given to it in Clause 23.10 to 23.18 (Confidentiality);		
"Dispute"	means any dispute, difference or question of interpretation arising out of or in connection with this Contract, including any dispute, difference or question of interpretation relating to the Services, failure to agree in accordance with the Variation Procedure or any matter where this Contract directs the Parties to resolve an issue by reference to the Dispute Resolution Procedure;		
"Dispute Notice"	means a written notice served by one Party on the other stating that the Party serving the notice believes that there is a Dispute;		
"Dispute Resolution Procedure"	means the dispute resolution procedure set out in Contract Schedule 6 (Dispute Resolution Procedure);		
"Dynamic Purchasing Agreement" or "DPA"	means the Dynamic Purchasing Agreement between the Authority and the Supplier, referred to in the Contract Order Form;		
"DPA Schedule"	means a schedule to the DPA;		
"Documentation"	means all documentation as: a) is required to be supplied by the Supplier to the Customer under this Contract;		
	 would reasonably be required by a competent third party capable of Good Industry Practice contracted by the Customer to develop, configure, build, deploy, run, maintain, upgrade and test the individual systems that provide the Services; 		
	c) is required by the Supplier in order to provide the Services; and/or		
	 has been or shall be generated for the purpose of providing the Services; 		
"DOTAS"	has the meaning given to it in DPA Schedule 1 (Definitions);		
"DPA 2018"	means Data Protection Act 2018;		
"Due Diligence Information"	means any information supplied to the Supplier by or on behalf of the Customer prior to the Contract Commencement Date;		
"Employer"	means the person or organisation that enters into a contract of employment with a person who becomes a Learner on a Skills Bootcamp Course delivered by or on behalf of the Supplier;		

"Employer-Provider"	means an employer that delivers one or more Skills Bootcamp Courses to Learners that are its employees;		
"Employee Liabilities"	means all claims, actions, proceedings, orders, demands, complaints, investigations (save for any claims for personal injury which are covered by insurance) and any award, compensation, damages, tribunal awards, fine, loss, order, penalty, disbursement, payment made by way of settlement and costs, expenses and legal costs reasonably incurred in connection with a claim or investigation including in relation to the following:		
	 a) redundancy payments including contractual or enhanced redundancy costs, termination costs and notice payments; b) unfair, wrongful or constructive dismissal compensation; 		
	 compensation for discrimination on grounds of sex, race, disability, age, religion or belief, gender reassignment, marriage or civil partnership, pregnancy and maternity or sexual orientation or claims for equal pay; 		
	 compensation for less favourable treatment of part time workers or fixed term employees; 		
	a) outstanding debts and unlawful deduction of wages including any PAYE and National Insurance Contributions in relation to payments made by the Customer or the Replacement Supplier to a Transferring Supplier Employee which would have been payable by the Supplier or the Sub-Contractor if such payment should have been made prior to the Service Transfer Date;		
	f) claims whether in tort, contract or statute or otherwise;		
	 g) any investigation by the Equality and Human Rights Commission or other enforcement, regulatory or supervisory body and of implementing any requirements which may arise from such investigation; 		
"Employment Regulations"	means the Transfer of Undertakings (Protection of Employment) Regulations 2006 (SI 2006/246) as amended or replaced or any other Regulations implementing the Acquired Rights Directive;		
"Equality and Human Rights Commission"	means the statutory and non-departmental public body established under the Equality Act 2006 as the national equality body for Great Britain;		
"Evidence Pack"	means a file or dossier of information, either in physical or electronic form, that evidences a Learner's progress on a Skills Bootcamp Course;		
"Environmental Information Regulations or EIRs"	has the meaning given to it in DPA Schedule 1 (Definitions);		
"Environmental Policy"	means to conserve energy, water, wood, paper and other resources, reduce waste and phase out the use of ozone depleting substances and minimise the release of greenhouse gases, volatile organic compounds and other substances damaging to health and the environment, including any written environmental policy of the Customer;		

"Exit Plan"	means the exit plan described in paragraph 5 of Contract Schedule 10 (Exit Management);
"Expedited Dispute Timetable"	means the timetable set out in paragraph 5 of Contract Schedule 6 (Dispute Resolution Procedure);
"FOIA"	has the meaning given to it in DPA Schedule 1 (Definitions);
"Force Majeure"	means any event outside the reasonable control of either Party affecting its performance of its obligations under this Contract, arising from acts, events, omissions, happenings or non-happenings beyond its reasonable control and which are not attributable to any wilful act, neglect or failure to take reasonable preventative actions by that Party, including riots, war terrorism, acts of government, local government or regulatory bodies, fire, flood, earthquake, or other natural disaster but excluding any industrial dispute relating to the Supplier or Supplier Personnel or any other failure in the Supplier's or Sub-contractor's supply chain;;
"Force Majeure Notice"	means a written notice served by the Affected Party on the other Party stating that the Affected Party believes that there is a Force Majeure Event;
"Former Supplier"	means a supplier supplying the Services to the Customer before the Relevant Transfer Date that are the same as or substantially similar to the Services (or any part of the Services) and shall include any subcontractor of such supplier (or any sub-contractor of any such subcontractor);
"Fraud"	has the meaning given to it in DPA Schedule 1 (Definitions);
"General Anti-Abuse Rule"	has the meaning given to it in DPA Schedule 1 (Definitions);
"General Change in Law"	Has the meaning given to it in DPA Schedule 1 (Definitions);
"Geographical Filter"	refers to three types of area in England: (i) Regions, (ii) Mayoral Combined Authority areas, and (iii) Upper Tier Local Authority areas;
"Good Industry Practice" "Government"	has the meaning given to it in DPA Schedule 1 (Definitions); has the meaning given to it in DPA Schedule 1 (Definitions);
"Government Procurement Card"	means the Government's preferred method of purchasing and payment for low value goods or services https://www.gov.uk/government/publications/government procurement-card2;
"Halifax Abuse Principle"	has the meaning given to it in DPA Schedule 1 (Definitions);
"HMRC"	means His Majesty's Revenue and Customs;
"Holding Company"	has the meaning given to it in DPA Schedule 1 (Definitions);
"ICT Policy"	means the Customer's policy in respect of information and communications technology, referred to in the Contract Order Form, which is in force as at the Contract Commencement Date (a copy of which has been supplied to the Supplier), as updated from time to time;
"Implementation Plan"	means the plan devised by the Supplier as to how it will mobilise the Services;
"Information"	has the meaning given to it in DPA Schedule 1 (Definitions);

"Insolvency Event"

means, in respect of the Supplier or DPA Guarantor or Contract Guarantor (as applicable):

- a proposal is made for a voluntary arrangement within Part I of the Insolvency Act 1986 or of any other composition scheme or arrangement with, or assignment for the benefit of, its creditors; or
- a shareholders' meeting is convened for the purpose of considering a resolution that it be wound up or a resolution for its winding-up is passed (other than as part of, and exclusively for the purpose of, a bona fide reconstruction or amalgamation); or
- c) a petition is presented for its winding up (which is not dismissed within fourteen (14) Working Days of its service) or an application is made for the appointment of a provisional liquidator or a creditors' meeting is convened pursuant to section 98 of the Insolvency Act 1986; or
- a receiver, administrative receiver or similar officer is appointed over the whole or any part of its business or assets; or
- e) an application order is made either for the appointment of an administrator or for an administration order, an administrator is appointed, or notice of intention to appoint an administrator is given; or
- f) it is or becomes insolvent within the meaning of section 123 of the Insolvency Act 1986; or
- g) being a "small company" within the meaning of section 382(3) of the Companies Act 2006, a moratorium comes into force pursuant to Schedule A1 of the Insolvency Act 1986; or
- where the Supplier or DPA Guarantor or Contract Guarantor is an individual or partnership, any event analogous to those listed in limbs (a) to (g) (inclusive) occurs in relation to that individual or partnership; or
- any event analogous to those listed in limbs (a) to (h)
 (inclusive) occurs under the law of any other jurisdiction;

"Intellectual Property Rights" or "IPR"

- copyright, rights related to or affording protection similar to copyright, rights in databases, patents and rights in inventions, semi-conductor topography rights, trade marks, rights in internet domain names and website addresses and other rights in trade or business names, designs, KnowHow, trade secrets and other rights in Confidential Information;
- b) applications for registration, and the right to apply for registration, for any of the rights listed at (a) that are capable of being registered in any country or jurisdiction; and
- c) all other rights having equivalent or similar effect in any country or jurisdiction;
- "IPR Claim" means any claim of infringement or alleged infringement (including the defence of such infringement or alleged infringement) of any IPR, used to provide the Services or as otherwise provided and/or licensed by the Supplier (or to which the Supplier has provided access) to the Customer in the fulfilment of its obligations under this Contract;

"IPR Register"

has the meaning given to it in Clause 22.21;

means

a)

"Joint Controllers" means where	two or more Controllers jointly determine the purposes and means of processing;
"Key Sub-Contract"	means each Sub-Contract with a Key Sub-Contractor;
"Key Sub-Contractor"	means any Sub-Contractor:
	 a) nominated as such as part of the Selection Questionnaire (SQ);
	b) which, in the opinion of the Customer, performs (or would perform if appointed) a critical role in the provision of all or any part of the Services which shall include but is not limited to the delivery of the Services to the Learners by way of education and training; and/or
	 with a Sub-Contract with a contract value which at the time of appointment exceeds (or would exceed if appointed) 20% of the aggregate Contract Charges forecast to be payable under this Contract;
"Know-How" means all ideas, co	ncepts, schemes, information, knowledge, techniques, methodology, and anything else in the nature of knowhow relating to the Services but excluding know-how already in the other Party's possession before the Contract Commencement Date;
"Law"	means any law, subordinate legislation within the meaning of Section 21(1) of the Interpretation Act 1978, bye-law, enforceable right within the meaning of Section 2 of the European Communities Act 1972, regulation, order, regulatory policy, mandatory guidance or code of practice, judgment of a relevant court of law, or directives or requirements, with which the Supplier is bound to comply;
"Learner" "Losses"	means an applicant who is accepted on to a Skills Bootcamp course; means all losses, liabilities, damages, costs, expenses (including legal fees), disbursements, costs of investigation, litigation, settlement, judgment, interest and penalties whether arising in contract, tort (including negligence), breach of statutory duty, misrepresentation or otherwise and " Loss " shall be interpreted accordingly;
"Man Day"	means 7.5 Man Hours, whether or not such hours are worked consecutively and whether or not they are worked on the same day;
"Man Hours"	means the hours spent by the Supplier Personnel properly working on the provision of the Services including time spent travelling (other than to and from the Supplier's offices, or to and from the Sites) but excluding lunch breaks;
"Management Charge"	means a charge levied by the Authority on the Supplier where the Customer is a contracting authority other than the Authority itself;
"Milestone"	means each of the Delivery Plan Trajectory Points as set out in the Delivery Plan.
"Milestone Date"	means the date on which a Milestone is to be achieved by the Supplier;
"Milestone Payment"	means the payment of the relevant sum of Contract Charges that the Supplier will receive for achieving the relevant Milestone at the relevant Delivery Plan Trajectory Point;
"Month"	means a calendar month and " Monthly " shall be interpreted accordingly;
"Occasion of Tax Non-	means:

Compliance"

a)

any tax return of the Supplier submitted to a Relevant Tax Authority on or after 1 October 2012 which is found on or after 1 April 2013 to be incorrect as a result of:

i) a Relevant Tax Authority successfully challenging the Supplier under the General Anti-Abuse Rule or the Halifax Abuse Principle or under any tax rules or legislation in any jurisdiction that have an effect equivalent or similar to the General Anti-Abuse Rule or the Halifax Abuse Principle; ii) the failure of an avoidance scheme which the Supplier was involved in, and which was, or should have been, notified to a Relevant Tax Authority under DOTAS or any equivalent.

 any tax return of the Supplier submitted to a Relevant Tax Authority on or after 1 October 2012 which gives rise, on or after 1 April 2013, to a criminal conviction in any jurisdiction for tax related offences which is not spent at the Contract Commencement Date or to a civil penalty;

"Open Book Data " means complete and accurate financial and non-financial information which is sufficient to enable the Customer to verify the Contract Charges already paid or payable and Contract Charges forecast to be paid during the remainder of this Contract, including details and all assumptions relating to:

- the Supplier's Costs broken down against each Service and/or Deliverable, including actual capital expenditure (including capital replacement costs) and the unit cost and total actual costs of all services;
- b) operating expenditure relating to the provision of the Services manpower resources broken down into the number and grade/role of all Supplier Personnel (free of any contingency) together with a list of agreed rates against each manpower grade; a list of Costs underpinning those rates for each manpower grade, being the agreed rate less the Supplier's Profit Margin;
- c) Overheads;
- d) all interest, expenses and any other third party financing costs incurred in relation to the provision of the Services;
- e) the Supplier Profit achieved over the Contract Period and on an annual basis;
- f) confirmation that all methods of Cost apportionment and Overhead allocation are consistent with and not more onerous than such methods applied generally by the Supplier;
- an explanation of the type and value of risk and contingencies associated with the provision of the Services, including the amount of money attributed to each risk and/or contingency; and
- h) the actual Costs profile for each Service Period.

means the order for the provision of the Services placed by the Customer with the Supplier in accordance with the DPA and under the terms of this Contract;

"Order"

"Other Supplier"	is not	ified to	upplier to the Customer (other than the Supplier) which the Supplier from time to time and/or of which the uld have been aware;
"Overhead"	the Su indire resea penal facilit	upplier's ct corpo rch and ties) bu ies and	e amounts which are intended to recover a proportion of 's or the Key Sub-Contractor's (as the context requires) porate costs (including financing, marketing, advertising, d development and insurance costs and any fines or ut excluding allowable indirect costs apportioned to d administration in the provision of Supplier Personnel ngly included within limb (a) of the definition of "Costs";
"Parent Company"	Suppl busin simila Comp	ier and ess activ r busine pany" sha	company which is the ultimate Holding Company of the which is either responsible directly or indirectly for the ivities of the Supplier or which is engaged by the same or less to the Supplier. The term "Holding or Parent hall have the meaning ascribed by the Companies Act statutory re-enactment or amendment thereto;
"Party"			ustomer or the Supplier and "Parties" shall mean both of
"Performance Measures"	set ou Mana	s the Tie it in Tab gement	
"Prohibited Act"		-	f the following:
	a)	workir	rectly or indirectly offer, promise or give any person ing for or engaged by the Customer and/or the Customer y other public body a financial or other advantage to:
		i)	induce that person to perform improperly a relevant function or activity; or
		ii)	reward that person for improper performance of a relevant function or activity;
	b)	financ impro	ectly or indirectly request, agree to receive or accept any cial or other advantage as an inducement or a reward for oper performance of a relevant function or activity in ection with this Contract;
	c)	comm	nitting any offence:
		i)	under the Bribery Act 2010 (or any legislation repealed or revoked by such Act); or
		ii)	under legislation or common law concerning fraudulent acts; or iii) defrauding, attempting to defraud or conspiring to defraud the Customer; or
		iv) an	ny activity, practice or conduct which would constitute one of the offences listed under (c) above if such activity, practice or conduct had been carried out in the UK;

"Protected Measures"	means appropriate technical and organisational measures which may include: pseudonymising and encrypting Personal Data, ensuring confidentiality, integrity, availability and resilience of systems and services, ensuring that availability of and access to Personal Data can be restored in a timely manner after an incident, and regularly assessing and evaluating the effectiveness of the such measures adopted by it including those outlined in Contract Schedule 4 (Security);	
"Project Specific IPR"	means:	
	 a) Intellectual Property Rights in items created by the Supplier (or by a third party on behalf of the Supplier) specifically for the purposes of this Contract and updates and amendments of these items including (but not limited to) database schema; and/or 	
	 b) IPR in or arising as a result of the performance of the Supplier's obligations under this Contract and all updates and amendments to the same; but shall not include the Supplier Background IPR; 	
"Quality Standards"	means any:	
	 a) standards published by BSI British Standards, the National Standards Body of the United Kingdom, the International Organisation for Standardisation or other reputable or equivalent bodies (and their successor bodies) that a skilled and experienced operator in the same type of industry or business sector as the Supplier would reasonably and ordinarily be expected to comply with; 	
	 b) standards detailed in the specification in DPA Schedule 2 Part 1 (Services); 	
	 standards detailed by the Customer in the Contract Order Form or agreed between the Parties from time to time; 	
	 relevant Government codes of practice and guidance applicable from time to time. 	
"Recipient"	has the meaning given to it in Clauses 23.10 to 23.18 (Confidentiality);	
"Rectification Plan"	means the rectification plan pursuant to the Rectification Plan Process;	
"Rectification Plan Process"	means the process set out in Clause 27.3 (Rectification Plan Process);	
"Registers"	has the meaning given to in Contract Schedule 10 (Exit Management);	
"Regulations"	has the meaning given to it in DPA Schedule 1 (Definitions);	
"Related Supplier"	means any person who provides services to the Customer which are related to the Services from time to time;	
"Relevant Conviction"	means a Conviction that is relevant to the nature of the Services to be provided or as specified in the Contract Order Form;	
"Relevant Requirements"	means all applicable Law relating to bribery, corruption and fraud, including the Bribery Act 2010 and any guidance issued by the Secretary of State for Justice pursuant to section 9 of the Bribery Act 2010;	

"Relevant Tax Authority"	means HMRC, or, if applicable, the tax authority in the jurisdiction in which the Supplier is established;
"Relevant Transfer"	means a transfer of employment to which the Employment Regulations applies;
"Relevant Transfer Date"	means, in relation to a Relevant Transfer, the date upon which the Relevant Transfer takes place;
"Replacement Services"	means any services which are substantially similar to any of the Services and which the Customer receives in substitution for any of the Services following the Contract Expiry Date, whether those services are provided by the Customer internally and/or by any third party;
"Replacement Sub- Contractor"	means a sub-contractor of the Replacement Supplier to whom Transferring Supplier Employees will transfer on a Service Transfer Date (or any sub-contractor of any such sub-contractor);
"Replacement Supplier"	means any third party provider of Replacement Services appointed by or at the direction of the Customer from time to time or where the Customer is providing Replacement Services for its own account, shall also include the Customer;
"Request for Information"	means a request for information or an apparent request relating to this Contract or the provision of the Services or an apparent request for such information under the FOIA or the EIRs;
"Security Management Plan"	means the Supplier's security management plan prepared pursuant to paragraph 4 of Contract Schedule 4 (Security) a draft of which has been provided by the Supplier to the Customer in accordance with paragraph 4 of Contract Schedule 4 (Security) and as updated from time to time;
"Security Policy"	means the Customer's security policy, referred to in the Contract Order Form and / or in contract Schedule 2 (Services, Performance Management and Contract Management), in force as at the Contract Commencement Date (a copy of which has been supplied to the Supplier), as updated from time to time and notified to the Supplier;
"Security Policy Framework"	the current HMG Security Policy DPA that can be found at https://www.gov.uk/government/publications/securitypolicy-DPA;
"Service Failure"	means a failure to substantially meet the Performance Measures set out in Contract Schedule 2 Part 2 (Performance Management);
"Service Transfer"	means any transfer of the Services (or any part of the Services), for whatever reason, from the Supplier or any Sub-Contractor to a Replacement Supplier or a Replacement Sub-Contractor;
"Service Transfer Date"	means the date of a Service Transfer;
"Services"	means the services to be provided by the Supplier to the Customer as referred to Schedule 2 Part A (Services);
"Skills Bootcamp Course"	means a course delivered by or on behalf of the Supplier under this Contract;
"Skills Bootcamp Guidance"	means guidance produced by the Authority relating to skills bootcamps which can be found at <u>Skills Bootcamps funding and performance management - GOV.UK (www.gov.uk)</u>

"Sites"	 means any premises (including the Customer Premises, the Supplier's premises or third party premises) or any virtual environment from, to or at which: a) the Services are (or are to be) provided; or 	
	 b) the Supplier manages, organises or otherwise directs the provision or the use of the Services. 	
"Specific Change in Law"	means a Change in Law that relates specifically to the business of the Customer and which would not affect a Comparable Supply;	
"Staffing Information"	has the meaning give to it in Contract Schedule 5 (Staff Transfer);	
"Sub-category 1"	refers the Institute for Apprenticeships and Technical Education occupational pathways which are set out at <u>Occupational Maps /</u> Institute for Apprenticeships and Technical Education;	
"Sub-category 2"	refers to the extent to which delivery of the Skills Bootcamp can correspond to Green Skills and Pathways to Accelerated Apprenticeships;	
"Sub-Contract"	means any contract or agreement (or proposed contract or agreement), other than this Contract or the DPA, pursuant to which a third party:	
	a) provides the Services (or any part of them);	
	 provides facilities or services necessary for the provision of the Services (or any part of them); and/or 	
	 c) is responsible for the management, direction or control of the provision of the Services (or any part of them); 	
"Sub-Contractor"	means any person other than the Supplier, who is a party to a SubContract and the servants or agents of that person;	
"Supplier"	means the person, firm or company with whom the Customer enters into this Contract as identified in the Contract Order Form;	
"Supplier Assets"	means all assets and rights used by the Supplier to provide the Services in accordance with this Contract but excluding the Customer Assets;	
"Supplier Background IPR"	means	
	a) Intellectual Property Rights owned by the Supplier before the Contract Commencement Date, for example those subsisting in the Supplier's standard development tools, program components or standard code used in computer programming or in physical or electronic media containing the Supplier's Know-How or generic business methodologies; and/or	
	 b) Intellectual Property Rights created by the Supplier independently of this Contract; 	

"Supplier's	means	
Confidential	a) any information, however it is conveyed, that relates to the	
Information"	business, affairs, developments, IPR of the Supplier (including the Supplier Background IPR) trade secrets, Know- How, and/or personnel of the Supplier;	
	 any other information clearly designated as being confidential (whether or not it is marked as "confidential") or which ought reasonably to be considered to be confidential and which comes (or has come) to the Supplier's attention or into the Supplier's possession in connection with this Contract; 	
	c) information derived from any of the above.	
"Supplier Equipment"	means the Supplier's hardware, computer and telecoms devices, equipment, plant, materials and such other items supplied and used by the Supplier (but not hired, leased or loaned from the Customer) in the performance of its obligations under this Contract;	
"Supplier Personnel"	means all directors, officers, employees, agents, consultants and contractors of the Supplier and/or of any Sub-Contractor engaged in the performance of the Supplier's obligations under this Contract;	
"Supplier Profit"	means, in relation to a period or a Milestone (as the context requires), the difference between the total Contract Charges (in nominal cash flow terms but excluding any Deductions) and total Costs (in nominal cash flow terms) for the relevant period or in relation to the relevant Milestone;	
"Supplier Profit Margin" "Supplier Representative"	means, in relation to a period or a Milestone (as the context requires), the Supplier Profit for the relevant period or in relation to the relevant Milestone divided by the total Contract Charges over the same period or in relation to the relevant Milestone and expressed as a percentage; means the representative appointed by the Supplier named in the	
	Contract Order Form;	
"Template Contract Order Form"	means the Template Contract Order Form in Annex 1 of DPA Schedule 4 (Template Contract Order Form and Template Contract Terms);	
"Template Contract Terms"	means the template terms and conditions in Annex 2 of DPA Schedule 4 (Template Order Form and Template Contract Terms);	
"Tender"	means the tender submitted by the Supplier to the Customer and annexed to or referred to in Contract Schedule 5;	
"Termination Notice"	means a written notice of termination given by one Party to the other, notifying the Party receiving the notice of the intention of the Party giving the notice to terminate this Contract on a specified date and setting out the grounds for termination;	
"Test Issue"	means any variance or non-conformity of the Services or Deliverables from their requirements as set out in the Contract;	
"Third Party IPR"	means Intellectual Property Rights owned by a third party which is or will be used by the Supplier for the purpose of providing the Services;	
"Transferring Customer Employees"	those employees of the Customer to whom the Employment Regulations will apply on the Relevant Transfer Date;	

"Transferring Former Supplier Employees"	in relation to a Former Supplier, those employees of the Former Supplier to whom the Employment Regulations will apply on the Relevant Transfer Date;
"Transferring Supplier Employees"	means those employees of the Supplier and/or the Supplier's SubContractors to whom the Employment Regulations will apply on the Service Transfer Date;
"Undelivered Services"	has the meaning given to it in Clause 8.5 (Services);
"Undisputed Sums Time Period"	has the meaning given to it Clause 31.1 (Termination of Customer Failure to Pay);
"Valid Invoice"	means an invoice issued by the Supplier to the Customer that complies with the invoicing procedure in paragraph 7 (Invoicing Procedure) of Contract Schedule 3 (Contract
	Charges, Payment and Invoicing);
"Variation"	has the meaning given to it in Clause 13.3 (Variation Procedure);
"Variation Form"	means the form set out in Schedule 11 (Variation Form);
"VAT"	has the meaning given to it in DPA Schedule 1 (Definitions);
"Worker"	means any one of the Supplier Personnel which the Customer, in its reasonable opinion, considers is an individual to which Procurement Policy Note 08/15 (Tax Arrangements of Public Appointees)
	https://www.gov.uk/government/publications/procuremen tpolicy-note-0815-tax-arrangements-of-appointees
"Working Day"	means any day other than a Saturday or Sunday or public holiday in England and Wales unless specified otherwise by Parties in this Contract;

CONTRACT SCHEDULE 2: SERVICES, PERFORMANCE MANAGEMENT AND CONTRACT MANAGEMENT

1. INTRODUCTION

1.1 This Contract Schedule 2 Part 1 specifies the Services to be provided under this Contract, Part 2 sets out the requirements for Performance Management and Part 3 sets out the requirements for Contract Management and Management Information:

Please refer to:

- DPA Schedule 2 Service Requirements
- DPS Second Competitions Tender Documents 4a 4h Contract Order Forms

PART 1: THE SERVICES

- 1. The Supplier will provide the Services as detailed in the Contract Order Form and/or as amended by special terms set out below in this Part 1.
- 2. The Supplier will provide any additional supplementary services required and specified by Contracting Authorities and as outlined in the Customer Needs under supplementary requirements.

PART 2: PERFORMANCE MANAGEMENT

In this section the words below have the following meaning:

"Contract Performance Review"	means a monthly review of the Supplier's performance under this Contract.
"Complete or Completion"	(a) means as follows;
	 (b) for All Learners – completion by the Learner of their training and final assessment;
	(c) for Independent Learners – an offer of an interview for a live vacancy which will utilise the skills obtained on the Skills Bootcamp, meet the criteria set out in the Service Requirements; and reflects the salary expectations and location set out in the Initial Advice and Guidance process
	(d) for Self-Employed Learners -written confirmation from the SelfEmployed Leaners of how the Skills Bootcamp training has been or will be applied to enable them to secure new work of contract, which meet the criteria set out in the Service Requirements; and
	(e) for Co-Funded Learners – an offer of an interview for a new role or responsibility or an offer of a new role or responsibility which in either case meets the criteria set out in the Service Requirements.
	 (f) Including any definition as defined within the payment terms/profile
"Drop Out"	means a Learner who starts but does not Complete a Skills Bootcamp
"Delivery Plan"	means the plan setting out key timings and trajectories for the Supplier's delivery during the Contract Period, as agreed by the Customer and including but not limited to
	(a) the Delivery Plan Trajectory Points; and
	(b) how many Learners will start or Complete at each of those Delivery Plan Trajectory Points.
"Delivery Plan Trajectory Point"	means, but is not limited to, the anticipated dates on which Milestone Payments in relation to specified numbers of Learners would be scheduled to be paid by the Customer to the Supplier as follows: (a) Commencement Date
	(b) Mid-course Milestone Date
	(c) Course Completion and offer of interview Milestone Date
	(d) Positive Outcome (or equivalent) Milestone Date
	(e) Finishing the Course;
	(f) Completion including offer of an interview Milestone Date.
	(g) Suppliers should note a completed delivery plan (which sets out the delivery trajectories) will need to be provided at contract mobilisation requested by contract managers.
"Improvement Plan"	means a plan for improvement that the Customer can request from the Supplier within ten (10) Working Days in the event of failure.

"Management Information"	means the data and information relevant to the Services and performance the Supplier shall collate and provide to the Customer in accordance with this Schedule.
"Management Information Report"	means the report containing the Management Information that the Supplier is required to submit to the Customer in accordance with this Schedule.
"Performance Management" means	how the Customer will measure the Supplier's performance and progress against the Service Requirements (Schedule 2: Part 1), and the Delivery Plan.
"Performance Manager"	means the person the Supplier will appoint to ensure that the Contract is delivered as specified in the Contract and that Service Levels, Recruitment Targets and Performance Measures are achieved.
"Performance Measures" means the	Tier 1 measures set out in Table 1 and the Tier 2 measures set out in Table 3.
"Reporting Period" means the report	ing period that occurs every calendar month from X of each month to the X of the following month and will commence on the Contract Commencement Date.

"Service Level" means the Service Levels as set out in Table 1 of this Schedule by which the Contractor's performance will be measured.

1. PERFORMANCE MEASURES

- 1.0. This section sets out the Performance Measures against which the Parties shall measure the Supplier's performance.
- 1.1. The objective of the Performance Measures is to:
 - 1.1.1 ensure that the Services are of a consistently high quality and meet the requirements of the Customer;
 - 1.1.2 provide a mechanism whereby the Customer can attain meaningful recognition of inconvenience and/or loss resulting from the Supplier's failure to deliver the Services; and
 - 1.1.3 incentivise the Supplier to meet the Performance Measures and to remedy any failure to meet the required standards expeditiously.

Performance Measures Targets

- 1.2. The Supplier shall ensure compliance with the Tier 1 Performance Measures listed in Table 1 and the Tier 2 Performance Measures in Table 3.
- 1.3. The Customer reserves the right to revise Performance Measures and introduce subject specific Performance Measures as required.
- 1.4. The Supplier and the Customer shall monitor the Supplier's performance against each of the Tier 1 and Tier 2 Performance Measures.
- 1.5. The Supplier shall complete and return the monthly Customer Reporting Template outlining performance against the Performance Measures to date and confirm whether they have been achieved.
- 1.6. If the Supplier fails to meet any one Performance Measure in any Reporting Period, the Customer reserves the right to take action in line with Paragraph 3 (Consequence of Service Failure in relation to a Tier 1

Performance Measure) and in line with Paragraph 5 (Consequence of Service Failure in relation to a Tier 2 Performance Measure) in relation to a Tier 2 Performance Measure.

1. TIER 1 PERFORMANCE MEASURES

1.1. The Tier 1 Performance Measures are set out in Table 1 below.

Table 1: Tier 1 Performance Measures

Performance Measure Description	Performance Measure Metric:
Tier 1 Performance Measure 1: Completions	<mark>Green</mark> : 80% or more of Learners
Learners who successfully Complete the Skills Bootcamp Course, of those Learners who Start the Skills Bootcamp. For the purposes of this measure, Completion must be achieved within 8 weeks of	<mark>Amber</mark> : Between 79 – 40% of Learners
the training finishing.	Red <mark>: Less than 40% of Learners</mark>
Tier 1 Performance Measure 2: Positive Outcomes Learners who successfully achieve a Positive	<mark>Green</mark> : 75% of Learners
Outcome within six months of Completing the Skills Bootcamp, of those Learners who Complete the Skills Bootcamp	<mark>Amber</mark> : 74 – 40% of Learners
	Red <mark>: Less than 40% of Learners</mark>
Tier 1 Performance Measure 3: Data Requests from DfE	Data returns will be checked on an ongoing basis to review if they are on time, accurate and complete.
The Supplier must submit timely, accurate and complete data and evidence.	Green: 90% or more of data returns are timely, accurate and complete
This includes:	
 The Supplier shall submit the Contract Monitoring Plan (CMP) Report on a monthly basis on a date specified by the Customer. 	Amber: 89 – 40% of data returns are timely, accurate and complete
 The Learner Datasheet (LD) must be submitted each time the Supplier makes an ILR submission to provide supporting information for the ILR submission. 	Red <mark>: Less than 40% of data returns are timely, accurate and complete</mark>
2. The Customer shall monitor the Supplier's perform	nance against each of the Performance Measures listed i

1.2. The Customer shall monitor the Supplier's performance against each of the Performance Measures listed in Table 1 – Tier 1 Performance Measures. The Supplier's performance will be measured and its performance rated in accordance with the criteria set out in Table 2 below.

Table 2: Tier 1 Performance Measures Rating and Performance Management Implications

Rating	Criteria
<mark>Green</mark> – Good	The Supplier is meeting or exceeding the Tier 1 Performance Measure Target
Amber - Improvement Needed	The performance of the Supplier is below the Tier 1 Performance Measure Target

- 1.3. If at the measurement or reporting point for an individual Tier 1 Performance Measure, the Supplier;
 - 1.3.1 achieves a rating of "Green Good" for an individual Tier 1 Performance Measure Target no further action will be required;
 - 1.3.2 achieves a rating of "Amber Improvement Needed" or "Red Unacceptable" for any Tier
 1 Performance Measure Target, the Supplier will incur a Service Failure and the provisions of Paragraph 3 of this Schedule will apply;
 - 1.3.3 achieves a rating of "Red Unacceptable", the Supplier will not be entitled to participate in any Call Off Competitions under the DPA while the Supplier's performance against the relevant Performance Measure Target remains "Red Unacceptable". If the Supplier achieves a rating of "Red Unacceptable" at Completion, then the Supplier will not be entitled to participate in any Further Competitions for Skills Bootcamps for 12 months;
 - 1.3.4 if the Supplier's performance remains "Red Unacceptable" at the next agreed measurement or reporting point at which the Tier 1 Performance is Measure is being considered, the Customer reserves the right to terminate this Contract and / or the DPA with regards to the Supplier.
- 1.4. In line with the cross-government transparency agenda the Customer reserves the right to make the Supplier's performance against the Performance Measures in Table 1 of this Schedule available in the public domain, which may include publishing them on gov.uk and including them in any related transparency reporting

2. CONSEQUENCE OF SERVICE FAILURE IN RELATION TO A TIER 1 PERFORMANCE MEASURE

- 2.1. Without prejudice to any other rights or remedies arising under this Contract, including under Clause 30 (Termination) for material Default, if the Supplier incurs a Service Failure in relation to any Tier 1 Performance Measure in any Reporting Period, the Supplier acknowledges and agrees that the Customer shall have the right to exercise (in its absolute and sole discretion) all or any of the following remedial actions:
 - 2.1.1 The Customer shall be entitled to require the Supplier, and the Supplier agrees to prepare and provide to the Customer, a plan for improvement (an "Improvement Plan") within ten (10) Working Days of a written request by the Customer for such Improvement Plan. Such Improvement Plan shall be subject to the Customer's prior approval and the Supplier will be required to implement any approved Improvement Plan, as soon as reasonably practicable;
 - 2.1.2 The Customer shall be entitled to require the Supplier, and the Supplier agrees to attend, within a reasonable time one (1) or more meetings at the request of the Customer in order to resolve the issues raised by the Customer in its notice to the Supplier requesting such meetings;
 - 2.1.3 Without prejudice to Paragraph 3.1.1, the Customer shall be entitled to serve a notice of improvement ("Improvement Notice") on the Supplier and the Supplier shall implement such requirements for improvement as set out in the Improvement Notice;
 - 2.1.4 The Customer shall be entitled to issue interim Performance Measures and/or milestones in order to monitor the Supplier's implementation of any Improvement Plan or Improvement Notice.
- 2.2. If at the date stated in the Improvement Plan or Improvement Notice by which time the Service Failures(s) should have been remedied, the Customer is not satisfied that the Supplier has:
 - 2.2.1 implemented the actions set out in the Improvement Notice (if an Improvement Notice has been served) and as a consequence the Service Failure is ongoing; and/or

- 2.2.2 implemented an Improvement Plan approved by the Customer and as a consequence the Service Failure is ongoing; and/or
- 2.2.3 met the interim performance measures and/or milestones;

and the relevant original Service Failure was originally "Amber – Requires Improvement", the Service Failure will be escalated to be "Red – Unacceptable" and the provisions of Paragraph 2.3.3 and if relevant Paragraph 2.3.4 will apply. If the original Service Failure was "Red – Unacceptable", Paragraph 2.3.4 will apply.

3. TIER 2 PERFORMANCE MEASURES

3.1. The Tier 2 Performance Measures are set out in Table 3 below:

Table 3: 1	Tier 2	Performance	Measures
------------	--------	-------------	----------

Service Level Description	Service Level Metric	Final date for measure to be assessed
Tier 2 Performance Measure 5: Drop outs Learners who leave a Skills Bootcamp before they finish the training, of those Learners who Start the Skills Bootcamp	Green: 15% or fewer of Learners Amber : 16 – 25% of Learners Red: Nore than 25% of Learners	This measure will be assessed on an ongoing basis
Tier 2 Performance Measure 6: meeting Delivery Plan Trajectory Points Likelihood of achieving the Tier 1 performance measure targets.	Green : The Supplier is likely or very likely Amber : The Supplier is unlikely he Supplier is very unlikely Red : T	This measure will be assessed on an ongoing basis
Tier 2 Performance Measure 7: Social Value Total percentage of opportunities created for the workforce employed under the contract to enable or promote access to apprenticeships, T level industry placements or other learning and development opportunities at level 2, 3, and 4+ (or equivalent) or other activities to support relevant sector related skills growth and sustainability in the contract workforce. (Please note that the contract workforce can include contractors and T-level placements.)	Green: 10% and above improvement the within lifetime of the contract from the starti g position Amber : above 0 to less than 10% Rec: 0 or decline	This measure will be assessed at the end of the contract term. Ongoing reporting will be required throughout the contract.

Contractor to declare percentage position at contract start and achieve a percentage point improvement on that starting position by contract end. i.e., If at contract start the 20% are from under-represented groups, a 10% improvement would mean that by contract end 22% are from underrepresented groups.		
Tier 2 Performance Measure 8: Employer Engagement The Supplier has engaged employers at the design, delivery and post-Skills Bootcamp stage as set out in Table 5 below.	The Supplier should be able to evidence Employer Engagement at the Design Stage, Learner Recruitment during the Delivery Stage and Post Skills Bootcamp Stage on every Skills Bootcamp. An Employer Engagement Pro Forma will be sent to all Suppliers to complete at the commencement of the Contract. The Supplier should list the Employers they have engaged (updated from bid stage) and include the role of each Employer and contact details. The Supplier will then need to provide evidence at four Points (as set out in Table 5 below) that Employers have been engaged in the way set out in the Pro Forma and the Supplier has achieved a minimum level of Employer satisfaction for the Skills Bootcamp. Green: the Supplier has met all Points to date Amber: the Supplier has not met one of the Points to date Rec: the Supplier has not met two or more Points to date	This measure will be assessed on an ongoing basis

3.2. The Customer shall monitor the Supplier's performance against each of the Performance Measures listed in Table 3 – Tier 2 Performance Measures. The Supplier's performance will be measured and its performance rated in accordance with the criteria set out in Table 4 below:

Table 4: Tier 2 Performance Measures Rating and Performance Management Implications

Rating	Criteria
<mark>Green</mark> - Good	The Supplier is meeting or exceeding the Tier 2 Performance Measure Target

A <mark>mber</mark> - Improvement Needed	The performance of the Supplier is below the Tier 2 Performance Measure Target
<mark>Red</mark> - Unacceptable	The performance of the Supplier is significantly below the Tier 2 Performance Measure Target

- 3.3. If at the agreed measurement or reporting point for an individual Tier 2 Performance Measure, the Supplier:
 - 3.3.1 achieves a rating of "Green Good" for an individual Tier 2 Performance Measure Target, no further action will be required;
- 3.4. achieves a rating of "Amber Improvement Needed" or "Red Unacceptable" for any Tier 2 Performance Measure Target, the Supplier will incur a Service Failure and the provisions of Paragraph 5 of this Schedule will apply. In line with the cross-government transparency agenda the Customer reserves the right to make the Supplier's performance against the Performance Measures in Table 3 of this Schedule available in the public domain, which may include publishing them on gov. uk and including them in any related transparency reporting.

4. CONSEQUENCE OF SERVICE FAILURE IN RELATION TO A TIER 2 PERFORMANCE MEASURE

- 4.1. Without prejudice to any other rights or remedies arising under this Contract, including under Clause 30 (Termination) for material Default, if the Supplier incurs a Service Failure in relation to any Tier 2 Performance Measure in any Reporting Period, the Supplier acknowledges and agrees that the Customer shall have the right to exercise (in its absolute and sole discretion) all or any of the following remedial actions:
 - 4.1.1 the Service Failure for the relevant Tier 2 Performance Measure is assessed as being "Amber Requires Improvement", the Supplier will, where instructed to do so by the Customer, attend Contract Performance Review meetings with the Customer at an interval specified by the Customer until the next date on which the relevant Tier 2 Performance Measure will be assessed and reported on;
 - 4.1.2 If the relevant Tier 2 Performance Measure remains "Amber Requires Improvement" at the time the Tier 2 Performance Measure is reassessed, the provisions set out in Paragraph 3 of this Schedule will apply as if the references to "Tier 1 Performance Measures" have been replaced with "Tier 2 Performance Measures";
 - 4.1.3 If the performance of the Supplier in relation to the relevant Tier 2 Performance Measure worsens to "Red Unacceptable" at the time the Tier 2 Performance Measure is reassessed, the Tier 2 Performance Measure will be immediately escalated to a Tier 1 Performance Measure Service Failure and Paragraph 2.3.2 will apply;
 - 4.1.4 If after complying with Paragraph 5.4.2, the Tier 2 Performance Measure remains "Amber Requires Improvement", the Tier 2 Performance Measure will be immediately escalated to a Tier 1 Performance Measure Service Failure and Paragraph 2.3.2 will apply;
 - 4.1.5 If the Service Failure for the relevant Tier 2 Performance Measure is assessed as being "Red Unacceptable", the Performance Measure will be immediately escalated to a Tier 1 Performance Measure Service Failure and Paragraph 2.3.2 will apply.

Table 5: Tier 2: Employer Engagement Performance Measure – Full Details

Monitoring Stage	When this will be measured	How this will be measured - An Employer Engagement Pro Forma will be sent to all Suppliers for the Call-Off Contract to complete on mobilisation. The Supplier will list the Employers engaged with each of the Supplier's Skills Bootcamps (updated from bid stage) within that Call-Off Contract with the role of each Employer and contact details provided. The supplier must include satisfactory explanation of the differences between Employers cited in bids or letters of support, and an updated list where some Employers have dropped out with an explanation of any dropouts and replacement Employers identified to ensure sufficient engagement and interviews for vacancies can be secured further on.
Point 1 - Understandi ng skills needs and learner screening	At or before Commencem ent payment	 Evidence confirming how at least one employer contributed to the learner screening process and that employer skills need informed the learner recruitment process, such as (but not limited to) one of the following learner screening activities: Confirming agreement with the employer regarding which learner eligibility requirements are needed in addition to the standard Skills Bootcamps eligibility requirements to ensure successful completion of the Skills Bootcamp OR confirmation that no further eligibility requirements are needed as agreed with the employer based on their skills needs; Co-designing recruitment criteria, interview criteria, or pre-interview recruitment assessments (for example situational judgement tests if applicable) with the employer; And/or the employer conducting learner recruitment activities, including interviews. Depending on the form of Employer engagement from the above list examples of acceptable forms of evidence for this stage may include:- email/written statement or confirmation from the Employer; photograph of the employer engaging with the Learner screening process with supporting email/written confirmation from the Supplier explaining what the photo indicates. The Customer reserves the right to amend/expand this list as necessary and where it does so will confirm the requirements.

Point 2 - Course	At or before Commencem	Evidence confirming that at least one employer engaged in at least one of the ways below to ensure effective course content alignment with existing skills need:	
Content Agreement	ent payment	 Engagement on the part of the employer in the co-design or co-development of the Skills Bootcamp to ensure that course provision will meet their existing skills need and what form this engagement took; Contribution of feedback by the employer on the course design process regarding ensuring that the Skills Bootcamp's course design would meet their existing need with confirmation of how this was implemented. For example, has the supplier carried out a needs analysis with the employer's business in order to develop the Skills Bootcamp, and the solution proposed as a result is approved by the employer [this line is from the current EE KPI Guidance around the training needs analysis]. 	

Point 3 - Course Delivery	At or before completion payment	 Evidence confirming that at least one employer was engaged in the delivery of the Skills Bootcamp and what form this took, for example via (but not limited to) one or more of the following activities: Employer presentation(s); Employer panel talk(s); Employer-delivered course content; Employer-led visit(s) to a workplace Learner mentoring or coaching session(s); Providing Learners with feedback on their work/performance; Employer-led insight day(s); Employer-led interview training and/or mock interview(s). Depending on the form of Employer engagement from the above list examples of acceptable forms of evidence for this stage may include:- Email/written statement or confirmation from the employer confirming they participated in one or more of such course delivery activities; Photograph of the employer engaging with the course delivery process via one or more of the above course delivery activities or equivalent with supporting email/written confirmation from the Supplier explaining what the photo indicates. The Customer reserves the right to amend/expand this list as necessary and where it does so will confirm the requirements.
Point 4 - Course and Candidate Suitability Review	At or before Outcome payment	 Evidence via an Employer survey (Employer Satisfaction Questionnaire), confirming the level of satisfaction held by at least one Employer on: the course's overall effectiveness in upskilling or reskilling Learners to the required standard in the required skills, knowledge and behaviours as outlined in the original tender; the suitability of Learner candidates who attended guaranteed interviews provided by the Employer.

Acceptable evidence format for this stage:-
 Course Content was approved by the employer as meeting the requirements of actual vacancies they hold or expect to hold within the next 12 months, or that the Skills Bootcamp meets the needs to train their own employees for a new role or new responsibilities.
Depending on the form of Employer engagement from the above list examples of acceptable forms of evidence for this stage may include:-
email/written statement or confirmation from the Employer;
 photograph of the Employer engaging with the course content agreement process with supporting email/written confirmation from the Supplier explaining what the photo indicates.
The Customer reserves the right to amend/expand this list as necessary and where it does so will confirm the requirements.

via the supplier.	np Employer Satisfaction Questionnaire by the employer, which is then submitted as evidence end/expand this list as necessary and where it does so will confirm the requirements.
[Provisional] Employer Satisfaction Qu	estionnaire Part A requirements:-
Part A – Course Review: Within the questionnaire, the Employe their needs overall with a score from 0	er will be asked their satisfaction levels with the Skills Bootcamp and to what extent it met -4.
Part A: Scoring Matrix	
4: Excellent	The Supplier has fully met the needs of the Employer to an excellent standard, providing significant value and benefit to the Employer
3: Good	The Supplier has met the needs of the Employer to a good and satisfactory standard, providing good value and benefit to the Employer
2: Unsatisfactory	The Supplier has failed to meet the needs of the Employer, in many ways and/or materially in one or more ways
1: Poor	The Supplier failed to meet the needs of the Employer, in a significant number of ways and/or inadequate or no supporting evidence has been provided to support the response
0: Unacceptable	The Employer details are absent or incomplete and/ or the employer is not involved in the Skills Bootcamp.
The RAG rating for measuring success	will be:-
Green - Good	Suppliers scoring 3 (Good) or above
Amber – Improvement Needed	Suppliers scoring 2 (Unsatisfactory) or above
Red	Suppliers scoring 1 (Poor) or below

[Provisional] Employer Satisfaction Questionnaire Part B requirements:-

Part B – Candidate Suitability Review:

Within the questionnaire, the Employer will be asked their satisfaction levels regarding the suitability of Learner candidates that attended the guaranteed interviews, that they led for the agreed roles under consideration (in answer to the question below or equivalent) and assessed using the scoring matrix below.

following statement:		rview candidates via indicating your level of agreement with the s for the Employer's vacancies at the guaranteed interview stage.'	
Part B Scoring Matrix			
4	4 Strongly Agree		
3		Agree	
2		Neither Agree nor Disagree	
1 Disagree		Disagree	
0 Strongly Disagree		Strongly Disagree	
The RAG rating for measuring success	s will be:-		
Green - Good Suppliers scoring 3 (Agree) or above) or above	
Amber – Improvement Needed	vement Needed Suppliers scoring 2 (Neither Agree nor Disagree) or above		
Red -	Suppliers scoring 1 (Disagree) or below		

PART 3: CONTRACT MANAGEMENT AND MANAGEMENT INFORMATION

1. PERFORMANCE MANAGEMENT REVIEW MEETINGS

- 1.1 The Supplier will work with the Customer and/or Other Contracting Authorities as appropriate to establish and maintain an effective and beneficial working relationship and to ensure that this Call Off Contract(s) is being delivered to at least the minimum required standard as specified in Schedule 2 (Services, Performance Management and Contract Management).
- 1.2 This Call Off Contract will be contract managed (including performance management) in accordance with the terms of this Call Off Contract. Where the Supplier has been awarded one or more Call Off Contracts, the Customer reserves the right to consolidate the contract management of each Call Off Contract into one contract management process and manage it through the DPA. For the avoidance of doubt, each Call Off Contract would still be contract managed in accordance with its own terms.
- 1.3 The Customer will conduct regular formal Performance Management Review Meetings at a frequency (usually monthly) determined by the Customer from time to time to monitor, measure and review the Supplier's performance under this Call Off Contract.
- 1.4 The objectives of the Performance Management Review Meetings will be to manage the Supplier's performance under the Call Off Contract awarded to the Supplier and shall include but not be limited to: -
 - (a) Monitoring and reviewing the Supplier's compliance with its obligations under this Call Off Contract;
 - (b) reviewing the performance of the Supplier against the Tier 1 and Tier 2 Performance Measures set out in the Call Off Contract, and conformance by the Supplier to all other standards and policies set out elsewhere in the Call Off Contract;
 - (c) reviewing Improvement Plans;
 - (d) agreeing any changes to the Call Off Contract, which have been raised through the Variation Procedure as detailed in Clause13 (Change);
 - (e) reviewing the Service Requirements set out in Schedule 2 Part 1 (Services);
 - (f) resolving operational and contractual problems, which may have been raised using the Dispute Resolution Procedure as set out in Schedule 6 (Dispute Resolution Procedure);
 - (g) identifying and recording Service Failures;
 - (h) the Supplier's System Assurance in accordance with Paragraph 2 of this Schedule.
- 1.5 Any agenda for the Performance Management Review Meetings shall be produced by the Customer in conjunction with the Supplier.
- 1.6 The Supplier shall supply all information requested by the Customer for the purpose of conducting the Performance Management Review Meetings.
- 1.7 The Supplier shall provide suitable representatives with the necessary authority to consider service performance and to ensure that any issues impacting the Customer are discussed and resolved.
- 1.8 The Customer shall make, retain and distribute a record of the meetings.

2 SUPPLIER'S SYSTEM ASSURANCE

- 2.1 The Supplier shall comply with the Customer's requirements for Supplier's Systems Assurance as described in this Paragraph 2;
- 2.2 The Supplier shall have systems in place to:
 - (a) Detect and prevent duplicate Contract Charges;
 - (b) Prepare and submit accurate, valid, supported, timely Contract Charges and cost information;
 - (c) Carry out effective monitoring of Sub-Contractors in accordance with Clause 18 and Schedule 14 (Sub-Contracting) of the Call Off Contract;
 - (d) Quality assure the content of Contract Monitoring Reports and Management Information; and
 - (e) Respond to compliance monitoring and Supplier Review Meetings with an appropriate action plan.
 - 2.3 The Customer will discuss any Supplier's System Assurance issues at the Performance Management Review Meetings. Any discussion on the Supplier's System Assurance will focus on the Supplier's internal control systems for:
 - (a) Governance Supplier systems for tracking and reporting performance including anti-fraud measures;
 - (b) Service delivery Supplier systems for ensuring the activities are delivered to the required standards and to prevent any delays to achieving the requirements for Milestone Payments;
 - (c) Claim procedures and payment Supplier systems to support claims for payment, including appropriate segregation of duties.

3 PERFORMANCE RECORDS

- 3.1 The Supplier shall keep appropriate documents and records (including but not limited to Learner eligibility records, attendance sheets, training records, complaints received) in relation to the Services being delivered. The records and documents of the Supplier shall be available for inspection by the Customer and/or its nominee at any time and the Customer and/or its nominee may make copies of any such records and documents.
- 3.2 In addition to the requirement in Paragraph 3.1 maintain appropriate documents and records, the Supplier shall provide to the Customer with such supporting documentation as the Customer may reasonably require in order to verify the level of the performance of the Supplier.

4 CONTRACT MONITORING PLAN AND LEARNER DATASHEET

- 4.1 The Supplier shall collect a range of data and report on its performance against the Performance Measures set out in this Schedule and the Service Levels set out in Table 1 of this Schedule.
- 4.2 The Supplier shall submit the Contract Monitoring Plan (CMP) Report on a monthly basis. on a date specified by the Customer. The Learner Datasheet (LD) must be submitted each time the Supplier makes an ILR submission to provide supporting information for the ILR submission. The Customer may amend the reporting frequency and format in respect of any or all Contract Monitoring or Learner Datasheets or waive the requirement for any aspect of the Contract Monitoring or Learner Datasheets to be reported upon, by giving the Supplier not less than one (1) months' notice in writing.

- 4.3 The content and format of the CMP and LD shall be specified by the Customer but unless otherwise stated, it shall be presented in a tabular format or MS Excel and it shall include the formulas and calculations used by the Supplier to calculate the performance.
- 4.4 The Supplier shall include in its CMP suitable commentary and narrative statements in respect of the report performance, including specifying any reasons for reported failures.
- 4.5 The Customer will use the data presented by the Supplier and the CMP and LD to inform Supplier Reviews and contract management discussions, progress against Milestones and Performance Measures, auditing, reporting performance to other government departments, for Quality Assurance purposes as well as to support an independent evaluation of the Supplier. The Customer will also use independent evaluation of the Supplier to help assess whether the Supplier has achieved the relevant Delivery Plan Trajectory as set out in the Delivery Plan. The Customer will share data and the contents of the CMP for these purposes.
- 4.6 The Supplier ensures it (and its Sub-Contractors do likewise) receives the necessary consent and puts in place data sharing agreements where necessary (with Learners, Sub-Contractors, third parties and the like), in relation to the collation and sharing of any data that is reported in its CMP and LD.
- 4.7 In addition to the Contract Monitoring, the Supplier must respond to ad-hoc requests for information relating to the Activities and or Services from the Customer. Suppliers should note a completed delivery plan (which sets out the delivery trajectories) will need to be provided at contract mobilisations requested by contract managers.
- 4.8 The Authority may require additional information from a supplier outside these stated requirements to validate the service requirements are being met. Any such requirement will be communicated as part of contract management.
- 4.9 Where any discrepancies, omissions or errors are identified in the CMP or LD, the Supplier shall correct and rectify the CMP or LD within 5 (five) Working Days of the date on which the discrepancy, omission or error was identified.
- 4.10 For the avoidance of doubt, the Supplier is responsible for the collection of data including data collected by Sub-Contractors and ensuring it is collected in a timely, secure, consistent and compliant manner.
- 4.11 As the Supplier will be required to provide personal information about Learners, the Supplier must ensure this data collection template is returned to the Customer using the encrypted system; Galaxkey.
- 4.12 Contract Performance will continue to be assessed beyond the contract end date for the Outcome payment milestone and performance measures relating to this.

5 DELIVERY PLAN TRAJECTORIES

- 5.1 In accordance with the Performance Measures set out at Part 2 of this Schedule, the Customer will assess if the Supplier has achieved the Delivery Plan trajectories. If the Supplier does not achieve a Delivery Plan Trajectory, the Customer reserves the right to apply any or all of the following:
 - (a) immediately require the Supplier to stop all or any part of the Services or not to continue to progress to deliver any further element of the Services (including those being provided under other call off contracts);
 - (b) require the Supplier to:
 - (i) revise and resubmit their Delivery Plan
 - (ii) reschedule any activity;

- (iii) attend additional Supplier Review meetings with the Customer, submit reports, report on progress, provide additional resources and take the necessary action to provide assurances to the Customer that the failure to achieve the Delivery Plan Trajectory will not adversely affect the overall delivery of the Services within the stated timescale for the Skills Bootcamp, or any amended timescale as agreed by the Customer;
- (iv) recover any Contract Charges already paid to the Supplier in relation to any Milestone Payment the requirements for which were not actually achieved by the Supplier prior to the Milestone Payment having been made;
- (v) If the Supplier fails to achieve any one of the Delivery Plan Trajectories and at the same time is delivering Services under another call off contract, the Customer reserves the right to apply any or all of the following in relation to that call off contract:
- (vi) immediately require the Supplier to stop all or any part of the Services or not to continue to progress to deliver any further element of the Services, including delaying the Contract Commencement Date;
- (c) require the Supplier to:
 - (i) revise and resubmit their Delivery Solution;
 - (ii) reschedule any activity;
 - (iii) attend meetings with the Customer, submit reports, report on progress, provide additional resources and take the necessary action to provide assurances to the Customer that the failure to achieve the Delivery Plan Trajectory on one call off contract will not adversely affect the overall delivery of the Services within the stated timescale for the Skills Bootcamp, or any amended timescale as agreed by the Customer in relation to any other Call Off Contract.

Annex 1: Contract Monitoring Plan (provided in tender documentation as CMP document)

CONTRACT SCHEDULE 3: CONTRACT CHARGES, PAYMENT AND INVOICING

1. GENERAL PROVISIONS

- 1.1 This Contract Schedule 3 details:
 - 1.1.1 the Contract Charges the Services under this Contract; and
 - 1.1.2 the payment terms/profile for the Contract Charges;
 - 1.1.3 the invoicing procedure; and
 - 1.1.4 the procedure applicable to any adjustments of the Contract Charges.

2. CONTRACT CHARGES

- 2.1 The Contract Charges which are applicable to this Contract are set out in Annex 1 of this Contract Schedule 3.
- 2.2 The Supplier acknowledges and agrees that the Contract Charges cannot be increased during the Contract Period.

3. COSTS AND EXPENSES

- 3.1 The Contract Charges include all costs and expenses relating to the Services and/or the Supplier's performance of its obligations under this Contract and no further amounts shall be payable by the Customer to the Supplier in respect of such performance, including in respect of matters such as:
 - 3.1.1 any incidental expenses that the Supplier incurs, including travel, subsistence and lodging, document or report reproduction, shipping, desktop or office equipment costs required by the Supplier Personnel, network or data interchange costs or other telecommunications charges; or
 - 3.1.2 any amount for any services provided or costs incurred by the Supplier prior to the Contract Commencement Date.

4. PAYMENT TERMS/PAYMENT PROFILE

4.1 The payment terms/profile which are applicable to this Contract are set out in Annex 2 of this Contract Schedule 3.

5. PAYMENT AND INVOICING PROCEDURES

- 5.1 Paragraphs 5.2 to 5.4 shall apply where the Customer is the Department for Education.
- 5.2 The Payment Profile for the Call Off Contract will be based upon the Milestones outlined in Part A of Schedule 3 (Pricing and Charging Structure) of the DPA which will be amended to reflect the detailed specific category requirement specified in the Call for Competition Statement of Requirements.
- 5.3 The Supplier must submit promptly to the Customer an accurate Individualised Learning Record ("ILR") in relation to each individual Learner in accordance with this Contract and the ILR technical documents, guidance and requirements as amended and updated from time to time.
- 5.4 Provided that the Supplier has complied with its obligation to complete the ILR, as set out in clause 14 and has fulfilled the criteria for the relevant Milestone Payment, the Customer shall pay the Milestone Payment on the 14th Working Day of the month following the date on which the Supplier submitted data onto the ILR related to the Milestone Payment. The Supplier must submit data by the 4th Working Day of the month to be paid on the 14th Working Day of

that same month, otherwise the Supplier will be paid on the 14th Working Day of the following month. For the avoidance of doubt, there is no requirement for the Supplier to submit an invoice for a Milestone Payment.

- 5.5 The provisions in clauses 14.14 14.16 apply in relation to the Contract Charges and the Customer may reduce and/or recover Contract Charges accordingly.
- 5.6 Paragraphs 5.7 5.12 shall apply where the Customer is not the Department for Education unless that Customer specifies an alternative payment and/or invoicing process in the Call for Competition.
- 5.7 The Customer shall pay all sums properly due and payable to the Supplier in cleared funds within thirty (30) days of receipt of a Valid Invoice, submitted to the address specified by the Customer and in accordance with the provisions of this Contract.
- 5.8 The Supplier shall ensure that each invoice (whether submitted electronically through a purchaseto-pay (P2P) automated system (or similar) or in a paper form, as the Customer may specify (but, in respect of paper form, subject to paragraph 5.3 below)):
 - 5.8.1 contains:
 - (a) all appropriate references, including the unique order reference number set out in the Contract Order Form; and
 - (b) a detailed breakdown of the Services, including the Milestone(s) (if any) and Deliverable(s) within this Contract to which the Services relate, against the applicable due and payable Contract Charges; and
 - 5.8.2 shows separately:
 - (a) any form of Service Credits due to the Customer; and
 - (b) the VAT added to the due and payable Contract Charges in accordance with Clause 14.22 of this Contract (VAT) and the tax point date relating to the rate of VAT shown; and
 - 5.8.3 is exclusive of any Management Charge (and the Supplier shall not attempt to increase the Contract Charges or otherwise recover from the Customer as a surcharge the Management Charge levied on it by the Authority); and
 - 5.8.4 it is supported by any other documentation reasonably required by the Customer to substantiate that the invoice is a Valid Invoice.
- 5.9 If the Customer is a Central Government Body, the Customer's right to request paper form invoicing shall be subject to procurement policy note 11/15 (available at <u>Procurement policy</u> <u>note 11/15: unstructured electronic invoices Publications GOV.UK</u> which sets out the policy in respect of unstructured electronic invoices submitted by the Supplier to the Customer (as may be amended from time to time).
- 5.10 The Supplier shall accept the Government Procurement Card as a means of payment for the Services where such card is agreed with the Customer to be a suitable means of payment. The Supplier shall be solely liable to pay any merchant fee levied for using the Government Procurement Card and shall not be entitled to recover this charge from the Customer.
- 5.11 All payments due by one Party to the other shall be made within thirty (30) days of receipt of a Valid Invoice unless otherwise specified in this Contract, in cleared funds, to such bank or building society account as the recipient Party may from time to time direct.
- 5.12 The Supplier shall submit invoices directly to the Customer's billing address set out in the Contract Order Form.

6. ADJUSTMENT OF CONTRACT CHARGES

- 6.1 The Contract Charges shall only be varied:
 - 6.1.1 due to a Specific Change in Law in relation to which the Parties agree that a change is required to all or part of the Contract Charges in accordance with Clause 13.1 to 13.2 of this Contract (Legislative Change); or
 - 6.1.2 and/ or in accordance with DMP Schedule 3 (DMP Prices and Charging structure), Part A Pricing Matrix Quarterly Adjustment Schedule.

ANNEX 1: CONTRACT CHARGES

The Contract Charges are as set out in the Summary of Funding taken from the Suppliers Pricing Sheet.

ANNEX 2: PAYMENT TERMS/PROFILE

1.1 Payment by the Customer of the Contract Charges is to be made in accordance with the payment profile set out in the Milestone Table below:

PAYMENT PROFILE: MILESTONE TABLE

For the following categories: Digital; Sales, Marketing and Procurement; Construction and the Built Environment; Engineering and Manufacturing; Business and Administration; and Creative and Design

Commencement	Contract Delivery		Outcome
	Mid-course	Course completion and interview offered	On job offer (or equivalent)
40%	0%	30%	30%

Paid on the learner completing 14 qualifying days of training and on completion of the initial assessment as set out in the service requirements). We expect the 14 qualifying days of the training course to include a minimum of 10 guided learning hours. Suppliers must achieve 20% of Starts by 31 st March 2024, in line with delivery plans. The Supplier should report on the next monthly reporting cycle any new learner Starts via the ILR, Contract Monitoring Plan and Learner Datasheet. More detail on reporting is set out in the funding guidance.	 Course completion and interview offered: Completion of the training and completion of a final assessment (final assessment as set out in the service requirements), and; For the independent learner: an offer of an interview with an employer for a live vacancy that will utilise the skills botained on the Skills Bootcamp, and which meets the criteria set out under the service requirements and reflects the salary expectations and location set out in the IAG process; For the co-funded learner: an offer of an interview which meets the criteria set out under the service requirements For the co-funded learner: an offer of an interview which meets the criteria set out under the service requirements For the self-employed learner: written confirmation of how the new training has been/will be applied to get new work or contracts which meet the criteria set out under the service requirements If an independent learner plans to be selfemployed, they can achieve a self-employed learner completion; and a self-employed learner who plans to become employed can achieve an independent learner completion. Suppliers must achieve Completions by 31st July 2025, and have outcomes achieved by 31st January 2026 in line with delivery plans. The Supplier should report on the next monthly reporting cycle any new learner Completions via the ILR, Contract Monitoring Plan and Learner Datasheet. 	 For the independent learner: An offer for a job which meets the criteria set out in the Contract Order Form Annex 1 service requirements. For the co-funded learner: the offer or commencement of a new role/responsibility within the current organisation which meets the criteria set out in the Contract Order Form Annex 1 service requirements. For the self-employed learner: Learner has secured new work/new contracts which meet the criteria set out in the Contract Order Form Annex 1 service requirements. If an independent learner plans to be selfemployed, they can achieve a self- employed learner outcome; and a self- employed learner outcome. Suppliers should note that the learner must have finished the training element of the Skills Bootcamp for the Supplier to be able to claim a Positive Outcome payment. Suppliers must achieve Positive Outcomes within the six months after the training finishes. The Supplier should report on the next monthly reporting cycle any new learner Positive Outcomes via the ILR, Contract Monitoring Plan and Learner Datasheet.

Pathways to Accelerated Apprenticeships

Commencement	Contract Delivery		Outcome	
	Mid-course	Course completion and interview offered	On job offer (or equivalent)	
40%	0%	30%	30%	
Paid on the learner completing 14 qualifying days of training and on completion of the initial assessment (initial assessment as set out in the service requirements). We expect the 14 qualifying days of the training course to include a minimum of 10 guided learning hours. Suppliers must achieve 20% of Starts by 31 st March 2024, in line with delivery plans. The Supplier should report on the next monthly reporting cycle any new learner Starts via the ILR, Contract Monitoring Plan and Learner	offered: Completion of a assessment as requirements), For the i offer of a employe or other will utilis the boot criteria s requirem salary ex set out in For the o offer of a the criter service r For the s written c new train	etion and interview the training and a final assessment (final s set out in the service and; ndependent learner: an an interview with an er for an apprenticeship live job vacancy which e the skills obtained on camp and meets the set out under the service nents and reflects the xpectations and location in the IAG process; co-funded learner: an an interview which meets ria set out under the requirements self-employed learner: confirmation of how the ning has been/will be to get new work or	 For the independent learner: An offer for an accelerated apprenticeship or for a job (which is not an apprenticeship) that meets the criteria set out in the Contract Order Form Annex 1 service requirements. For the co-funded learner: the offer or commencement of a new role/responsibility within the current organisation which meets the criteria set out in the Contract Order Form Annex 1 service requirements For the self-employed learner: Learner has secured new work/new contracts which meet the criteria set out in the Contract Order Form Annex 1 service requirements. If an independent learner plans to be self-employed learner outcome; and a selfemployed learner who plans to become 	

Datasheet. More detail on reporting is set out in the funding guidance.	contracts which meet the criteria set out under the service requirements If an independent learner plans to be self-employed, they can achieve a self- employed learner completion; and a self-employed learner who plans to become employed can achieve an independent learner completion.	employed can achieve an independent learner outcome. The offer of an apprenticeship that has not been accelerated cannot be used to claim for the Outcome payment in this model. Suppliers should note that the learner must have completed the training element of the Skills Bootcamp to be able to claim an outcome payment.
	Suppliers must achieve Completions by 31 st July 2025, and have outcomes achieved by 31 st January 2026 in line with delivery plans. The Supplier should report on the next monthly reporting cycle any new learner Completions via the U.B. Contract	Suppliers must achieve Positive Outcomes within the six months after the training finishes. The Supplier should report on the next monthly reporting cycle any new learner Positive Outcomes via the ILR, Contract Monitoring Plan and Learner Datasheet.

For Category Transport and Logistics – HGV Driving (Pathway A-C)

Commencement	Contract Delivery		Outcome
Eligibility & suitability	Mid-course	Course completion and interview offered	On job offer (or equivalent)
20%	30%	20%	30%

Eligibility and suitability checks: including attained provisional licence.	Passed Theory test, completed the practical training and taken a practical test.	Passed Practical Test and passed CPC mod 4 and evidence of job interview.	Evidence of a job offer / commencement of new role / self-employed outcome: • For the independent
Suppliers must achieve 20% of Starts by 31 st March 2024, in line with delivery plans. The Supplier should report on the next monthly reporting cycle any new learner Starts via the ILR, Contract Monitoring Plan and Learner Datasheet. More detail on reporting is	Completion of the training passed the theory test and taken a practical test.	Passed the practical test and passed the CPC mod 4, and: • For the independent learner: an offer of an interview with an employer for a live vacancy which will utilise the skills obtained	 learner: An offer for a job which meets the criteria set out in the Contract Order Form Annex 1 service requirements. For the co-funded learner: the offer or commencement of a new role/responsibility within the current organisation which meets the criteria set out in the service requirements.

set out in the funding guidance.	on the Skills Bootcamp and meets the criteria set out under the service requirements and reflects the salary expectations and location set out in the Initial Advice and Guidance (IAG) process; • For the cofunded learner: an offer of an interview which meets the criteria set out under the service requirements • For the selfemployed learner: written confirmation of how the new training has been/will be applied to get new work or contracts which meet the criteria set out under the service requirements If an independent learner plans to be self-employed learner completion; and a self-employed learner who plans to become employed can achieve an independent learner completion.	 For the self-employed learner: Learner has secured new work/new contracts which meet the criteria set out in the service requirements. If an independent learner plans to be self-employed, they can achieve a selfemployed learner outcome; and a self-employed learner who plans to become employed can achieve an independent learner outcome. Suppliers should note that the learner must have finished the training element of the Skills Bootcamp for the Supplier to be able to claim a Positive Outcome payment. Suppliers must achieve Positive Outcomes within the six months after the training finishes. The Supplier should report on the next monthly reporting cycle any new learner Positive Outcomes via the ILR, Contract Monitoring Plan and Learner Datasheet.

]
	Suppliers must achieve Completions by 31 st July 2025,	
	by 01 - 001y 2020,	

	and have outcomes achieved by 31 st January 2026 in line with delivery plans. The Supplier should report on the next monthly reporting cycle any new learner Completions via the ILR, Contract Monitoring Plan and Learner Datasheet.	
--	---	--

For Category Transport and Logistics - HGV driving (Pathway D-H) and (Transport Manager)

Commencement	Contract Delivery		Outcome
	Mid-course	Course completion and interview offered	On job offer (or equivalent)
40%	0%	30%	30%

Paid on the learner completing 14 qualifying days of training and on completion of the initial assessment as set out in the service requirements). We expect the 14 qualifying days of the training course to include a minimum of 10 guided learning hours. Suppliers must achieve 20% of Starts by 31 st March 2024, in line with delivery plans. The Supplier should report on the next monthly reporting cycle any new learner Starts via the ILR, Contract Monitoring Plan and Learner	 Course completion and interview offered: Completion of the training and completion of a final assessment (final assessment as set out in the service requirements), and; For the independent learner: an offer of an interview with an employer for a live vacancy that will utilise the skills obtained on the Skills Bootcamp and which meets the criteria set out under the service requirements and reflects the salary expectations and location set out in the IAG process; For the co-funded learner: an offer of an interview which meets the criteria set out under the service requirements. For the self-employed learner: written confirmation of how the new training has been/will be applied to get new work or contracts which meet the criteria set out under the service requirements. If an independent learner plans to be selfemployed, they can achieve a self- 	 For the independent learner: An offer for a job which meets the criteria set out in the Contract Order Form Annex 1 service requirements For the co-funded learner: the offer or commencement of a new role/responsibility within the current organisation which meets the criteria set out in the Contract Order Form Annex 1 service requirements For the self-employed learner: Learner has secured new work/new contracts which meet the criteria set out in the Contract Order Form Annex 1 service requirements
Datasheet. More detail on reporting is set out in the funding guidance. guidance.	employed learner completion; and a selfemployed learner who plans to become employed can achieve an independent learner completion. Suppliers must achieve Completions by 31 st July 2025, and have outcomes achieved by 31 st January 2026 in line with delivery plans. The Supplier should report on the next monthly reporting cycle any new learner Completions via the ILR, Contract Monitoring Plan and Learner Datasheet.	If an independent learner plans to be self-employed, they can achieve a selfemployed learner outcome; and a self-employed learner who plans to become employed can achieve an independent learner outcome. Suppliers should note that the learner must have finished the training element of the Skills Bootcamp for the Supplier to be able to claim a Positive Outcome payment. Suppliers must achieve Positive Outcomes within the six months after the training finishes. The Supplier should report on the next monthly reporting cycle any new learner Positive Outcomes via the ILR, Contract Monitoring Plan and Learner Datasheet.

For Category Education and Early Years (Early Years Educator)

Commencement	Contract Delivery		Outcome
	Mid-course	Course completion and interview offered	On job offer (or equivalent)
40%	0%	30%	30%
 Paid on: the learner completing 14 qualifying days of training, on completion of the initial assessment (initial assessment as set out in the service requirement s) and, 	offered: Completion of a completion of a (final assessmu- service require • For the learner intervie for an appren job (wh appren that wi obtaine Bootca	etion and interview the training and a final assessment ent as set out in the ements), and; e independent r: an offer of an ew with an employer accelerated nticeship or other live nich is not an nticeship) vacancy Il utilise the skills ed on the Skills amp and which meets teria set out in the	 For the independent learner: An offer for an accelerated apprenticeship or for a job (which is not an apprenticeship) that meets the criteria set out in the Contract Order Form Annex 1 service requirements. For the co-funded learner: the offer or commencement of a new role/responsibility within the current organisation which meets the criteria set out in the Contract Order Form Annex 1 service requirements

 evidence of 	service requirements and	• For the self-employed learner:
DBS	reflects the salary	Learner has secured new
applications	expectations and location set	work/new contracts which
submitted for	out in the IAG process;	meet the criteria set out in the
learners or an	• For the co-funded learner:	Contract Order Form Annex 1
existing and	an offer of an interview for a	service requirements.
relevant	new role/responsibility within	
DBS	the current organisation	If an independent learner plans to be
certificate.	which meets the criteria set	self-employed, they can achieve a
	out in the service	self-employed learner outcome; and a
We expect the 14	requirements;	self-employed learner who plans to
qualifying days of the	 For the self-employed 	become employed can achieve an
training course to	learner: written confirmation	independent learner outcome.
include a minimum of	of how the new training has	
10 guided learning	been/will be applied to get	The offer of an apprenticeship that has
hours.	new work or contracts which	not been accelerated cannot be used
	meet the criteria set out in	to claim for the Outcome payment in
Suppliers must	the service requirements.	this model. Suppliers should note that
achieve 20% of		the learner must have completed the
	If an independent learner plans to be	training element of the Skills Bootcamp
	self-employed, they can achieve a	to be able to claim an outcome
	self-employed learner completion;	payment.
	and a self-employed learner who	
	plans to become employed can	Suppliers must achieve Positive
	achieve an independent learner	Outcomes within the six months after
new learner Starts via	completion.	the training finishes. Please note this is
the ILR, Contract		not six months after the full Completion
Monitoring Plan and	Suppliers must achieve	milestone, but after the training
Learner Datasheet.	Completions by 31 st July 2025, and	element has finished. The Supplier
More detail on	have outcomes achieved by 31 st	should report on the next monthly
reporting is set out in	January 2026 in line with delivery	reporting cycle any new learner
the funding guidance.	plans. The Supplier should report on	Positive Outcomes via the ILR,
	the next monthly reporting cycle any	Contract Monitoring Plan and Learner
	new learner Completions via the ILR,	Datasheet.
	Contract Monitoring Plan and	
	Learner Datasheet.	
	1	

- 1.2 At each Milestone Date specified in the above table, the Customer will assess if the Supplier has achieved the relevant Milestone. If the Supplier does not achieve a Milestone, the Customer reserves the right to apply any or all of the following:
 - 1.2.1 immediately require the Supplier to stop or not start any part of the Services (including those being provided under other call off contracts).
 - 1.2.2 require the Supplier to:
 - 1.2.2.1 revise and resubmit their Delivery and Implementation Plan;
 - 1.2.2.2 reschedule any activity;
 - 1.2.2.3 attend meetings with the Customer and/or its QA Function, submit reports, report on progress, provide additional resources and take the necessary action to provide assurances

to the Customer that the Milestone will be achieved by a revised Milestone Date with time being of the essence;

- 1.2.3 withhold any Contract Charges that would have been due to the Supplier if the Supplier had achieved the Milestone;
- 1.2.4 recover any Contract Charges already paid to the Supplier.
- 1.3 If the Supplier fails to achieve any one of the Milestones and at the same time is delivering Services under another call off contract, the Supplier reserves the right to also take action under that call off contract.
- 1.4 The customer reserves the right to review the milestones and where it does so will confirm the requirements.

2. The following will not be accepted as an offer of an interview :

2.1 Where:

2.1.1 Learners are invited to employer meet and greet sessions including but not limited to an employer insight day

2.1.2 Learners are invited to meet employers as part of an initial screening process

3. Interviews by recruitment agencies will only be accepted where:

(I) the interview is for live vacancies and the recruitment agency would be the employer and or contractor for the learner if the learner is successful

OR

(ii) where employers engaged by the supplier have contracted with a recruitment agency to interview candidates for relevant job vacancies/roles on their behalf

3.1 If suppliers cannot provide either of these pieces of evidence at milsetone 2, we will only pay milestone 2 once milestone 3 has been evidenced in line with agreed time limits.

4. Suppliers must provide evidence of the below for payment for payment milestone 2:

(I) Confirmation sent to the supplier from the employer or apprenticeship provider that the learner has been offered an interview with an employer to include details of company name, job role, date and time of interview. e.g. email exchange of employer engagement on an interview OR

(ii) In the case where a learner has independently arranged an interview, a declaration from the learner that they independently arranged an interview confirmation from the learner of the details of the interview providing one of the below:

(a)Learner declaration (signed by the Learner) confirming the details of the interview (company name, job role, date and time of interview)

(b) Voice recording of conversation between Supplier and Learner confirming the details of the interview (company name, job role, date and time of interview)

(c)Email from Learner confirming the details of the interview (company name, job role, date and time of interview

(d)Text (or other message service) from Learner confirming the details of the interview(company name, job role, date and time of interview)

5. The Supplier must have agreed a sufficient number of job interviews with Employers to (I) provide all Learners with an offer for a Job Interview and (ii) meet the payment milestone 3 KPI

CONTRACT SCHEDULE 4: SECURITY

1.	DEFINITIONS	
	"BPSS"	a level of security clearance described as pre-employment
	"Baseline Contractor Staff Security Standar	d" checks in the National Vetting Policy. Further information can be found at:
		https://www.gov.uk/government/publications/government
	"****	-baseline-Contractor Staff-security-standard
	"CCSC" "Certified Cyber Security Consultancy"	is NCSC's approach to assessing the services provided by consultancies and confirming that they meet NCSC's standards. This approach builds on the strength of CLAS and
		certifies the competence of suppliers to deliver a wide and
		complex range of cyber security consultancy services to both the public and private sectors. See website:
		https://www.ncsc.gov.uk/scheme/certified-
		<u>cyberconsultancy</u>
	"CCP"	is a NCSC scheme in consultation with government, industry
	"Certified Professional"	and academia to address the growing need for specialists in the cyber security profession and are building a community
		of recognised professionals in both the UK public and private sectors. See website:
		https://www.ncsc.gov.uk/scheme/certified-professional
	"CC"	the Common Criteria scheme provides assurance that a
	"Common Criteria"	developer's claims about the security features of their product are valid and have been independently tested against recognised criteria.
	"Cloud Security Principles"	Is a set of guidance on cloud security principles, including
		their goals and technical implementation:
		https://www.ncsc.gov.uk/collection/cloud/the-
		<u>cloudsecurity-principles</u>
	"CPA"	is an 'information assurance scheme' which evaluates
	"Commercial Product Assurance"	commercial off the shelf (COTS) products and their developers against published security and development
	[formerly called "CESG Product Assurance"	standards. These CPA certified products can be used by
		government, the wider public sector and industry. See website:
		https://www.ncsc.gov.uk/scheme/commercialproduct-
		assurance-cpa
	"Cyber Essentials"	Cyber Essentials is the government backed, industry
	"Cyber Essentials Plus"	supported scheme to help organisations protect themselves against common cyber-attacks. Cyber Essentials and Cyber Essentials Plus are levels within the scheme.
		There are a number of certification bodies that can be approached
		for further advice on the scheme; the link below points to one of
		these providers: <u>https://www.iasme.co.uk/apply-for-self-</u>
		<u>assessment/</u>

"Department's Data" "Department's Information"	 is any data or information owned or retained in order to meet departmental business objectives and tasks, including: (a) any data, text, drawings, diagrams, images or sounds (together with any repository or database made up of any of these components) which are embodied in any electronic, magnetic, optical or tangible media, and which are: (i) supplied to the Supplier by or on behalf of the Customer; or (ii) which the Supplier is required to generate, process, store or transmit pursuant to this Contract; or
	 (b) any Personal Data for which the Department is the Data Controller;
"Authority", "Customer" "Department"	means the Department for Education
"Departmental Security Standards"	means the Customer's security policy or any standards, procedures, process or specification for security that the Supplier is required to deliver.
"Digital Marketplace / GCloud"	the Digital Marketplace is the online framework for identifying and procuring cloud technology and people for digital projects. Cloud services (e.g. web hosting or IT health checks) are on the G-Cloud framework.
"FIPS 140-2"	this is the Federal Information Processing Standard (FIPS) Publication 140-2, (FIPS PUB 140-2), entitled 'Security Requirements for Cryptographic Modules'. This document is the de facto security standard used for the accreditation of cryptographic modules.
"Good Industry Practice" "Industry Good Practice"	means the exercise of that degree of skill, care, prudence, efficiency, foresight and timeliness as would be expected from a leading company within the relevant industry or business sector.
"Good Industry Standard" "Industry Good Standard"	means the implementation of products and solutions, and the exercise of that degree of skill, care, prudence, efficiency, foresight and timeliness as would be expected from a leading company within the relevant industry or business sector.

"GSC" "GSCP"	means the Government Security Classification Policy which establishes the rules for classifying HMG information. The policy is available at: <u>https://www.gov.uk/government/publications/government</u> <u>-security-classifications</u>
"Guided Learning"	'Guided Learning is the activity of a Learner being taught or instructed by – or otherwise participating in education or training (including placement, onsite or practical education or training) under the Immediate Guidance or Supervision of – a lecturer, supervisor, tutor or other appropriate provider of education or training, with the simultaneous presence of the Learner and that person in the same physical or virtual space. This could include, but is not limited to, simultaneous presence: in a classroom, in a virtual classroom or breakout room, on site, placement, or other physical learning environment.'
"HMG"	means His Majesty's Government
"ICT"	means Information and Communications Technology (ICT) is used as an extended synonym for information technology (IT), used to describe the bringing together of enabling technologies used to deliver the end-to-end solution
"Interview"	A job interview is an interview consisting of a conversation between a job applicant and a representative of an employer which is conducted to assess whether the applicant should be hired.
"ISO/IEC 27001" "ISO 27001"	is the International Standard for Information Security Management Systems Requirements
"ISO/IEC 27002" "ISO 27002"	is the International Standard describing the Code of Practice for Information Security Controls.
"ISO 22301"	is the International Standard describing for Business Continuity
"IT Security Health Check (ITSHC)" "IT Health Check (ITHC)" "Penetration Testing"	means an assessment to identify risks and vulnerabilities in systems, applications and networks which may compromise the confidentiality, integrity or availability of information held on that IT system.
"Need-to-Know"	the Need-to-Know principle is employed within HMG to limit the distribution of classified information to those people with a clear 'need to know' in order to carry out their duties.

"NCSC"	The National Cyber Security Centre (NCSC) formerly CESG is the UK government's National Technical Authority for Information Assurance. The NCSC website is <u>https://www.ncsc.gov.uk</u>
"OFFICIAL"	the term 'OFFICIAL' is used to describe the baseline level of
"OFFICIAL-SENSITIVE"	'security classification' described within the Government Security Classification Policy (GSCP) which details the level of protection to be afforded to information by HMG, for all routine public sector business, operations and services.
	The 'OFFICIAL–SENSITIVE' caveat is used to identify a limited subset of OFFICIAL information that could have more damaging consequences (for individuals, an organisation or government generally) if it were lost, stolen or published in the media, as described in the Government Security Classification Policy.
"RBAC"	Role Based Access Control, a method of restricting a
"Role Based Access Control"	person's or process' access to information depending on the role or functions assigned to them.
"Secure Sanitisation"	Secure sanitisation is the process of treating data held on storage media to reduce the likelihood of retrieval and reconstruction to an acceptable level. Some forms of
sanitisation will allow you to re-use the media,	while others are destructive in nature and render the media unusable.
	Secure sanitisation was previously covered by "Information Assurance Standard No. 5 - Secure Sanitisation" ("IS5") issued by the former CESG. Guidance can now be found at: <u>https://www.ncsc.gov.uk/guidance/secure-</u> <u>sanitisationstorage-media</u>
	The disposal of physical documents and hardcopy materials advice can be found at: <u>https://www.cpni.gov.uk/securedestruction</u>
"Security and Information Risk Advisor"	the Security and Information Risk Advisor (SIRA) is a role
"CCP SIRA"	defined under the NCSC Certified Professional (CCP)
"SIRA" https://www.ncsc.gov.uk/articles/about-	Scheme. See also: certifiedprofessional-scheme
"Service Standards"	The Service Standard helps teams to create and run great public services ensuring user needs are at the centre of design https://www.gov.uk/service-manual/service-standard
"SPF"	This is the definitive HMG Security Policy which describes
"HMG Security Policy Framework"	the expectations of the Cabinet Secretary and Government's Official Committee on Security on how HMG organisations and third parties handling HMG information and other assets will apply protective security to ensure HMG

can function effectively, efficiently and securely. https://www.gov.uk/government/publications/securitypolicyframework "Technology Code of Practice" The Technology Code of Practice is a set of criteria to help government design, build and operate technology services effectively and efficiently. https://www.gov.uk/guidance/the-technology-code-ofpractice "Tailored Assurance" is an 'information assurance scheme' which provides [formerly called "CTAS", or, assurance for a wide range of HMG, MOD, Critical National Infrastructure (CNI) and public sector customers procuring "CESG Tailored Assurance"] IT systems, products and services, ranging from simple software components to national infrastructure networks. https://www.ncsc.gov.uk/documents/ctas-principlesandmethodology

- 1.1. The Supplier shall comply with the relevant HMG security policy framework, NCSC guidelines, Cloud Security Principles, and where applicable Departmental Security Standards which include but are not constrained to the following paragraphs.
- 1.2. If the Supplier is providing a Technology Solution and / or as the Supplier will be handling information at OFFICIAL on behalf of the Customer, the requirements under Cabinet Office Procurement Policy Note Use of Cyber Essentials Scheme certification Action Note 09/14 25 May 2016, or any subsequent updated document, are mandated; that "contractors supplying products or services to HMG shall have achieved, and retain certification at the appropriate level, under the HMG Cyber Essentials Scheme". The certification scope must be relevant to the services supplied to, or on behalf of, the Customer.
- 1.3. The Supplier shall be able to demonstrate conformance to, and show evidence of such conformance to the ISO/IEC 27001 (Information Security Management Systems Requirements) standard, including the application of controls from ISO/IEC 27002 (Code of Practice for Information Security Controls).
- 1.4. The Supplier shall follow the UK Government Security Classification Policy (GSCP) in respect of any Departmental Data being handled in the course of providing these Services, and will handle this data in accordance with its security classification. (In the event where the Supplier has an existing Protective Marking Scheme then the Supplier may continue to use this but must map the HMG security classifications against it to ensure the correct controls are applied to the Departmental Data).
- 1.5. Departmental Data being handled in the course of providing an ICT solution or service must be segregated from all other data on the Supplier's or Sub-Contractor's own IT equipment to protect the Departmental Data and enable the data to be identified and securely deleted when required. In the event that it is not possible to segregate any Departmental Data then the Supplier and any SubContractor shall be required to ensure that it is stored in such a way that it is possible to securely delete the data in line with Paragraph 1.16.
- 1.6. The Supplier shall have in place and maintain physical security, in line with those outlined in ISO/IEC 27002 including, but not limited to, entry control mechanisms (e.g. door access) to premises and sensitive areas.
- 1.7. The Supplier shall have in place and maintain an access control policy and process for the logical access (e.g. identification and authentication) to ICT systems to ensure only authorised Staff have access to Departmental Data. This policy should include appropriate segregation of duties and if applicable role based access controls (RBAC). User credentials that give access to Departmental Data or systems shall be considered to be sensitive data and must be protected accordingly.

- 1.8. The Supplier shall have in place and shall maintain procedural, Staff, physical and technical safeguards to protect Departmental Data, including but not limited to:
 - i. physical security controls;
 - ii. good industry standard policies and processes;
 - iii. malware protection;
 - iv. boundary access controls including firewalls, application gateways, etc;
 - v. maintenance and use of fully supported software packages in accordance with vendor recommendations;
 - vi. use of secure device configuration and builds;
 - vii. software updates and patching regimes including malware signatures, for operating systems, network devices, applications and services;
 - viii. user identity and access controls, including the use of multi-factor authentication for sensitive data and privileged account accesses;
 - ix. Services provided to the Customer must capture audit logs for security events in an electronic format at the application, service and system level to meet the Customer's logging and auditing requirements, plus logs shall be:
 - a. retained and protected from tampering for a minimum period of six months;
 - b. made available to the Customer on request.
- 1.9. Any data in transit using either physical or electronic transfer methods across public space or cyberspace, including mail and courier systems, or third party provider networks must be protected via encryption which has been certified to FIPS 140-2 standard or a similar method approved by the Customer prior to being used for the transfer of any Departmental Data.
- 1.10. Storage of Departmental Data on any portable devices or media shall be limited to the absolute minimum required to deliver the stated business requirement and shall be subject to Paragraph 1.11 and 1.12 below.
- 1.11. Any portable removable media (including but not constrained to pen drives, flash drives, memory sticks, CDs, DVDs, or other devices) which handle, store or process Departmental Data to deliver and support the service, shall be under the control and configuration management of the Suppler or SubContractors providing the Services, shall be both necessary to deliver the Services and shall be encrypted using a product which has been certified to FIPS140-2 standard or another encryption standard that is acceptable to the Customer.
- 1.12. All portable ICT devices, including but not limited to laptops, tablets, smartphones or other devices, such as smart watches, which handle, store or process Departmental Data to deliver and support the Services, shall be under the control and configuration management of the Supplier or Sub-Contractors providing the Services, and shall be necessary to deliver the Services. These devices shall be full-disk encrypted using a product which has been certified to FIPS140-2 standard or another encryption standard that is acceptable to the Customer.
- 1.13. Whilst in the Supplier's care all removable media and hardcopy paper documents containing Departmental Data must be handled securely and secured under lock and key when not in use and shall be securely destroyed when no longer required, using either a cross-cut shredder or a professional secure disposal organisation.
- 1.14. When necessary to hand carry removable media and/or hardcopy paper documents containing Departmental Data, the media or documents being carried shall be kept under cover and transported in such a way as to ensure that no unauthorised person has either visual or physical access to the material being carried. This requirement shall apply equally regardless of whether the material is being carried inside or outside of company premises. The term 'under cover' means that the information is

carried within an opaque folder or envelope within official premises and buildings and within a closed briefcase or other similar bag or container when outside official premises or buildings.

- 1.15. In the event of termination of the Contract due to expiry, as a result of an Insolvency Event or for Default by the Supplier, all products and materials provided, created or resulting from provision of the Services shall not be considered as the Supplier's assets and must be returned to the Customer and written assurance obtained from an appropriate officer of the Supplier that these assets regardless of location and format have been fully sanitised throughout the Supplier's organisation in line with paragraph 1.16.
- 1.16. At the end of the Contract Period or in the event of equipment failure or obsolescence, all Departmental information and data, in either hardcopy or electronic format, that is physically held or logically stored on the Supplier's ICT infrastructure must be securely sanitised or destroyed and accounted for in accordance with the current HMG policy using a NCSC approved product or method. Where sanitisation or destruction is not possible for legal, regulatory or technical reasons, such as a Storage Area Network (SAN) or shared backup tapes, then the Supplier or Sub-Contractor shall protect the Customer's information and data until the time, which may be long after the end of the contract, when it can be securely cleansed or destroyed. Evidence of secure destruction will be required in all cases.
- 1.17. In the event of termination of Contract due to expiry, the Supplier will ensure that any purchased domain name (or domain names) in connection with the provision of the Services transfers ownership of those domains to the Customer free of charge as part of Exit Plan activity. Agreement on domain name will be made between the Government Digital Service and the Customer.
- 1.18. Access by the Supplier or Sub-Contractor Staff to Departmental Data shall be confined to those individuals who have a "need-to-know" in order to carry out their role; and have undergone mandatory pre-employment screening, to a minimum of HMG Baseline Contractor Staff Security Standard (BPSS); or hold an appropriate National Security Vetting clearance as required by the Customer. All Supplier or Sub-Contractor Staff must complete this process before access to Departmental Data is permitted. Any Supplier Staff who will be in contact with children or vulnerable adults must, in addition to any security clearance, have successfully undergone an Enhanced DBS (Disclosure and Barring Service) check prior to any contact.
- 1.19. All Supplier or Sub-Contractor Staff who handle Departmental Data must have annual awareness training in protecting information.
- 1.20. Notwithstanding any other provisions as to business continuity and disaster recovery in the Contract, the Supplier shall, as a minimum, have in place robust Business Continuity arrangements and processes including IT disaster recovery plans and procedures that conform to ISO 22301 to ensure that the delivery of the Services is not adversely affected in the event of an incident. An incident shall be defined as any situation that might, or could lead to, a disruption, loss, emergency or crisis to the services delivered. If a ISO 22301 certificate is not available, the Supplier will provide evidence of the effectiveness of their ISO 22301 conformant Business Continuity arrangements and processes including IT disaster recovery plans and procedures. This should include evidence that the Supplier has tested or exercised these plans within the last 12 months and produced a written report of the outcome, including required actions.
- 1.21. Any suspected or actual breach of the confidentiality, integrity or availability of Departmental Data being handled in the course of providing this service, or any non-compliance with these Departmental Security Standards for Contractors, or other Security Standards pertaining to the solution, shall be recorded as an incident. This includes any non-compliance with the Departmental Security Standards and these provisions, or other security standards pertaining to the solution. Incidents shall be reported to the Customer immediately, wherever practical, even if unconfirmed or when full details are not known, but always within 24 hours of discovery. If incident reporting has been delayed by more than 24 hours, the Supplier should provide an explanation about the delay. Incidents shall be reported through the Customer's nominated system or service owner. Incidents shall be investigated by the Supplier with outcomes being notified to the Customer.

- 1.22. The Supplier shall ensure that any IT systems and hosting environments that are used to handle, store or process Departmental Data shall be subject to independent IT Health Checks (ITHC) using a NCSC approved ITHC provider before go-live and periodically (at least annually) thereafter. The findings of the ITHC relevant to the Services being provided are to be shared with the Customer and all necessary remedial work carried out. In the event of significant security issues being identified, a follow up remediation test may be required.
- 1.23. The Supplier or Sub-Contractors providing the Services will provide the Customer with full details of any storage of Departmental Data outside of the UK or any future intention to host Departmental Data outside the UK or to perform any form of ICT management, support or development function from outside the UK. The Supplier or Sub-Contractor will not go ahead with any such proposal without the prior written agreement from the Customer.
- 1.24. The Customer reserves the right to audit the Supplier or Sub-Contractors providing the Services within a mutually agreed timeframe but always within seven days of a request to audit being given. The audit shall cover the overall scope of the Services being provided and the Supplier's, and any Sub-Contractors, compliance with the clauses contained in this Section.
- 1.25. The Supplier shall use best endeavours to procure that all these Departmental Security Standards for Contractors shall be complied with by any third-party suppliers, Sub-Contractors or partners who could potentially access Departmental Data in the course of providing the Services.
- 1.26. The Supplier and Sub-Contractors shall undergo appropriate security assurance activities as determined by the Customer. The Supplier and Sub-Contractors shall support the provision of appropriate evidence of assurance and the production of the necessary security documentation such as completing the Authority Security Assurance Model (DSAM) process or the Business Service Assurance Model (BSAM). This will include obtaining any necessary professional security resources required to support the Supplier's and Sub-Contractor's security assurance activities such as: a NCSC Certified Cyber Security Consultancy (CCSC) or NCSC Certified Professional (CCP) Security and Information Risk Advisor (SIRA)
- 1.27. Where the Supplier is delivering an ICT solution to the Customer they shall design and deliver solutions and services that are compliant with the HMG Security Policy Framework in conjunction with current NCSC Information Assurance Guidance and Authority's Policy. The Supplier will provide the Customer with evidence of compliance for the solutions and Services to be delivered. The Customer's expectation is that the Supplier shall provide written evidence of:
 - i. Compliance with HMG Minimum Cyber Security Standard.
 - ii. Any existing security assurance for the Services to be delivered, such as: ISO/IEC 27001 / 27002 or an equivalent industry level certification.
 - iii. Any existing HMG security accreditations or assurance that are still valid including: details of the awarding body; the scope of the accreditation; any caveats or restrictions to the accreditation; the date awarded, plus a copy of the residual risk statement. iv. Documented progress in achieving any security assurance or accreditation activities including whether documentation has been produced and submitted. The Supplier shall provide details of who the awarding body or organisation will be and date expected.

ANNEX 1: SECURITY MANAGEMENT PLAN

Not included

CONTRACT SCHEDULE 5: STAFF TRANSFER

1. DEFINITIONS

In this Contract Schedule 5, the following definitions shall apply:

- "Admission Agreement" The agreement to be entered into by which the supplier agrees to participate in the Schemes as amended from time to time;
- "Eligible Employee" any Fair Deal Employee who at the relevant time is an eligible employee as defined in the Admission Agreement;

"Employee Liabilities" all claims, actions, proceedings, orders, demands, complaints, investigations (save for any claims for personal injury which are covered by insurance) and any award, compensation, damages, tribunal awards, fine, loss, order, penalty, disbursement, payment made by way of settlement and costs, expenses and legal costs reasonably incurred in connection with a claim or investigation related to employment including in relation to the following:

- (a) redundancy payments including contractual or enhanced redundancy costs, termination costs and notice payments;
- (b) unfair, wrongful or constructive dismissal compensation;
- (c) compensation for discrimination on grounds of sex, race, disability, age, religion or belief, gender reassignment, marriage or civil partnership, pregnancy and maternity or sexual orientation or claims for equal pay;
- (d) compensation for less favourable treatment of parttime workers or fixed term employees;
- (e) outstanding employment debts and unlawful deduction of wages including any PAYE and national insurance contributions;
- (f) employment claims whether in tort, contract or statute or otherwise;

any investigation relating to employment matters by the Equality and Human Rights Commission or other enforcement, regulatory or supervisory body and of implementing any requirements which may arise from such investigation;

"Fair Deal Employees" those Transferring Customer Employees who are on the Relevant Transfer Date entitled to the protection of New

"Former Supplier" a supplier sup	Fair Deal and any Transferring Former Supplier Employees who originally transferred pursuant to a Relevant Transfer under the Employment Regulations (or the predecessor legislation to the Employment Regulations), from employment with a public sector employer and who were once eligible to participate in the Schemes and who at the Relevant Transfer Date become entitled to the protection of New Fair Deal; plying services to the Customer before the Relevant Transfer Date that are the same as or substantially similar to the Services (or any part of the Services) and shall include any sub-contractor of such supplier (or any sub-contractor of any such subcontractor);	
"New Fair Deal" the revised Fair Deal position set out in the HM Treasury guidance: "Fair Deal for staff pensions: staff transfer from central government" issued in October 2013;		
"Notified Sub-Contractor" a Sub	-Contractor identified in the Annex to this Contract Schedule 5 to whom Transferring Customer Employees and/or Transferring Former Supplier Employees will transfer on a Relevant Transfer Date;	
"Replacement Sub-Contractor" a sub-contractor of the Replacement Supplier to whom		
	Transferring Supplier Employees will transfer on a Service Transfer Date (or any sub-contractor of any such subcontractor);	
"Relevant Transfer"	a transfer of employment to which the Employment	
	Regulations applies;	
"Relevant Transfer Date"	in relation to a Relevant Transfer, the date upon which the Relevant Transfer takes place;	
"Schemes"	the Principal Civil Service Pension Scheme available to employees of the civil service and employees of bodies under the Superannuation Act 1972, as governed by rules adopted by Parliament; the Partnership Pension Account and its (i) III health Benefits Scheme and (ii) Death Benefits Scheme; the Civil Service Additional Voluntary Contribution Scheme; and the 2015 New Scheme (with effect from a date to be notified to the Supplier by the Minister for the Cabinet Office);	
"Service Transfer"	any transfer of the Services (or any part of the Services), for whatever reason, from the Supplier or any Sub- Contractor to a Replacement Supplier or a Replacement	

Sub-Contractor;

"Service Transfer Date" the da

"Staffing Information"

the date of a Service Transfer;

in relation to all persons identified on the Supplier's Provisional Supplier Personnel List or Supplier's Final Supplier Personnel List, as the case may be, such information as the Customer may reasonably request (subject to all applicable provisions of the DPA), but including in an anonymised format:

- (a) their ages, dates of commencement of employment or engagement and gender;
- (b) details of whether they are employed, selfemployed contractors or consultants, agency workers or otherwise;
- (c) the identity of the employer or relevant contracting party;
- (d) their relevant contractual notice periods and any other terms relating to termination of employment, including redundancy procedures, and redundancy payments;
- (e) their wages, salaries and profit sharing

arrangements as applicable;

- (f) details of other employment-related benefits, including (without limitation) medical insurance, life assurance, pension or other retirement benefit schemes, share option schemes and company car schedules applicable to them;
- (g) any outstanding or potential contractual, statutory or other liabilities in respect of such individuals (including in respect of personal injury claims);
- (h) details of any such individuals on long term sickness absence, parental leave, maternity leave or other authorised long term absence;
- copies of all relevant documents and materials relating to such information, including copies of relevant contracts of employment (or relevant standard contracts if applied generally in respect of such employees); and
- (j) any other "employee liability information" as such term is defined in regulation 11 of the Employment Regulations;

"Supplier's Final Supplier a list provided by the Supplier of all Supplier Personnel Personnel List" who will transfer under the Employment Regulations on the Relevant Transfer Date;

"Supplier's Provisional Supplier a list prepared and updated by the Supplier of all Supplier **Personnel List"** Personnel who are engaged in or wholly or mainly

	assigned to the provision of the Services or any relevant part of the
	Services which it is envisaged as at the date of such list will no longer be provided by the Supplier;
"Transferring Customer Employees"	those employees of the Customer to whom the Employment Regulations will apply on the Relevant Transfer Date;
"Transferring Former Supplier Employees"	in relation to a Former Supplier, those employees of the Former Supplier to whom the Employment Regulations will apply on the Relevant Transfer Date; and
"Transferring Supplier Employees"	those employees of the Supplier and/or the Supplier's Sub- Contractors to whom the Employment Regulations will apply on the Service Transfer Date.

2. INTERPRETATION

Where a provision in this Contract Schedule 5 imposes an obligation on the Supplier to provide an indemnity, undertaking or warranty, the Supplier shall procure that each of its Sub-Contractors shall comply with such obligation and provide such indemnity, undertaking or warranty to the Customer, Former Supplier, Replacement Supplier or Replacement Sub-Contractor, as the case may be.

PART A

TRANSFERRING CUSTOMER EMPLOYEES AT COMMENCEMENT OF SERVICES

1. RELEVANT TRANSFERS

- 1.1. The Customer and the Supplier agree that:
 - 1.1.1. the commencement of the provision of the Services or of each relevant part of the Services will be a Relevant Transfer in relation to the Transferring Customer Employees; and
 - 1.1.2. as a result of the operation of the Employment Regulations, the contracts of employment between the Customer and the Transferring Customer Employees (except in relation to any terms disapplied through operation of regulation 10(2) of the Employment Regulations) will have effect on and from the Relevant Transfer Date as if originally made between the Supplier and/or any Notified Sub-Contractor and each such Transferring Customer Employee.
 - 1.2. The Customer shall comply with all its obligations under the Employment Regulations and shall perform and discharge all its obligations in respect of the Transferring Customer Employees in respect of the period arising up to (but not including)the Relevant Transfer Date (including the payment of all remuneration, benefits, entitlements and outgoings, all wages, accrued but untaken holiday pay, bonuses,

commissions, payments of PAYE, national insurance contributions and pension contributions which in any case are attributable in whole or in part to the period up to (but not including) the Relevant Transfer Date) and any necessary apportionments in respect of any periodic payments shall be made between:

- (i) the Customer; and
- (ii) the Supplier and/or any Notified Sub-Contractor (as appropriate).

2. CUSTOMER INDEMNITIES

- 2.1. Subject to Paragraph 2.2, the Customer shall indemnify the Supplier and any Notified Sub-Contractor against any Employee Liabilities in respect of any Transferring Customer Employee (or, where applicable any employee representative as defined in the Employment Regulations) arising from or as a result of:
 - 2.1.1. any act or omission by the Customer occurring before the Relevant Transfer Date;
 - 2.1.2. the breach or non-observance by the Customer before the Relevant Transfer Date of:
 - a any collective agreement applicable to the Transferring Customer Employees; and/or
 - b any custom or practice in respect of any Transferring Customer Employees which the Customer is contractually bound to honour;
 - 2.1.3. any claim by any trade union or other body or person representing the Transferring Customer Employees arising from or connected with any failure by the Customer to comply with any legal obligation to such trade union, body or person arising before the Relevant Transfer Date;
 - 2.1.4. any proceeding, claim or demand by HMRC or other statutory authority in respect of any financial obligation including, but not limited to, PAYE and primary and secondary national insurance contributions:
 - (a) in relation to any Transferring Customer Employee, to the extent that the proceeding, claim or demand by HMRC or other statutory authority relates to financial obligations arising before the Relevant Transfer Date; and
 - (b) in relation to any employee who is not a Transferring Customer Employee and in respect of whom it is later alleged or determined that the Employment Regulations applied so as to transfer his/her employment from the Customer to the Supplier and/or any Notified Sub-Contractor as appropriate, to the extent that the proceeding, claim or demand by the HMRC or other statutory authority relates to financial obligations arising before the Relevant Transfer Date.
 - 2.1.5. a failure of the Customer to discharge, or procure the discharge of, all wages, salaries and all other benefits and all PAYE tax deductions and national insurance contributions relating to the Transferring Customer Employees arising before the Relevant Transfer Date;
 - 2.1.6. any claim made by or in respect of any person employed or formerly employed by the Customer other than a Transferring Customer Employee for whom it is alleged the Supplier and/or any Notified Sub-Contractor as appropriate may be liable by virtue of the Employment Regulations and/or the Acquired Rights Directive; and
 - 2.1.7. any claim made by or in respect of a Transferring Customer Employee or any appropriate employee representative (as defined in the Employment Regulations) of any Transferring Customer Employee relating to any act or omission of the Customer in relation to its obligations under regulation 13 of the Employment Regulations, except to the extent that the liability arises from the failure by the Supplier or any Sub-Contractor to comply with regulation 13(4) of the Employment Regulations.
- 2.2. The indemnities in Paragraph 2.1 shall not apply to the extent that the Employee Liabilities arise or are attributable to an act or omission of the Supplier or any Sub-Contractor (whether or not a Notified

SubContractor) whether occurring or having its origin before, on or after the Relevant Transfer Date including any Employee Liabilities:

- 2.2.1. arising out of the resignation of any Transferring Customer Employee before the Relevant Transfer Date on account of substantial detrimental changes to his/her working conditions proposed by the Supplier and/or any Sub-Contractor to occur in the period from (and including) the Relevant Transfer Date; or
- 2.2.2. arising from the failure by the Supplier or any Sub-Contractor to comply with its obligations under the Employment Regulations.
- 2.3. If any person who is not identified by the Customer as a Transferring Customer Employee claims, or it is determined in relation to any person who is not identified by the Customer as a Transferring Customer Employee, that his/her contract of employment has been transferred from the Customer to the Supplier and/or any Notified Sub-Contractor pursuant to the Employment Regulations or the Acquired Rights Directive then:
 - 2.3.1. the Supplier shall, or shall procure that the Notified Sub-Contractor shall, within 5 Working Days of becoming aware of that fact, give notice in writing to the Customer; and
 - 2.3.2. the Customer may offer (or may procure that a third party may offer) employment to such person within 15 Working Days of receipt of the notification by the Supplier and/or any Notified Sub-Contractor, or take such other reasonable steps as the Customer considers appropriate to deal with the matter provided always that such steps are in compliance with Law.
- 2.4. If an offer referred to in Paragraph 2.3.2 is accepted, or if the situation has otherwise been resolved by the Customer, the Supplier shall, or shall procure that the Notified Sub-Contractor shall, immediately release the person from his/her employment or alleged employment.
- 2.5. If by the end of the 15 Working Day period specified in Paragraph 2.3.2:
 - 2.5.1. no such offer of employment has been made;

2.5.2. such offer has been made but not accepted; or 2.5.3 the situation has not otherwise been resolved, the Supplier and/or any Notified Sub-Contractor may within 5 Working Days give notice to terminate the employment or alleged employment of such person.

- 2.6. Subject to the Supplier and/or any Notified Sub-Contractor acting in accordance with the provisions of Paragraphs 2.3 to 2.5 and in accordance with all applicable proper employment procedures set out in applicable Law, the Customer shall indemnify the Supplier and/or any Notified Sub-Contractor (as appropriate) against all Employee Liabilities arising out of the termination pursuant to the provisions of Paragraph 2.5 provided that the Supplier takes, or procures that the Notified Sub-Contractor takes, all reasonable steps to minimise any such Employee Liabilities.
- 2.7. The indemnity in Paragraph 2.6:
 - 2.7.1. shall not apply to:
 - (a) any claim for:
 - discrimination, including on the grounds of sex, race, disability, age, gender reassignment, marriage or civil partnership, pregnancy and maternity or sexual orientation, religion or belief; or
 - (ii) equal pay or compensation for less favourable treatment of part-time workers or fixed-term employees, in any case in relation to any alleged act or omission of the Supplier and/or any Sub-Contractor; or
 - (b) any claim that the termination of employment was unfair because the Supplier and/or Notified Sub-Contractor neglected to follow a fair dismissal procedure; and

- 2.7.2. shall apply only where the notification referred to in Paragraph 2.3.1 is made by the Supplier and/or any Notified Sub-Contractor (as appropriate) to the Customer within 6 months of the Contract Commencement Date.
- 2.8. If any such person as is referred to in Paragraph 2.3 is neither re-employed by the Customer nor dismissed by the Supplier and/or any Notified Sub-Contractor within the time scales set out in Paragraph 2.5 such person shall be treated as having transferred to the Supplier and/or any Notified Sub-Contractor and the Supplier shall, or shall procure that the Notified Sub-Contractor shall, comply with such obligations as may be imposed upon it under applicable Law.

3. SUPPLIER INDEMNITIES AND OBLIGATIONS

- 3.1. Subject to Paragraph 3.2 the Supplier shall indemnify the Customer against any Employee Liabilities in respect of any Transferring Customer Employee (or, where applicable any employee representative as defined in the Employment Regulations) arising from or as a result of:
 - 3.1.1. any act or omission by the Supplier or any Sub-Contractor whether occurring before, on or after the Relevant Transfer Date;
 - 3.1.2. the breach or non-observance by the Supplier or any Sub-Contractor on or after the Relevant Transfer Date of:
 - (a) any collective agreement applicable to the Transferring Customer Employees; and/or
 - (b) any custom or practice in respect of any Transferring Customer Employees which the Supplier or any Sub-Contractor is contractually bound to honour;
 - 3.1.3. any claim by any trade union or other body or person representing any Transferring Customer Employees arising from or connected with any failure by the Supplier or any SubContractor to comply with any legal obligation to such trade union, body or person arising on or after the Relevant Transfer Date;
 - 3.1.4. any proposal by the Supplier or a Sub-contractor made before the Relevant Transfer Date to make changes to the terms and conditions of employment or working conditions of any Transferring Customer Employees to their material detriment on or after their transfer to the Supplier or the relevant Sub-Contractor (as the case may be) on the Relevant Transfer Date, or to change the terms and conditions of employment or working conditions of any person who would have been a Transferring Customer Employee but for their resignation (or decision to treat their employment as terminated under regulation 4(9) of the Employment Regulations) before the Relevant Transfer Date as a result of or for a reason connected to such proposed changes;
 - 3.1.5. any statement communicated to or action undertaken by the Supplier or any SubContractor to, or in respect of, any Transferring Customer Employee before the Relevant Transfer Date regarding the Relevant Transfer which has not been agreed in advance with the Customer in writing;
 - 3.1.6. any proceeding, claim or demand by HMRC or other statutory authority in respect of any financial obligation including, but not limited to, PAYE and primary and secondary national insurance contributions:
 - (a) in relation to any Transferring Customer Employee, to the extent that the proceeding, claim or demand by HMRC or other statutory authority relates to financial obligations arising on or after the Relevant Transfer Date; and
 - (b) in relation to any employee who is not a Transferring Customer Employee, and in respect of whom it is later alleged or determined that the Employment Regulations applied so as to transfer his/her employment from the Customer to the Supplier or a Sub-Contractor, to the extent that the proceeding, claim or

demand by HMRC or other statutory authority relates to financial obligations arising on or after the Relevant Transfer Date;

- 3.1.7. a failure of the Supplier or any Sub-Contractor to discharge or procure the discharge of all wages, salaries and all other benefits and all PAYE tax deductions and national insurance contributions relating to the Transferring Customer Employees in respect of the period from (and including) the Relevant Transfer Date; and
- 3.1.8. any claim made by or in respect of a Transferring Customer Employee or any appropriate employee representative (as defined in the Employment Regulations) of any Transferring Customer Employee relating to any act or omission of the Supplier or any Sub-Contractor in relation to their obligations under regulation 13 of the Employment Regulations, except to the extent that the liability arises from the Customer's failure to comply with its obligations under regulation 13 of the Employment Regulations.
- 3.2. The indemnities in Paragraph 3.1 shall not apply to the extent that the Employee Liabilities arise or are attributable to an act or omission of the Customer whether occurring or having its origin before, on or after the Relevant Transfer Date including, without limitation, any Employee Liabilities arising from the Customer's failure to comply with its obligations under the Employment Regulations.
- 3.3. The Supplier shall comply, and shall procure that each Sub-Contractor shall comply, with all its obligations under the Employment Regulations (including its obligation to inform and consult in accordance with regulation 13 of the Employment Regulations) and shall perform and discharge, and shall procure that each Sub-Contractor shall perform and discharge, all its obligations in respect of the Transferring Customer Employees, from (and including) the Relevant Transfer Date (including the payment of all remuneration, benefits, entitlements and outgoings, all wages, accrued but untaken holiday pay, bonuses, commissions, payments of PAYE, national insurance contributions and pension contributions which in any case are attributable in whole or in part to the period from and including the Relevant Transfer Date) and any necessary apportionments in respect of any periodic payments shall be made between the Customer and the Supplier.

4. INFORMATION

4.1. The Supplier shall, and shall procure that each Sub-Contractor shall, promptly provide to the Customer in writing such information as is necessary to enable the Customer to carry out its duties under regulation 13 of the Employment Regulations. The Customer shall promptly provide to the Supplier and each Notified Sub-Contractor in writing such information as is necessary to enable the Supplier and each Notified Sub-Contractor to carry out their respective duties under regulation 13 of the Employment Regulations.

5. PRINCIPLES OF GOOD EMPLOYMENT PRACTICE

- 5.1. The Parties agree that the Principles of Good Employment Practice issued by the Cabinet Office in December 2010 apply to the treatment by the Supplier of employees whose employment begins after the Relevant Transfer Date, and the Supplier undertakes to treat such employees in accordance with the provisions of the Principles of Good Employment Practice.
- 5.2. The Supplier shall, and shall procure that each Sub-Contractor shall, comply with any requirement notified to it by the Customer relating to pensions in respect of any Transferring Customer Employee as set down in:
 - 5.2.1. the Cabinet Office Statement of Practice on Staff Transfers in the Public Sector of January 2000, revised 2007;
 - 5.2.2. HM Treasury's guidance "Staff Transfers from Central Government: A Fair Deal for Staff Pensions of 1999;
 - 5.2.3. HM Treasury's guidance "Fair deal for staff pensions: procurement of Bulk Transfer Agreements and Related Issues" of June 2004; and/or

- 5.2.4. the New Fair Deal.
- 5.3. Any changes embodied in any statement of practice, paper or other guidance that replaces any of the documentation referred to in Paragraphs 5.1 or 5.2 shall be agreed in accordance with the Variation Procedure.

6. PENSIONS

6.1. The Supplier shall, and shall procure that each of its Sub-Contractors shall, comply with the pensions provisions in the following Annex.

ANNEX TO PART A: PENSIONS

1. PARTICIPATION

- 1.1 The Supplier undertakes to enter into the Admission Agreement.
- 1.2 The Supplier and the Customer:
 - 1.2.1 undertake to do all such things and execute any documents (including the Admission Agreement) as may be required to enable the Supplier to participate in the Schemes in respect of the Fair Deal Employees;
 - 1.2.2 agree that the Customer is entitled to make arrangements with the
 - 1.2.3 body responsible for the Schemes for the Customer to be notified if the Supplier breaches the Admission Agreement;
 - 1.2.4 notwithstanding Paragraph 1.2.2 of this Annex, the Supplier shall notify the Customer in the event that it breaches the Admission Agreement; and
 - 1.2.5 agree that the Customer may terminate this Contract for material default in the event that the Supplier breaches the Admission Agreement.
- 1.3 The Supplier shall bear its own costs and all costs that the Customer reasonably incurs in connection with the negotiation, preparation and execution of documents to facilitate the Supplier participating in the Schemes.

2. FUTURE SERVICE BENEFITS

- 2.1 The Supplier shall procure that the Fair Deal Employees, shall be either admitted into, or offered continued membership of, the relevant section of the Schemes that they currently contribute to, or were eligible to join immediately prior to the Relevant Transfer Date and the Supplier shall procure that the Fair Deal Employees continue to accrue benefits in accordance with the provisions governing the relevant section of Schemes for service from (and including) the Relevant Transfer Date.
- 2.2 The Supplier undertakes that should it cease to participate in the Schemes for whatever reason at a time when it has Eligible Employees, that it will, at no extra cost to the Customer, provide to any Fair Deal Employee who immediately prior to such cessation remained an Eligible Employee with access to an occupational pension scheme certified by the Government Actuary's Department or any actuary nominated by the Customer in accordance with relevant guidance produced by the Government Actuary's Department to those provided by the Schemes at the relevant date.
- 2.3 The Parties acknowledge that the Civil Service Compensation Scheme and the Civil Service Injury Benefit Scheme (established pursuant to section 1 of the Superannuation Act 1972) are not covered by the protection of New Fair Deal.

3. FUNDING

- 3.1 The Supplier undertakes to pay to the Schemes all such amounts as are due under the Admission Agreement and shall deduct and pay to the Schemes such employee contributions as are required by the Schemes.
- 3.2 The Supplier shall indemnify and keep indemnified the Customer on demand against any claim by, payment to, or loss incurred by, the Schemes in respect of the failure to account to the Schemes for payments received and the non-payment or the late payment of any sum payable by the Supplier to or in respect of the Schemes.

4. PROVISION OF INFORMATION

The Supplier and the Customer respectively undertake to each other:

- 4.1 to provide all information which the other Party may reasonably request concerning matters referred to in this Annex and set out in the Admission Agreement, and to supply the information as expeditiously as possible; and
- 4.2 not to issue any announcements to the Fair Deal Employees prior to the Relevant Transfer Date concerning the matters stated in this Annex without the consent in writing of the other Party (not to be unreasonably withheld or delayed).

5. INDEMNITY

5.1 The Supplier undertakes to the Customer to indemnify and keep indemnified the Customer on demand from and against all and any Losses whatsoever arising out of or in connection with any liability towards the Fair Deal Employees arising in respect of service on or after the Relevant Transfer Date which relate to the payment of benefits under an occupational pension scheme (within the meaning provided for in section 1 of the Pension Schemes Act 1993) or the Schemes.

6. EMPLOYER OBLIGATION

6.1 The Supplier shall comply with the requirements of the Pensions Act 2008 and the Transfer of Employment (Pension Protection) Regulations 2005.

7. SUBSEQUENT TRANSFERS

The Supplier shall:

- 7.1 not adversely affect pension rights accrued by any Fair Deal Employee in the period ending on the date of the relevant future transfer;
- 7.2 provide all such co-operation and assistance as the Schemes and the Replacement Supplier and/or the Customer may reasonably require to enable the Replacement Supplier to participate in the Schemes in respect of any Eligible Employee and to give effect to any transfer of accrued rights required as part of participation under New Fair Deal; and
- 7.3 for the period either:
 - 7.3.1 after notice (for whatever reason) is given, in accordance with the other provisions of this Contract, to terminate the Contract or any part of the Services; or
 - 7.3.2 after the date which is two (2) years prior to the date of expiry of this Contract, ensure that no change is made to pension, retirement and death benefits provided for or in respect of any person who will transfer to the Replacement Supplier or the Customer, no category of earnings which were not previously pensionable are made pensionable and the contributions (if any) payable by such employees are not reduced without (in any case) the prior approval of the Customer (such approval not to be unreasonably

withheld). Save that this subparagraph shall not apply to any change made as a consequence of participation in an Admission Agreement.

PART B

TRANSFERRING FORMER SUPPLIER EMPLOYEES AT COMMENCEMENT OF SERVICES

1. RELEVANT TRANSFERS

- 1.1 The Customer and the Supplier agree that:
 - 1.1.1 the commencement of the provision of the Services or of any relevant part of the Services will be a Relevant Transfer in relation to the Transferring Former Supplier Employees; and
 - 1.1.2 as a result of the operation of the Employment Regulations, the contracts of employment between each Former Supplier and the Transferring Former Supplier Employees (except in relation to any terms disapplied through the operation of regulation 10(2) of the Employment Regulations) shall have effect on and from the Relevant Transfer Date as if originally made between the Supplier and/or Notified Sub-Contractor and each such Transferring Former Supplier Employee.
- 1.2 Subject to Paragraph 6, the Customer shall procure that each Former Supplier shall comply with all its obligations under the Employment Regulations and shall perform and discharge all its obligations in respect of all the Transferring Former Supplier Employees in respect of the period up to (but not including) the Relevant Transfer Date (including the payment of all remuneration, benefits, entitlements and outgoings, all wages, accrued but untaken holiday pay, bonuses, commissions, payments of PAYE, national insurance contributions and pension contributions which in any case are attributable in whole or in part in respect of the period up to (but not including) the Relevant Transfer Date) and the Supplier shall make, and the Customer shall procure that each Former Supplier makes, any necessary apportionments in respect of any periodic payments.

2. FORMER SUPPLIER INDEMNITIES

- 2.1 Subject to Paragraphs 2.2 and 6, the Customer shall procure that each Former Supplier shall indemnify the Supplier and any Notified Sub-Contractor against any Employee Liabilities in respect of any Transferring Former Supplier Employee (or, where applicable any employee representative as defined in the Employment Regulations) arising from or as a result of:
 - 2.1.1 any act or omission by the Former Supplier arising before the Relevant Transfer Date;
 - 2.1.2 the breach or non-observance by the Former Supplier arising before the Relevant Transfer Date of:
 - (a) any collective agreement applicable to the Transferring Former Supplier Employees; and/or
 - (b) any custom or practice in respect of any Transferring Former Supplier Employees which the Former Supplier is contractually bound to honour;
 - 2.1.3 any proceeding, claim or demand by HMRC or other statutory authority in respect of any financial obligation including, but not limited to, PAYE and primary and secondary national insurance contributions:
 - (a) in relation to any Transferring Former Supplier Employee, to the extent that the proceeding, claim or demand by HMRC or other statutory authority relates to financial obligations arising before the Relevant Transfer Date; and

- (b) in relation to any employee who is not a Transferring Former Supplier Employee and in respect of whom it is later alleged or determined that the Employment Regulations applied so as to transfer his/her employment from the Former Supplier to the Supplier and/or any Notified Sub-Contractor as appropriate, to the extent that the proceeding, claim or demand by HMRC or other statutory authority relates to financial obligations in respect of the period to (but excluding) the Relevant Transfer Date;
- 2.1.4 a failure of the Former Supplier to discharge or procure the discharge of all wages, salaries and all other benefits and all PAYE tax deductions and national insurance contributions relating to the Transferring Former Supplier Employees in respect of the period to (but excluding) the Relevant Transfer Date;
- 2.1.5 any claim made by or in respect of any person employed or formerly employed by the Former Supplier other than a Transferring Former Supplier Employee for whom it is alleged the Supplier and/or any Notified Sub-Contractor as appropriate may be liable by virtue of this Contract and/or the Employment Regulations and/or the Acquired Rights Directive; and
- 2.1.6 any claim made by or in respect of a Transferring Former Supplier Employee or any appropriate employee representative (as defined in the Employment Regulations) of any Transferring Former Supplier Employee relating to any act or omission of the Former Supplier in relation to its obligations under regulation 13 of the Employment Regulations, except to the extent that the liability arises from the failure by the Supplier or any Sub-Contractor to comply with regulation 13(4) of the Employment Regulations.
- 2.2 The indemnities in Paragraph 2.1 shall not apply to the extent that the Employee Liabilities arise or are attributable to an act or omission of the Supplier or any Sub-Contractor whether occurring or having its origin before, on or after the Relevant Transfer Date including, without limitation, any Employee Liabilities:
 - 2.2.1 arising out of the resignation of any Transferring Former Supplier Employee before the Relevant Transfer Date on account of substantial detrimental changes to his/her working conditions proposed by the Supplier or any Sub-Contractor to occur in the period from (and including) the Relevant Transfer Date; or
 - 2.2.2 arising from the failure by the Supplier and/or any Sub-Contractor to comply with its obligations under the Employment Regulations.
- 2.3 If any person who is not identified by the Customer as a Transferring Former Supplier Employee claims, or it is determined in relation to any person who is not identified by the Customer as a Transferring Former Supplier Employee, that his/her contract of employment has been transferred from a Former Supplier to the Supplier and/or any Notified SubContractor pursuant to the Employment Regulations or the Acquired Rights Directive then:
 - 2.3.1 the Supplier shall, or shall procure that the Notified Sub-Contractor shall, within 5 Working Days of becoming aware of that fact, give notice in writing to the Customer and, where required by the Customer, to the Former Supplier; and
 - 2.3.2 the Former Supplier may offer (or may procure that a third party may offer) employment to such person within 15 Working Days of the notification by the Supplier and/or the Notified Sub-Contractor or take such other reasonable steps as the Former Supplier considers appropriate to deal with the matter provided always that such steps are in compliance with applicable Law.
- 2.4 If an offer referred to in Paragraph 2.3.2 is accepted, or if the situation has otherwise been resolved by the Former Supplier and/or the Customer, the Supplier shall, or shall procure that the

Notified Sub-Contractor shall, immediately release the person from his/her employment or alleged employment.

- 2.5 If by the end of the 15 Working Day period specified in Paragraph 2.3.2:
 - 2.5.1 no such offer of employment has been made;
 - 2.5.2 such offer has been made but not accepted; or 2.5.3 the situation has not otherwise been resolved, the Supplier and/or any Notified Sub-Contractor may within 5 Working Days give notice to terminate the employment or alleged employment of such person.
- 2.6 Subject to the Supplier and/or any Notified Sub-Contractor acting in accordance with the provisions of Paragraphs 2.3 to 2.5 and in accordance with all applicable proper employment procedures set out in Law, the Customer shall procure that the Former Supplier indemnifies the Supplier and/or any Notified Sub-Contractor (as appropriate) against all Employee Liabilities arising out of the termination pursuant to the provisions of Paragraph 2.5 provided that the Supplier takes, or shall procure that the Notified Sub-Contractor takes, all reasonable steps to minimise any such Employee Liabilities.
- 2.7 The indemnity in Paragraph 2.6:
 - 2.7.1 shall not apply to: (a)

any claim for:

i discrimination, including on the grounds of sex, race, disability, age,

gender reassignment, marriage or civil partnership, pregnancy and maternity or sexual orientation, religion or belief; or

ii equal pay or compensation for less favourable treatment of parttime workers or fixedterm employees,

in any case in relation to any alleged act or omission of the Supplier and/or any Sub-Contractor; or

- (b) any claim that the termination of employment was unfair because the Supplier and/or Notified Sub-Contractor neglected to follow a fair dismissal procedure; and
- 2.7.2 shall apply only where the notification referred to in Paragraph 2.3.1 is made by the Supplier and/or any Notified Sub-Contractor (as appropriate) to the Customer and, if applicable, the Former Supplier, within 6 months of the Contract Commencement Date.
- 2.8 If any such person as is described in Paragraph 2.3 is neither re-employed by the Former Supplier nor dismissed by the Supplier and/or any Notified Sub-Contractor within the time scales set out in Paragraph 2.5, such person shall be treated as having transferred to the Supplier or Notified Sub-Contractor and the Supplier shall, or shall procure that the Notified Sub-Contractor shall, comply with such obligations as may be imposed upon it under the Law.

3. SUPPLIER INDEMNITIES AND OBLIGATIONS

- 3.1 Subject to Paragraph 3.2, the Supplier shall indemnify the Customer and/or the Former Supplier against any Employee Liabilities in respect of any Transferring Former Supplier Employee (or, where applicable any employee representative as defined in the Employment Regulations) arising from or as a result of:
 - 3.1.1 any act or omission by the Supplier or any Sub-Contractor whether occurring before, on or after the Relevant Transfer Date;
 - 3.1.2 the breach or non-observance by the Supplier or any Sub-Contractor on or after the Relevant Transfer Date of:

- (a) any collective agreement applicable to the Transferring Former Supplier Employee; and/or
- (b) any custom or practice in respect of any Transferring Former Supplier Employees which the Supplier or any Sub-Contractor is contractually bound to honour;
- 3.1.3 any claim by any trade union or other body or person representing any Transferring Former Supplier Employees arising from or connected with any failure by the Supplier or a Sub-Contractor to comply with any legal obligation to such trade union, body or person arising on or after the Relevant Transfer Date;
- 3.1.4 any proposal by the Supplier or a Sub-Contractor prior to the Relevant Transfer Date to make changes to the terms and conditions of employment or working conditions of any Transferring Former Supplier Employees to their material detriment on or after their transfer to the Supplier or a Sub-Contractor (as the case may be) on the Relevant Transfer Date, or to change the terms and conditions of employment or working conditions of any person who would have been a Transferring Former Supplier Employee but for their resignation (or decision to treat their employment as terminated under regulation 4(9) of the Employment Regulations) before the Relevant Transfer Date as a result of or for a reason connected to such proposed changes;
- 3.1.5 any statement communicated to or action undertaken by the Supplier or a SubContractor to, or in respect of, any Transferring Former Supplier Employee before the Relevant Transfer Date regarding the Relevant Transfer which has not been agreed in advance with the Customer and/or the Former Supplier in writing;
- 3.1.6 any proceeding, claim or demand by HMRC or other statutory authority in respect of any financial obligation including, but not limited to, PAYE and primary and secondary national insurance contributions:
 - (a) in relation to any Transferring Former Supplier Employee, to the extent that the proceeding, claim or demand by HMRC or other statutory authority relates to financial obligations arising on or after the Relevant Transfer Date; and
 - (b) in relation to any employee who is not a Transferring Former Supplier Employee, and in respect of whom it is later alleged or determined that the Employment Regulations applied so as to transfer his/her employment from the Former Supplier to the Supplier or a Sub-Contractor, to the extent that the proceeding, claim or demand by the HMRC or other statutory authority relates to financial obligations arising on or after the Relevant Transfer Date;
- 3.1.7 a failure of the Supplier or any Sub-Contractor to discharge or procure the discharge of all wages, salaries and all other benefits and all PAYE tax deductions and national insurance contributions relating to the Transferring Former Supplier Employees in respect of the period from (and including) the Relevant Transfer Date; and
- 3.1.8 any claim made by or in respect of a Transferring Former Supplier Employee or any appropriate employee representative (as defined in the Employment Regulations) of any Transferring Former Supplier Employee relating to any act or omission of the Supplier or any Sub-Contractor in relation to obligations under regulation 13 of the Employment Regulations, except to the extent that the liability arises from the

Former Supplier's failure to comply with its obligations under regulation 13 of the Employment Regulations.

3.2 The indemnities in Paragraph 3.1 shall not apply to the extent that the Employee Liabilities arise or are attributable to an act or omission of the Former Supplier whether occurring or having its origin before, on or after the Relevant Transfer Date including, without limitation,

any Employee Liabilities arising from the Former Supplier's failure to comply with its obligations under the Employment Regulations.

3.3 The Supplier shall comply, and shall procure that each Sub-Contractor shall comply, with all its obligations under the Employment Regulations (including without limitation its obligation to inform and consult in accordance with regulation 13 of the Employment Regulations) and shall perform and discharge, and shall procure that each Sub-Contractor shall perform and discharge, all its obligations in respect of all the Transferring Former Supplier Employees, on and from the Relevant Transfer Date (including the payment of all remuneration, benefits, entitlements and outgoings, all wages, accrued but untaken holiday pay, bonuses, commissions, payments of PAYE, national insurance contributions and pension contributions which in any case are attributable in whole or in part to the period from (and including) the Relevant Transfer Date) and any necessary apportionments in respect of any periodic payments shall be made between the Supplier and the Former Supplier.

4. INFORMATION

4.1 The Supplier shall, and shall procure that each Sub-Contractor shall, promptly provide to the Customer and/or at the Customer's direction, the Former Supplier, in writing such information as is necessary to enable the Customer and/or the Former Supplier to carry out their respective duties under regulation 13 of the Employment Regulations. Subject to Paragraph 6, the Customer shall procure that the Former Supplier shall promptly provide to the Supplier and each Notified Sub-Contractor in writing such information as is necessary to enable the Supplier and each Notified Sub-Contractor to carry out their respective duties under regulation 13 of the Employment Regulations.

5. PRINCIPLES OF GOOD EMPLOYMENT PRACTICE

- 5.1 The Supplier shall, and shall procure that each Sub-Contractor shall, comply with any requirement notified to it by the Customer relating to pensions in respect of any Transferring Former Supplier Employee as set down in:
 - 5.1.1 the Cabinet Office Statement of Practice on Staff Transfers in the Public Sector of January 2000, revised 2007;
 - 5.1.2 HM Treasury's guidance "Staff Transfers from Central Government: A Fair Deal for Staff Pensions of 1999;
 - 5.1.3 HM Treasury's guidance: "Fair deal for staff pensions: procurement of Bulk Transfer Agreements and Related Issues" of June 2004; and/or
 - 5.1.4 the New Fair Deal.
- 5.2 Any changes embodied in any statement of practice, paper or other guidance that replaces any of the documentation referred to in Paragraph 5.1 shall be agreed in accordance with the Variation Procedure.

6. PROCUREMENT OBLIGATIONS

6.1 Notwithstanding any other provisions of this Part B, where in this Part B the Customer accepts an obligation to procure that a Former Supplier does or does not do something, such obligation shall be limited so that it extends only to the extent that the Customer's contract with the

Former Supplier contains a contractual right in that regard which the Customer may enforce, or otherwise so that it requires only that the Customer must use reasonable endeavours to procure that the Former Supplier does or does not act accordingly.

7. PENSIONS

7.1 The Supplier shall, and shall procure that each Sub-Contractor shall, comply with the pensions provisions in the following Annex in respect of any Transferring Former Supplier Employees who transfer from the Former Supplier to the Supplier.

ANNEX TO PART B: PENSIONS

1. PARTICIPATION

- 1.1 The Supplier undertakes to enter into the Admission Agreement.
- 1.2 The Supplier and the Customer:
 - 1.2.1 undertake to do all such things and execute any documents (including the Admission Agreement) as may be required to enable the Supplier to participate in the Schemes in respect of the Fair Deal Employees;
 - 1.2.2 agree that the Customer is entitled to make arrangements with the body responsible for the Schemes for the Customer to be notified if the Supplier breaches the Admission Agreement;
 - 1.2.3 notwithstanding Paragraph 1.2.2 of this Annex, the Supplier shall notify the Customer in the event that it breaches the Admission Agreement; and
 - 1.2.4 agree that the Customer may terminate this Contract for material default in the event that the Supplier breaches the Admission Agreement.
- 1.3 The Supplier shall bear its own costs and all costs that the Customer reasonably incurs in connection with the negotiation, preparation and execution of documents to facilitate the Supplier participating in the Schemes.

2. FUTURE SERVICE BENEFITS

- 2.1 If the Supplier is rejoining the Schemes for the first time, the Supplier shall procure that the Fair Deal Employees shall be either admitted to or offered continued membership of the relevant section of the Schemes that they became eligible to join on the Relevant Transfer Date and shall continue to accrue or accrue benefits in accordance with the provisions governing the relevant section of the Schemes for service from (and including) the Relevant Transfer Date.
- 2.2 If staff have already been readmitted to the Schemes, the Supplier shall procure that the Fair Deal Employees, shall be either admitted into, or offered continued membership of, the relevant section of the Schemes that they currently contribute to, or were eligible to join immediately prior to the Relevant Transfer Date and the Supplier shall procure that the Fair Deal Employees continue to accrue benefits in accordance with the provisions governing the relevant section of the Schemes for service from (and including) the Relevant Transfer Date.
- 2.3 The Supplier undertakes that should it cease to participate in the Schemes for whatever reason at a time when it has Eligible Employees, that it will, at no extra cost to the Customer, provide to any Fair Deal Employee who immediately prior to such cessation remained an Eligible Employee with access to an occupational pension scheme certified by the Government Actuary's Department or any actuary nominated by the Customer in accordance with relevant guidance produced by the Government Actuary's Department to those provided by the Schemes at the relevant date.
- 2.4 The Parties acknowledge that the Civil Service Compensation Scheme and the Civil Service Injury Benefit Scheme (established pursuant to section 1 of the Superannuation Act 1972) are not covered by the protection of New Fair Deal.

3. FUNDING

3.1 The Supplier undertakes to pay to the Schemes all such amounts as are due under the Admission Agreement and shall deduct and pay to the Schemes such employee contributions as are required by the Schemes. 3.2 The Supplier shall indemnify and keep indemnified the Customer on demand against any claim by, payment to, or loss incurred by the Schemes in respect of the failure to account to the Schemes for payments received and the non-payment or the late payment of any sum payable by the Supplier to or in respect of the Schemes.

4. PROVISION OF INFORMATION

The Supplier and the Customer respectively undertake to each other:

- 4.1 to provide all information which the other Party may reasonably request concerning matters (i) referred to in this Annex and (ii) set out in the Admission Agreement, and to supply the information as expeditiously as possible; and
- 4.2 not to issue any announcements to the Fair Deal Employees prior to the Relevant Transfer Date concerning the matters stated in this Annex without the consent in writing of the other Party (not to be unreasonably withheld or delayed).

5. INDEMNITY

5.1 The Supplier undertakes to the Customer to indemnify and keep indemnified the Customer on demand from and against all and any Losses whatsoever arising out of or in connection with any liability towards the Fair Deal Employees arising in respect of service on or after the Relevant Transfer Date which relate to the payment of benefits under an occupational pension scheme (within the meaning provided for in section 1 of the Pension Schemes Act 1993) or the Schemes.

6. EMPLOYER OBLIGATION

6.1 The Supplier shall comply with the requirements of the Pensions Act 2008 and the Transfer of Employment (Pension Protection) Regulations 2005.

7. SUBSEQUENT TRANSFERS

The Supplier shall:

- 7.1 not adversely affect pension rights accrued by any Fair Deal Employee in the period ending on the date of the relevant future transfer;
- 7.2 provide all such co-operation and assistance as the Schemes and the Replacement Supplier and/or the Customer may reasonably require to enable the Replacement Supplier to participate in the Schemes in respect of any Eligible Employee and to give effect to any transfer of accrued rights required as part of participation under the New Fair Deal; and
- 7.3 for the period either
 - 7.3.1 after notice (for whatever reason) is given, in accordance with the other provisions of this Contract, to terminate the Contract or any part of the Services; or
 - 7.3.2 after the date which is two (2) years prior to the date of expiry of this Contract,
 - 7.3.3 ensure that no change is made to pension, retirement and death benefits provided for or in respect of any person who will transfer to the Replacement Supplier or the Customer, no category of earnings which were not previously pensionable are made pensionable and the contributions (if any) payable by such employees are not reduced without (in any case) the prior approval of the Customer (such approval not

to be unreasonably withheld). Save that this subparagraph shall not apply to any change made as a consequence of participation in an Admission Agreement.

PART C NO TRANSFER OF EMPLOYEES AT COMMENCEMENT OF SERVICES

1. PROCEDURE IN THE EVENT OF TRANSFER

- 1.1 The Customer and the Supplier agree that the commencement of the provision of the Services or of any part of the Services will not be a Relevant Transfer in relation to any employees of the Customer and/or any Former Supplier.
- 1.2 If any employee of the Customer and/or a Former Supplier claims, or it is determined in relation to any employee of the Customer and/or a Former Supplier, that his/her contract of employment has been transferred from the Customer and/or the Former Supplier to the Supplier and/or any Sub-Contractor pursuant to the Employment Regulations or the Acquired Rights Directive then:
 - 1.2.1 the Supplier shall, and shall procure that the relevant Sub-Contractor shall, within five(5) Working Days of becoming aware of that fact, give notice in writing to the Customer and, where required by the Customer, give notice to the Former Supplier; and
 - 1.2.2 the Customer and/or the Former Supplier may offer (or may procure that a third party may offer) employment to such person within fifteen (15) Working Days of the notification by the Supplier or the Sub-Contractor (as appropriate) or take such other reasonable steps as the Customer or Former Supplier (as the case may be) considers appropriate to deal with the matter provided always that such steps are in compliance with applicable Law.
- 1.3 If an offer referred to in Paragraph 1.2.2 is accepted (or if the situation has otherwise been resolved by the Customer and/or the Former Supplier), the Supplier shall, or shall procure that the Sub-Contractor shall, immediately release the person from his/her employment or alleged employment.
- 1.4 If by the end of the fifteen (15) Working Day period specified in Paragraph 1.2.2:
 - 1.4.1 no such offer of employment has been made;
 - 1.4.2 such offer has been made but not accepted; or 1.4.3 the situation has not otherwise been resolved, the Supplier and/or the Sub-Contractor may within five (5) Working Days give notice to terminate the employment or alleged employment of such person.

2. INDEMNITIES

- 2.1 Subject to the Supplier and/or the relevant Sub-Contractor acting in accordance with the provisions of Paragraphs 1.2 to 1.4 and in accordance with all applicable employment procedures set out in applicable Law and subject also to Paragraph 2.4, the Customer shall:
 - 2.1.1 indemnify the Supplier and/or the relevant Sub-Contractor against all Employee Liabilities arising out of the termination of the employment of any employees of the Customer referred to in Paragraph 1.2 made pursuant to the provisions of Paragraph 1.4 provided that the Supplier takes, or shall procure that the Notified Sub-Contractor takes, all reasonable steps to minimise any such Employee Liabilities; and
 - 2.1.2 subject to paragraph 3, procure that the Former Supplier indemnifies the Supplier and/or any Notified Sub-Contractor against all Employee Liabilities arising out of termination of the employment of the employees of the Former Supplier made pursuant to the provisions of Paragraph 1.4 provided that the Supplier takes, or shall

procure that the relevant Sub-Contractor takes, all reasonable steps to minimise any such Employee Liabilities.

- 2.2 If any such person as is described in Paragraph 1.2 is neither re employed by the Customer and/or the Former Supplier as appropriate nor dismissed by the Supplier and/or any SubContractor within the fifteen (15) Working Day period referred to in Paragraph 1.4 such person shall be treated as having transferred to the Supplier and/or the Sub-Contractor (as appropriate) and the Supplier shall, or shall procure that the Sub-Contractor shall, comply with such obligations as may be imposed upon it under Law.
- 2.3 Where any person remains employed by the Supplier and/or any Sub-Contractor pursuant to Paragraph 2.2, all Employee Liabilities in relation to such employee shall remain with the Supplier and/or the Sub-Contractor and the Supplier shall indemnify the Customer and any Former Supplier, and shall procure that the Sub-Contractor shall indemnify the Customer and any Former Supplier, against any Employee Liabilities that either of them may incur in respect of any such employees of the Supplier and/or employees of the Sub-Contractor.
- 2.4 The indemnities in Paragraph 2.1:
 - 2.4.1 shall not apply to:
 - (a) any claim for:
 - discrimination, including on the grounds of sex, race, disability, age, gender reassignment, marriage or civil partnership, pregnancy and maternity or sexual orientation, religion or belief; or
 - (ii) equal pay or compensation for less favourable treatment of parttime workers or fixed-term employees,

in any case in relation to any alleged act or omission of the Supplier and/or any Sub-Contractor; or

- (b) any claim that the termination of employment was unfair because the Supplier and/or any Sub-Contractor neglected to follow a fair dismissal procedure; and
- 2.4.2 shall apply only where the notification referred to in Paragraph 1.2.1 is made by the Supplier and/or any Sub-Contractor to the Customer and, if applicable, Former Supplier within 6 months of the Contract Commencement Date.

3. PROCUREMENT OBLIGATIONS

3.1 Where in this Part C the Customer accepts an obligation to procure that a Former Supplier does or does not do something, such obligation shall be limited so that it extends only to the extent that the Customer's contract with the Former Supplier contains a contractual right in that regard which the Customer may enforce, or otherwise so that it requires only that the Customer must use reasonable endeavours to procure that the Former Supplier does or does not accordingly.

PART D EMPLOYMENT EXIT PROVISIONS

1. PRE-SERVICE TRANSFER OBLIGATIONS

- 1.1 The Supplier agrees that within twenty (20) Working Days of the earliest of:
 - 1.1.1 receipt of a notification from the Customer of a Service Transfer or intended Service Transfer;
 - 1.1.2 receipt of the giving of notice of early termination or any Partial Termination of this Contract ;
 - 1.1.3 the date which is twelve (12) Months before the end of the Term; and

1.1.4 receipt of a written request of the Customer at any time (provided that the Customer shall only be entitled to make one such request in any six (6) month period),

it shall provide in a suitably anonymised format so as to comply with the DPA, the Supplier's Provisional Supplier Personnel List, together with the Staffing Information in relation to the Supplier's Provisional Supplier Personnel List and it shall provide an updated Supplier's Provisional Supplier Personnel List at such intervals as are reasonably requested by the Customer.

- 1.2 At least thirty (30) Working Days prior to the Service Transfer Date, the Supplier shall provide to the Customer or at the direction of the Customer to any Replacement Supplier and/or any Replacement Sub-Contractor:
 - 1.2.1 the Supplier's Final Supplier Personnel List, which shall identify which of the Supplier Personnel are Transferring Supplier Employees; and
 - 1.2.2 the Staffing Information in relation to the Supplier's Final Supplier Personnel List (insofar as such information has not previously been provided).
- 1.3 The Customer shall be permitted to use and disclose information provided by the Supplier under Paragraphs 1.1 and 1.2 for the purpose of informing any prospective Replacement Supplier and/or Replacement Sub-Contractor.
- 1.4 The Supplier warrants, for the benefit of the Customer, any Replacement Supplier, and any Replacement Sub-Contractor that all information provided pursuant to Paragraphs 1.1 and 1.2 shall be true and accurate in all material respects at the time of providing the information.
- 1.5 From the date of the earliest event referred to in Paragraph 1.1, the Supplier agrees, that it shall not, and agrees to procure that each Sub-Contractor shall not, assign any person to the provision of the Services who is not listed on the Supplier's Provisional Supplier Personnel List and shall not without the approval of the Customer (not to be unreasonably withheld or delayed):
 - 1.5.1 replace or re-deploy any Supplier Personnel listed on the Supplier Provisional Supplier Personnel List other than where any replacement is of equivalent grade, skills, experience and expertise and is employed on the same terms and conditions of employment as the person he/she replaces;
 - 1.5.2 make, promise, propose or permit any material changes to the terms and conditions of employment of the Supplier Personnel (including any payments connected with the termination of employment);
 - 1.5.3 increase the proportion of working time spent on the Services (or the relevant part of the Services) by any of the Supplier Personnel save for fulfilling assignments and projects previously scheduled and agreed;
 - 1.5.4 introduce any new contractual or customary practice concerning the making of any lump sum payment on the termination of employment of any employees listed on the Supplier's Provisional Supplier Personnel List;
 - 1.5.5 increase or reduce the total number of employees so engaged, or deploy any other person to perform the Services (or the relevant part of the Services); or
 - 1.5.6 terminate or give notice to terminate the employment or contracts of any persons on the Supplier's Provisional Supplier Personnel List save by due disciplinary process, and shall promptly notify, and procure that each SubContractor shall promptly notify, the Customer or, at the direction of the Customer, any Replacement Supplier and any Replacement Sub-Contractor of any notice to terminate employment given by the Supplier or relevant SubContractor or received from any persons listed on the Supplier's Provisional Supplier Personnel List regardless of when such notice takes effect.

- 1.6 During the Term, the Supplier shall provide, and shall procure that each Sub-Contractor shall provide, to the Customer any information the Customer may reasonably require relating to the manner in which Services are organised, which shall include:
 - 1.6.1 the numbers of employees engaged in providing the Services;
 - 1.6.2 the percentage of time spent by each employee engaged in providing the Services; and
 - 1.6.3 a description of the nature of the work undertaken by each employee by location.
- 1.7 The Supplier shall provide, and shall procure that each Sub-Contractor shall provide, all reasonable cooperation and assistance to the Customer, any Replacement Supplier and/or any Replacement Sub-Contractor to ensure the smooth transfer of the Transferring Supplier Employees on the Service Transfer Date including providing sufficient information in advance of the Service Transfer Date to ensure that all necessary payroll arrangements can be made to enable the Transferring Supplier Employees to be paid as appropriate. Without prejudice to the generality of the foregoing, within five (5) Working Days following the Service Transfer Date, the Supplier shall provide, and shall procure that each SubContractor shall provide, to the Customer or, at the direction of the Customer, to any Replacement Supplier and/or any Replacement Sub-Contractor (as appropriate), in respect of each person on the Supplier's Final Supplier Personnel List who is a Transferring Supplier Employee:
 - 1.7.1 the most recent month's copy pay slip data;
 - 1.7.2 details of cumulative pay for tax and pension purposes;
 - 1.7.3 details of cumulative tax paid;
 - 1.7.4 tax code;
 - 1.7.5 details of any voluntary deductions from pay; and
 - 1.7.6 bank/building society account details for payroll purposes.

2. EMPLOYMENT REGULATIONS EXIT PROVISIONS

2.1 The Customer and the Supplier acknowledge that subsequent to the commencement of the provision of the Services, the identity of the provider of the Services (or any part of the

Services) may change (whether as a result of termination or Partial Termination of this Contract or otherwise) resulting in the Services being undertaken by a Replacement Supplier and/or a Replacement Sub-Contractor. Such change in the identity of the Supplier of such Services may constitute a Relevant Transfer to which the Employment Regulations and/or the Acquired Rights Directive will apply. The Customer and the Supplier further agree that, as a result of the operation of the Employment Regulations, where a Relevant Transfer occurs, the contracts of employment between the Supplier and the Transferring Supplier Employees (except in relation to any contract terms disapplied through operation of regulation 10(2) of the Employment Regulations) will have effect on and from the Service Transfer Date as if originally made between the Replacement Supplier and/or a Replacement Sub-Contractor (as the case may be) and each such Transferring Supplier Employee.

2.2 The Supplier shall, and shall procure that each Sub-Contractor shall, comply with all its obligations in respect of the Transferring Supplier Employees arising under the Employment Regulations in respect of the period up to (but not including) the Service Transfer Date and shall perform and discharge, and procure that each Sub-Contractor shall perform and discharge, all its obligations in respect of all the Transferring Supplier Employees arising in respect of the period up to (and including) the Service Transfer Date (including the payment of all remuneration, benefits, entitlements and outgoings, all wages, accrued but untaken holiday pay, bonuses, commissions, payments of PAYE, national insurance contributions and pension contributions which in any case are attributable in whole or in part to the period ending on (and including) the Service Transfer Date) and any necessary apportionments in respect of any periodic payments shall be made between: (i) the Supplier and/or the Sub-Contractor (as appropriate); and (ii) the Replacement Supplier and/or Replacement Sub-Contractor.

- 2.3 Subject to Paragraph 2.4, where a Relevant Transfer occurs the Supplier shall indemnify the Customer and/or the Replacement Supplier and/or any Replacement Sub-Contractor against any Employee Liabilities in respect of any Transferring Supplier Employee (or, where applicable any employee representative as defined in the Employment Regulations) arising from or as a result of:
 - 2.3.1 any act or omission of the Supplier or any Sub-Contractor whether occurring before, on or after the Service Transfer Date;
 - 2.3.2 the breach or non-observance by the Supplier or any Sub-Contractor occurring on or before the Service Transfer Date of:
 - (a) any collective agreement applicable to the Transferring Supplier Employees; and/or
 - (b) any other custom or practice with a trade union or staff association in respect of any Transferring Supplier Employees which the Supplier or any Sub-Contractor is contractually bound to honour;
 - 2.3.3 any claim by any trade union or other body or person representing any Transferring Supplier Employees arising from or connected with any failure by the Supplier or a Sub-Contractor to comply with any legal obligation to such trade union, body or person arising on or before the Service Transfer Date;
 - 2.3.4 any proceeding, claim or demand by HMRC or other statutory authority in respect of any financial obligation including, but not limited to, PAYE and primary and secondary national insurance contributions:
 - (a) in relation to any Transferring Supplier Employee, to the extent that the proceeding, claim or demand by HMRC or other statutory authority

relates to financial obligations arising on and before the Service Transfer Date; and

- (b) in relation to any employee who is not a Transferring Supplier Employee, and in respect of whom it is later alleged or determined that the Employment Regulations applied so as to transfer his/her employment from the Supplier to the Customer and/or Replacement Supplier and/or any Replacement Sub-Contractor, to the extent that the proceeding, claim or demand by HMRC or other statutory authority relates to financial obligations arising on or before the Service Transfer Date;
- 2.3.5 a failure of the Supplier or any Sub-Contractor to discharge or procure the discharge of all wages, salaries and all other benefits and all PAYE tax deductions and national insurance contributions relating to the Transferring Supplier Employees in respect of the period up to (and including) the Service Transfer Date);
- 2.3.6 any claim made by or in respect of any person employed or formerly employed by the Supplier or any Sub-Contractor other than a Transferring Supplier Employee for whom it is alleged the Customer and/or the Replacement Supplier and/or any Replacement Sub-Contractor may be liable by virtue of this Contract and/or the Employment Regulations and/or the Acquired Rights Directive; and
- 2.3.7 any claim made by or in respect of a Transferring Supplier Employee or any appropriate employee representative (as defined in the Employment

Regulations) of any Transferring Supplier Employee relating to any act or omission of the Supplier or any Sub-Contractor in relation to its obligations under regulation 13 of the Employment Regulations, except to the extent that the liability arises from the failure by the Customer and/or Replacement Supplier to comply with regulation 13(4) of the Employment Regulations.

- 2.4 The indemnities in Paragraph 2.3 shall not apply to the extent that the Employee Liabilities arise or are attributable to an act or omission of the Replacement Supplier and/or any Replacement Sub-Contractor whether occurring or having its origin before, on or after the Service Transfer Date, including any Employee Liabilities:
 - 2.4.1 arising out of the resignation of any Transferring Supplier Employee before the Service Transfer Date on account of substantial detrimental changes to his/her working conditions proposed by the Replacement Supplier and/or any Replacement Sub-Contractor to occur in the period on or after the Service Transfer Date; or
 - 2.4.2 arising from the Replacement Supplier's failure, and/or Replacement SubContractor's failure, to comply with its obligations under the Employment Regulations.
- 2.5 If any person who is not a Transferring Supplier Employee claims, or it is determined in relation to any person who is not a Transferring Supplier Employee, that his/her contract of employment has been transferred from the Supplier or any Sub-Contractor to the Replacement Supplier and/or Replacement Sub-Contractor pursuant to the Employment Regulations or the Acquired Rights Directive, then:
 - 2.5.1 the Customer shall procure that the Replacement Supplier shall, or any Replacement Sub-Contractor shall, within five (5) Working Days of becoming aware of that fact, give notice in writing to the Supplier; and
 - 2.5.2 the Supplier may offer (or may procure that a Sub-Contractor may offer) employment to such person within fifteen (15) Working Days of the notification by the Replacement Supplier and/or any and/or Replacement Sub-Contractor or take such other reasonable steps as it considers appropriate to deal with the matter provided always that such steps are in compliance with Law.
- 2.6 If such offer is accepted, or if the situation has otherwise been resolved by the Supplier or a Sub-Contractor, the Customer shall procure that the Replacement Supplier shall, or procure that the Replacement Sub-Contractor shall, immediately release or procure the release of the person from his/her employment or alleged employment.
- 2.7 If after the fifteen (15) Working Day period specified in Paragraph 2.5.2 has elapsed:
 - 2.7.1 no such offer of employment has been made;
 - 2.7.2 such offer has been made but not accepted; or 2.7.3 the

situation has not otherwise been resolved

the Customer shall advise the Replacement Supplier and/or Replacement Sub-Contractor, as appropriate that it may within five (5) Working Days give notice to terminate the employment or alleged employment of such person.

2.8 Subject to the Replacement Supplier and/or Replacement Sub-Contractor acting in accordance with the provisions of Paragraphs 2.5 to 2.7, and in accordance with all applicable proper employment procedures set out in applicable Law, the Supplier shall indemnify the Replacement Supplier and/or Replacement Sub-Contractor against all Employee Liabilities arising out of the termination pursuant to the provisions of Paragraph 2.7 provided that the

Replacement Supplier takes, or shall procure that the Replacement Sub-Contractor takes, all reasonable steps to minimise any such Employee Liabilities.

- 2.9 The indemnity in Paragraph 2.8:
 - 2.9.1 shall not apply to:
 - (a) any claim for:
 - discrimination, including on the grounds of sex, race, disability, age, gender reassignment, marriage or civil partnership, pregnancy and maternity or sexual orientation, religion or belief; or
 - (ii) equal pay or compensation for less favourable treatment of parttime workers or fixed-term employees,

in any case in relation to any alleged act or omission of the Replacement Supplier and/or Replacement Sub-Contractor; or

- (b) any claim that the termination of employment was unfair because the Replacement Supplier and/or Replacement Sub-Contractor neglected to follow a fair dismissal procedure; and
- 2.9.2 shall apply only where the notification referred to in Paragraph 2.5.1 is made by the Replacement Supplier and/or Replacement Sub-Contractor to the Supplier within six (6) months of the Service Transfer Date.
- 2.10 If any such person as is described in Paragraph 2.5 is neither re-employed by the Supplier or any Sub-Contractor nor dismissed by the Replacement Supplier and/or Replacement Sub-Contractor within the time scales set out in Paragraphs 2.5 to 2.7, such person shall be treated as a Transferring Supplier Employee and the Replacement Supplier and/or Replacement Sub-Contractor shall comply with such obligations as may be imposed upon it under applicable Law.
- 2.11 The Supplier shall comply, and shall procure that each Sub-Contractor shall comply, with all its obligations under the Employment Regulations and shall perform and discharge, and shall procure that each Sub-Contractor shall perform and discharge, all its obligations in respect of the Transferring Supplier Employees before and on the Service Transfer Date (including the payment of all remuneration, benefits, entitlements and outgoings, all wages, accrued but untaken holiday pay, bonuses, commissions, payments of PAYE, national insurance contributions and pension contributions which in any case are attributable in whole or in part in respect of the period up to (and including) the Service Transfer Date) and any necessary apportionments in respect of any periodic payments shall be made between:
 - 2.11.1 the Supplier and/or any Sub-Contractor; and
 - 2.11.2 the Replacement Supplier and/or the Replacement Sub-Contractor.
- 2.12 The Supplier shall, and shall procure that each Sub-Contractor shall, promptly provide to the Customer and any Replacement Supplier and/or Replacement Sub-Contractor, in writing such information as is necessary to enable the Customer, the Replacement Supplier and/or Replacement Sub-Contractor to carry out their respective duties under regulation 13 of the Employment Regulations. The Customer shall procure that the Replacement Supplier and each Sub-Contractor in writing such information as is necessary to enable the Supplier and each Sub-Contractor to carry out their respective duties under regulation 13 of the Employment Regulations.

- 2.13 Subject to Paragraph 2.14, where a Relevant Transfer occurs the Customer shall procure that the Replacement Supplier indemnifies the Supplier on its own behalf and on behalf of any Replacement Sub-Contractor and its sub-contractors against any Employee Liabilities in respect of each Transferring Supplier Employee (or, where applicable any employee representative (as defined in the Employment Regulations) of any Transferring Supplier Employee) arising from or as a result of:
 - 2.13.1 any act or omission of the Replacement Supplier and/or Replacement SubContractor;
 - 2.13.2 the breach or non-observance by the Replacement Supplier and/or Replacement Sub-Contractor on or after the Service Transfer Date of:
 - (a) any collective agreement applicable to the Transferring Supplier Employees; and/or
 - (b) any custom or practice in respect of any Transferring Supplier
 Employees which the Replacement Supplier and/or Replacement Sub-Contractor is contractually bound to honour;
 - 2.13.3 any claim by any trade union or other body or person representing any Transferring Supplier Employees arising from or connected with any failure by the Replacement Supplier and/or Replacement Sub-Contractor to comply with any legal obligation to such trade union, body or person arising on or after the Relevant Transfer Date;
 - 2.13.4 any proposal by the Replacement Supplier and/or Replacement Sub-Contractor to change the terms and conditions of employment or working conditions of any Transferring Supplier Employees on or after their transfer to the Replacement Supplier or Replacement Sub-Contractor (as the case may be) on the Relevant Transfer Date, or to change the terms and conditions of employment or working

conditions of any person who would have been a Transferring Supplier Employee but for their resignation (or decision to treat their employment as terminated under regulation 4(9) of the Employment Regulations) before the Relevant Transfer Date as a result of or for a reason connected to such proposed changes;

- 2.13.5 any statement communicated to or action undertaken by the Replacement Supplier or Replacement Sub-Contractor to, or in respect of, any Transferring Supplier Employee on or before the Relevant Transfer Date regarding the Relevant Transfer which has not been agreed in advance with the Supplier in writing;
- 2.13.6 any proceeding, claim or demand by HMRC or other statutory authority in respect of any financial obligation including, but not limited to, PAYE and primary and secondary national insurance contributions:
 - in relation to any Transferring Supplier Employee, to the extent that the proceeding, claim or demand by HMRC or other statutory authority relates to financial obligations arising after the Service Transfer Date; and
 - (b) in relation to any employee who is not a Transferring Supplier Employee, and in respect of whom it is later alleged or determined that the Employment Regulations applied so as to transfer his/her employment from the Supplier or Sub-Contractor, to the Replacement Supplier or Replacement Sub-Contractor to the extent that the proceeding, claim or demand by HMRC or other statutory authority relates to financial obligations arising after the Service Transfer Date;
- 2.13.7 a failure of the Replacement Supplier or Replacement Sub-Contractor to discharge or procure the discharge of all wages, salaries and all other benefits and all PAYE tax deductions and national insurance contributions relating to the Transferring

Supplier Employees in respect of the period from (and including) the Service Transfer Date; and

- 2.13.8 any claim made by or in respect of a Transferring Supplier Employee or any appropriate employee representative (as defined in the Employment Regulations) of any Transferring Supplier Employee relating to any act or omission of the Replacement Supplier or Replacement Sub-Contractor in relation to obligations under regulation 13 of the Employment Regulations.
- 2.14 The indemnities in Paragraph 2.13 shall not apply to the extent that the Employee Liabilities arise or are attributable to an act or omission of the Supplier and/or any SubContractor (as applicable) whether occurring or having its origin before, on or after the

Relevant Transfer Date, including any Employee Liabilities arising from the failure by the Supplier and/or any Sub-Contractor (as applicable) to comply with its obligations under the Employment Regulations.

ANNEX TO SCHEDULE 5: LIST OF NOTIFIED SUB-CONTRACTORS

CONTRACT SCHEDULE 6: DISPUTE RESOLUTION PROCEDURE

1. **DEFINITIONS**

1.1. In this Contract Schedule 6, the following definitions shall apply:

"CEDR"	the Centre for Effective Dispute Resolution of International Dispute Resolution Centre, 70 Fleet Street, London, EC4Y 1EU;
"Counter Notice" has the	meaning given to it in paragraph of this Contract Schedule 6;
"Exception" a deviation	of project tolerances in accordance with PRINCE2 methodology in respect of this Contract or in the supply of the Services;
"Expert"	the person appointed by the Parties in accordance with paragraph 5.2 of this Contract Schedule 6; and
"Mediation Notice" has the meaning given to it in paragraph 3.2 of this Contract Schedule 6;	
"Mediator"	the independent third party appointed in accordance with paragraph 4.2 of this Contract Schedule 6.

2. INTRODUCTION

- 2.1. If a Dispute arises then:
 - 2.1.1. the representative of the Customer and the Supplier Representative shall attempt in good faith to resolve the Dispute; and
 - 2.1.2. if such attempts are not successful within a reasonable time either Party may give to the other a Dispute Notice.
- 2.2. The Dispute Notice shall set out:
 - 2.2.1. the material particulars of the Dispute;
 - 2.2.2. the reasons why the Party serving the Dispute Notice believes that the Dispute has arisen; and
 - 2.2.3. if the Party serving the Dispute Notice believes that the Dispute should be dealt with under the Expedited Dispute Timetable as set out in paragraph 2.6 of this Contract Schedule 6, the reason why.
- 2.3. Unless agreed otherwise in writing, the Parties shall continue to comply with their respective obligations under this Contract regardless of the nature of the Dispute and notwithstanding the referral of the Dispute to the Dispute Resolution Procedure.
- 2.4. Subject to paragraph 3.2 of this Contract Schedule 6, the Parties shall seek to resolve Disputes:
 - 2.4.1. first by commercial negotiation (as prescribed in paragraph 3 of this Contract Schedule 6);
 - 2.4.2. then by mediation (as prescribed in paragraph 4 of this Contract Schedule 6); and
 - 2.4.3. lastly by recourse to arbitration (as prescribed in paragraph 6 of this Contract Schedule6) or litigation (in accordance with Clause 47 of this Contract (Governing Law and Jurisdiction)).

- 2.5. Specific issues shall be referred to Expert Determination (as prescribed in paragraph 5 of this Contract Schedule 6) where specified under the provisions of this Contract and may also be referred to Expert Determination where otherwise appropriate as specified in paragraph 5 of this Contract Schedule 6.
- 2.6. In exceptional circumstances where the use of the times in this Contract Schedule 6 would be unreasonable, including (by way of example) where one Party would be materially disadvantaged by a delay in resolving the Dispute, the Parties may agree to use the Expedited Dispute Timetable. If the Parties are unable to reach agreement on whether to use of the Expedited Dispute Timetable within five (5) Working Days of the issue of the Dispute Notice, the use of the Expedited Dispute Timetable shall be at the sole discretion of the Customer.
- 2.7. If the use of the Expedited Dispute Timetable is determined in accordance with paragraph 2.5 or is otherwise specified under the provisions of this Contract, then the following periods of time shall apply in lieu of the time periods specified in the applicable paragraphs:
 - 2.7.1. in paragraph 3.2.3, ten (10) Working Days;
 - 2.7.2. in paragraph 4.2, ten (10) Working Days;
 - 2.7.3. in paragraph 5.2, five (5) Working Days; and
 - 2.7.4. in paragraph 6.2, ten (10) Working Days.
- 2.8. If at any point it becomes clear that an applicable deadline cannot be met or has passed, the Parties may (but shall be under no obligation to) agree in writing to extend the deadline. Any agreed extension shall have the effect of delaying the start of the subsequent stages by the period agreed in the extension.

3. COMMERCIAL NEGOTIATIONS

- 3.1. Following the service of a Dispute Notice, the Customer and the Supplier shall use reasonable endeavours to resolve the Dispute as soon as possible, by discussion between the Customer Representative and the Supplier Representative.
- 3.2. If:
 - 3.2.1. either Party is of the reasonable opinion that the resolution of a Dispute by commercial negotiation, or the continuance of commercial negotiations, will not result in an appropriate solution;
 - 3.2.2. the Parties have already held discussions of a nature and intent (or otherwise were conducted in the spirit) that would equate to the conduct of commercial negotiations in accordance with this paragraph 3 of this Contract Schedule 6; or
 - 3.2.3. the Parties have not settled the Dispute in accordance with paragraph 3.1 of this Contract Schedule 6 within thirty (30) Working Days of service of the Dispute Notice,

either Party may serve a written notice to proceed to mediation (a "**Mediation Notice**") in accordance with paragraph 4 of this Contract Schedule 6.

4. MEDIATION

- 4.1. If a Mediation Notice is served, the Parties shall attempt to resolve the dispute in accordance with CEDR's Model Mediation Agreement which shall be deemed to be incorporated by reference into this Contract.
- 4.2. If the Parties are unable to agree on the joint appointment of a Mediator within thirty (30) Working Days from service of the Mediation Notice then either Party may apply to CEDR to nominate the Mediator.

- 4.3. If the Parties are unable to reach a settlement in the negotiations at the mediation, and only if the Parties so request and the Mediator agrees, the Mediator shall produce for the Parties a nonbinding recommendation on terms of settlement. This shall not attempt to anticipate what a court might order but shall set out what the Mediator suggests are appropriate settlement terms in all of the circumstances.
- 4.4. Any settlement reached in the mediation shall not be legally binding until it has been reduced to writing and signed by, or on behalf of, the Parties (in accordance with the Variation Procedure where appropriate). The Mediator shall assist the Parties in recording the outcome of the mediation.

5. **EXPERT DETERMINATION**

- 5.1. If a Dispute relates to any aspect of the technology underlying the provision of the Services or otherwise relates to a financial technical or other aspect of a technical nature (as the Parties may agree) and the Dispute has not been resolved by discussion or mediation, then either Party may request (which request will not be unreasonably withheld or delayed) by written notice to the other that the Dispute is referred to an Expert for determination.
- 5.2. The Expert shall be appointed by agreement in writing between the Parties, but in the event of a failure to agree within ten (10) Working Days, or if the person appointed is unable or unwilling to act, the Expert shall be appointed on the instructions of the relevant professional body.
- 5.3. The Expert shall act on the following basis:
 - 5.3.1. he/she shall act as an expert and not as an arbitrator and shall act fairly and impartially;
 - 5.3.2. the Expert's determination shall (in the absence of a material failure to follow the agreed procedures) be final and binding on the Parties;
 - 5.3.3. the Expert shall decide the procedure to be followed in the determination and shall be requested to make his/her determination within thirty (30) Working Days of his appointment or as soon as reasonably practicable thereafter and the Parties shall assist and provide the documentation that the Expert requires for the purpose of the determination;
 - 5.3.4. any amount payable by one Party to another as a result of the Expert's determination shall be due and payable within twenty (20) Working Days of the Expert's determination being notified to the Parties;
 - 5.3.5. the process shall be conducted in private and shall be confidential; and
 - 5.3.6. the Expert shall determine how and by whom the costs of the determination, including his/her fees and expenses, are to be paid.

6. **ARBITRATION**

- 6.1. The Customer may at any time before court proceedings are commenced refer the Dispute to arbitration in accordance with the provisions of paragraph 6.4 of this Contract Schedule 6.
- 6.2. Before the Supplier commences court proceedings or arbitration, it shall serve written notice on the Customer of its intentions and the Customer shall have fifteen (15) Working Days following receipt of such notice to serve a reply (a "Counter Notice") on the Supplier requiring the Dispute to be referred to and resolved by arbitration in accordance with paragraph 6.4 of

this Contract Schedule 6 or be subject to the jurisdiction of the courts in accordance with Clause 47 of this Contract (Governing Law and Jurisdiction). The Supplier shall not commence any court proceedings or arbitration until the expiry of such fifteen (15) Working Day period.

6.3. If:

- 6.3.1. the Counter Notice requires the Dispute to be referred to arbitration, the provisions of paragraph 6.4 of this Contract Schedule 6 shall apply;
- 6.3.2. the Counter Notice requires the Dispute to be subject to the exclusive jurisdiction of the courts in accordance with Clause 47 of this Contract (Governing Law and Jurisdiction), the Dispute shall be so referred to the courts and the Supplier shall not commence arbitration proceedings;
- 6.3.3. the Customer does not serve a Counter Notice within the fifteen (15) Working Days period referred to in paragraph 6.2 of this Contract Schedule 6, the Supplier may either commence arbitration proceedings in accordance with paragraph 6.4 of this Contract Schedule 6 or commence court proceedings in the courts in accordance with Clause 47 of this Contract (Governing Law and Jurisdiction) which shall (in those circumstances) have exclusive jurisdiction.
- 6.4. In the event that any arbitration proceedings are commenced pursuant to paragraphs 6.1 to6.3 of this Contract Schedule 6, the Parties hereby confirm that:
 - 6.4.1. all disputes, issues or claims arising out of or in connection with this Contract (including as to its existence, validity or performance) shall be referred to and finally resolved by arbitration under the Rules of the London Court of International Arbitration ("LCIA") (subject to paragraphs 6.4.5 to 6.4.7 of this Contract Schedule 6);
 - 6.4.2. the arbitration shall be administered by the LCIA;
 - 6.4.3. the LCIA procedural rules in force at the date that the Dispute was referred to arbitration shall be applied and are deemed to be incorporated by reference into this Contract and the decision of the arbitrator shall be binding on the Parties in the absence of any material failure to comply with such rules;
 - 6.4.4. if the Parties fail to agree the appointment of the arbitrator within ten (10) days from the date on which arbitration proceedings are commenced or if the person appointed is unable or unwilling to act, the arbitrator shall be appointed by the LCIA;
 - 6.4.5. the chair of the arbitral tribunal shall be British;
 - 6.4.6. the arbitration proceedings shall take place in London and in the English language; and
 - 6.4.7. the seat of the arbitration shall be London.

7. URGENT RELIEF

- 7.1. Either Party may at any time take proceedings or seek remedies before any court or tribunal of competent jurisdiction:
 - 7.1.1. for interim or interlocutory remedies in relation to this Contract or infringement by the other Party of that Party's Intellectual Property Rights; and/or
 - 7.1.2. where compliance with paragraph 2.1 of this Contract Schedule 6 and/or referring the Dispute to mediation may leave insufficient time for that Party to commence proceedings before the expiry of the limitation period.

CONTRACT SCHEDULE 7: PROCESSING PERSONAL DATA AND DATA SUBJECTS

This Schedule shall be completed by the Controller, who may take account of the view of the Processors, however the final decision as to the content of this Schedule shall be with the Controller at its absolute discretion.

- The contact details of the Controller's Data Protection Officer are: Emma Wharram, Deputy Director -Departmental Data Protection Officer, dataprotection.office@education.gov.uk, Department for Education, 7 & 8 Wellington Place, Wellington Street, Leeds, LS1 4AW
- 2. The contact details of the Processor's Data Protection Officer are: [Insert Supplier Contact details at Award]

- 3. The Processor shall comply with any further written instructions with respect to processing by the Controller.
- 4. Any such further instructions shall be incorporated into this Schedule.

Details	
The Parties acknowledge that for the purposes of the Data Protection Legislation, the Customer is the Controller and the Contractor is the Processor in accordance with Clause 23.22.	
The processing of personal data relating to applicants for, Learners and employers taking part in Skills Bootcamp(s) is required to ensure the mandatory requirements of the Service Requirements are delivered by the contractor.	
Where this information may already be collected or processed by the processor, this is to be considered separate processing activities and the contractor will be the data controller for those processing activities and out of scope for this contract.	
From the commencement date of the call off agreement until the Expiry Date.	
Personal data will be collected, stored, used and disclosed by the processor	
For the purposes of:	
 Recruitment of Learners to enrol on Skills Bootcamps courses Provision of wraparound support for Learners 	
 Monitoring and Assessment of Learners undertaking Skills Bootcamps and 	
meeting the evidence requirements of the Call Off Contract	
 All Applicants: Date of Birth Gender Disability/long-term health condition Ethnicity Marriage and civil partnership status Sexual orientation 	

	 Gender re-assignment Pregnancy and maternity Religion or belief Learners:
	 Name (first name and surname) National Insurance number Postcode Email address Telephone number
	Employer Contacts: Name Email address Telephone Number
Categories of Data Subject	 Applicants for a Skills Bootcamp Learners taking part in Skills Bootcamps. Employers
Plan for return and destruction of the data once the processing is complete UNLESS requirement under union or member state law to preserve that type of data	Unless further written instructions are provided by the Controller: Personal Data will be retained for six years past the end of the financial year in which the last payment is made, to support audit requirements, at which point it will be securely destroyed in line with the Service Requirements.
Plan for return and destruction of the data once the processing is complete UNLESS requirement under union or member state law to preserve that type of data	 Name Email address Telephone Number Applicants for a Skills Bootcamp Learners taking part in Skills Bootcamps. Employers Unless further written instructions are provided by the Controller: Personal Data will be retained for six years past the end of the financial year in which the last payment is made, to support audit requirements, at which point it will be securely destroyed in line with the Service

CONTRACT SCHEDULE 8: JOINT CONTROLLER AGREEMENT

[Guidance: It is anticipated that the arrangement will be Controller and Processor. Insert this schedule only where a Joint Controller arrangement applies in Schedule 7]

[In this Annex the Parties must outline each party's responsibilities for:

- providing information to data subjects under <u>Article 13 and 14</u> of the GDPR.
- responding to data subject requests under <u>Articles 15-22</u> of the GDPR
- notifying the Information Commissioner (and data subjects) where necessary about data breaches
- maintaining records of processing under <u>Article 30</u> of the GDPR
- carrying out any required Data Protection Impact Assessment
- The agreement must include a statement as to who is the point of contact for data subjects.

The essence of this relationship shall be published.

You may wish to incorporate some clauses equivalent to those specified in Clauses 23.22 - 23.36

You may also wish to include an additional clause apportioning liability between the parties arising out of data protection; of data that is jointly controlled.

Where there is a Joint Control relationship, but no controller to processor relationship under the Call Off Contract, this completed Contract Schedule 8 should be used instead of Clause 23.22 - 23.36]

CONTRACT SCHEDULE 9: NOT USED

CONTRACT SCHEDULE 10: EXIT MANAGEMENT

1. **DEFINITIONS**

1.1. In this Contract Schedule 10, the following definitions shall apply:

	e fonotning deminitions shan appry.
"Exclusive Assets"	means those Supplier Assets used by the Supplier or a Key Sub-Contractor which are used exclusively in the provision of the Services;
"Exit Information"	has the meaning given to it in paragraph 4.1 of this Contract Schedule 10;
"Exit Manager"	means the person appointed by each Party pursuant to paragraph 3.4 of this Contract Schedule 10 for managing the Parties' respective obligations under this Contract Schedule 10;
"Net Book Value"	means the net book value of the relevant Supplier Asset(s) calculated in accordance with the depreciation policy of the Supplier set out in the letter in the agreed form from the Supplier to the Costumer of even date with this Contract;
"Non-Exclusive Assets"	means those Supplier Assets (if any) which are used by the Supplier or a Key Sub Contractor in connection with the Services but which are also used by the Supplier or Key Sub-Contractor for other purposes;
"Registers"	means the register and configuration database referred to in paragraphs 3.1.1 and 3.1.2 of this Contract Schedule 10;
"Termination Assistance"	means the activities to be performed by the Supplier pursuant to the Exit Plan, and any other assistance required by the Customer pursuant to the Termination Assistance Notice;
"Termination Assistance Notice"	has the meaning given to it in paragraph 6.1 of this Contract Schedule 10;
"Termination Assistance Period"	means in relation to a Termination Assistance Notice, the period specified in the Termination Assistance Notice for which the Supplier is required to provide the Termination Assistance as such period may be extended pursuant to paragraph 6.2 of this Contract Schedule 10;
"Transferable Assets"	means those of the Exclusive Assets which are capable of legal transfer to the Customer;
"Transferable Contracts"	means the Sub-Contracts, licences for Supplier Background IPR, Project Specific IPR, licences for

	Third Party IPR or other agreements which are necessary to enable the Customer or any Replacement Supplier to provide the Services or the Replacement Services, including				
	in releva	relation nt Documenta	to tion:	licences	all
"Transferring Assets"	has the	e meaning give of this Contract	en to it in		
"Transferring Contracts"		ning given to i chedule 10.	t in parag	raph 9.2.3 of tł	nis Contract

2. INTRODUCTION

- 2.1. This Contract Schedule 10 describes provisions that should be included in the Exit Plan, the duties and responsibilities of the Supplier to the Customer leading up to and covering the Contract Expiry Date and the transfer of service provision to the Customer and/or a Replacement Supplier.
- 2.2. The objectives of the exit planning and service transfer arrangements are to ensure a smooth transition of the availability of the Services from the Supplier to the Customer and/or a Replacement Supplier at the Contract Expiry Date.

3. OBLIGATIONS DURING THE CONTRACT PERIOD TO FACILITATE EXIT

- 3.1. During the Contract Period, the Supplier shall:
 - 3.1.1. create and maintain a Register of all:
 - (a) Supplier Assets, detailing their: make, model and
 - i) asset number;
 - ii) ownership and status as either Exclusive Assets or Non Exclusive Assets;
 - iii) Net Book Value;
 - iv) condition and physical location; and
 - v) use (including technical specifications);
 - (b) Sub-Contracts and other relevant agreements (including relevant software licences, maintenance and support agreements and equipment rental and lease agreements) required for the performance of the Services;
 - (c) Information and data relating to Learners (including Learner files) and course material used for the Skill Bootcamps; and
 - (d) Supplier Background IPR, Third Party IPR, Project Specific IPR, Customer Background IPR and Customer Data.
 - 3.1.2. Customer Data. create and maintain a configuration database detailing the technical infrastructure and operating procedures through which the Supplier provides the Services, which shall contain sufficient detail to permit the Customer and/or Replacement Supplier to understand how the Supplier provides the Services

and to enable the smooth transition of the Services with the minimum of disruption;

- 3.1.3. agree the format of the Registers with the Customer as part of the process of agreeing the Exit Plan; and
- 3.1.4. at all times keep the Registers up to date, in particular in the event that Assets, SubContracts or other relevant agreements and information relating to the Learners (including Learner files) and / or course material used for the Skill Bootcamps are added to or removed from the Services.
- 3.2. The Supplier shall:
 - 3.2.1. procure that all Exclusive Assets listed in the Registers are clearly marked to identify that they are exclusively used for the provision of the Services under this Contract; and
 - 3.2.2. (unless otherwise agreed by the Customer in writing) procure that all licences for Third Party IPR and all Sub-Contracts shall be assignable and/or capable of novation at the request of the Customer to the Customer (and/or its nominee) and/or any Replacement Supplier upon the Supplier ceasing to provide the Services (or part of them) without restriction (including any need to obtain any consent or approval) or payment by the Customer.
- 3.3. Where the Supplier is unable to procure that any Sub-Contract or other agreement referred to in paragraph 3.2.2 of this Contract Schedule 10 which the Supplier proposes to enter into after the Contract Commencement Date is assignable and/or capable of novation to the Customer (and/or its nominee) and/or any Replacement Supplier without restriction or payment, the Supplier shall promptly notify the Customer of this and the Parties shall (acting reasonably and without undue delay) discuss the appropriate action to be taken which, where the Customer so directs, may include the Supplier seeking an alternative SubContractor or provider of services to which the relevant agreement relates.
- 3.4. Each Party shall appoint a person for the purposes of managing the Parties' respective obligations under this Contract Schedule 10 and provide written notification of such appointment to the other Party within three (3) Months of the Contract Commencement Date. The Supplier's Exit Manager shall be responsible for ensuring that the Supplier and its employees, agents and Sub-Contractors comply with this Contract Schedule 10. The Supplier shall ensure that its Exit Manager has the requisite authority to arrange and procure any resources of the Supplier as are reasonably necessary to enable the Supplier to comply with the requirements set out in this Contract Schedule 10. The Parties' Exit Managers will liaise with one another in relation to all issues relevant to the termination of this Contract and all matters connected with this Contract Schedule 10 and each Party's compliance with it.

4. OBLIGATIONS TO ASSIST ON RE-TENDERING OF SERVICES

- 4.1. On reasonable notice at any point during the Contract Period, the Supplier shall provide to the Customer and/or its potential Replacement Supplier's (subject to the potential Replacement Supplier's entering into reasonable written confidentiality undertakings), the following material and information in order to facilitate the preparation by the Customer of any invitation to tender and/or to facilitate any potential Replacement Supplier's undertaking due diligence:
 - 4.1.1. details of the Service(s);
 - 4.1.2. a copy of the Registers, updated by the Supplier up to the date of delivery of such Registers;
 - 4.1.3. an inventory of Customer Data in the Supplier's possession or control;
 - 4.1.4. details of any key terms of any third party contracts and licences, particularly as regards charges, termination, assignment and novation;
 - 4.1.5. a list of on-going and/or threatened disputes in relation to the provision of the Services;

4.1.6. all information relating to Transferring Supplier Employees required to be provided by the Supplier under this Contract; and

4.1.7. such other material and information as the Customer shall reasonably require, (together,

the "Exit Information").

- 4.2. The Supplier acknowledges that the Customer may disclose the Supplier's Confidential Information to an actual or prospective Replacement Supplier or any third party whom the Customer is considering engaging to the extent that such disclosure is necessary in connection with such engagement (except that the Customer may not under this paragraph 4.2 of this Contract Schedule 10 disclose any Supplier's Confidential Information which is information relating to the Supplier's or its Sub-Contractors' prices or costs).
- 4.3. The Supplier shall:
 - 4.3.1. notify the Customer within five (5) Working Days of any material change to the Exit Information which may adversely impact upon the provision of any Services and shall consult with the Customer regarding such proposed material changes; and
 - 4.3.2. provide complete updates of the Exit Information on an as-requested basis as soon as reasonably practicable and in any event within ten (10) Working Days of a request in writing from the Customer.
- 4.4. The Supplier may charge the Customer for its reasonable additional costs to the extent the Customer requests more than four (4) updates in any six (6) month period.
- 4.5. The Exit Information shall be accurate and complete in all material respects and the level of detail to be provided by the Supplier shall be such as would be reasonably necessary to enable a third party to:
 - 4.5.1. prepare an informed offer for those Services; and
 - 4.5.2. not be disadvantaged in any subsequent procurement process
 - 4.5.3. compared to the Supplier (if the Supplier is invited to participate).

5. EXIT PLAN

- 5.1. The Supplier shall, within three (3) Months after the Contract Commencement Date, deliver to the Customer an Exit Plan which:
 - 5.1.1. sets out the Supplier's proposed methodology for achieving an orderly transition of the Services from the Supplier to the Customer and/or its Replacement Supplier on the expiry or termination of this Contract;
 - 5.1.2. complies with the requirements set out in paragraph 5.3 of this Contract Schedule 10;
 - 5.1.3. sets out an orderly transfer of the items recorded in the Registers including but not limited to the IPR rights as referred to in Paragraph 3.1.1 (d).
 - 5.1.4. is otherwise reasonably satisfactory to the Customer.
- 5.2. The Parties shall use reasonable endeavours to agree the contents of the Exit Plan. If the Parties are unable to agree the contents of the Exit Plan within twenty (20) Working Days of

its submission, then such Dispute shall be resolved in accordance with the Dispute Resolution Procedure.

- 5.3. Unless otherwise specified by the Customer or Approved, the Exit Plan shall set out, as a minimum:
 - 5.3.1. how the Exit Information is obtained;
 - 5.3.2. the management structure to be employed during both transfer and cessation of the Services;

- 5.3.3. the management structure to be employed during the Termination Assistance Period;
- 5.3.4. a detailed description of both the transfer and cessation processes, including a timetable;
- 5.3.5. how the Services will transfer to the Replacement Supplier and/or the Customer, including details of the processes, documentation, data transfer, systems migration, security and the segregation of the Customer's technology components from any technology components operated by the Supplier or its Sub-Contractors (where applicable);
- 5.3.6. details of contracts (if any) which will be available for transfer to the Customer and/or the Replacement Supplier upon the Contract Expiry Date together with any reasonable costs required to effect such transfer (and the Supplier agrees that all assets and contracts used by the Supplier in connection with the provision of the Services will be available for such transfer);
- 5.3.7. proposals for the training of key members of the Replacement Supplier's personnel in connection with the continuation of the provision of the Services following the Contract Expiry Date charged at rates agreed between the Parties at that time;
- 5.3.8. proposals for providing the Customer or a Replacement Supplier copies of all documentation:
 - (a) used in the provision of the Services and necessarily required for the continued use thereof, in which the Intellectual Property Rights are owned by the Supplier; and
 - (b) relating to the use and operation of the Services;
- 5.3.9. proposals for the assignment or novation of the provision of all services, leases, maintenance agreements and support agreements utilised by the Supplier in connection with the performance of the supply of the Services;
- 5.3.10. proposals for the identification and return of all Customer Property in the possession of and/or control of the Supplier or any third party (including any SubContractor);
- 5.3.11. proposals for the disposal of any redundant Services and materials;
- 5.3.12. procedures to deal with requests made by the Customer and/or a Replacement Supplier for Staffing Information pursuant to Contract Schedule 10 (Staff Transfer);
- 5.3.13. how each of the issues set out in this Contract Schedule 10 will be addressed to facilitate the transition of the Services from the Supplier to the Replacement Supplier and/or the Customer with the aim of ensuring that there is no disruption to or degradation of the Services during the Termination Assistance Period; and
- 5.3.14. proposals for the supply of any other information or assistance reasonably required by the Customer or a Replacement Supplier in order to effect an orderly handover of the provision of the Services.

6. TERMINATION ASSISTANCE

- 6.1. The Customer shall be entitled to require the provision of Termination Assistance at any time during the Contract Period by giving written notice to the Supplier (a "Termination Assistance Notice") at least four (4) Months prior to the Contract Expiry Date or as soon as reasonably practicable (but in any event, not later than one (1) month) following the service by either Party of a Termination Notice. The Termination Assistance Notice shall specify:
 - 6.1.1. the date from which Termination Assistance is required;
 - 6.1.2. the nature of the Termination Assistance required; and

- 6.1.3. the period during which it is anticipated that Termination Assistance will be required, which shall continue no longer than twelve (12) Months after the date that the Supplier ceases to provide the Services.
- 6.2. The Customer shall have an option to extend the Termination Assistance Period beyond the period specified in the Termination Assistance Notice provided that such extension shall not extend for more than six (6) Months after the date the Supplier ceases to provide the Services or, if applicable, beyond the end of the Termination Assistance Period and provided that it shall notify the Supplier to such effect no later than twenty (20) Working Days prior to the date on which the provision of Termination Assistance is otherwise due to expire. The Customer shall have the right to terminate its requirement for Termination Assistance by serving not less than (20) Working Days' written notice upon the Supplier to such effect.

7. TERMINATION ASSISTANCE PERIOD

- 7.1. Throughout the Termination Assistance Period, or such shorter period as the Customer may require, the Supplier shall:
 - 7.1.1. continue to provide the Services (as applicable) and, if required by the Customer pursuant to paragraph 6.1 of this Contract Schedule 10, provide the Termination Assistance;
 - 7.1.2. in addition to providing the Services and the Termination Assistance, provide to the Customer any reasonable assistance requested by the Customer to allow the Services to continue without interruption following the termination or expiry of this Contract and to facilitate the orderly transfer of responsibility for and conduct of the Services to the Customer and/or its Replacement Supplier;
 - 7.1.3. use all reasonable endeavours to reallocate resources to provide such
 - 7.1.4. assistance as is referred to in paragraph 7.1.2 of this Contract Schedule 10 without additional costs to the Customer;
 - 7.1.5. provide the Services and the Termination Assistance at no detriment to the Service Level Performance Measures, save to the extent that the Parties agree otherwise in accordance with paragraph 7.3; and
 - 7.1.6. at the Customer's request and on reasonable notice, deliver up-to-date Registers to the Customer.
- 7.2. Without prejudice to the Supplier's obligations under paragraph 7.1.3 of this Contract Schedule 10, if it is not possible for the Supplier to reallocate resources to provide such assistance as is referred to in paragraph 7.1.2 of this Contract Schedule 10 without additional costs to the Customer, any additional costs incurred by the Supplier in providing such reasonable assistance which is not already in the scope of the Termination Assistance or the Exit Plan shall be subject to the Variation Procedure.
- 7.3. If the Supplier demonstrates to the Customer's reasonable satisfaction that transition of the Services and provision of the Termination Assist during the Termination Assistance Period will have a material, unavoidable adverse effect on the Supplier's ability to meet one or more particular Service Level Performance Measure(s), the Parties shall vary the relevant Service Level Performance Measure(s) and/or the applicable Service Credits to take account of such adverse effect.

8. TERMINATION OBLIGATIONS

- 8.1. The Supplier shall comply with all of its obligations contained in the Exit Plan.
- 8.2. Upon termination or expiry (as the case may be) or at the end of the Termination Assistance Period (or earlier if this does not adversely affect the Supplier's performance of the Services

and the Termination Assistance and its compliance with the other provisions of this Contract Schedule 10), the Supplier shall:

- 8.2.1. cease to use the Customer Data;
- 8.2.2. provide the Customer and/or the Replacement Supplier with a complete and uncorrupted version of the Customer Data in electronic form (or such other format as reasonably required by the Customer);
- 8.2.3. erase from any computers, storage devices and storage media that are to be retained by the Supplier after the end of the Termination Assistance Period all Customer Data and promptly certify to the Customer that it has completed such deletion;
- 8.2.4. return to the Customer such of the following as is in the Supplier's possession or control:
 - (a) all materials created by the Supplier under this Contract in which the IPRs are owned by the Customer;
 - (b) any equipment which belongs to the Customer;
 - (c) any items that have been on-charged to the Customer, such as consumables; and
 - (d) any sums prepaid by the Customer in respect of Services not Delivered by the Contract Expiry Date; and
 - (e) all Information and data relating to Learners (including Learner files) and course material used for the Skill Bootcamps
- 8.2.5. vacate any Customer Premises;
- 8.2.6. remove the Supplier Equipment together with any other materials used by the Supplier to supply the Services and shall leave the Sites in a clean, safe and tidy condition. The Supplier is solely responsible for making good any damage to the Sites or any objects contained thereon, other than fair wear and tear, which is caused by the Supplier and/or any Supplier Personnel;
- 8.2.7. provide access during normal working hours to the Customer and/or the Replacement Supplier for up to twelve (12) Months after expiry or termination to:
 - (a) such information relating to the Services as remains in the possession or control of the Supplier; and
 - (b) such members of the Supplier Personnel as have been involved in the design, development and provision of the Services and who are still employed by the Supplier, provided that the Customer and/or the Replacement Supplier shall pay the reasonable costs of the Supplier actually incurred in responding to requests for access under this paragraph.
- 8.3. Upon termination or expiry (as the case may be) or at the end of the Termination Assistance Period (or earlier if this does not adversely affect the Supplier's performance of the Services and the Termination Assistance and its compliance with the other provisions of this Contract Schedule 10), each Party shall return to the other Party (or if requested, destroy or delete) all Confidential Information of the other Party and shall certify that it does not retain the other Party's Confidential Information save to the extent (and for the limited period) that such information needs to be retained by the Party in question for the purposes of providing or receiving any Services or termination services or for statutory compliance purposes.
- 8.4. Except where this Contract provides otherwise, all licences, leases and authorisations granted by the Customer to the Supplier in relation to the Services shall be terminated with effect from the end of the Termination Assistance Period.

9. ASSETS AND SUB-CONTRACTS

- 9.1. Following notice of termination of this Contract and during the Termination Assistance Period, the Supplier shall not, without the Customer's prior written consent:
 - 9.1.1. terminate, enter into or vary any Sub-Contract;
 - 9.1.2. (subject to normal maintenance requirements) make material modifications to, or dispose of, any existing Supplier Assets or acquire any new Supplier Assets; or
 - 9.1.3. terminate, enter into or vary any licence for software in connection with the provision of the Services.
- 9.2. Within twenty (20) Working Days of receipt of the up-to-date Registers provided by the Supplier pursuant to paragraph 7.1.5 of this Contract Schedule 10, the Customer shall provide written notice to the Supplier setting out:
 - 9.2.1. which, if any, of the Transferable Assets the Customer requires to be transferred to the Customer and/or the Replacement Supplier ("**Transferring Assets**");
 - 9.2.2. which, if any, of:
 - (a) the Exclusive Assets that are not Transferable Assets; and
 - (b) the Non-Exclusive Assets, the Customer and/or the Replacement Supplier requires the continued use of; and
 - 9.2.3. which, if any, of Transferable Contracts the Customer requires to be assigned or novated to the Customer and/or the Replacement Supplier
 - 9.2.4. (the **"Transferring Contracts"**), in order for the Customer and/or its Replacement Supplier to provide the Services from the expiry of the Termination Assistance Period. Where requested by the Customer and/or its Replacement Supplier, the Supplier shall provide all reasonable assistance to the Customer and/or its Replacement Supplier to enable it to determine which Transferable Assets and Transferable Contracts the Customer and/or its Replacement Supplier requires to provide the Services or the Replacement Services.
- 9.3. With effect from the expiry of the Termination Assistance Period, the Supplier shall sell the Transferring Assets to the Customer and/or its nominated Replacement Supplier for a consideration equal to their Net Book Value, except where the cost of the Transferring Asset has been partially or fully paid for through the Contract Charges at the Contract Expiry Date, in which case the Customer shall pay the Supplier the Net Book Value of the Transferring Asset less the amount already paid through the Contract Charges.
- 9.4. Risk in the Transferring Assets shall pass to the Customer or the Replacement Supplier (as appropriate) at the end of the Termination Assistance Period and title to the Transferring Assets shall pass to the Customer or the Replacement Supplier (as appropriate) on payment for the same.
- 9.5. Where the Supplier is notified in accordance with paragraph 9.2.2 of this Contract Schedule 10 that the Customer and/or the Replacement Supplier requires continued use of any Exclusive Assets that are not Transferable Assets or any Non-Exclusive Assets, the Supplier shall as soon as reasonably practicable:
 - 9.5.1. procure a non-exclusive, perpetual, royalty-free licence (or licence on such other terms that have been agreed by the Customer) for the Customer and/or the Replacement Supplier to use such assets (with a right of sub-licence or assignment on the same terms); or failing which
 - 9.5.2. procure a suitable alternative to such assets and the Customer or the Replacement Supplier shall bear the reasonable proven costs of procuring the same.

- 9.6. The Supplier shall as soon as reasonably practicable assign or procure the novation to the Customer and/or the Replacement Supplier of the Transferring Contracts. The Supplier shall execute such documents and provide such other assistance as the Customer reasonably requires to effect this novation or assignment.
- 9.7. The Customer shall:
 - 9.7.1. accept assignments from the Supplier or join with the Supplier in procuring a novation of each Transferring Contract; and
 - 9.7.2. once a Transferring Contract is novated or assigned to the Customer and/or the Replacement Supplier, carry out, perform and discharge all the obligations and liabilities created by or arising under that Transferring Contract and exercise its rights arising under that Transferring Contract, or as applicable, procure that the Replacement Supplier does the same.
- 9.8. The Supplier shall hold any Transferring Contracts on trust for the Customer until such time as the transfer of the relevant Transferring Contract to the Customer and/or the Replacement Supplier has been effected.
- 9.9. The Supplier shall indemnify the Customer (and/or the Replacement Supplier, as applicable) against each loss, liability and cost arising out of any claims made by a counterparty to a Transferring Contract which is assigned or novated to the Customer (and/or Replacement Supplier) pursuant to paragraph 9.6 of this Contract Schedule 10 in relation to any matters arising prior to the date of assignment or novation of such Transferring Contract.

10. SUPPLIER PERSONNEL

- 10.1. The Customer and Supplier agree and acknowledge that in the event of the Supplier ceasing to provide the Services or part of them for any reason, Contract Schedule 10 (Staff Transfer) shall apply.
- 10.2. The Supplier shall not take any step (expressly or implicitly and directly or indirectly by itself or through any other person) to dissuade or discourage any employees engaged in the provision of the Services from transferring their employment to the Customer and/or the Replacement Supplier.
- 10.3. During the Termination Assistance Period, the Supplier shall give the Customer and/or the Replacement Supplier reasonable access to the Supplier's personnel to present the case for transferring their employment to the Customer and/or the Replacement Supplier.
- 10.4. The Supplier shall immediately notify the Customer or, at the direction of the Customer, the Replacement Supplier of any period of notice given by the Supplier or received from any person referred to in the Staffing Information, regardless of when such notice takes effect.
- 10.5. The Supplier shall not for a period of twelve (12) Months from the date of transfer re-employ or re-engage or entice any employees, Supplier's or Sub-Contractors whose employment or engagement is transferred to the Customer and/or the Replacement Supplier, unless approval has been obtained from the Customer which shall not be unreasonably withheld.

11. CHARGES

11.1. Except as otherwise expressly specified in this Contract, the Supplier shall not make any charges for the services provided by the Supplier pursuant to, and the Customer shall not be obliged to pay for costs incurred by the Supplier in relation to its compliance with, this Contract Schedule 10 including the preparation and implementation of the Exit Plan, the Termination Assistance and any activities mutually agreed between the Parties to carry on after the expiry of the Termination Assistance Period.

12. **APPORTIONMENTS**

- 12.1. All outgoings and expenses (including any remuneration due) and all rents, royalties and other periodical payments receivable in respect of the Transferring Assets and Transferring Contracts shall be apportioned between the Customer and the Supplier and/or the Replacement Supplier and the Supplier (as applicable) as follows:
 - 12.1.1. the amounts shall be annualised and divided by 365 to reach a daily rate;
 - 12.1.2. the Customer shall be responsible for (or shall procure that the Replacement Supplier shall be responsible for) or entitled to (as the case may be) that part of the value of the invoice pro rata to the number of complete days following the transfer, multiplied by the daily rate; and
 - 12.1.3. the Supplier shall be responsible for or entitled to (as the case may be) the rest of the invoice.
- 12.2. Each Party shall pay (and/or the Customer shall procure that the Replacement Supplier shall pay) any monies due under paragraph 12.1 of this Contract Schedule 10 as soon as reasonably practicable.

CONTRACT SCHEDULE 11: VARIATION FORM

No of Contract Order Form being varied:

.....

Variation Form No:

.....

BETWEEN:

[insert name of Customer] ("the Customer") and

[insert name of Supplier] ("the Supplier")

1 This Contract is varied as follows and shall take effect on the date signed by both Parties:

[Insert details of the Variation]

- 2 Words and expressions in this Variation shall have the meanings given to them in this Contract.
- 3 This Contract, including any previous Variations, shall remain effective and unaltered except as amended by this Variation.

Signed by an authorised signatory for and on behalf of the Customer

Signature	
Date	
Name (in Capitals)	
Address	
Signed by an authorised signator	y to sign for and on behalf of the Supplier
Signature	
Date	
Name (in Capitals)	
Address	

CONTRACT SCHEDULE 12: ALTERNATIVE AND/OR ADDITIONAL CLAUSES

1. INTRODUCTION

1.1 This Contract Schedule 12 specifies the range of Alternative Clauses and Additional Clauses that may be requested in the Contract Order Form and, if requested in the Contract Order Form, shall apply to this Contract.

2. CLAUSES SELECTED

- 2.1 The Customer may, in the Contract Order Form, request the following Alternative Clauses
 - 2.1.1 Financial Limits (see paragraph 4.5 of this Contract Schedule 12).
- 2.2 The Customer may, in the Contract Order Form, request the following Additional Clauses should apply:
 - 2.2.1 Security Measures (see paragraph 5.1 of this Contract Schedule 12);

3. IMPLEMENTATION

3.1 The appropriate changes have been made in this Contract to implement the Alternative and/or Additional Clauses specified in paragraph 2.1 of this Contract Schedule 12 and the Additional Clauses specified in paragraphs 2.2 and 2.2.1 of this Contract Schedule 12 shall be deemed to be incorporated into this Contract.

4. ALTERNATIVE CLAUSES

5.

4.1 FINANCIAL LIMITS

In Clause 25.3 remove the monetary amount and the percentage stated therein and replace respectively with:

[enter monetary amount in words] [£ X]

[enter percentage in words] [£ X]

ADDITIONAL CLAUSES: GENERAL

5.1 SECURITY MEASURES

5.1.1 The following definitions to be added to Contract Schedule 1 (Definitions) to the Contract Order Form and the Contract Terms:

"Document" includes specifications, plans, drawings, photographs and books;

"Secret Matter" means any matter connected with or arising out of the performance of this Contract which has been, or may hereafter be, by a notice in writing given by the Customer to the Supplier be designated 'top secret', 'secret', or 'confidential';

"**Servant**" where the Supplier is a body corporate shall include a director of that body and any person occupying in relation to that body the position of director by whatever name called.

5.1.2 The following new Clause [58] shall apply:

58. [SECURITY MEASURES]

- 58.1. The Supplier shall not, either before or after the completion or termination of this Contract, do or permit to be done anything which it knows or ought reasonably to know may result in information about a secret matter being:
 - 58.1.1. without the prior consent in writing of the Customer, disclosed to or acquired by a person who is an alien or who is a British subject by

virtue only of a certificate of naturalisation in which his name was included;

- 58.1.2. disclosed to or acquired by a person as respects whom the Customer has given to the Supplier a notice in writing which has not been cancelled stating that the Customer requires that secret matters shall not be disclosed to that person;
- 58.1.3. without the prior consent in writing of the Customer, disclosed to or acquired by any person who is not a servant of the Supplier; or
- 58.1.4. disclosed to or acquired by a person who is an employee of the Supplier except in a case where it is necessary for the proper performance of this Contract that such person shall have the information.
- 58.2. Without prejudice to the provisions of Clause 58.1, the Supplier shall, both before and after the completion or termination of this Contract, take all reasonable steps to ensure:
 - 58.2.1. no such person as is mentioned in Clauses 58.1, 58.1.1 or 58.1.2 hereof shall have access to any item or document under the control of the Supplier containing information about a secret matter except with the prior consent in writing of the Customer;
 - 58.2.2. that no visitor to any premises in which there is any item to be supplied under this Contract or where Services are being supplied shall see or discuss with the Supplier or any person employed by him any secret matter unless the visitor is authorised in writing by the Customer so to do;
 - 58.2.3. that no photograph of any item to be supplied under this Contract or any portions of the Services shall be taken except insofar as may be necessary for the proper performance of this Contract or with the prior consent in writing of the Customer, and that no such photograph shall, without such consent, be published or otherwise circulated;
 - 58.2.4 . that all information about any secret matter and every document model or other item which contains or may reveal any such information is at all times strictly safeguarded, and that, except insofar as may be necessary for the proper performance of this Contract or with the prior consent in writing of the Customer, no copies of or extracts from any such document, model or item shall be made or used and no designation of description which may reveal information about the nature or contents of any such document, model or item shall be placed thereon; and
 - 58.2.5. that if the Customer gives notice in writing to the Supplier at any time requiring the delivery to the Customer of any such document, model or item as is mentioned in Clause 58.2.3, that document, model or item (including all copies of or extracts therefrom) shall forthwith be delivered to the Customer who shall be deemed to be the owner thereof and accordingly entitled to retain the same.
- 58.3. The decision of the Customer on the question whether the Supplier has taken or is taking all reasonable steps as required by the foregoing provisions of Clause 58 shall be final and conclusive.

- 58.4. If and when directed by the Customer, the Supplier shall furnish full particulars of all people who are at any time concerned with any secret matter.
- 58.6. If, at any time either before or after the expiry or termination of this Contract, it comes to the notice of the Supplier that any person acting without lawful authority is seeking or has sought to obtain information concerning this Contract or anything done or to be done in pursuance thereof, the matter shall be forthwith reported by the Supplier to the Customer and the report shall, in each case, be accompanied by a statement of the facts, including, if possible, the name, address and occupation of that person, and the Supplier shall be responsible for making all such arrangements as it may consider appropriate to ensure that if any such occurrence comes to the knowledge of any person employed by it, that person shall forthwith report the matter to the Supplier with a statement of the facts as aforesaid.
- 58.7. The Supplier shall place every person employed by it, other than a SubContractor, who in its opinion has or will have such knowledge of any secret matter as to appreciate its significance, under a duty to the Supplier to observe the same obligations in relation to that matter as are imposed on the Supplier by Clauses 58.1 and 58.2 and shall, if directed by the Customer, place every person who is specified in the direction or is one of a class of people so specified, under the like duty in relation to any secret matter which may be specified in the direction, and shall at all times use its best endeavours to ensure that every person upon whom obligations are imposed by virtue of Clause 58 observes the said obligations, and the Supplier shall give such instructions and information to every such person as may be necessary for that purpose, and shall, immediately upon becoming aware of any act or omission which is or would be a breach of the said obligations, report the facts to the Supplier with all necessary particulars.
- 58.8. The Supplier shall, if directed by the Customer, include in the Sub-Contract provisions in such terms as the Customer may consider appropriate for placing the Sub-Contractor under obligations in relation to secrecy and security corresponding to those placed on the Supplier by Clause 58, but with such variations (if any) as the Customer may consider necessary. Further the Supplier shall:
 - 58.8.1. give such notices, directions, requirements and decisions to its SubContractors as may be necessary to bring the provisions relating to secrecy and security which are included in Sub-Contracts under Clause 58 into operation in such cases and to such extent as the Customer may direct;
 - 58.8.2. if there comes to its notice any breach by the Subcontractor of the obligations of secrecy and security included in their Sub-Contracts in pursuance of Clause 58, notify such breach forthwith to the Customer; and
 - 58.8.3. if and when so required by the Customer, exercise its power to determine the Sub-Contract under the provision in that SubContract which corresponds to Clause 58.11.
- 58.9. The Supplier shall give the Customer such information and particulars as the Customer may from time to time require for the purposes of satisfying the Customer that the obligations imposed by or under the foregoing provisions of Clause 58 have been and are being observed and as to what the Supplier has done or is doing or proposes to do to secure the observance of those obligations and to prevent any breach thereof, and the Supplier shall secure

that a representative of the Customer duly authorised in writing shall be entitled at reasonable times to enter and inspect any premises in which anything is being done or is to be done under this Contract or in which there is or will be any item to be supplied under this Contract, and also to inspect any document or item in any such premises or which is being made or used for the purposes of this Contract and that any such representative shall be given all such information as he may require on the occasion of, or arising out of, any such inspection.

- 58.10. Nothing in Clause 58 shall prevent any person from giving any information or doing anything on any occasion when it is, by virtue of any enactment, the duty of that person to give that information or do that thing.
- 58.11. If the Customer shall consider that any of the following events has occurred:
 - 58.11.1. that the Supplier has committed a breach of, or failed to comply with any of, the foregoing provisions of Clause 58; or
 - 58.11.2. that the Supplier has committed a breach of any obligations in relation to secrecy or security imposed upon it by any other contract with the Customer, or with any department or person acting on behalf of the Crown; or
 - 58.11.3. that by reason of an act or omission on the part of the Supplier, or of a person employed by the Supplier, which does not constitute such a breach or failure as is mentioned in 58.11.2, information about a secret matter has been or is likely to be acquired by a person who, in the opinion of the Customer, ought not to have such information;

and shall also decide that the interests of the State require the termination of this Contract, the Customer may by notice in writing terminate this Contract forthwith.

- 58.12. A decision of the Customer to terminate this Contract in accordance with the provisions of Clause 58.11 shall be final and conclusive and it shall not be necessary for any notice of such termination to specify or refer in any way to the event or considerations upon which the Customer's decision is based.
- 58.13. Supplier's notice
 - 58.13.1. The Supplier may within five (5) Working Days of the termination of this Contract in accordance with the provisions of Clause 58.11, give the Customer notice in writing requesting

the Customer to state whether the event upon which the Customer's decision to terminate was based is an event mentioned in Clauses 58.11, 58.11.1 or 58.11.2 and to give particulars of that event; and

58.13.2. the Customer shall within ten (10) Working Days of the receipt of such a request give notice in writing to the Supplier containing such a statement and particulars as are required by the request.

58.14. Matters pursuant to termination

- 58.14.1. The termination of this Contract pursuant to Clause 58.11 shall be without prejudice to any rights of either party which shall have accrued before the date of such termination;
- 58.14.2. The Supplier shall be entitled to be paid for any work or thing done under this Contract and accepted but not paid for by the

Customer at the date of such termination either at the price which would have been payable under this Contract if this Contract had not been terminated, or at a reasonable price;

- 58.14.3. The Customer may take over any work or thing done or made under this Contract (whether completed or not) and not accepted at the date of such termination which the Customer may by notice in writing to the Supplier given within thirty (30) Working Days from the time when the provisions of Clause 58 shall have effect, elect to take over, and the Supplier shall be entitled to be paid for any work or thing so taken over a price which, having regard to the stage which that work or thing has reached and its condition at the time it is taken over, is reasonable. The Supplier shall in accordance with directions given by the Customer, deliver any work or thing taken over under this Clause, and take all such other steps as may be reasonably necessary to enable the Customer to have the full benefit of any work or thing taken over under this Clause; and
- 58.14.4. Save as aforesaid, the Supplier shall not be entitled to any payment from the Customer after the termination of this Contract.
- 58.15. If, after notice of termination of this Contract pursuant to the provisions of 58.11:
 - 58.15.1. the Customer shall not within ten (10) Working Days of the receipt of a request from the Supplier, furnish such a statement and particulars as are detailed in Clause 58.13.1; or
 - 58.15.2. the Customer shall state in the statement and particulars detailed in Clause 58.13.2. that the event upon which the Customer's decision to terminate this Contract was based is an event mentioned in Clause 58.11.3,

the respective rights and obligations of the Supplier and the Customer shall be terminated in accordance with the following provisions:

58.15.3. the Customer shall take over from the Supplier at a fair and reasonable price all unused and undamaged materials, boughtout parts and components and articles in course of manufacture

in the possession of the Supplier upon the termination of this Contract under the provisions of Clause 58.11 and properly provided by or supplied to the Supplier for the performance of this Contract, except such materials, bought-out parts and components and articles in course of manufacture as the Supplier shall, with the concurrence of the Customer, elect to retain;

58.15.4. the Supplier shall prepare and deliver to the Customer within an agreed period or in default of agreement within such period as the Customer may specify, a list of all such unused and undamaged materials, bought-out parts and components and articles in course of manufacture liable to be taken over by or previously belonging to the Customer and shall deliver such materials and items in accordance with the directions of the Customer who shall pay to the Supplier fair and reasonable

handling and delivery charges incurred in complying with such directions;

- 58.15.5. the Customer shall indemnify the Supplier against any commitments, liabilities or expenditure which are reasonably and properly chargeable by the Supplier in connection with this Contract to the extent to which the said commitments, liabilities or expenditure would otherwise represent an unavoidable loss by the Supplier by reason of the termination of this Contract;
- 58.15.6. if hardship to the Supplier should arise from the operation of Clause 58.15 it shall be open to the Supplier to refer the circumstances to the Customer who, on being satisfied that such hardship exists shall make such allowance, if any, as in its opinion is reasonable and the decision of the Customer on any matter arising out of this Clause 58.15 shall be final and conclusive; and

subject to the operation of Clauses 58.15.3, 58.15.4, 58.15.5 and 58.15.6 termination of this Contract shall be without prejudice to any rights of either party that may have accrued before the date of such termination.

CONTRACT SCHEDULE 13: CONTRACT TENDER

NOTE: the Supplier's quality tender response for the Call-Off Contract will be attached to the back of the contract

CONTRACT SCHEDULE 14: SUB-CONTRACTING

General

- 1.1. The Supplier must select its Sub-Contractors fairly and without discrimination and must comply with any procurement rules that apply when doing so, including the Public Contracts Regulations 2015 where the Supplier is a contracting authority under those regulations.
- 1.2. The Supplier must ensure that proposed Sub-Contractors are of adequate financial standing and have sufficient capacity and capability to deliver the Services that are to be Sub-Contracted.
- 1.3. The Supplier must have a direct contractual relationship by way of a Sub- Contract with all of its SubContractors. The Supplier must ensure that its Sub-Contractors do not sub-contract any of the Services further to other Suppliers unless the Customer has given its expression written permission for further Sub-Contracting
- 1.4. The Supplier must not enter into any agreement for Brokerage in relation to the Services under this Contract.

Requirements for Sub-Contracts

- 1.5. The Supplier must enter into a written Sub-Contract with any supplier that the Supplier is proposing to use as a Sub-Contractor. The Supplier and the Sub-Contractor must enter into the Sub-Contract before the Sub-Contractor commences the delivery of the proposed sub-contracted Services.
- 1.6. The Sub-Contract must include:
 - 1.6.1. terms and conditions substantially the same as those set out in this Contract;
 - 1.6.2. payment provisions such that the Supplier must pay the Sub- Contractor within 30 days of receiving a valid invoice;
 - 1.6.3. an obligation on the Sub-Contractor to obtain express written permission from the Supplier before enrolling any Learners;
 - 1.6.4. an obligation on the Sub-Contractor to participate in any rectification plan as appropriate in the event that the Sub-Contractor has committed an act which would constitute a material Default of this Contract (whether or not it is also a breach of the Sub-Contract); and
 - 1.6.5. sanctions on the Sub-Contractor for material Default of the Sub- Contract substantially the same as those set out in this Contract.
- 1.7. Sub-Contracts must be available at all times for the Customer to inspect on request.

Supplier Obligations

- 1.8. The Supplier must make payment to any Sub-Contractor within 30 days of receiving a valid invoice in accordance with the required provisions of the Sub- Contract set out at paragraph 1.8.2 of this Schedule.
- 1.9. The Supplier must manage and monitor its Sub-Contractors in accordance with the relevant SubContract to ensure that the Sub-Contractors deliver the sub-contracted Services to the standard set out in Clause 3 (Service Delivery) of this Contract.
- 1.10. The Supplier must inform the Customer whenever a Sub-Contractor goes into administration or liquidation.
- 1.11. The Supplier must have a contingency plan in place to ensure that there is continuity of Services for existing Learners in the event of any circumstances that may arise that render a Sub-Contractor unable to deliver the sub- contracted Services including but not limited to the expiry or termination of the Sub-Contract.
- 1.12. The Supplier must ensure that a Sub-Contractor that has committed an act which constitutes a material Default of this Contract (whether or not it is also a breach of its Sub-Contract) participates in

any rectification plan as appropriate.

- 1.13. The Supplier must carry out an investigation at its own cost if there is any evidence of a Sub-Contractor having irregular financial or delivery activity and notify the Customer of this and of the outcome of any such investigation.
- 1.14. The Supplier must not enter new sub-contracting arrangements or increase the value of existing arrangements, without the written consent of the Customer, if any of the following circumstances apply:
 - 1.14.1. Ofsted has rated the Contractor's management and leadership as inadequate;
 - 1.14.2. the outcome of the Contractor's annual financial health assessment is inadequate, unless the Customer has provided its written consent to the proposed sub-contracting arrangement in advance of a Sub- Contract being entered into;
 - 1.14.3. a Sub-Supplier has been inspected and judged to be inadequate by Ofsted; or
 - 1.14.4. a Sub-Contractor is subject to ongoing intervention or investigation by the Customer.

Financial Thresholds

- 1.15. Where the annual value of this Contract in any Academic Year exceeds £5 million and the Supplier wants to sub-contract one or more of the Services the following provisions apply:
 - 1.15.1. The Supplier will advertise on Contracts Finder all sub-contract opportunities arising from or in connection with the provision of the Services above a minimum threshold of £25,000 that arise during the Contract Period.
 - 1.15.2. Once a Sub-Contract has been awarded, the Supplier will update the notice on Contracts Finder with the details of the successful Sub-Contractor.
 - 1.15.3. In addition to any other management information requirements set out in this Contract, the Supplier agrees and acknowledges that it will, at no charge, provider timely, full, accurate and complete SME Management Information ("MI) Reports to the Customer which incorporate the following:
 - (a) The total revenue received directly from the Contract
 - (b) The total value of Services under the Contract that have been sub- contracted (including revenues for non-SMEs/non-VCSEs); and
 - (c) The total value of sub-contracted revenues to SMEs and VCSEs.
 - 1.15.4. The SME Management Information Reports will be provided in the correct format as required by the Customer and any guidance issued by the Customer from time to time.
- 1.16. Where the Customer requests it, the Supplier must obtain an annual report from an external auditor if the aggregate total of Services in respect of which the Supplier has entered into Sub-Contracts exceeds or is anticipated to exceed £50,000 in any single Contract Year. The aggregate of the sums paid by the Supplier to its Sub-Contractors relates solely to Services that the Supplier is required to delivery under this Call Off Contract.

Departmental Rights

1.17. In the event that the Supplier fails to comply with any of the requirements set out in this Schedule the Customer reserves the right to take such remedial action under this Contract as it considers appropriate in the circumstances, which may include but is not limited to requiring the Supplier to terminate an existing Sub-Contract and/ or prohibiting the Supplier from entering into any new SubContract to deliver the Services under this Contract.

Key Sub-Contractor

2.1. The following are agreed between the Parties to be Key Sub-Contracts at the Contract Commencement Date:

2.1.1. [insert full name details]

CONTRACT SCHEDULE 15: FINANCIAL DISTRESS

1. DEFINITIONS

1.1 In this Schedule, the following definitions shall apply:

"Applicable Financial Indicators"	means the financial indicators from Paragraph 5.1 of this Schedule which are to apply to the Monitored Suppliers as set out in Paragraph 5.2 of this Schedule;
"Board"	means the Supplier's board of directors;
"Board Confirmation"	means written confirmation from the Board in accordance with Paragraph 8 of this Schedule;
"Credit Rating Level"	a credit rating level as specified in Annex 1 of this Schedule;
"Credit Rating Threshold"	the minimum Credit Rating Level for each entity in the FDE Group as set out in Annex 2 of this Schedule;
"FDE Group"	means the [Supplier, Key Sub-Contractors, [the Guarantor] and the [Monitored Suppliers]];
"Financial Indicators"	in respect of the Supplier, Key Sub-Contractors and the Guarantor, means each of the financial indicators set out at Paragraph 5.1 of this Schedule; and in respect of each Monitored Supplier, means those Applicable Financial Indicators;
"Financial Target Thresholds"	means the target thresholds for each of the Financial Indicators set out at Paragraph 5.1 of this Schedule;
"Monitored Suppliers" means	those entities specified at Paragraph 5.2 of this Schedule; "Rating

Agencies" the rating agencies listed in Annex 1 of this Schedule.

2. WARRANTIES AND DUTY TO NOTIFY

- 2.1 The Supplier warrants and represents to the Customer for the benefit of the Customer that as at the Contract Commencement Date:
 - 2.1.1 the long term credit ratings issued for each entity in the FDE Group by each of the Rating Agencies are as set out in Annex 2 of this Schedule; and
 - 2.1.2 the financial position or, as appropriate, the financial performance of each of the Supplier, Guarantor and Key Sub-Contractors satisfies the Financial Target Thresholds.
- 2.2 The Supplier shall promptly notify (or shall procure that its auditors promptly notify) the Customer in writing if there is any downgrade in the credit rating issued by any Rating Agency for any entity in the FDE Group (and in any event within 5 Working Days of the occurrence of the downgrade).
- 2.3 The Supplier shall:
 - 2.3.1 regularly monitor the credit ratings of each entity in the FDE Group with the Rating Agencies;
 - 2.3.2 monitor and report on the Financial Indicators for each entity in the FDE Group against the

Financial Target Thresholds at least at the frequency set out for each at Paragraph 5.1

(where specified) and in any event, on a regular basis and no less than once a year within one hundred and twenty (120) days after the Accounting Reference Date; and

- 2.3.3 promptly notify (or shall procure that its auditors promptly notify) the Customer in writing following the occurrence of a Financial Distress Event or any fact, circumstance or matter which could cause a Financial Distress Event (and in any event, ensure that such notification is made within 10 Working Days of the date on which the Supplier first becomes aware of the Financial Distress Event or the fact, circumstance or matter which could cause a Financial Distress Event).
- 2.4 For the purposes of determining whether a Financial Distress Event has occurred pursuant to the provisions of Paragraphs 3.1.10, and for the purposes of determining relief under Paragraph 7.1, the credit rating of an FDE Group entity shall be deemed to have dropped below the applicable Credit Rating Threshold if:
 - 2.4.1 any of the Rating Agencies have given a Credit Rating Level for that entity which is below the applicable Credit Rating Threshold; or
 - 2.4.2 a Rating Agency that is specified as holding a Credit Rating for an entity as set out at Annex 2 of this Schedule ceases to hold a Credit Rating for that entity.
- 2.5 Each report submitted by the Supplier pursuant to Paragraph 2.3.2 shall:
 - 2.5.1 be a single report with separate sections for each of the FDE Group entities;
 - 2.5.2 contain a sufficient level of information to enable the Customer to verify the calculations that have been made in respect of the Financial Indicators;
 - 2.5.3 include key financial and other supporting information (including any accounts data that has been relied on) as separate annexes;
 - 2.5.4 be based on the audited accounts for the date or period on which the Financial Indicator is based or, where the Financial Indicator is not linked to an accounting period or an accounting reference date, on unaudited management accounts prepared in accordance with their normal timetable; and
 - 2.5.5 include a history of the Financial Indicators reported by the Supplier in graph form to enable the Customer to easily analyse and assess the trends in financial performance.

3. FINANCIAL DISTRESS EVENTS

- 3.1 The following shall be Financial Distress Events:
 - 3.1.1 the credit rating of an FDE Group entity dropping below the applicable Credit Rating Threshold;
 - 3.1.2 an FDE Group entity issuing a profits warning to a stock exchange or making any other public announcement, in each case about a material deterioration in its financial position or prospects;
 - 3.1.3 there being a public investigation into improper financial accounting and reporting, suspected fraud or any other impropriety of an FDE Group entity;
 - 3.1.4 an FDE Group entity committing a material breach of covenant to its lenders;
 - 3.1.5 a Key Sub-Contractor notifying the Customer that the Supplier has not satisfied any material sums properly due under a specified invoice and not subject to a genuine dispute;

3.1.6 any FDE Group entity extends the filing period for filing its accounts with the Registrar of Companies so that the filing period ends more than 9 months after its accounting reference

date without an explanation to the Customer which the Customer (acting reasonably) considers to be adequate;

- 3.1.7 any FDE Group entity is late to file its annual accounts without a public notification or an explanation to the Customer which the Customer, acting reasonably, considers to be adequate;
- 3.1.8 the directors and/or external auditors of any FDE Group entity conclude that a material uncertainty exists in relation to that FDE Group entity's going concern in the annual report including in a reasonable but plausible downside scenario. This includes, but is not limited to, commentary about liquidity and trading prospects in the reports from directors or external auditors;
- 3.1.9 any of the following:
 - any FDE Group entity makes a public announcement which contains adverse commentary with regards to that FDE Group entity's liquidity and trading and trading prospects, such as but not limited to, a profit warning or ability to trade as a going concern;
 - (ii) commencement of any litigation against an FDE Group entity with respect to financial indebtedness greater than £5m or obligations under a service contract with a total contract value greater than £5m;
 - (iii) non-payment by an FDE Group entity of any financial indebtedness;
 - (iv) any financial indebtedness of an FDE Group entity becoming due as a result of an event of default;
 - (v) the cancellation or suspension of any financial indebtedness in respect of an FDE Group entity; or
 - (vi) the external auditor of an FDE Group entity expressing a qualified opinion on, or including an emphasis of matter in, its opinion on the statutory accounts of that FDE entity;

in each case which the Customer reasonably believes (or would be likely reasonably to believe) could directly impact on the continued performance and delivery of the Services in accordance with this Contract; and

3.1.10 any [one] of the Financial Indicators set out at Paragraph 5 for any of the FDE Group entities failing to meet the required Financial Target Threshold.

4. CONSEQUENCES OF FINANCIAL DISTRESS EVENTS

- 4.1 Immediately upon notification by the Supplier of a Financial Distress Event (or if the Customer becomes aware of a Financial Distress Event without notification and brings the event to the attention of the Supplier), the Supplier shall have the obligations and the Customer shall have the rights and remedies as set out in Paragraphs 4.3 to 4.5.
- 4.2 In the event of a late or non-payment of a Key Sub-Contractor pursuant to Paragraph 3.1.5, the Customer shall not exercise any of its rights or remedies under Paragraph 4.3 without first giving the Supplier 10 Working Days to:

- 4.2.1 rectify such late or non-payment; or
- 4.2.2 demonstrate to the Customer's reasonable satisfaction that there is a valid reason for late or non-payment.
- 4.3 The Supplier shall (and shall procure that any Monitored Supplier, the Guarantor and/or any relevant Key Sub-Contractor shall):
 - 4.3.1 at the request of the Customer, meet the Customer as soon as reasonably practicable (and in any event within 3 Working Days of the initial notification (or awareness) of the Financial Distress Event or such other period as the Customer may permit and notify to the Supplier in writing) to review the effect of the Financial Distress Event on the continued performance and delivery of the Services in accordance with this Contract; and
 - 4.3.2 where the Customer reasonably believes (taking into account the discussions and any representations made under Paragraph 4.3.1 that the Financial Distress Event could impact on the continued performance and delivery of the Services in accordance with this Contract:
 - submit to the Customer for its approval, a draft Financial Distress Remediation Plan as soon as reasonably practicable (and in any event, within 10 Working Days of the initial notification (or awareness) of the Financial Distress Event or such other period as the Customer may permit and notify to the Supplier in writing); and
 - to the extent that it is legally permitted to do so and subject to Paragraph 4.7, provide such information relating to the Supplier, any Monitored Supplier, Key Sub-Contractors and/or the Guarantor as the Customer may reasonably require in order to understand the risk to the Services, which may include forecasts in relation to cash flow, orders and profits and details of financial measures being considered to mitigate the impact of the Financial Distress Event.
- 4.4 The Customer shall not withhold its approval of a draft Financial Distress Remediation Plan unreasonably. If the Customer does not approve the draft Financial Distress Remediation Plan, it shall inform the Supplier of its reasons and the Supplier shall take those reasons into account in the preparation of a further draft Financial Distress Remediation Plan, which shall be resubmitted to the Customer within 5 Working Days of the rejection of the first draft. This process shall be repeated until the Financial Distress Remediation Plan is either:
 - 4.4.1 approved by the Customer;
 - 4.4.2 referred, by notice sent by either Party to the other Party explaining why it thinks the Financial Distress Remediation Plan has not been approved, to commercial negotiation led by senior representatives who have authority to agree the Financial Distress Remediation Plan to be held within 28 days of the date of the notice; or
 - 4.4.3 finally rejected by the Customer.
- 4.5 Following approval of the Financial Distress Remediation Plan by the Customer, the Supplier shall:
 - 4.5.1 on a regular basis (which shall not be less than fortnightly):
 - (i) review and make any updates to the Financial Distress Remediation Plan as the Supplier may deem reasonably necessary and/or as may be reasonably requested by the Customer, so that the plan remains adequate, up to date and ensures the continued performance and delivery of the Services in accordance with this Contract; and
 - (ii) provide a written report to the Customer setting out its progress against the Financial Distress Remediation Plan, the reasons for any changes made to the

Financial Distress Remediation Plan by the Supplier and/or the reasons why the Supplier may have decided not to make any changes;

- 4.5.2 where updates are made to the Financial Distress Remediation Plan in accordance with Paragraph 4.5.1 submit an updated Financial Distress Remediation Plan to the Customer for its approval, and the provisions of Paragraphs 4.4 and 4.5.1 shall apply to the review and approval process for the updated Financial Distress Remediation Plan; and
- 4.5.3 comply with the Financial Distress Remediation Plan (including any updated Financial Distress Remediation Plan) and ensure that it achieves the financial and performance requirements set out in the Financial Distress Remediation Plan.
- 4.6 Where the Supplier reasonably believes that the relevant Financial Distress Event under Paragraph 4.1 (or the circumstance or matter which has caused or otherwise led to it) no longer exists, it shall notify the Customer and the Parties may agree that the Supplier shall be relieved of its obligations under Paragraph 4.5.
- 4.7 The Supplier shall use reasonable endeavours to put in place the necessary measures to ensure that the information specified at Paragraph 4.3.2(ii) is available when required and on request from the Customer and within reasonable timescales. Such measures may include:
 - 4.7.1 obtaining in advance written authority from Key Sub-Contractors, the Guarantor and/or Monitored Suppliers authorising the disclosure of the information to the Customer and/or entering into confidentiality agreements which permit disclosure;
 - 4.7.2 agreeing in advance with the Customer, Key Sub-Contractors, the Guarantor and/or Monitored Suppliers a form of confidentiality agreement to be entered by the relevant parties to enable the disclosure of the information to the Customer;
 - 4.7.3 putting in place any other reasonable arrangements to enable the information to be lawfully disclosed to the Customer (which may include making price sensitive information available to Customer nominated personnel through confidential arrangements, subject to their consent); and
 - 4.7.4 disclosing the information to the fullest extent that it is lawfully entitled to do so, including through the use of redaction, anonymisation and any other techniques to permit disclosure of the information without breaching a duty of confidentiality.

5. FINANCIAL INDICATORS

5.1 Subject to the calculation methodology set out at Annex 3 of this Schedule, the Financial Indicators and the corresponding calculations and thresholds used to determine whether a Financial Distress Event has occurred in respect of those Financial Indicators, shall be as follows:

Financial Indicator Calculation	¹ Financial Target Threshold:	Monitoring and Reporting Frequency
---------------------------------	---	---------------------------------------

			۱ ۱
1 Operating Margin OR The higher of (a) the Operating Margin for the most recent 12 month period and (b) the average Operating Margin for the last two 12 month periods	[Operating Margin = Operating Profit / Revenue]	5% Red 10% Green	Tested and reported yearly in arrears within 120 days of each accounting reference date based upon figures for the 12 months ending on the relevant accounting reference date
2 Free Cash Flow to Net Debt Ratio OR Net Debt to EBITDA Ratio	[Free Cash Flow to Net Debt Ratio = Free Cash Flow / Net Debt] OR [Net Debt to EBITDA ratio = Net Debt / EBITDA]	Free Cash Flow to Net Debt Ratio 5% Red, 15% Green Net Debt to EBITDA Ratio 3.5% Red, 2.5% Green	Tested and reported yearly in arrears within 120 days of each accounting reference date] based upon [Free Cash Flow / EBITDA] for the 12 months ending on, and Net Debt at, the relevant accounting reference date
3 Net Debt + Net Pension Deficit to EBITDA ratio	[Net Debt + Net Pension Deficit to EBITDA Ratio = (Net Debt + Net Pension Deficit) / EBITDA]	5% Red 4% Green	Tested and reported yearly in arrears within 120 days of each accounting reference date based upon EBITDA for the 12 months ending on, and the Net Debt and Net Pension Deficit at, the relevant accounting reference date
4 [Net Interest Paid Cover]	[Net Interest Paid Cover = Earnings Before Interest and Tax / Net Interest Paid]	3% Red 4.5% Green	Tested and reported yearly in arrears within 120 days of each accounting reference date based upon figures for the 12 months ending on the relevant [accounting reference date / half year end]
5 [Acid Ratio]	[Acid Ratio = (Current Assets – Inventories) / Current Liabilities]	0.8 Red 1 Green	Tested and reported yearly in arrears within 120 days of each accounting reference date based upon figures at the relevant accounting reference date
6 [Net Asset value]	[Net Asset Value = Net Assets]	- 0	Tested and reported yearly in arrears within 120 days of each accounting reference date based upon figures at the relevant accounting reference date

7 [Group Exposure Ratio]	[Group Exposure / Gross Assets]	50% Red 25% Green	Tested and reported yearly in arrears within 120 days of each accounting reference date based upon figures at the relevant accounting reference date
-----------------------------	------------------------------------	----------------------	--

Key: ¹ – See Annex 3 of this Schedule which sets out the calculation methodology to be used in the calculation of each Financial Indicator.

5.2 Monitored Suppliers

Monitored Supplier	Applicable Financial Indicators (these are the Financial Indicators from the table in Paragraph 0 which are to apply to the Monitored Suppliers)
Group Member, Sub-contractor, Relevant Parent Company	[1] [2] [3][4][5][6][7]

6. TERMINATION RIGHTS

- 6.1 The Customer shall be entitled to terminate this Contract under Clause 30.3.1 (Termination in relation to Financial Standing) if:
 - 6.1.2 the Supplier fails to notify the Customer of a Financial Distress Event in accordance with Paragraph 2.3.3;
 - 6.1.3 the Supplier fails to comply with any part of Paragraph 4.3;
 - 6.1.4 the Customer finally rejects a Financial Distress Remediation Plan (or any updated Financial Distress Remediation Plan) in accordance with Paragraphs 4.4 to 4.5.1; and/or
 - 6.1.5 the Supplier fails to comply with the terms of the Financial Distress Remediation Plan (or any updated Financial Distress Remediation Plan) in accordance with Paragraph 4.5.3.

7. PRIMACY OF CREDIT RATINGS

- 7.1 Without prejudice to the Supplier's obligations and the Customer's rights and remedies under Paragraph 2, if, following the occurrence of a Financial Distress Event pursuant to any of Paragraphs 3.1.2 to 3.1.10 the Rating Agencies review and report subsequently that the credit ratings for the FDE Group entities do not drop below the relevant Credit Rating Thresholds specified for those entities in Annex 2 of this Schedule, then:
 - 7.1.1 the Supplier shall be relieved automatically of its obligations under Paragraphs 4.3 to 4.5; and
 - 7.1.2 the Customer shall not be entitled to require the Supplier to provide financial information in accordance with Paragraph 4.3.2(ii)4(ii).

8. BOARD CONFIRMATION

8.1 Subject to Paragraph 8.4 of this Schedule, the Supplier shall within one hundred and twenty (120) days after each Accounting Reference Date or within 15 months of the previous Board Confirmation (whichever is the earlier) provide a Board Confirmation to the Customer in the form set out at Annex

4 of this Schedule, confirming that to the best of the Board's knowledge and belief, it is not aware of and has no knowledge:

- 8.1.1 that a Financial Distress Event has occurred since the later of the Contract Commencement Date or the previous Board Confirmation or is subsisting; or
- 8.1.2 of any matters which have occurred or are subsisting that could reasonably be expected to cause a Financial Distress Event.
- 8.2 The Supplier shall ensure that in its preparation of the Board Confirmation it exercises due care and diligence and has made reasonable enquiry of all relevant Supplier Personnel and other persons as is reasonably necessary to understand and confirm the position.
- 8.3 In respect of the first Board Confirmation to be provided under this Contract, the Supplier shall provide the Board Confirmation within 15 months of the Contract Commencement Date if earlier than the timescale for submission set out in Paragraph 8.1 of this Schedule.
- 8.4 Where the Supplier is unable to provide a Board Confirmation in accordance with Paragraphs 8.1 to 8.3 of this Schedule due to the occurrence of a Financial Distress Event or knowledge of subsisting matters which could reasonably be expected to cause a Financial Distress Event, it will be sufficient for the Supplier to submit in place of the Board Confirmation, a statement from the Board of Directors to the Customer (and where the Supplier is a Strategic Supplier, the Supplier shall send a copy of the statement to the Cabinet Office Markets and Suppliers Team) setting out full details of any Financial Distress Events that have occurred and/or the matters which could reasonably be expected to cause a Financial Distress Event.

ANNEX 1: RATING AGENCIES AND THEIR STANDARD RATING SYSTEM

NOT USED

ANNEX 2: CREDIT RATINGS AND CREDIT RATING THRESHOLDS

NOT USED

ANNEX 3: CALCULATION METHODOLOGY FOR FINANCIAL INDICATORS

The Supplier shall ensure that it uses the following general and specific methodologies for calculating the Financial Indicators against the Financial Target Thresholds:

General methodology

Financial Indicator	Specific Methodology
1 [Operating Margin]	The elements used to calculate the Operating Margin should be shown on the face of the Income Statement in a standard set of financial statements. Figures for Operating Profit and Revenue should exclude the entity's share of the results of any joint ventures or Associates. Where an entity has an operating loss (i.e. where the operating profit is negative), Operating Profit should be taken to be zero.

2 [Free Cash Flow to Net Debt	[" Free Cash Flow " = Net Cash Flow from Operating Activities – Capital Expenditure
Ratio]	"Capital Expenditure" = Purchase of property, plant & equipment + purchase of intangible assets
	" Net Debt " = Bank overdrafts + Loans and borrowings + Finance Leases + Deferred consideration payable – Cash and cash equivalents
	The majority of the elements used to calculate the Free Cash Flow to Net Debt Ratio should be shown on the face of the Statement of Cash Flows and the Balance Sheet in a standard set of financial statements.
	<u>Net Cash Flow from Operating Activities</u> : This should be stated after deduction of interest and tax paid.
	<u>Capital expenditure</u> : The elements of capital expenditure may be described slightly differently but will be found under 'Cash flows from investing activities' in the Statement of Cash Flows; they should be limited to the purchase of fixed assets (including intangible assets) for the business and exclude acquisitions. The figure should be shown gross without any deduction for any proceeds of sale of fixed assets.
	<u>Net Debt</u> : The elements of Net Debt may also be described slightly differently and should be found either on the face of the Balance Sheet or in the

- 1. *Terminology*: The terms referred to in this Annex are those used by UK companies in their financial statements. Where the entity is not a UK company, the corresponding items should be used even if the terminology is slightly different (for example a charity would refer to a surplus or deficit rather than a profit or loss).
- 1. *Groups*: Where the entity is the holding company of a group and prepares consolidated financial statements, the consolidated figures should be used.
- 2. *Foreign currency conversion*: Figures denominated in foreign currencies should be converted at the exchange rate in force at the relevant date for which the Financial Indicator is being calculated.
- 3. *Treatment of non-underlying items:* Financial Indicators should be based on the figures in the financial statements before adjusting for non-underlying items.

Specific Methodology

	 relevant note to the financial statements. All interest bearing liabilities (other than retirement benefit obligations) should be treated as borrowings as should, where disclosed, any liabilities (less any assets) in respect of any hedges designated as linked to borrowings (but not non-designated hedges). Borrowings should also include balances owed to other group members. Deferred consideration payable should be included in Net Debt despite typically being non-interest bearing. Cash and cash equivalents should include short-term financial investments shown in current assets. Where Net debt is negative (i.e. an entity has net cash), the relevant Financial Target Threshold should be treated as having been met.] OR ["Net Debt" = Bank overdrafts + Loans and borrowings + Finance leases + Deferred consideration payable – Cash and cash equivalents "EBITDA" = Operating profit + Depreciation charge + Amortisation charge
	The majority of the elements used to calculate the Net Debt to EBITDA Ratio should be shown on the face of the Balance sheet, Income statement and Statement of Cash Flows in a standard set of financial statements but will otherwise be found in the notes to the financial statements.
OR	<u>Net Debt</u> : The elements of Net Debt may be described slightly differently and should be found either on the face of the Balance Sheet or in the relevant note to the financial statements. All interest bearing liabilities (other than retirement benefit obligations) should be included as borrowings as should, where disclosed, any liabilities (less any assets) in respect of any hedges designated as linked to borrowings (but not non-designated hedges). Borrowings should also include balances owed to other group members.
[Net Debt to EBITDA Ratio]	Deferred consideration payable should be included in Net Debt despite typically being non-interest bearing.
	Cash and cash equivalents should include short-term financial investments shown in current assets.
	Where Net debt is negative (i.e. an entity has net cash), the relevant Financial Target Threshold should be treated as having been met.
	<u>EBITDA</u> : Operating profit should be shown on the face of the Income Statement and, for the purposes of calculating this Financial Indicator, should include the entity's share of the results of any joint ventures or Associates. The depreciation and amortisation charges for the period may be found on the face of the Statement of Cash Flows or in a Note to the Accounts. Where EBITDA is negative, the relevant Financial Target Threshold should be treated as not having been met (unless Net Debt is also negative, in which case the relevant Financial Target Threshold should be treated as having been met).]

3 ["Net Debt + Net Pension Deficit to EBITDA ratio] ["Net Pension Deficit" = Bank overdrafts + Loans and borrowings + Finance leases + Deferred consideration payable - Cash and cash equivalents "Net Pension Deficit" = Retirement Benefit Obligations - Retirement Benefit Assets
--

"EBITDA" = Operating profit + Depreciation charge + Amortisation charge
The majority of the elements used to calculate the Net Debt + Net Pension Deficit to EBITDA Ratio should be shown on the face of the Balance sheet, Income statement and Statement of Cash Flows in a standard set of financial statements but will otherwise be found in the notes to the financial statements.
<u>Net Debt</u> : The elements of Net Debt may be described slightly differently and should be found either on the face of the Balance Sheet or in the relevant note to the financial statements. All interest bearing liabilities (other than retirement benefit obligations) should be included as borrowings as should, where disclosed, any liabilities (less any assets) in respect of any hedges designated as linked to borrowings (but <i>not</i> non-designated hedges). Borrowings should also include balances owed to other group members.
Deferred consideration payable should be included in Net Debt despite typically being non-interest bearing.
Cash and cash equivalents should include short-term financial investments shown in current assets.
<u>Net Pension Deficit</u> : Retirement Benefit Obligations and Retirement Benefit Assets may be shown on the face of the Balance Sheet or in the notes to the financial statements. They may also be described as pension benefits / obligations, post-employment obligations or other similar terms.
Where 'Net Debt + Net Pension Deficit' is negative, the relevant Financial Target Threshold should be treated as having been met.
<u><i>EBITDA</i></u> : Operating profit should be shown on the face of the Income Statement and, for the purposes of calculating this Financial Indicator, should include the entity's share of the results of any joint ventures or Associates.
The depreciation and amortisation charges for the period may be found on the face of the Statement of Cash Flows or in a Note to the Accounts.
Where EBITDA is negative, the relevant Financial Target Threshold should be treated as not having been met (unless 'Net Debt + Net Pension Deficit' is also negative, in which case the relevant Financial Target Threshold should be regarded as having been met).]

4 [Net Interest Paid Cover]	["Earnings Before Interest and Tax" = Operating profit "Net Interest Paid" = Interest paid – Interest received Operating profit should be shown on the face of the Income Statement in a standard set of financial statements and, for the purposes of calculating this Financial Indicator, should include the entity's share of the results of any joint ventures or Associates. Interest received and interest paid should be shown on the face of the Cash Flow statement. Where Net interest paid is negative (i.e. the entity has net interest received), the relevant Financial Target Threshold should be treated as having been met.]
5 [Acid Ratio]	[All elements that are used to calculate the Acid Ratio are available on the face of the Balance Sheet in a standard set of financial statements.]
6 [Net Asset value]	[Net Assets are shown (but sometimes not labelled) on the face of the Balance Sheet of a standard set of financial statements. Net Assets are sometimes called net worth or 'Shareholders' Funds'. They represent the net assets available to the shareholders. Where an entity has a majority interest in another entity in which there are also minority or non-controlling interests (i.e. where it has a subsidiary partially owned by outside investors), Net Assets should be taken inclusive of minority or non-controlling interests (as if the entity owned 100% of such entity).]

	["Group Exposure" = Balances owed by Group Undertakings + Contingent liabilities assumed in support of Group Undertakings "Gross Assets" = Fixed Assets + Current Assets <u>Group Exposure</u> : Balances owed by (i.e. receivable from) Group Undertakings are shown within Fixed assets or Current assets either on the face of the Balance Sheet or in the relevant notes to the financial statements. In many cases there may be no such balances, in particular where an entity is not a member of a group or is itself the ultimate holding company of the group.
7 [Group Exposure Ratio]	Contingent liabilities assumed in support of Group Undertakings are shown in the Contingent Liabilities note in a standard set of financial statements. They include guarantees and security given in support of the borrowings of other group companies, often as part of group borrowing arrangements. Where the contingent liabilities are capped, the capped figure should be taken as their value. Where no cap or maximum is specified, the relevant Financial Target Threshold should automatically be regarded as not having been met. In many cases an entity may not have assumed any contingent liabilities in support of Group Undertakings, in particular where an entity is not a member of a group or is itself the ultimate holding company of the group. <u>Gross Assets</u> : Both Fixed assets and Current assets are shown on the face of the Balance Sheet]

ANNEX 4: BOARD CONFIRMATION

Supplier Name:

Contract Reference Number:

The Board of Directors acknowledge the requirements set out at paragraph 8 of Schedule 15 (Financial Distress) and confirm that the Supplier has exercised due care and diligence and made reasonable enquiry of all relevant Supplier Personnel and other persons as is reasonably necessary to enable the Board to prepare this statement.

The Board of Directors confirms, to the best of its knowledge and belief, that as at the date of this Board Confirmation it is not aware of and has no knowledge:

- (a) that a Financial Distress Event has occurred since the later of the previous Board Confirmation and the Contract Commencement Date or is subsisting; or
- (b) of any matters which have occurred or are subsisting that could reasonably be expected to cause a Financial Distress Event

On behalf of the Board of Directors:

Chair	
Signed	
Date	
Director	
Signed	
Date	
	176

SUPPLIER'S QUALITY TENDER RESPONSE



Department for Education

DfE Skills Bootcamps Dynamic Purchasing System

Document 5a. Technical Response Template

1

Question 1

Important Note to Potential Suppliers: Please refer to Document 5 Technical Response Instructions for full information on the question requirements.

QUESTION 1 - Project Plan & Mobilisation

Complete Q1.1,1.2,1.3,1.4 and 1.5 for each Skills Bootcamp being proposed

Q1.1 Please complete <u>all</u> the boxes below (state N/A if not applicable to the Skills Bootcamp you are proposing)

Skills Bootcamp Name	Pathway to Accelerated Apprenticeships - OLE	Category and Subcategory	Category 2 - Engineering & Manufacturing Lot - Pathway to Accelerated Apprenticeships in Engineering & Manufacturing Core Subject – Maintenance, Installation & Repair
Lead Provider Name:	Intertrain UK Ltd.	Delivery Partner(s) and what they will be delivering:	N/A

Region of Delivery:	National to include: East Midlands East of England London North-east North-west South- east South-west West Midlands Yorkshire and the Humber.	Qualification(s) Gained or Mapped to:	The training planned is currently delivered as part of the Level 3 Rail Engineering Technician Standard Apprenticeship, which forms part of their OtJ hours and can be mapped to the Standard.
Will training be accredited? If yes, by which Awarding organisation?	Yes, by Sentinel (NWR competencies) and by internal competency tickets.	If the training is not mapped or accredited, please identify the method for recognising and recording progress and achievement.	N/A
Skill Level:	Level 3	Mode(s) of Delivery:	Remote/Classroom/Blended
Number of Guided Learning Hours* (GLH):	114	Number of Independent Learning Hours:	0
Number of learners (across the lifetime of the Skills Bootcamp):	8 learners	Size of Cohorts:	12 – maximum 4 – minimum (regulatory)

Identify and/or quantify the measurement of	Certification (Either internal or external certificate, or a notification to Sentinel for rail-based courses)	What occupation(s) do you expect	Apprentice Rail Engineering Technician.
successful completion of the training element	Register to confirm learner has attended the full Skills Bootcamp (for rail-based courses the trainer will mark all assessments at the end of the course so will identify on	your learners to move into following the	

of the Skills Bootcamp.	the register whether the learner is competent (C) or not competent (NC). Skills Bootcamp tracker where dedicated team track the learner's journey from beginning to end and collate the above information to confirm that the learner has successfully completed the programme.	Skills Bootcamp?	
With reference to the IFATE occupational maps, if there are relevant occupational standards, does your Skills Bootcamp cover some of the content set out in the occupational standard(s)?	Yes Rail engineering technician - Rail Systems Engineering and manufacturing OCC0318c	Where Engineering Skills Bootcamp delivery requires you to be registered with an accredited body e.g., RTAS (Rail Training Assurance Scheme), provide your registration number and where	Apprentice Rail Engineering Technician.

		applicable grade.	
Where the delivery is in a Green Skills subcategory, state if the Skills Bootcamp will deliver skills:	N/A	If the Skills Bootcamp is delivering in Early Years, please state if the Skills Bootcamp will include a work placement. If	N/A

Skills definition, □ in	so, please specify the number of	
Prioritised Green Sectors,	days the placement will be.	
☐ that are 'Direct Green'. **		

* Guided Learning Hours are the time a learner spends being taught or instructed by, or otherwise participating in education or training under the immediate, real-time guidance of a lecturer, supervisor, tutor or other appropriate supplier of education or training, whether online or in person. Prerecorded content does not count towards GLHs.

** Green Skills: definition – "skills that directly meet the needs of the green economy – support the transition from high to low carbon, directly contribute to the achievement of the UK's net zero emissions target and help mitigate climate risks"; Direct green – "skills that are <u>specific</u> to a Prioritised Green Sector or role within that sector. For example, a heat pump installer needing specific skills to be able to do that role."

Response requirement:

- Complete all of the boxes put N/A if not applicable to show you have dealt with the question.
- Complete Q1.1 boxes for each Skills Bootcamp being proposed.

Q1.2 Employer Requirements

Describe the knowledge^[i], skills^[ii] and behaviours^[iii] that are required by one of your employers, or those you require if you are an employer upskilling/training your own employees.

Knowledge [can be learnt and applied – can be technical, practical, and theoretical and evidenced by the learner] (e.g., knowing relevant sections of health and safety legislation and how to adhere to them):

<u>Skills</u> [can be demonstrated and learnt – can be practical, technical, and shown by the learner] (e.g., problem solving, Customer service skills):

Behaviours [can be observed and developed – can be social, cultural, and interpersonal] (e.g., resilience, empathy, punctuality):

Knowledge is the technical detail and understanding the learners need to achieve and understand to complete their learning. The knowledge gained in the Skills Bootcamps course should improve the learner's success and help achieve positive outcomes.

Skills are the application of the knowledge the learner receives when attending the Skills Bootcamp course and is shown through both experience and on the job training. Skills are not a repetition of activities.

Behaviours are what can be seen and demonstrated by learners, these can be learnt and should be transferable.

Response requirement:

• Set out a response to Knowledge, to Skills and to Behaviours required by one of your employers (or you if you are an employer upskilling your own employee). Examples should be included as part of the response, as opposed to making reference to any other questions/responses within the bid.

Maximum 1000 characters 959 characters

Knowledge

- Safe and professional working practices to industry requirements
- Engineering techniques to maintain, renew and construct the railway
- Effectively contribute to engineering solutions, awareness of 3rd party requirements and corporate policies
- Awareness of applicable commercial principles and implications.

Skills

- Adhere to safe working practices
- Prepare for high standard of technical work using drawings and schedules
- Deliver high standard technical work for maintenance, construction and renewal
- Identify and report problems
- Correct utilisation and storage of resources, tools and equipment •

Communicate effectively electronically and

verbally. Behaviours

- Professionalism and mutual respect
- Risk aware

• Self-disciplined and motivated

- Reliable and safe to industry standards
- Efficient individually and in a team
- Receptive to feedback
- Prepared to make a personal commitment to their employer, the industry and its professional standards.

Q1.3 - Assessment for Learning

Describe your initial^[1] and final^[1] assessments and how they will show the comparable increase in the knowledge, skills, and behaviour of learners at the start and end of a Skills Bootcamp. These may be in a written, oral, or practical form that is appropriate to assess such knowledge, skills, and behaviour.

Describe how you will demonstrate learner success on completion of the Skills Bootcamp. E.g., qualification gained, Skills learnt, Portfolio etc.

Initial assessment is the assessment used prior to a learner joining a course or training programme by Providers with learners to ascertain the learner's starting points. Initial assessment (IA) usually considers Maths and English levels but can also include IT. Initial Assessments are used by practitioners to plan their teaching.

Final assessment is the assessment activity used at the end of a unit, module, or time period. These often take the form of a test/exam, portfolio, or final project.

Response requirement:

- Describe your initial and final assessment and how they will show the comparable increase in the knowledge, skills, and behaviour of learners at the start and end of a Skills Bootcamp.
- Describe how you will demonstrate learner success on completion of the Skills Bootcamp.

Maximum 750 characters 718 characters

- Initial Assessment: Before starting, learners complete an Individual Learner Plan (ILP), with questions on prior knowledge/skills/behaviours
- **Final Assessment**: Practical, e.g identify hazards on a live railway, and knowledge assessments (learners are also assessed after each individual element)
- **Showing increases in knowledge/skills/behaviour**: The ILP records learner progress throughout a comprehensive record of achievement with clear evidence of knowledge/skills/behaviour at the start and end
- **Demonstrating learner success upon completion**: If the learner achieves competence, skills are added to their Sentinel competency card. If competency cannot be demonstrated, further training/assessment is undertaken

Q1.4 - High Level Overview of Weeks / Sessions / Modules (please identify which):

Please list the key topic areas by week, session, or module (whichever you feel is more appropriate and succinct) using bullet points or short sentences.

See example below for an 8-week Skills Bootcamp:

Week 1 – Introduction to Drylining within the construction industry (theory)

Week 2 – Introduction to Drylining within the construction industry (practical)

Week 3 – CSCS card theory and application

Week 3 – Explore the use of Health and Safety in a construction workplace.

Week 4 – Drylining Systems

Week 5 – Demonstrate skills in Drylining, including fitting and finishing.

Week 6 – Develop job ready practical skills for within the construction industry. Week

7-8 Progress into employment in Drylining.

Response requirement:

- Provide a high-level overview, listing key topic areas.
- This can be done by week, session, or module as appropriate.
- Bullet points or short sentences are acceptable.

Maximum 1250 characters

875 characters

Week 1 - Safety Awareness, Manual Handling, Fire Awareness, Emergency First Aid at Work, Personal Track Safety (PTS) Practical, Overhead Line Construction (OLEC 1)/Employer Engagement. **Week 2** – Overhead Line Construction (OLEC 2)

- OLE Systems
- Documentation, Design Drawings & Reporting
- Installation Methods & Techniques (including practical assessment)
- General Safety
- Theory Assessment

Week 3 – Road Rail Vehicle (RRV) & Mobile Elevated Working Platform (MEWP) Awareness, Overhead Line (OLE) Bonding, Fabrication, Basic Signalling & Track Awareness, Isolation HV & ATF, Design, Assurance, Pile Setting, Topo Setting.

Week 4 - Working at Height & Harness Training, Direct Current Conductor Rail (DCCR), Behaviours (FTBS/Situational & Incidents/Awareness & Change), Safety (Driving/Alpha & Wellbeing), Emergency Response, Environment, Employability Skills/Employer Engagement.

Q1.5 - Week/ Session / Module Plan

Please complete the table below providing detail for 20% of your proposed^[1] curriculum. Please see the examples below for guidance.

- For column one, please identify whether you are outlining by week, session, or module, this is entirely up to you.
- For column two, please identify the learning aim(s)^[iii] and objective(s)^[iiii] for that week, session, or module. Consider how the aims and objectives for each week, session or module will support learners in successfully completing their Skills Bootcamps course and to meet employer need.
- For column three, please identify what activities learners will be doing in weeks, sessions, or modules. Consider what activities learners should do to support them in developing the knowledge, skills and behaviours needed for the Skills Bootcamp.
- For column four, please identify what methods of assessing or checking you will use to see if learners are making progress during their time on the Skills Bootcamp, and against the aims and objectives. The assessment activity

here does not have to be formal or accredited but should enable you to track learners progress between initial and final assessment.

Maximum 200 characters for each learning aims cell, 400 characters for each learning activities cell, and 200 characters for the progress cells.

Wee	ek/Session	Learning aims/objectives	What activities are the learners completing and how is it being delivered (mode, e.g., online, face to face, blended)?	How will you know if the learner is making progress? How are you going to check for this?
Line	2 – Overhead ruction 2 2)	 Learners can: Explain functions of main OLE systems, how to work safely and limits of authority Complete OLE construction activities Interpret renewal, installation and enhancement documentation 	 Face-to-face activities Tutor led presentation - OLE electrification system Q&A on main responsibilities of the Electrical Control Operator (ECO) Group discussion on drawings, diagrams and colour coding Trainer demonstration on tools and equipment Practical exercises – including cut and splice OLE wires, install in-line fittings, use a laser height and stagger gauge 	Trainer checks knowledge in demo/Q&A/discussion and identifies gaps Trainer/peer observation and feedback on practicals Initial, formative and summative assessments Progress review: 1-2-1 and update ILP

Session – Emergency First Aid at Work (Qualsafe Accredited)	 Learners can: Describe the roles and responsibilities of a first aider Respond in emergencies Respond to unresponsive/ choking/bleeding casualties and those suffering from shock and minor injuries 	 Face-to-face activities Group discussions: Roles and responsibilities of a first aider How to minimise risk of infection Need for consent Group discussion on when to administer CPR 	Trainer/peer observation and feedback on practical exercises Q&A/discussions – Trainer checks knowledge/identifies gaps Initial, summative, formative, knowledge tests Progress review: 1-2-1/update ILP
		 Practical exercises How to give CPR using a manikin How to put someone in recovery position Monitor breathing • Use a defibrillator (AED) Q&As: How to conduct a scene survey How to summon assistance 	

^{II} Changes may still be made to the proposed curriculum before delivery commences.

Aims are the overarching intentions of the course or programme and what learners should leave with once completed. Learni outcomes/objectives are derived from these and articulate clear measures which help to determine if the aims have been met.

Objectives are brief statements that describe what learners will be expected to learn by the end of their Skills Bootcamps. T Provider objectives will be different from the learner and linked to the quality of the Skills Bootcamps delivery and ensuring learner have the best possible outcomes when attending a Skills Bootcamp.

Weighting (for Q1 as a total) – 20% (Where Pathway to Accelerated Apprenticeships delivery taking place – 15%)

Please write your response to this question in this Word document template and upload it to the relevant question answer area in the Jaggaer e-Sourcing Portal within the Technical Envelope.

Please ensure your document response is named as per the response naming guidance in Document 1 Procurement Document A Invitation to Participate.



Department for Education

DfE Skills Bootcamps Dynamic Purchasing System

Document 5a. Technical Response Template

1

Question 1

Important Note to Potential Suppliers: Please refer to Document 5 Technical Response Instructions for full information on the question requirements.

QUESTION 1 - Project Plan & Mobilisation

Complete Q1.1,1.2,1.3,1.4 and 1.5 for each Skills Bootcamp being proposed

Q1.1 Please complete <u>all</u> the boxes below (state N/A if not applicable to the Skills Bootcamp you are proposing)

Skills Bootcamp Name	Pathway to Accelerated Apprenticeship - Signalling	Category and Subcategory	Category 2 - Engineering & Manufacturing Lot - Pathway to Accelerated Apprenticeships in Engineering & Manufacturing Core Subject – Maintenance, Installation & Repair
Lead Provider Name:	Intertrain UK Ltd.	Delivery Partner(s) and what they will be delivering:	N/A
Region of Delivery:	National to include: • East Midlands	Qualification(s) Gained or	The training planned is currently delivered as part

	 East of England London North-east North-west • South-east South-west West Midlands Yorkshire and the Humber 	Mapped to:	of the Level 3 Rail Engineering Technician Standard Apprenticeship, which forms part of their OtJ hours and can be mapped to the Standard.
Will training be accredited? If yes, by which Awarding organisation?	Yes, by Sentinel (NWR competencies) and by internal competency tickets.	If the training is not mapped or accredited, please identify the method for recognising and recording progress and achievement.	N/A
Skill Level:	Level 3	Mode(s) of Delivery:	Remote / Classroom / Blended
Number of Guided Learning Hours* (GLH):	150	Number of Independent Learning Hours:	0
Number of learners (across the lifetime of the Skills Bootcamp):		Size of Cohorts:	10 – maximum 4 – minimum (regulatory)

Identify and/or quantify the measurement of successful completion of the training element of the Skills Bootcamp.Certification (Either internal or external certificate, or a notification to Sentinel for rail-based courses)Register to confirm learner has attended the full Skills Bootcamp (for rail-based courses the trainer will mark all assessments at the end of the course so will identify on	· · · · · · · · · · · · · · · · ·	Apprentice Rail Engineering Technician
--	-----------------------------------	---

the register whether the learner is competent (C) or not competent (NC).		
Skills Bootcamp tracker where dedicated team track the learner's journey from beginning to end and collate the above information to confirm that the learner has successfully completed the programme.		

With reference to the IFATE occupational maps, if there are relevant occupational standards, does your Skills Bootcamp cover some of the content set out in the occupational standard(s)?	OCC0318c	Where Engineering Skills Bootcamp delivery requires you to be registered with an accredited body e.g., RTAS (Rail Training Assurance Scheme), provide your registration number and where applicable grade.	RTAS Number – 0460, NSAR Assured, Silver Award
Where the delivery is in a Green Skills subcategory, state if the Skills Bootcamp will deliver skills:	N/A	If the Skills Bootcamp is delivering in Early Years, please state if the Skills Bootcamp will include a work placement. If	N/A

Skills definition, □ in	so, please specify the number of	
Prioritised Green Sectors,	days the placement will be.	
☐ that are 'Direct Green'. **		

* Guided Learning Hours are the time a learner spends being taught or instructed by, or otherwise participating in education or training under the immediate, real-time guidance of a lecturer, supervisor, tutor or other appropriate supplier of education or training, whether online or in person. Prerecorded content does not count towards GLHs.

** Green Skills: definition – "skills that directly meet the needs of the green economy – support the transition from high to low carbon, directly contribute to the achievement of the UK's net zero emissions target and help mitigate climate risks"; Direct green – "skills that are <u>specific</u> to a Prioritised Green Sector or role within that sector. For example, a heat pump installer needing specific skills to be able to do that role."

Response requirement:

- Complete all of the boxes put N/A if not applicable to show you have dealt with the question.
- Complete Q1.1 boxes for each Skills Bootcamp being proposed.

Q1.2 Employer Requirements

Describe the knowledge^[i], skills^[ii] and behaviours^[iii] that are required by one of your employers, or those you require if you are an employer upskilling/training your own employees.

Knowledge [can be learnt and applied – can be technical, practical, and theoretical and evidenced by the learner] (e.g., knowing relevant sections of health and safety legislation and how to adhere to them):

<u>Skills</u> [can be demonstrated and learnt – can be practical, technical, and shown by the learner] (e.g., problem solving, Customer service skills):

Behaviours [can be observed and developed – can be social, cultural, and interpersonal] (e.g., resilience, empathy, punctuality):

Knowledge is the technical detail and understanding the learners need to achieve and understand to complete their learning. The knowledge gained in the Skills Bootcamps course should improve the learner's success and help achieve positive outcomes.

Skills are the application of the knowledge the learner receives when attending the Skills Bootcamp course and is shown through both experience and on the job training. Skills are not a repetition of activities.

Behaviours are what can be seen and demonstrated by learners, these can be learnt and should be transferable.

Response requirement:

• Set out a response to Knowledge, to Skills and to Behaviours required by one of your employers (or you if you are an employer upskilling your own employee). Examples should be included as part of the response, as opposed to making reference to any other questions/responses within the bid.

Maximum 1000 characters 968 characters

Knowledge

- Safe and Professional working practices to industry requirements
- Engineering techniques to maintain, renew and construct the Railway
- Effectively contribute to engineering solutions, awareness of 3rd party requirements and corporate policies
- Awareness of applicable commercial principles and implications

Skills

- Adhere to safe working practices
- Prepare for high standard of technical work using drawings and schedules
- Deliver high standard technical work for maintenance, construction and renewal
- Identify and report problems
- Correct utilisation and storage of resources, tools and equipment
- Communicate effectively electronically and verbally

Behaviours

- Safety conscious/risk aware
- Inclusive treat all equally
- Self-disciplined and motivated
- Reliable and safe to industry standards
- Efficient individually and in a team
- Receptive to feedback
- Prepared to make a personal commitment to their employer, the industry and its professional standards

Q1.3 - Assessment for Learning

Describe your initial^[1] and final^[1] assessments and how they will show the comparable increase in the knowledge, skills, and behaviour of learners at the start and end of a Skills Bootcamp. These may be in a written, oral, or practical form that is appropriate to assess such knowledge, skills, and behaviour.

Describe how you will demonstrate learner success on completion of the Skills Bootcamp. E.g., qualification gained, Skills learnt, Portfolio etc.

Initial assessment is the assessment used prior to a learner joining a course or training programme by Providers with learners to ascertain the learner's starting points. Initial assessment (IA) usually considers Maths and English levels but can also include IT. Initial Assessments are used by practitioners to plan their teaching.

Final assessment is the assessment activity used at the end of a unit, module, or time period. These often take the form of a test/exam, portfolio, or final project.

Response requirement:

- Describe your initial and final assessment and how they will show the comparable increase in the knowledge, skills, and behaviour of learners at the start and end of a Skills Bootcamp.
- Describe how you will demonstrate learner success on completion of the Skills Bootcamp.

Maximum 750 characters 718 characters

- Initial Assessment: Before starting, learners complete an Individual Learner Plan (ILP), with questions on prior knowledge/skills/behaviours.
- Final Assessment: Practical, e.g identify hazards on a live railway, and knowledge assessments (learner are also assessed after each individual element)
- Showing increases in knowledge/skills/behaviour: The ILP records learner progress throughout a comprehensive record of achievement with clear evidence of knowledge/skills/behaviour at the start and end
- **Demonstrating learner success upon completion:** If the learner achieves competence, skills are added to their Sentinel competency card. If competency cannot be demonstrated, further training/assessment is undertaken

Q1.4 - High Level Overview of Weeks / Sessions / Modules (please identify which):

Please list the key topic areas by week, session, or module (whichever you feel is more appropriate and succinct) using bullet points or short sentences.

See example below for an 8-week Skills Bootcamp:

Week 1 – Introduction to Drylining within the construction industry (theory)

Week 2 – Introduction to Drylining within the construction industry (practical)

Week 3 – CSCS card theory and application

Week 3 – Explore the use of Health and Safety in a construction workplace.

Week 4 – Drylining Systems

Week 5 – Demonstrate skills in Drylining, including fitting and finishing.

Week 6 – Develop job ready practical skills for within the construction industry. Week

7-8 Progress into employment in Drylining.

Response requirement:

- Provide a high-level overview, listing key topic areas.
- This can be done by week, session, or module as appropriate.
- Bullet points or short sentences are acceptable.

Maximum 1250 characters

1244 characters

Week 1 - Safety Awareness, Manual Handling, Fire Awareness, Emergency First Aid at Work, Personal Track Safety (PTS) Practical, Overhead Line Construction (OLEC 1)/Employer Engagement:

- Rule Book GE/RT 8000 Series
- Guide to PTS
- PTS Key Point Card
- Safe System of Work
- Track visit

Week 2 - Electronic Principles 1 (EP1), Diagram Reading

- Basic circuit readings
- Diagram symbols/equipment
- Track circuits/axle counters
- Level crossings, telecoms, networks
- Power supplies/batteries
- Operations of relays

Week 3 – Safety Precautions when Working with Electrical Equipment (SPWEE)

- Electricity at Work Regs
- Safe Working Practices
- Working on voltages not above 1000V AC
- Local/protective isolation
- Methods to test isolation

Weeks 3 & 4 - Basic Signalling 1 (BS1)

- Jargon
- Signalling overview/signal types

Week 5 - Working at Height & Harness, Direct Current Conductor Rail (DCCR), Behaviours (FTBS/Situational & Incidents/Awareness & Change), Safety (Driving/Alpha & Wellbeing), Emergency Response, Employability Skills/Employer Engagement:

- Principles/components of DC electrified system
- Hazards, risks, defects
- How to cross DC rail
- Emergency response
- Inspecting tools
- Application/removal of conductor rail shields Risk control areas

Q1.5 - Week/ Session / Module Plan

Please complete the table below providing detail for 20% of your proposed^[1] curriculum. Please see the examples below for guidance.

- For column one, please identify whether you are outlining by week, session, or module, this is entirely up to you.
- For column two, please identify the learning aim(s)^[iii] and objective(s)^[iiii] for that week, session, or module. Consider how the aims and objectives for each week, session or module will support learners in successfully completing their Skills Bootcamps course and to meet employer need.
- For column three, please identify what activities learners will be doing in weeks, sessions, or modules. Consider what activities learners should do to support them in developing the knowledge, skills and behaviours needed for the Skills Bootcamp.
- For column four, please identify what methods of assessing or checking you will use to see if learners are making progress during their time on the Skills Bootcamp, and against the aims and objectives. The assessment activity here does not have to be formal or accredited but should enable you to track learners progress between initial and final assessment.

Maximum 200 characters for each learning aims cell, 400 characters for each learning activities cell, and 200 characters for the progress cells.

Session	Learning aims/objectives	What activities are the learners completing and how is it being delivered (mode, e.g., online, face to face, blended)?	How will you know if the learner is making progress? How are you going to check for this?

Session – Basic Signalling 1 & 2	Learners can define signalling jargon, types of signals and equipment, basic diagrams and the development of track circuits	 All face-to-face activities Tutor led session on signalling jargon Group discussion -why we need signalling, different types of signals/ equipment Individual exercise - learners identify basic diagram symbols Group exercise - learners explain development of track circuits/axle counters Q&A session - AWS/TPWS equipment operations and European Railway Traffic Management System 	Q&A – Trainer checks knowledge Group discussions – Trainer spots knowledge gaps Trainer/peer observation of exercises Initial/formative/summative assessments Progress review: 1-2-1/update ILP
Session – Practical Installation	Learners can explain the railway alphabet, diagram symbols, types of signalling cables and the use of basic hand tools.	 All face-to-face activities Tutor led session - railway alphabet and symbols on diagrams Group discussion - types of railway signalling cables/ 	Q&A – Trainer checks knowledge Group discussions – Trainer identifies gaps Trainer/peer observation of

	Practical exercise - learners demonstrate use of basic hand tools	practicals Initial/summative assessment and test 1-2-1 progress review/update ILP
--	---	--

^{II} Changes may still be made to the proposed curriculum before delivery commences.

Aims are the overarching intentions of the course or programme and what learners should leave with once completed. Learning outcomes/objectives are derived from these and articulate clear measures which help to determine if the aims have been met.

Objectives are brief statements that describe what learners will be expected to learn by the end of their Skills Bootcamps. The Provider objectives will be different from the learner and linked to the quality of the Skills Bootcamps delivery and ensuring learners have the best possible outcomes when attending a Skills Bootcamp.

Weighting (for Q1 as a total) – 20% (Where Pathway to Accelerated Apprenticeships delivery taking place – 15%)

Please write your response to this question in this Word document template and upload it to the relevant question answer area in the Jaggaer e-Sourcing Portal within the Technical Envelope.

Please ensure your document response is named as per the response naming guidance in Document 1 Procurement Document A Invitation to Participate.



Department for Education

DfE Skills Bootcamps Dynamic Purchasing System

Document 5a. Technical Response Template

1

Question 1

Important Note to Potential Suppliers: Please refer to Document 5 Technical Response Instructions for full information on the question requirements.

QUESTION 1 - Project Plan & Mobilisation

Complete Q1.1,1.2,1.3,1.4 and 1.5 for each Skills Bootcamp being proposed

Q1.1 Please complete <u>all</u> the boxes below (state N/A if not applicable to the Skills Bootcamp you are proposing)

Skills Bootcamp Name	Pathway to Accelerated Apprenticeship - Track	Category and Subcategory	Category 2 - Engineering & Manufacturing, Lot - Pathway to Accelerated Apprenticeships in Engineering & Manufacturing Core Subject – Maintenance, Installation & Repair
Lead Provider Name:	Intertrain UK Ltd.	Delivery Partner(s) and what they will be delivering:	N/A

Region of Delivery:	National to include: • East Midlands • East of England • London • North-East • North-West • South-East • South-West • West Midlands • Yorkshire and the Humber	Qualification(s) Gained or Mapped to:	The training planned is currently delivered as part of the Level 3 Rail Engineering Technician Standard Apprenticeship, which forms part of their OtJ hours and can be mapped to the Standard. Rail Engineering Operative – Track Level 2 Code: OCC0317e
Will training be accredited? If yes, by which Awarding organisation?	Yes, by Sentinel (NWR competencies) and by internal competency tickets.	If the training is not mapped or accredited, please identify the method for recognising and recording progress and achievement.	N/A
Skill Level:	Level 3	Mode(s) of Delivery:	Remote / Classroom / Blended
Number of Guided Learning Hours* (GLH):	114 Hours Track	Number of Independent Learning Hours:	0
Number of learners (across the lifetime of the Skills Bootcamp):		Size of Cohorts:	12 – maximum 4 – minimum (regulatory)

Identify and/or quantify the measurement of successful completion of the	Certification (Either internal or external certificate, or a notification to Sentinel for rail-based courses)	What occupation(s) do you expect your learners to move into	Apprentice Rail Engineering Technician
training element of the Skills Bootcamp.	Register to confirm learner has attended the full Skills Bootcamp (for rail-based courses the trainer will mark all assessments at the end of the course so will identify on the register whether the learner is competent (C) or not competent (NC). Skills Bootcamp tracker where dedicated team track the learner's journey from beginning to end and collate the above information to confirm that the learner has successfully completed the programme.	following the Skills Bootcamp?	

With reference to the IFATE occupational maps, if there are relevant occupational standards, does your Skills Bootcamp cover some of the content set out in the occupational standard(s)?	Yes Rail Engineering Operative – Track Level 2 Code: OCC0317e	Where Engineering Skills Bootcamp delivery requires you to be registered with an accredited body e.g., RTAS (Rail Training Assurance Scheme), provide your registration number and where applicable grade.	RTAS Number – 0460, NSAR Assured, Silver Award
Where the delivery is in a Green Skills subcategory, state if the Skills	N/A	<i>If the Skills Bootcamp is delivering in Early Years, please state if the Skills</i>	N/A

Bootcamp will deliver skills:	Bootcamp will include a work placement. If	
the Green Skills	so, please specify the	
definition,	number of days the	
Prioritised Green	placement will be.	
Sectors, that are		
'Direct Green'. **		

* Guided Learning Hours are the time a learner spends being taught or instructed by, or otherwise participating in education or training under the immediate, real-time guidance of a lecturer, supervisor, tutor or other appropriate supplier of education or training, whether online or in person. Prerecorded content does not count towards GLHs.

** Green Skills: definition – "skills that directly meet the needs of the green economy – support the transition from high to low carbon, directly contribute to the achievement of the UK's net zero emissions target and help mitigate climate risks"; Direct green – "skills that are <u>specific</u> to a Prioritised Green Sector or role within that sector. For example, a heat pump installer needing specific skills to be able to do that role."

Response requirement:

- Complete all of the boxes put N/A if not applicable to show you have dealt with the question.
- Complete Q1.1 boxes for each Skills Bootcamp being proposed.

Q1.2 Employer Requirements

Describe the knowledge^[i], skills^[ii] and behaviours^[iii] that are required by one of your employers, or those you require if you are an employer upskilling/training your own employees.

Knowledge [can be learnt and applied – can be technical, practical, and theoretical and evidenced by the learner] (e.g., knowing relevant sections of health and safety legislation and how to adhere to them):

<u>Skills</u> [can be demonstrated and learnt – can be practical, technical, and shown by the learner] (e.g., problem solving, Customer service skills):

Behaviours [can be observed and developed – can be social, cultural, and interpersonal] (e.g., resilience, empathy, punctuality):

Knowledge is the technical detail and understanding the learners need to achieve and understand to complete their learning. The knowledge gained in the Skills Bootcamps course should improve the learner's success and help achieve positive outcomes.

Skills are the application of the knowledge the learner receives when attending the Skills Bootcamp course and is shown through both experience and on the job training. Skills are not a repetition of activities.

Behaviours are what can be seen and demonstrated by learners, these can be learnt and should be transferable.

Response requirement:

• Set out a response to Knowledge, to Skills and to Behaviours required by one of your employers (or you if you are an employer upskilling your own employee). Examples should be included as part of the response, as opposed to making reference to any other questions/responses within the bid.

Maximum 1000 characters 919 characters

Knowledge

- Safe and Professional working practices to industry requirements
- Engineering techniques to maintain, renew and construct the Railway
- Effectively contribute to engineering solutions, awareness of 3rd party requirements and corporate policies
- Awareness of applicable commercial principles and implications.

Skills

- Adhere to safe working practices
- Prepare for high standard of technical work using drawings and schedules
- Deliver high standard technical work for maintenance, construction and renewal
- Identify and report problems
- Correct utilisation and storage of resources, tools and equipment
- Communicate effectively electronically and verbally.

Behaviours

- Safety conscious
- Treat all equally
- Risk aware
- Self-disciplined and motivated
- Efficient individually and in a team
- Receptive to feedback
- Prepared to make a personal commitment to their employer, the industry and its professional standards.

Q1.3 - Assessment for Learning

Describe your initial^[1] and final^[1] assessments and how they will show the comparable increase in the knowledge, skills, and behaviour of learners at the start and end of a Skills Bootcamp. These may be in a written, oral, or practical form that is appropriate to assess such knowledge, skills, and behaviour.

Describe how you will demonstrate learner success on completion of the Skills Bootcamp. E.g., qualification gained, Skills learnt, Portfolio etc.

Initial assessment is the assessment used prior to a learner joining a course or training programme by Providers with learners to ascertain the learner's starting points. Initial assessment (IA) usually considers Maths and English levels but can also include IT. Initial Assessments are used by practitioners to plan their teaching.

Final assessment is the assessment activity used at the end of a unit, module, or time period. These often take the form of a test/exam, portfolio, or final project.

Response requirement:

- Describe your initial and final assessment and how they will show the comparable increase in the knowledge, skills, and behaviour of learners at the start and end of a Skills Bootcamp.
- Describe how you will demonstrate learner success on completion of the Skills Bootcamp.

Maximum 750 characters 718 characters

- Initial Assessment: Before starting, learners complete an Individual Learner Plan (ILP), with questions on prior knowledge/skills/behaviours
- **Final Assessment**: Practical, e.g. identify hazards on a live railway, and knowledge assessments (learner are also assessed after each individual element)
- Showing increases in knowledge/skills/behaviour: The ILP records learner progress throughout a comprehensive record of achievement with clear evidence of knowledge/skills/behaviour at the start and end
- **Demonstrating learner success upon completion:** If the learner achieves competence, skills are added to their Sentinel competency card. If competency cannot be demonstrated, further training/assessment is undertaken

Q1.4 - High Level Overview of Weeks / Sessions / Modules (please identify which):

Please list the key topic areas by week, session, or module (whichever you feel is more appropriate and succinct) using bullet points or short sentences.

See example below for an 8-week Skills Bootcamp:

Week 1 – Introduction to Drylining within the construction industry (theory)

- Week 2 Introduction to Drylining within the construction industry (practical)
- Week 3 CSCS card theory and application
- Week 3 Explore the use of Health and Safety in a construction workplace.
- Week 4 Drylining Systems
- Week 5 Demonstrate skills in Drylining, including fitting and finishing.
- Week 6 Develop job ready practical skills for within the construction industry. Week
- 7-8 Progress into employment in Drylining.

Response requirement:

- Provide a high-level overview, listing key topic areas.
- This can be done by week, session, or module as appropriate.
- Bullet points or short sentences are acceptable.

Maximum 1250 characters

1214 characters

Week 1 - Safety Awareness, Manual Handling, Fire, Emergency First Aid, Personal Track Safety (PTS) Practical, Overhead Line Construction (OLEC 1)/Employer Engagement

- Rule Book GE/RT 8000 Series/Guide to Personal Track Safety/PTS Key Point Card NR9922
- Safe System of Work
- Track visit/assessment
- Managing an emergency
- Communication and casualty care
- Seizures, bleeding, shock, burns, choking, burns, CPR
- Defibrillator

Week 2 - Track Induction

- Plain line switches/crossings
- Joints
- Adjustment switches
- Drains, vegetation
- Equipment signals, points machines, cranks, rodding
- OHL equipment structures, catenaries; earth, red/yellow bonds

Week 3 - Track Renewals, Basic Signalling Awareness, Hand Trolley, Temporary Joints Installation (TR52)

- Signalling jargon
- Signal types

Week 4 - Working at Height & Harnesses, Direct Current Conductor Rail (DCCR), Behaviours (FTBS/Situational &

Incidents/Awareness & Change), Safety (Driving/Alpha & Wellbeing), Emergency Response, Employability Skills

- DC electrified system/DC conductor rail system/DC conductor rail safety
- Hazards/risks
- Transit of tools/equipment across
- Emergency procedures
- Tool inspections
- Risk control areas in NR/L3/MTC/EP0152 Ladders

Q1.5 - Week/ Session / Module Plan

Please complete the table below providing detail for 20% of your proposed^[1] curriculum. Please see the examples below for guidance.

- For column one, please identify whether you are outlining by week, session, or module, this is entirely up to you.
- For column two, please identify the learning aim(s)^[iii] and objective(s)^[iiii] for that week, session, or module. Consider how the aims and objectives for each week, session or module will support learners in successfully completing their Skills Bootcamps course and to meet employer need.
- For column three, please identify what activities learners will be doing in weeks, sessions, or modules. Consider what activities learners should do to support them in developing the knowledge, skills and behaviours needed for the Skills Bootcamp.
- For column four, please identify what methods of assessing or checking you will use to see if learners are making
 progress during their time on the Skills Bootcamp, and against the aims and objectives. The assessment activity
 here does not have to be formal or accredited but should enable you to track learners progress between initial and
 final assessment.

Maximum 200 characters for each learning aims cell, 400 characters for each learning activities cell, and 200 characters for the progress cells.

Session/Week	Learning aims/objectives	What activities are the learners completing and how is it being delivered (mode, e.g., online, face to face, blended)?	How will you know if the learner is making progress? How are you going to check for this?
Session – Personal Track Safety (PTS) Practical Event	 Learners can: Explain lifesaving rules, effects of incidents on others and personal responsibilities Deal with emergency situations Follow correct procedures when making an emergency call 	 All face-to-face activities Individual Activity - PTS Knowledge Check Q&A - lifesaving rules and which apply to personal track safety. 	Workbook completed Trainer/peer observation of role play Q&A – Trainer checks knowledge Group discussions – Trainer spots gaps Formative/summative assessment Progress reviews: 1-2-1/update ILP

Week 2– Track Induction	 Learners can: Describe each component part of track infrastructure Describe routine preventative maintenance on the 	 Tutor led session -history of the construction of railways Q&A - hazards on single and bi-directional lines Group discussion - why 	Q&A – Trainer checks knowledge Group discussion – Trainer spots knowledge gaps Trainer/peer observation of practicals Initial/formative/summative/assessments Progress reviews: 1-2-1/update ILP
	 permanent way Carry out maintenance activities within limits of their authority 	 essential Q&A - identification of rails Practical exercise – identifying track components and their uses Practical exercise – using hand tools safely and effectively and ensuring they are well maintained 	

^{II} Changes may still be made to the proposed curriculum before delivery commences.

Aims are the overarching intentions of the course or programme and what learners should leave with once completed. Learning outcomes/objectives are derived from these and articulate clear measures which help to determine if the aims have been met.

Objectives are brief statements that describe what learners will be expected to learn by the end of their Skills Bootcamps. The Provider objectives will be different from the learner and linked to the quality of the Skills Bootcamps delivery and ensuring learners have the best possible outcomes when attending a Skills Bootcamp.

Weighting (for Q1 as a total) – 20% (Where Pathway to Accelerated Apprenticeships delivery taking place – 15%)

Please write your response to this question in this Word document template and upload it to the relevant question answer area in the Jaggaer e-Sourcing Portal within the Technical Envelope.

Please ensure your document response is named as per the response naming guidance in Document 1 Procurement Document A Invitation to Participate.



DfE Skills Bootcamps Dynamic Purchasing System

1

Document 5b. Technical Response Template

Question 2

Important Note to Potential Suppliers: Please refer to Document 5 Technical Response Instructions for full information on the question requirements.

QUESTION 2 – Learner Engagement

Describe how you will engage learners throughout the Skills Bootcamp, including during:

- recruitment,
- delivery of training, and
- offer of an interview(s) to ensure they reach completion and achieve a positive outcome.

Response requirements:

Describe how you will:

- Recruit and support potential learners, including those in underrepresented groups. Include how you will provide high quality Information, Advice and Guidance, and screen potential learners to ensure the Skills Bootcamp is appropriate for them and that they fully understand the learning and commitment involved, and how you will ensure that your training is available and accessible to a diverse cohort, representative of the communities in which you will deliver.
- Support learners throughout the Skills Bootcamp to promote learner retention. This should include details of the Wraparound Support that you will provide before, during and after the completion of the training, e.g., mentoring, coaching, behavioural skills development, career advice, interview support including post-interview if not initially successful.
- Support learners throughout the Skills Bootcamp, including throughout the period learners are being linked to interviews, to promote positive outcomes. This should include interview support for learners moving into a new organisation, for existing employees to move into a new role or new responsibilities within their own organisation, or for a self-employed learner to secure new contracts and new opportunities. (As applicable for the learners being supported by the Skills Bootcamp.)

Weighting – 20% Maximum characters: 3750

Please write your response to this question in this Word document template and upload it to the relevant question answer area in the Jaggaer e-Sourcing Portal within the Technical Envelope.

Please ensure your document response is named as per the response naming guidance in Document 1 Procurement Document an Invitation to Participate.

3747 characters

Intertrain is part of the City & Guilds (C&G) Delivery Hub, combining the delivery expertise/resources of all C&G businesses. We will engage learners as follows.

Recruiting learners

- Contact 2,150 employers, plus subcontractors' networks
- Advertise on employer websites/internal comms
- Social media, e.g LinkedIn, Facebook Sourcing **new entrants**:
- Use referral partners e.g JCP, NCS, RFEA, work programme providers
- School Engagement Officer via in-school events, mock interviews
- Skills/career fairs
- Inspira career advice service
- Work with prisons
- Mailshots 20,000 customers
- Paid search/SEO

Sourcing employed workers:

- Engage employer network and industry specialists (Network Rail, TfL) to identify employees at risk of redundancy
 - Provide literature/hold IAG sessions **Sourcing longer-term unemployed**:
 - Target advertising for high-unemployment regions
 - Weekly liaison with DWP
 - Referrals from other courses **Sourcing self employed**:
 - Mailshot customer database
 - Paid search/SEO
 - Contribute to business forums
 - Work with partners, e.g Trade Up

Our Learning Experience Team (LET) hold IAG, Equality, Diversity and Inclusion and Safer Recruitment qualifications. Recruitment is fair/non-discriminatory, with equal opportunity for all, including those with learning difficulties/additional support needs, including:

- Additional time/flexibility in aptitude testing, e.g flexible location
- Applicants disclosing disability guaranteed interview
- Informing interview panel of specific adjustments

Reaching under-represented groups

- Train in prisons 82% job outcomes to date
- Engage charities, e.g BounceBack/Offploy for training prisoners on release
- Advocacy for under-represented e.g. deprived young people (St Giles Trust), disadvantaged Londoners (City of London)
 - Target marketing to address challenges/support available

IAG: Underpinned by our **Matrix Accreditation**, IAG is central for learner support/ensuring Skills Bootcamps enable choice of job/career options. We cover:

- Duration/time commitment/attendance
- Resources
- Assessment
- Accreditation/certification
- Interview expectations
- Benefits/outcomes of Skills Bootcamp

Screening/wraparound support for retention

We will offer support for all Skills Bootcamps, with 100% of learners progressing into employment.

Before

- Screening to ensure Skills Bootcamp appropriate:

 Identify previous learning issues/barriers
 - \circ Industry explanation \circ Outcomes
 - required \circ Progression routes \circ
 - Expectations of Skills Bootcamp
 - Assess Right to Work
 - Ensure only take one Skills Bootcamp per 12 months
 - Unsuccessful applicants signposted to alternative provision **During**:
 - Support from Trainer
 - Pastoral support
 - Safeguarding Lead
 - Individual Learning Plans
 - Mentoring/coaching
 - Employer meetings/industry talks
 - Soft skills/behaviours required for job After:
 - Exit interview
 - Contacts to support employment/progression
 - 3/6/9/12-month check-ins
 - Workplace mentor
 - Support to attend interview
 - Post interview if unsuccessful
 - Employer support for induction
 - Prisoners: Upon release, help with housing/probation etc contact within 24

hours

Self-employed - check-ins to offer support/guidance/career information

Learner support for positive outcomes

Our Skills Bootcamps are designed using skills shortage vacancies (SSV) data and employer skills/vacancy needs, so we confirm that 100% of learners are guaranteed an interview for a role matching the skills acquired.

Learners connect with their future employer from the start. Employers carry out early interviews to understand learners' goals and learners can ask questions, ensuring both learner/employer are comfortable.

Our LET supports learners with CVs/interview skills - we offer 10-min speed interviews with local employers to practise techniques.



DfE Skills Bootcamps Dynamic Purchasing System

1

Document 5c. Technical Response Template

Question 3

Important Note to Potential Suppliers: Please refer to Document 5 Technical Response Instructions for full information on the question requirements.

QUESTION 3 – Employer Engagement

3.1 Describe how you will:

- engage employers to ensure that the training is targeted to meet employer needs,
- ensure employers remain engaged with the Skills Bootcamp throughout delivery, and
- secure learners the offer of an interview and support them to achieve positive outcomes.

3.2 Provide contact details for <u>3 employers</u> that you are working with on the end-to-end design/development and delivery of the proposed Skills Bootcamp; and who intend to use the Skills Bootcamp to fill specific vacancies within their organisation. You must provide details for each procurement category you are bidding into.

Where you are an employer using a Skills Bootcamp to train your own employees, you are only required to provide your own employer contact details, making it clear that you are an employer supplier.

Response requirements:

Describe how you will work with employers to:

- Ensure you have identified sufficient employers/opportunities according to your learner numbers and how you will develop your Skills Bootcamps in line with specific needs to fill vacancies/roles in the organisations you are working with, including understanding what technical and soft skills recruiters are looking for in candidates.
- Ensure ongoing employer/organisational engagement throughout the lifetime of the Skills Bootcamp and give examples of the form this will take, including how you will engage Small and Medium Enterprises (SMEs) and / or SMEs in supply chains.
- Secure the offer of an interview for live vacancies, the offer of a new role or responsibilities, or new contract opportunities, relevant to the skills gained within the Skills Bootcamp.

Contact details for employer(s) (or your own organisation, in the case where you are an employer) should include:

- Name of employer organisation and contact
- The Skills Bootcamp category and subcategory this employer engagement is related to
- Employer email address
- Employer telephone number

Working with the employers listed in 3.2 will be condition of the contract, with any variations subject to DfE approval at contract mobilisation.

Maximum characters: 3750 (employer contact details <u>not</u> included in maximum character count)

Weighting – 20%

Please write your response to this question in this Word document template and upload it to the relevant question answer area in the Jaggaer e-Sourcing Portal within the Technical Envelope.

Please ensure your document response is named as per the response naming guidance in Document 1 Procurement Document an Invitation to Participate.

3735 characters

Sufficient employers/opportunities

Intertrain is part of the City & Guilds (C&G) Delivery Hub, combining the delivery expertise/resources of all C&G businesses, with a network of 2,150 rail/construction/engineering employers, and a subcontractor network of 1,000+ employers.

Our approach starts with identifying priority skills shortage vacancies (SSV), then targeting employers based on size/sector. We design Skills Bootcamps alongside employers and subcontractors to provide learners with the knowledge, skills and behaviours sought by employers/recruiters. DfE can be assured that all designs are based on insight into technical/soft skills shortages, guaranteeing positive outcomes:

- **Employer demand**: 16 employers have confirmed SSV to match the 1,740 learner places in our proposal, guaranteeing job outcomes/progression. We will continually monitor/prioritise demand, including monthly reviews
- Job postings: We engaged Lightcast, providers of information on occupations/skills demand/career pathways, to identify job postings specific to Skills Bootcamps within this category, showing huge demand e.g:
- **Rail Lineside Management**: This skill has grown in demand by 242% in relative frequency to all job postings, a huge increase, especially in South East/Yorkshire and Humber
- Rail Overhead Lines: This has increased by 67%, with shortages in London/South East /East
- Our market analysis: Our survey of 400 L&D managers outside our customer base measured awareness of/appetite for Skills Bootcamps:

 36% plan to deliver Skills Bootcamps in next 2 years

 $\,\circ\,$ 47% very likely to consider funded Skills Bootcamps – a chance to drive uptake

Industry insight

 Rail skills shortages - 120,000 more required in next 5-10 years
 Continued investment - £46bn rail infrastructure/£75bn HS2 over 10 years

Ensuring ongoing employer engagement

We meet monthly with each employer to review progress/identify improvements. **We actively engage them in delivery** so learners have full expectations of the job/market, preparing for transition to job/role:

- Induction from employers:

 Set expectations about job/market
 Is the industry suitable for learner?
 Importance of behavioural skills as well as technical
- Employer attendance at screening days
- Inviting learners to view their premises
- Delivery of sessions, where possible
- On-site observations of delivery
- MS Teams sessions
- Employer involvement in supporting those who are struggling

Engaging with SMEs

SMEs form the backbone of our supply chain - Wave 2 (W2) extension, 86% of employers were SME and 77% in Wave 3. We have engaged with 650+ SMEs as well as the self-employed.

We understand SME challenges and how to tailor support e.g. workforce planning to identify skills shortages. We match attendees to SSV so our designs reflect the unique requirements of smaller organisations, as well as large. Each SME has an Account Manager to discuss training, vacancies and support.

Guaranteed interviews and job offers

We ensure learners connect with future employers at the start - employers conduct interviews to understand learners' goals, and for learners to ensure the role is right. As our Skills Bootcamps are designed using SSV/employer skills/vacancy needs, we confirm that 100% are guaranteed an interview for a role matching their new skills.

Outcomes for W2/W2 extension from the C&G Delivery Hub are shown below (*we are collecting data until Sept 23 for W3).

	W2	W2 Ext*
Number of learners completed	233	258
% job interviews for learners	100%	100%
% successful outcomes within 6 months	93%	98%
Retention	100%	100%

For the self-employed, upon completion, our Contracts and Partnership Team contacts them for written confirmation on how the Skills Bootcamp has supported new opportunities.

Sub Category	Employer 1	Employer 2	Employer 3
Manufacturi ng	Name of employer organisation and contact:	Name of employer organisation and contact:	Name of employer organisation and contact:
	The Skills Bootcamp category and subcategory this employer engagement is related to: Category 2 - Engineering & Manufacturing, Engineering & Manufacturing (General) Employer email address:	The Skills Bootcamp category and subcategory this employer engagement is related to: Category 2 - Engineering & Manufacturing, Engineering & Manufacturing (General) Employer email address:	The Skills Bootcamp category and subcategory this employer engagement related to: Category 2 - Engineering & Manufacturing, Engineering & Manufacturing (General) Employer email address: Employer telephone number:

Manufacturi ng	The Skills Bootcamp category and subcategory this employer engagement is related to: Category 2 - Engineering & Manufacturing, Pathway to Accelerated Apprenticeships in Engineering & Manufacturing Employer email address:	The Skills Bootcamp category and subcategory this employer engagement is related to: Category 2 - Engineering & Manufacturing, Pathway to Accelerated Apprenticeships in Engineering & Manufacturing Employer email address:	The Skills Bootcamp category and subcategory this employer engagement is related to: Category 2 - Engineering & Manufacturing, Pathway to Accelerated Apprenticeships in Engineering & Manufacturing Employer email address:
	Employer telephone	Employer telephone	Employer telephone
	number:	number:	number:



DfE Skills Bootcamps Dynamic Purchasing System

1

Document 5d. Technical Response Template

Question 4

Important Note to Potential Suppliers: Please refer to Document 5 Technical Response Instructions for full information on the question requirements.

QUESTION 4 – Resources and Delivery

Describe how you will ensure that you have the right resources and systems in place to deliver successful Skills Bootcamps.

Response requirements:

Describe how you will use data/information to inform your training course(s) and learner numbers by describing how:

- The human resources and equipment that you will use will be sufficient and appropriate to successfully deliver the Skills Bootcamp provision for all learners including supporting data and evidencing requirements.
- Your proposal aligns with and does not duplicate the skills priorities and delivery of Mayoral Combined Authorities/Local Authorities and/or Local Enterprise Partnership(s) and their Skills Boards or comparable bodies.
- You will ensure that training is delivered by suitable, experienced, and appropriately qualified practitioners, and that you have sufficient trainer resource in place to effectively monitor and promote learner progress through all your proposed Skills Bootcamps.

Maximum characters: 3750

Weighting – 20% (Where Pathway to Accelerated Apprenticeships delivery taking place – 15%)

Please write your response to this question in this Word document template and upload it to the relevant question answer area in the Jaggaer e-Sourcing Portal within the Technical Envelope.

Please ensure your document response is named as per the response naming guidance in Document 1 Procurement Document an Invitation to Participate.

3744 characters

Intertrain is part of the City & Guilds (C&G) Delivery Hub, combining the delivery expertise/resources of all C&G businesses, with 20+ years' experience delivering specialist training, assessments, mentoring and vocational programmes, including Apprenticeships, Study/AEB programmes to support the unemployed. The C&G Delivery Hub:

- Is currently managing £36m of Apprenticeship provision over several years
- Has delivered £1.5m of Wave 2 (W2) +W2 extension provision

We have previously delivered 23 different Skills Bootcamps across several sectors. We will use this experience to monitor and evaluate/improve performance and ensure we have the correct resources in place to deliver further Skills Bootcamps.

Human resources/equipment

Together with our subcontractors, we will provide all Skills Bootcamp resources and equipment. Most sessions will be delivered by the C&G Delivery hub team of 200+ qualified Trainers at one of 20 C&G training centres or at a partner's centre. Resources include:

- Classrooms, simulated work environments, catering facilities
- Personal Protective Equipment
- Industry equipment, e.g. welding
- Learning resources- lesson plans/training materials from accreditation bodies
- IT equipment

We will manage all requests, book trainers/rooms and deal with enquiries, using our booking system. We conduct weekly resource planning of our pool of qualified trainers- using a supplier pool as contingency.

Supporting data/evidence

Having delivered highly successful Skills Bootcamps in W2, we received a W2 extension, Wave 3 (W3) and W3 extension. Our delivery has expanded, and our approach has matured into a streamlined, risk-free model.

	W2	W2 Ext*
Number of learners completed	233	258
% job interviews for learners	100%	100%
% successful outcomes within 6 months	93%	98%
Retention	100%	100%

*We have until Sept 2023 to obtain successful outcomes. (W3/W3 Extension data not available - training still in progress.)

Collaboration to align/prevent duplication

Links with local bodies in SCR/WMCA/GMCA/GLA mean we are updated on skills needs and not competing for the same employers/vacancies. We will align provision through:

- A national communications initiative with LEPs/LAs/MCAs, Skills Boards and partners to notify of our Skills Bootcamps activity
- Roundtables, project meetings, employer events and local/national skills agenda events
- Working nationally with local bodies regarding their skills/training requirements and planned procurement/funding allocation To prevent duplication:
- Planning/liaising with employers at national level

- Consulting Skills for Life website
- Gathering employer expressions of interest, confirming no duplication

Suitably experienced/qualified practitioners

We have more than enough suitably qualified/experienced practitioners to begin delivery from 2024. Learner progress will be monitored and promoted by ensuring all Trainers:

- Hold teaching/assessing qualifications
- Have current industry knowledge/competence
- Are safeguarding, PREVENT trained
- Complete 30 hours CPD annually, and attend at least 3 CPD events
- Conduct annual online exams/attend Sentinel courses
- Are assessed against robust quality criteria
- Are regularly observed in delivery, with 1-2-1 feedback on style/content/engagement
- Are internally audited to verify learning/assessment
- Share best practice through peer review, IQA strategy and Assessor standardisation

Our qualified Trainers equip learners with the skills and qualifications to achieve sustainable employment and career progression, ensuring learners are work ready.

Our Head of Quality & Compliance will monitor teaching, and our observation strategy is in line with Ofsted. Our Contracts and Partnership Manager will ensure subcontractors deliver to standard.



DfE Skills Bootcamps Dynamic Purchasing System

1

Document 5e. Technical Response Template

Question 5

Important Note to Potential Suppliers: Please refer to Document 5 Technical Response Instructions for full information on the question requirements.

QUESTION 5 - Quality Assurance & Continuous Improvement

Describe your quality assurance, continuous improvement, and risk management processes to ensure that you deliver a high-quality service. Include how you will:

- collect,
- manage,
- report, and
- use data to evidence and improve your own performance and meet the Department for Education's data requirements and how you will manage and mitigate risk. **Response requirements:**

Describe your approach to quality assurance and continuous improvement processes, including how you will:

- Ensure that all required data, including from subcontractors, is collected, and reported to the Department for Education, and ensure reporting is timely, accurate, and complete.
- Manage the key risks to delivery and what mitigations you propose to undertake in response to those risks.
- Implement robust and auditable procedures for logging, managing, escalating, and resolving issues and complaints.
- Use your data to evidence, review and improve your performance (e.g., learner dropouts, completions, outcomes, quality of training etc).

Maximum characters: 2500

Weighting – 10%

Please write your response to this question in this Word document template and upload it to the relevant question answer area in the Jaggaer e-Sourcing Portal within the Technical Envelope.

Please ensure your document response is named as per the response naming guidance in Document 1 Procurement Document an Invitation to Participate. 2491 characters

Intertrain is part of the City & Guilds Delivery Hub, combining the delivery expertise/resources of all C&G businesses. The Hub uses ISO9001 Quality Management System (QMS) and Quality Assurance (QA) according to ESFA/DfE Ofsted (rated good), NSAR, Apprenticeship Standards, reinforced by C&G's QA:

- Delivery observations with 121 feedback
- Internal learning/assessment audits
- Best practice shared via peer review, IQA strategy etc
- Stakeholder feedback

Collecting/Reporting Data

We have extensive experience collecting/managing/reporting data. Our MI System (in place since 2017), and MIS/Funding & Compliance team will:

- Collect learner/employer data to meet DfE's needs (ID/Right to Work, Companies House/credit)
- Submit accurate/timely ILR Return in line with ILR Freeze Schedule, using Funding Information Service/Provider Data Self-Assessment Toolkit to ensure zero funding errors on FRM reports
- Submit monthly reports to Senior Leadership to achieve best outcomes/completions
- Follow internal MI audit calendar for ILR/learner file reviews to ensure accuracy

We have zero service failures for late ILR returns, DfE audited with zero errors (quality/accuracy).

Subcontractor Quality

Subcontractors managed via monthly meetings, reviews, quality/compliance checks and robust SLAs. Our CPT will:

- Select/onboard, ensuring licensed under relevant Awarding Body
- Monitor subcontractor quality/report KPIs
- Hold monthly contract meetings **Risks**

A risk log is updated weekly, reviewed at contract/Governance meetings and shared with DfE. Typical risks:

Session cancelled at short notice

- Large/multi-skilled team staff gives cover
- 3-tier subcontractor system gives contingency
- Possibly re-book on open course

High drop-outs

- Clarify expectations
- Pastoral support/121s/IAG

Subcontractor underperformance

- Robust subcontractor policy
- Oversight by CPT in line with QA/Quality Management Framework

Risks on QMS registers are audited annually. *We hold an ESFA low-risk rating, with zero error ratings for eligibility.*

Resolving complaints

We will log, manage, escalate and resolve issues/complaints in line with DfE requirements.

Data and Continuous Improvement (CI)

We implement CI through QA, self-assessment, employer/learner surveys, formative/summative assessment and monthly employer meetings, reporting against KPIs. Information is entered into our QI Plan, raised at monthly meetings and fed into annual SARs. informing our QI strategy. CI reporting format/frequency to be agreed with DfE.



DfE Skills Bootcamps Dynamic Purchasing System

Document 5f. Technical Response Template

Question 6

Important Note to Potential Suppliers: Please refer to Document 5 Technical Response Instructions for full information on the question requirements.

QUESTION 6 - Social Value [Social value theme 2 – Tackling Economic Inequality]

Please outline what positive commitments you, as a Potential Supplier, can make to increasing social value and reducing economic inequality **within your workforce.** Please outline how you would engage with stakeholders, measure, and monitor progress, and what innovative, sustainable approaches in the form of improved tools, techniques, and technologies you would deploy.

For the avoidance of doubt your response must make clear how the measures you are proposing are <u>in addition to the core delivery activity</u> that you are proposing to meet the requirements of the Specification.

Response requirements:

In answering the above question, please include in your response <u>all</u> the following elements:

- Enabling or promoting access by **your contract workforce** to apprenticeships, T Level industry placements or other learning and development opportunities at level 2, 3, and 4+ or equivalent).
- Other activities to support relevant sector related skills growth and sustainability **in the contract workforce** (illustrative examples: careers talks, curriculum support, literacy support, safety talks and volunteering).

Since 2013, the Social Value Act has required social value to be considered in the design of contracts, with government departments expected to report on the social impact of their major contracts. To facilitate this, we have included a specific social value question. Social Value is the additional value derived from the delivery of a contract where a wider community or public benefit can be created, for example to your own staff or to your local community. For Skills Bootcamps it is important to note social value cannot include the services already being delivered to learners as outlined in the Specification. Your response must therefore make clear how the measures you are proposing are in addition to meeting the requirements of the Specification. The scoring of this social value question is based on narrative responses from bidders and not on volumes.

The Social Value Act 2012 calls for all <u>public sector</u> commissioning to "have regard to" economic, social, and environmental well-being in connection with public services contracts; and for connected purposes. It requires that all public bodies in <u>England and Wales</u>, to consider how the services they <u>commission and procure</u> which are expected to cost more than the <u>thresholds</u> provided for in the Public Contracts Regulations might improve the social, economic and <u>environmental well-being</u> of the area.

Maximum characters: 1000

Weighting: 10%

Please write your response to this question in this Word document template and upload it to the relevant question answer area in the Jaggaer e-Sourcing Portal within the Technical Envelope.

Please ensure your document response is named as per the response naming guidance in Document 1 Procurement Document an Invitation to Participate.

1000 characters

In addition to core delivery, we will increase social value/reduce economic inequality within our workforce through:

Apprenticeships: We will enroll 2 Apprentices (Levels 1-4) to work across Skills Bootcamps, recruiting from economically disadvantaged areas/groups to give opportunities for a career/grow in a constructive, safe, supportive environment

Learning and development: We will:

- Host 8 CPD training days for engineering Trainers ensuring skills/knowledge progression, with access to 121/peer coaching to boost confidence and stimulate personal/career growth
- Enable staff to volunteer locally, using skills/experience to connect with communities we serve, developing interpersonal skills and sense of fulfilment

Also, as part of City & Guilds (a UK charity), any surplus we make from Skills Bootcamps will be reinvested in developing work-ready school leavers/inspiring managers/productive workforces.

At contract review meetings we will update DfE of progress in achieving our commitments.

			2: Pathway to Accelerated All sections (A, B and C) mu			
A) CONTENT MAPPING	In this section you will need to select	In this section you will need to select	Where in the Skills Bootcamp is this statement being covered?	Provide a short explanation on what you would expect most learners to be able to DO or KNOW	B) ADDITIONAL CONTENT	
	the level of coverage the Skills	the level of competency. This is whether the Skills	Indicate the module/unit titles and the wee	by the end of the Skills Bootcamp in relation to ek the linked KSB statement and how this will be		
	Bootcamp will provide against each KSB (not	Bootcamp will	of delivery.	further developed in the apprenticeship to reach full competency.		
	covered, partially covered or fully	with full or entry level occupational				
	covered).	competence for the linked KS8.			As well as specific content that aligns to an occupational star relevant content which is de	dard, Skills Bootcamps can include additional occupations amed of value to employers.
n this section please copy and paste the KSB's for the aligned occupational standard.					This may include wraparound support, work readiness and additi	onal content that supports greens skills/specific employer
Mapping to:	Please select one option from the drop down list.	Please select one option from the drop down list.	You only need to provide evidence of delivery for those statements where you have indicated coverage.	You only need to provide expected outcomes information for those statements where you have indicated coverage.	Please list any additional content beyond the KS8s being delivered .	Please provide a rationale for the inclusion of this conte how it will compliment the KSBs.
AAL ENGNEERING TECHNICIAN LVEL3- https://www.instituteforapprenticeships.org/apprenticeship-standards/rail engineering-technician-v1-1			50 words maximum per entry	100 words maximum per entry		100 words maximum per entry Bationale
inowledge afe and Professional working practices including legislation, regulation, industry procedures and safety	KS8 Coverage	KS8 Competency	W18W2 Safety Awareness, Manual	Expected Outcomes Skills Bootcamp	Additional Content Basic Signalling & Track Awareness	This session will provide an introduction to other pathw
equirements.			Handling, Fire Awareness, Emergency First Aid at Work, PTS E-Learning & Prep Day, PT AC DCCR Working at Height and Hamess	This Skills Bootcamp will provide the learner S with a full NWR Sentinel based competence for PTS AC which enables them to work on the rail		the industry which will enable a learner to understand h the railway operates, the dangers of other elements, an compliment the development of various KSB's.
			Training, Behaviours, Safety	infrastructure. The learner will also be awarded full competence in Emergency First Aid at Work,		compriment the best operant of vertices rule 2.
				Working at Heights & Harness Training. Learners will also be provided with the necessary		
				knowledge in all other training listed in which they will develop further in their apprenticeship. Accelerated Apprenticeship		
				Learners will practice their licence to practice NWR Sentinel based competency, along with		
				further developing their knowledge in all other training listed through their qualifications and employment.		
				employment.		
The scientific, technical, engineering, mathematical and design principles that are required to support the	Partially covered	Full level competency	W3&W4 Design, Fabrication, Assurance, Pi	e Skills Bootcamp	Emergency Response	Emergency response is a vital element in understanding
naintenance, renewal and construction of The Rail/way.			Setting, Topo Setting, OLE Bonding	The Skills Bootcamp will provide an introduction to Design, Fabrication, Assurance, Pile Setting, Topo Setting, OLE Bonding in which they will		action to take in the event of injury, damage or loss of of which form part of the safety culture within the inde
				Topo Setting, OLE Bonding in which they will build on this knowledge in their apprenticeship. Accelerated Apprenticeship		and provides the learner the opportunity to develop th behaviours.
				Learners develop their knowledge on Design, Fabrication, Assurance, Pile Setting, Topo		
				Setting, OLE Bonding, but also carry out various training and competencies, along with industry		
				experience with regards to scientific, technical, engineering, mathematical and design principles.		
	Partially covered	Entry level competency				
How to work effectively and contribute to engineering solutions and innovation including understanding and applying problem solving techniques and diagnostics, project planning and management, time management and autility management and assume switems.	Parceny covered	competency		These skills will be developed once in employment.	Environment	The session will compliment the knowledge required w working within the industry to reduce the impact that and jobs have on the environment. This will also provide
daman di suna additi suna ana ana ana adaman an	Not covered					and jobs have on the environment. This will also provi introduction to an element of their knowledge qualific
The importance of 3rd party requirements and client confidentiality and the need to understand and adhere to How the railway works commercially including contractual principles and implications.	Not covered			These skills will be developed once in These skills will be developed once in employment.		
					C) ACCELEBATED APPRENTICESHIP QUESTIONS Provide the typical duration and price of the ap	neatleachin ar nar MATE's accumational many
					Typical duration to gateway (full apprenticeship)	30 months
					Typical price of the apprenticeship (full apprenticeship)	£24,000.00
					State the impact Prior Learning from the Skills Bootcan	p will have on the apprenticeship duration and price. 27 months
					Estimated duration to gateway (accelerated apprenticeship)	£21,600.00
Skills					Estimated price of the apprenticeship (acclerated apprenticeship) Can you confirm that the Skills Bootcamp will provide a route requirement of the ac	to an accelerated apprenticeship, whilst ensuring the
Please copy all the skills statements from the IRATE website two themesives and others refer to adhering to refer undring matrices. Understand and comply with statistical	Partially covered	full level	W18W2 Selatu Augusta Ere Aug	Skille Bootramo	requirements of the ap Minimum 20% off the job training	prenticeship are met? Yes
Keep themselves and others safe by adhering to safe working practices. Understand and comply with statutory regulations and organisational safety requirements, including safe access to work locations.	Fartially covered	Full level competency	W18W2 Safety Awareness, Fire Awareness Emergency First Aid at Work, PTS E-Learnin & Prep Day, PTS AC, Working at Height and	g This Skills Bootcamp will provide the learner		
			Harness Training, Behaviours, Safety	PTS AC which enables them to work on the rail infrastructure. The learner will also be awarded		
				full competence in Emergency First Aid at Work, Working at Heights & Harness Training. Learners will also be provided with the necessary skills in		
				all other training listed in which they will develop further in their apprenticeship.		
				Accelerated Apprenticeship Learners will practice their licence to practice		
				NWR Sentinel based competency, along with further developing their skills in all other training listed through their qualifications and		
				employment.		
					Minimum 12 months duration	Yes
Plan a high standard of technical work: Gathering and interpreting information including drawings, plans, schedular needed for the development of rall engineering activity planning. Detailed inspection and performance 8 condition analysis of sausts, Han work to be undertaken including the appropriate resources.	Not covered			These skills will be developed once in employment.		
силалонт агаприя от азакся, тактички со не опнессаниет пслокещ сте арреоргание горонска.						
					Any requirements specific to the occupational standard	Yes
Deliver a high standard of technical work: Undertake engineering activities in relation to maintenance, construction / installation and or renewal of assets. Complete integrity & compliance checks on own work, instigate testing and identify where independent testing is required. Transfer responsibility of assets once work has	Not covered			These skills will be developed once in employment.		
instigate testing and identify where independent testing is required. Transfer responsibility of assets once work has been completed. Supervise their own work and that of others.						
Solve problems: identify problems and apply a structured approach and appropriate methods to problem solving	Partially covered	Full level	W3 OLEC2	Skills Bootcamp	Please explain your rationale for why you have chosen to cove in the expected amount of reduction on t The KSB's covered in the Skills Bootcamp have been selected as th	he apprenticeship (100 word maximum).
sove process. Identity process and apply a solucioned approach and appropriate memory to process solving and diagnosis.		competency		Learners will gain full NWR Sentinel based competencies for OLEC2 which includes	apprentice in so they can work in industry, particularly in the lice	se to practice elements. All other training selected pro-
				elements of problem solving and diagnosis and can support the learner in gaining employment in the industry.	as rail engineering technicians on the infrastructure. A vast major apprenticeship is aimed at KSB's, which makes the reduction of th	rty or the training provided through the Skills Bootcamp e apprenticeship justifiable.
				in the industry. Accelerated Apprenticeship The full NWR Sentinel based competency will be		
				used to enable to apprentice to fulfil their role in the industry and they will be able to further develop their skills.		
				www.chupi.undir.skins.		
Anage resources including the correct utilisation and storage of tools, materials and equipment, and the lifting of impulses of materials, commenters and equipment.	Partially covered	Full level	W2-W4 OLEC2, Cembre Rail Drill, MAG Drill	Skills Bootcamp		
nd moving of materials, components and equipment.		competency	Incl. Use of Cutting Fluid, Fixed Drill / Battery Drill	Learners will gain full NWR Sentinel based competencies for OLEC2 which includes elements of managing resources and materials		
				elements of managing resources and materials and can support the learner in gaining employment in the industry, Learners will also		
				operate and store tools and equipment correctly as part of the small plant training. Accelerated Apprenticeship		
				The full NWR Sentinel based competency will be used to enable to apprentice to fulfil their role in		
				the industry and they will be able to further develop their skills.	Explain how you will support learners to ensure Prior Learning	from the Skills Bootrame is recognized in shale
					Intial Assessment. This should also address where the Skills Bo maxin	otcamp provider is not the apprenticeship provider (I sum).
Communicate effectively. Use orall written, electronic and IT based methods and systems for the accurate	Partially covered	Full level	W2-W4 OLEC2. Isolations incl. Form C /	Skills Bootcamo	Intertrain's current skills scan assessment identifies learner's curr Intial Assessment. This should also address where the Skills Boot maximum).	ent knowledge skills and behaviours against the apprent camp provider is not the apprenticeship provider (1
Communicate effectively. Use oral, written, electronic and IT based methods and systems for the accurate I	Partially covered	Full level	V2-W4 OLEC2, Isolations incl. Form C / S	kills Bootcamp	Intertrain's current skills scan assessment identifies learner's curre	
Lommunicate effectively. Use oral, written, electronic and Li based methods and systems for the accurate is communication, reporting & recording of technical and other information, using correct terms, standards, iemplates and certifications.		competency F	orm B, Employability Skills, Employer L ingagement c	earners will gain full NWR Sentinel based ompetencies for OLEC2 which includes	apprenticeship standard to ensure that appropriate starting points designed to suit the employer's and the apprentice's needs. The p	are established, and a tailored training plan is for learning that learners will gain completing the
			e t	elements of communication and can support he learner in gaining employment in the	Skills Bootcamp will allow apprentices completing their apprentice	ship programme in a shorter time.
			c	ndustry. Learners will also create and brief ithers on the Form C / Form B requirements. earners will learn interview skills and		
			د م	ommunicate with employers. Accelerated Apprenticeship		
			1	he full NWR Sentinel based competency will we used to enable to apprentice to fulfil their		
			r f	ole in the industry and they will be able to urther develop their skills.		

	Specific browkedge & Stats Derived Lines. Derived Lines. Derived Lines. Derived Lines. Derived Lines. Derived Lines Lines and Derived Lines Lines Lines and Derived Lines Lines and Derived Lines Lines Lines Lines Lines and Derived Lines Lines Lines Lines Lines Lines and Derived Lines Line	Partially covered	competency	Harness Training, Fabrication, Design, Assurance, Pile Setting, Topo Setting, OLE Bonding	Stills Rotzamp Larners still gain kill NWR Sentterb based competencies for UEL2 and GALE2 and gain specific trowing and stills in various and will support them in becoming more emphasize. Acaderated Approxticably The Kill NVR Stimuto Based centers to hild their order the industry and they will be abit on further develop that stills in various agatest of GLE covered in the other training.		
	Behaviours						
ļ	At professionally demonstrating dependebility, determination, honesty and integrity. Respect others, act ethically and contribute to sustainable development.	Partially covered	Entry level competency		Skills Bootcamp Learners will carry out Behavioural training which includes free to be Safe culture, Situational & inclusion / Awareness & Change. This training will be added value to employers air torate a culture of safety and employens who are more safety function of the safety of the safety of the Learners will develop their behaviours through employment and through their industry.		
	lie risk awar so as to reduce risk through systematic monitoring and checking of information, concentration on the task, and awareness of changing circumstances on activity.	Partially covered	Full level		Skills addrama National Starters with full NNR Sentine Ibards competencies in PTS AC and OLCE, insurant but the lybo and the required table address and the library of the library of the starters and the library of the library of the Starters and the library of the library of the Starters and Starters and uncertain Starters and Starters and uncertain Starters and Starters and uncertain Starters and Starters and uncertain Concentration and Starters and uncertain Concentration and Starters and uncertain Ibrary TSA and Starters and uncertain Concentration and the starters and uncertaint Resentation and the starters and uncertaint Resentation and the starters and the starters and the requirements. Learners will develop their bahavious through the relaxing as this Lembedded through the training as this lembedded through the starters and starters and starters and starters and the starters and starters and starters and the starters and th		
	logiage and disciplined, and mathematically providing approach to work, able to make independent decisions whits twowing only limitations and when to ask for help or to excitate.	Partially covered		Incidents / Awareness & Change)	Salis Bootcamp Latence will carry set Belowiourd transing Statuscular Microsoft I and Salis Carlos, Salistancia Microsoft I Anameens & Change. The training will be added value to employers ail creates a culture of allely and employers who are more safety conscious. Accelerate a culture of allely Latences will develop their behaviours through employment and through their training as that is embedded throughout the holdry.		
	Mon relativ and safely, often without close supervision, to approved industry standards and safe working		competency		Learners will carry out Behavioural training which hickels for the basic culture. Charge the basic culture is the basic culture culture is the standard of the standard of the prophysers at it creates a culture of asien to prophysers at it creates a culture of asien to and employees who are more safely conscious. Accelerates a culture of asien to provide the standard of the standard hange employment at charge their behaviours through employment ad charge their training as this is embedded throughout the abative.		
	Wook effectively and efficiently, individually and as part of a team, maintaining effective relationships with colleagues, clients, supplers and the public.		competency	incidents / Awareness & Change)	Stills Bockcamp Learners will carry out. Behavioural training which includes Free to be Safe culture, Statutional & includes / Awareners & Change. The training will be added value to employers ait crosses a culture of safety and employers who are more safety concluse. Accedence AppendiceAhy Learners will develop their behaviours through employment and through their training as this is embedded throughout the industry.		
	Receptor to feedback, willing to kenn new skills and solgut to change. Identifying, canying out and recording CPD neexsary to maintain and enhance competence.	Partially covered	Entry level competency	Employability Skills	Skills sockamp The Skills kockamp offers employability this is not writing, interview skills and CPO, interview skills and CPO, interview skills and CPO, interview skills, skills, skills, skills, interview skills, skills, skills, skills, skills, interview skills, skills, skills, skills, skills, interview skills, skillskills,		

Prepared to make a personal commitment to their employer, the industry and its professional standards.	Partially covered	Entry level competency	WLIAWS Behaviour (ITIS / Staubiona I indicenter, Javaneses & Canage), Employability Skills, Employer Engagement	The skills bootcamp offers employability skills in CV writer, interview skills and CPD, which will support a learner will gaining employment in the industry, Learners will carry out Behavioural taining witch includes the to be skill extrum. Skuustonal & Indident / Awareness & Change. This is a creater a culture of skilly and employees who are more safety conscious. Accelerated Appenticably Learners will develop their behaviours through employment and through their			
				through employment and through their training as this is embedded throughout the industry.			
2: Pathway to Accelerated Apprenticeships All sections (A, B and C) must be completed.							

Mapping ter: Mare indicated coverage. Name indicated coverage. Mare indicated cove	Normal problem Normal proble					t be completed.		
Non- transmission Non- status Non- status <t< td=""><td>Normal problem Normal proble</td><td></td><td></td><td></td><td></td><td></td><td>ELADOTIONAL CONTEXT</td><td></td></t<>	Normal problem Normal proble						ELADOTIONAL CONTEXT	
Normalization Norm	Normalization Norm	A) CONTENT MAPPING	In this section you	in this section you	Where in the Skills Rootcamo is this	Provide a short explanation on what you would	B) ADDITIONAL CONTENT	
Note in the second s	Note of the second s					expect most learners to be able to DO or KNOW		
Notes Notes Name <	Normal problem Name Nam				Indicate the module/unit titles and the week	by the end of the Skills Bootcamp in relation to the linked KSB statement and how this will be		
Name Nam Name Name Name	Name		Bootcamp will	whether the Skills		further developed in the apprenticeship to reach		
Mark	Note of a constraint of a con		provide against	Bootcamp will		full competency.		
Market Barbard calabiant particular set and set a	Mark ControlMark		covered, partially	with full or entry				
Note of the sector of the	Note of the second		covered or fully					
And Add Add and and add Add Add Add Add Add Add Add Add Ad	And Add and ang and bill of Link and ang and and ang and ang		coresta.	linked KSB.			As well as specific content that aligns to an occupational stand relevant content which is deep	ard, Skills Bootcamps can include additional occupationally ned of value to employers
Sector Sector <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>								
Land DescriptionMarket is ParticularMarket is ParticularMarket is ParticularMarket is ParticularMarket is ParticularMarket is 	Image: product of the state	In this section please copy and paste the KSB's for the aligned occupational standard.					This may include wraparound support, work readiness and addition	al content that supports greens skills/specific employer nee
Land DescriptionMarket is ParticularMarket is Particu	Image: product of the state		Please select one	Please salert con	You only need to provide evideore of	You only need to provide experted outcomes		
Markamental Status S	Markamanuscular interventional and an analysis of the second s		option from the	option from the	delivery for those statements where you	information for those statements where you	Please list any additional content beyond the KSBs being	Please provide a rationale for the inclusion of this content a
ranze and set of the s	Particuluit Image Image <td>Mapping to: RAIL ENGINEERING TECHNICIAN LVEL 3- https://www.instituteforapprenticeships.org/apprenticeship-standards/rail</td> <td>drop down list.</td> <td>drop down list.</td> <td>have indicated coverage.</td> <td>have indicated coverage.</td> <td>delivered .</td> <td>how it will compliment the KSBs.</td>	Mapping to: RAIL ENGINEERING TECHNICIAN LVEL 3- https://www.instituteforapprenticeships.org/apprenticeship-standards/rail	drop down list.	drop down list.	have indicated coverage.	have indicated coverage.	delivered .	how it will compliment the KSBs.
Mathematical and the second	Mathematical constraints Marka Marka Marka Marka Marka Marka Marka Marka Shares Shar	engineering-technician-v1-1						
Mathematical and the second	Mathematical constraints Marka Marka Marka Marka Marka Marka Marka Marka Shares Shar							
Manute Manute Market Market Market Sample description Res Res Res Res Res Sample description Res Res Res Res Res Sample description Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res Res	Mathematical experimentation of the second of the secon							
		VCB Externat	Kill Courses	VER Competence	50 words maximum per entry	100 words maximum per entry	Additional Contract	100 words maximum per entry
Automatical and a set of a set	Autom Autom Autom Subscription	Knowledge						nectorial.
Image: Section of Section 2 and Section 2	Image: Section of the section of t	Safe and Professional working practices including legislation, regulation, industry procedures and safety requirements.	Partially covered	Full level competency	Fire Awareness, Emergency First Aid at	Will provide the learner with a full NWR Sentinel	Emergency Response	action to take in the event of injury, damage or loss of life.
Image: Section of the section of t	Jail 243 Jail 244 Jail				Work, PTS E-Learning & Prep Day, PTS AC,	based competence for PTS AC which enables		of which form part of the safety culture within the industry
Augusta	Name Name Problem Structure Problem S				Training, Behaviours, Safety, Basic Signalling	learner will also be awarded full competence in		
American Antipue and antipue anti	Automatical and a second se				1 (BS1), SPWEE.	Emergency First Aid at Work, Working at Heights		
And the second secon	Answer in the second					provided with the necessary knowledge in all		
Image: Section of the section of t	Automatical and a second se					further in their apprenticeship.		
Autor base and autor base aut	Augusta					Accelerated Apprenticeship		
And the state of a state of	Augusta					Learners will practice their licence to practice NWR Sentinel based competency, along with		
Name Norm	Name					further developing their knowledge in all other		
starters and administration function Since Sin	startersinterve<					employment.		
starters and administration function Since Sin	startersinterve<							
Markat was at all and during the large of the large o	manual waterial industry theoremis and is an industry is an industr							
Markat was at all and during the large of the large o	manual waterial industry theoremis and is an industry is an industr	The scientific, technical, engineering, mathematical and design principles that are required to support the	Partially covered	Entry level	W3-W4 Basic Signalling 1 (BS1)	Skills Bootcamp	Environment	The session will compliment the knowledge required when
Image: Source of the sector	Image: Section of the sectin of the section of the section of the	maintenance, renewal and construction of The Railway.		competency		The Skills bootcamp will provide an introduction		working within the industry to reduce the impact that proje
Image: Section of the sectin of the section of the section of the	Image: Section of Section Section of Section Sectin Section Section Section Section Section Section Section S					knowledge in their apprenticeship.		and jobs have on the environment. This will also provide an introduction to an element of their knowledge qualification
Image: Section of the sectin of the section of the section of th	Image: Section of the section of t					Accelerated Apprenticeship		
Image: Constraint of the section of	Image: Additional problem is and which is a problem is and which is a problem is and which is and which is and which is a problem is and which is a problem i					but also carry out various training and		
Image: Source of the standard energy of the standar	Image: Another stands and					competencies, along with industry experience with regards to scientific, technical, engineering.		
and equivale intermine the	and public pu					mathematical and design principles.		
under status data statusMarcial <t< td=""><td>uninterpretention interpret</br></br></br></br></td><td>How to work effectively and contribute to engineering solutions and innovation including understanding and</td><td>Not covered</td><td></td><td></td><td>These skills will be developed once in</td><td></td><td></td></t<>	uninterpretention interpretention interpretention interpretention interpretention interpretention interpretention interpretention interpretention interpretention interpretention interpretention interpretention 	How to work effectively and contribute to engineering solutions and innovation including understanding and	Not covered			These skills will be developed once in		
Name of the set	Name Nam Name Name Name	applying problem solving techniques and diagnostics, project planning and management, time management and quality management and assurance systems.				employment.		
Image: Control of the section of t	Image: Control of the control of t	The importance of 3rd party requirements and client confidentiality and the need to understand and adhere to						
Image: Section of the section of t	Image: Single state in the	How the railway works commercially including contractual principles and implications.	Not covered			These skills will be developed once in employment.		
Image: Section of the section of t	Image: Control of the control of t							
Image: Section of the section of t	Image: Control of the control of t							
Image: Control in the second	Index controlIndex							AND
Image: Constraint of the second se	Image: Control of the second							
Image: Constraint of the second se	Image: set of the set of							
Image: Control in the second secon	Image: Control interaction of the section of the sectin of the section of the s							
Image: Control in the second interpretation of the second	Image: Control interaction of the second interaction						State the impact Prior Learning from the Skills Bootcam	will have on the apprenticeship duration and price.
Index Index Index Index Index Index Index Note Index I	Image: Control in the section of t							27 months
In the stateInclude the stateInclude the stateInclude 	Image: Control and inclusion of the section of th						Estimated duration to gateway (accelerated apprenticeship)	F31 600 00
Interpretation of the behaviory particle. Under out out you had a constrained of the spectra of the spe	Instrument Instrum							
Interformation and density the density questers. Under gene with the density of the dens	Interformed intervention and density the density of the basis in the term intervention and density in the basis in the basis intervention and density in the basis interventin and densintervention and density in the basis intervention and	Skills					Can you confirm that the Skills Bootcamp will provide a route t	o an accelerated apprenticeship, whilst ensuring the below
Are bigs, Fride States, Beauty, Barl, S. & Bar	Are by by TA & for our performance Are by ta for our pe	Keep themselves and others safe by adhering to safe working practices. Understand and comply with statutory	Partially covered	Full level	W18W5 Safety Awareness, Fire Awareness,	Skills Bootcamp	requirements or the app	renticeship are met?
Image: Section of the finance f	Image: Description the set of the control the control the control the	regulations and organisational safety requirements, including safe access to work locations.		competency	Emergency First Aid at Work, PTS E-Learning & Pren Day, PTS &C. Working at Height and	This Skills Bootcamp will provide the learner with a full NWR Sectionel based competence for		
Interpretend property function water, faithing on property faithing on	Apple to proper to a trapper of a trapp					PTS AC which enables them to work on the rail		
Note the set of the set	Notes of particular direction in the source integration in the source integration is the source integration in the source integration is the so							
have been been been been been been been be	here and particular of reference on restrict from the service of restrict					Working at Heights & Harness Training, Learners		
Intersection Intersection <th< td=""><td>Andread Andread Andrea Andread Andread</td><td></td><td></td><td></td><td></td><td>will also be provided with the necessary skills in</td><td></td><td></td></th<>	Andread Andrea Andread Andread					will also be provided with the necessary skills in		
Interview	Internet of generative intervent of generative					all other training listed in which they will		
Autor descensing of white descension of white descense of white descension of white descension of white desce	And the second of school watch determines from which is a low ration of school watch determines from which determin					develop further in their apprenticeship.		
Name Name Name Name Name Name Name Name Name Name N	Name and protection of works during energing strategy and protection in devices on and					develop further in their apprenticeship. Accelerated Apprenticeship Learners will practice their licence to practice		
Name Image: Section of static sta	As a ph model of the development of a step opported storego provided in spectra statuting of the step opported storego provided in spectra statuting of the step opported storego provided in spectra statuting of the step opported storego provided in spectra statuting of the step opported storego provided in spectra statuting of the step opported storego provided in spectra statuting of the step opported storego provided in spectra statuting of the step opported storego provided in spectra statuting of the step opported storego provided in spectra statuting of the step opported storego provided in spectra statuting of the step opported storego provided in step opported storego provi					develop further in their apprenticeship. Accelerated Apprenticeship Learners will practice their licence to practice NVR Sentinel based competency, along with		
Nais Nationale of technical work to be exclusioned and integrating strategistical	Nie is begin staatie of technol wird in terperspectrate interviewing einformation technolog generation technolog g					develop further in their apprenticeship. Accelerated Apprenticeship Learners will prectice their licence to practice NWR Sentinel based competency, along with further developing their skills in all other training listed through their qualifications and		
his has fastable of tabulal out of baseling information including galaxies, parking leasting and leasting data tabulage data galaxies of tabulage leasting into tabulage galaxies, parking leasting data tabulage data galaxies of tabulage leasting into tabulage galaxies, parking leasting least	has has indiced a technical work in the integration including galaxies, joint, many including work in the work expension including galaxies, joint, many including work in the work expension including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration is a standard of the integration including galaxies, joint work is as work expension. The integration is a standard is the integration including galaxies, joint work is as wor					develop further in their apprenticeship. Accelerated Apprenticeship Learners will prectice their licence to practice NWR Sentinel based competency, along with further developing their skills in all other training listed through their qualifications and		
his has fastable of tabulal out of baseling information including galaxies, parking leasting and leasting data tabulage data galaxies of tabulage leasting into tabulage galaxies, parking leasting data tabulage data galaxies of tabulage leasting into tabulage galaxies, parking leasting least	has has indiced a technical work in the integration including galaxies, joint, many including work in the work expension including galaxies, joint, many including work in the work expension including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration including galaxies, joint work is as work expension. The integration is a standard of the integration including galaxies, joint work is as work expension. The integration is a standard is the integration including galaxies, joint work is as wor					develop further in their apprenticeship. Accelerated Apprenticeship Learners will prectice their licence to practice NWR Sentinel based competency, along with further developing their skills in all other training listed through their qualifications and		
independence and go faces, here work to be endersteen notable to approprint resource. Independence and go faces, here work to be endersteen notable to be properties resource. Independence and go faces, here work to be endersteen notable to be properties resource. Independence and go faces, here work to be endersteen notable to be properties resource. Independence and go faces, here work to be endersteen notable to be properties resource. Independence and go faces, here work to be endersteen notable to be properties resource. Independence and go faces, here work to be endersteen notable to be properties resource. Independence and go faces, here work to be endersteen notable to be properties resource. Independence and faces, here work to be endersteen notable to be properties resource. Independence and faces, here work to be endersteen notable to be properties resource. Independence and faces, here work to be endersteen notable to be properties resource. Independence and faces, here work to be endersteen notable to problem some response resource. Independence and faces, here work to be endersteen notable to problem some response resource and the information of the second resource faces and to be second resource faces and to be resource and the information of the second resource faces and to be resource and the information of the second resource and the second resource and the second resource and the information of the second resource and the information	indexidence analysis of same, Pier work to be evedenced to the development to be development. Development to be development. Development to be development to be development to be development t					develop further in their apprenticeship. Accelerated Apprenticeship Learners will prectice their licence to practice NWR Sentinel based competency, along with further developing their skills in all other training listed through their qualifications and	Manimum 196, of the late tradition	~
Interstanding of technical under latencing carries and sequences. Interstanding carries and sequences. Interstand	here a understand there when while on the set on the set on while on the set on while on the set on the set on while on the set on the	Plan a high standard of technical work: Castlening and Interpreting Information Including distwings, plans,	Partially covered	Entry level	WZ Diagram Reading	Gevelop futber in their apprenticelyio. Accelerated Appreciationally Learners will practice their licence to practice WMS Sectined based competency, along with futber developing their alkills in all other training liced through their qualifications and employment. Stallis Bootcamp	Minimum 20% off the job training	76
According dypermentation Index of dypermentation </td <td>According Agenetication According Agenetication In According Agenetication In In</td> <td>schedules needed for the development of rail engineering activity planning: Detailed inspection and performance &</td> <td>Partially covered</td> <td>Entry level competency</td> <td>W2 Diagram Reading</td> <td>Gevelop futbre in their apprenticable, Accelerated Appreciation(a) Learners will practice their learners by practice that the second competency, and with the second through their qualifications and employment. Skills Rootcamp Learners will be introduced to diagrams and how</td> <td>Minimum 20% of the job training</td> <td>Yes</td>	According Agenetication According Agenetication In According Agenetication In	schedules needed for the development of rail engineering activity planning: Detailed inspection and performance &	Partially covered	Entry level competency	W2 Diagram Reading	Gevelop futbre in their apprenticable, Accelerated Appreciation(a) Learners will practice their learners by practice that the second competency, and with the second through their qualifications and employment. Skills Rootcamp Learners will be introduced to diagrams and how	Minimum 20% of the job training	Yes
Image: Specific direction unit, Unitative regression unit direction on matteress, specific traits regression unit direction on matteress, specific traits regression unit direction unit	Image: Specific time rouge and roug	schedules needed for the development of rail engineering activity planning: Detailed inspection and performance &	Partially covered	Entry level competency	W2 Diagram Reading	Genetion further in their apprecision)in Automatical Comparison (Comparison) (Comparison) NNR Senting based comparison, storigk with NNR Senting based comparison, storigk with third developing their qualifications and employment. Salits Bootcamp Learners will be introduced to disparant and how Learners will be introduced to disparant and how	Minimum 20% of the job training	Yes
Checker - Night Description Since - Section Since - Section <td>Cherges Problems See Subjects See Subjects</td> <td>schedules needed for the development of rail engineering activity planning. Detailed inspection and performance &</td> <td>Partially covered</td> <td>Entry level competency</td> <td>W2 Diagram Resting</td> <td>decelop Junter in their appretication). Autorizante Appreciation (Junter 2006). Mind Senton Laward Comparison, Junter 2006). Mind Senton Laward Comparison, Junter 2006 Linear Annual Mark (Linear 2006). Sality Rotocamp Lammers will be introduced to diagrams and how to morepret them in Signality. This will easily the Interpretation Signality. This will easily the Interpretation of Signality. This will easily the Interpretation of Signality. This will easily the Interpretation of Signality.</td> <td>Monnum 20% of the job training</td> <td>Y6</td>	Cherges Problems See Subjects	schedules needed for the development of rail engineering activity planning. Detailed inspection and performance &	Partially covered	Entry level competency	W2 Diagram Resting	decelop Junter in their appretication). Autorizante Appreciation (Junter 2006). Mind Senton Laward Comparison, Junter 2006). Mind Senton Laward Comparison, Junter 2006 Linear Annual Mark (Linear 2006). Sality Rotocamp Lammers will be introduced to diagrams and how to morepret them in Signality. This will easily the Interpretation Signality. This will easily the Interpretation of Signality. This will easily the Interpretation of Signality. This will easily the Interpretation of Signality.	Monnum 20% of the job training	Y6
Onlow F a blanded of technologies upperformed and properties intelling to mainteneeus. Norweil Norweile Norweile Norweile N	Onlow F a bit and with a depositive depositive approaches approaches and sequences. Net owned Net owned Res value with with depositive data on in approaches and sequences. Net owned Net owne	schedules needed for the development of rail engineering activity planning. Detailed inspection and performance &	Partially covered	Entry level competency	W2 Diagram Reading	decidy Infer in their appretication). Advancement Appreciation to to protect NMS Sentenet Appreciation to the protect NMS Sentenet based comparence, using with function devices protect and the sentence of the protect and the sentence of the sentence of the sentence of the sentence of the sentence of the to interpret them in Signaling. This will reader the sentence of the sentence of the sentence of the interpret of the sentence of the sentence of the sentence of the interpret of the sentence of the sentence of the sentence of the interpret of the sentence of the sentence of the sentence of the interpret of the sentence of the sentence of the sentence of the interpret of the sentence of the sentence of the sentence of the interpret of the sentence of t	Minimum 20% off the job training	Yes
Onlow F a blanded of technologies upperformed and properties intelling to mainteneeus. Norweil Norweile Norweile Norweile N	Onlow F a bit and with a depositive depositive approaches approaches and sequences. Net owned Net owned Res value with with depositive data on in approaches and sequences. Net owned Net owne	schedules needed for the development of rail engineering activity planning. Detailed inspection and performance &	Partially covered	Entry level competency	W2 Diagram Resting	develop latter in their appretication). Andersenant Appreciation (Samp Samphanian) Institution (Samphanian) (Samphanian) Institution (Samphanian)	Minimum 20% off the job training	Yes
Onlow F a blanded of technologies upperformed and properties intelling to mainteneeus. Norweil Norweile Norweile Norweile N	Onlow F a bit and with a depositive depositive approaches approaches and sequences. Net owned Net owned Res value with with depositive data on in approaches and sequences. Net owned Net owne	schedules needed for the development of rail engineering activity planning. Detailed inspection and performance &	Partially covered	Entry level competency	W2 Diagram Reading	develop latter in their appretication). Andersenant Appreciation (Samp Samphanian) Institution (Samphanian) (Samphanian) Institution (Samphanian)	Minimum 20% of the job training	Y8
contruction and or presents of a start. Complete high of complexes reduction of a start one werk, many presents in and and presents of a start. Complete high of a complexes of a start in and or presents of a start in and or presents of a start. Complete high of a complexes of a start in and or presents of a start in and or presents of a start. Complete high of a start one werk, many presents of a start in and or presents of a start in a start in and or presents of a start in a start in and or presents of a start in	contruction and registering of complexes of wares. Complexe registering of complexes of wares. Complexe registering of complexes of wares	schedules needed for the development of rail engineering activity planning. Detailed inspection and performance &	Partially covered	Entry level competency	W2 Diagram Reading	develop latter in their appretication). Andersenant Appreciation (Samp Samphanian) Institution (Samphanian) (Samphanian) Institution (Samphanian)		Yes
Integrate trading and denty, where subjected trading is required by a structured sported to the construct and base of observes. The subject and sported to the construct and base of observes. The subject and sported to the construct and base of observes. The subject and sported to the construct and sport at the construct	Integrate transfer and denoty where rependenticing in sequences in all forms. Through responsibility of states does work has a first of others. Solve problems: Store problems: Store y toolers and approximate methods to problem solving and deponds. Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to problem solving Manage responsibility of toolers and approximate methods to proble	Listendues medied for the development of rail engineering activity planning. Detailed impection and performance & condition analysis of assets, Plan work to be understaten including the appropriate resources.		Entry level competency	W2 Diagram Resting	decieg) partners in their appretication). Autorisment Appreciation) as position MMS Sentent based competings, and using MMS Sentent based competings, and using their hands and their approximation and an analysisment Salith Rototamp Learners will be introduced to diagrams and how in the strength their approximation and their ments and approximation and their approximation and learners will prostice their diagrams and basis them to sufficient their diagrams and strengt solity and the working the homotomy and will anold guilds them to sufficient their diagrams and strengt solity them to shall their role as a technician.		Yes Yes
Sole problems: locating the contrast shad again printer methods to problem source Print (sole defended again printer) Print (sole defended again print p	Sole problems: locently problems and approprise methods to problem solicity Parally covered Forry inel Sole problems: locently problems and approprise methods to problem solicity Parally covered Forry inel Sole problems: locently problems and approprise methods to problem solicity Parally covered Forry inel Sole problems: locently covered Sole problems: locen	Unterskies mediaf for the development of rail engineering activity planning. Detailed impection and performance & condition analysis of assets, Plan work to be understaten including the appropriate resources. Deliver a high standard of technical work. Understate engineering activities in relation to maintenance, construction / installation and or relevant of starts. Complete integrity a completence checks on own work.		Entry level competency	W2 Diagram Reading	decidy Inter: It der appreticien), Ackelorient Apprechending im portice NMS Sententen Apprechending im portice NMS Sententen Basis (and per samme listel through their qualifications and amplyingent). Skills Bootcamp Larener suit lieb entroligenite in Switz analytic larener suit lieb entroligenite in Switz analytic inder Switz and Switz and Switz and Switz Market and Switz and Switz and Switz Accessed and Switz and Switz and Switz Market and Switz and Switz and Switz Switz and Switz and Switz and Switz and Switz Switz and Switz and Switz and Switz and Switz and Switz Switz and Switz and Switz and Switz and Switz and Switz Switz and Switz and S		Yes
Sole position: Servicy positions and spays at nuclicred spaporch and sparsation methods to prostem solelity Pure view Name Accurany Literates size in the competency of ESD at holds to prostem solelity of ES	Sole potencie Sole potencie<	Undersken mediad for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plan work to be understaten including the appropriate resource.		Entry level compatency	W2 Diagram Reading	decidy Inter: It der appreticien), Ackelorient Apprechending im portice NMS Sententen Apprechending im portice NMS Sententen Basis (and per samme listel through their qualifications and amplyingent). Skills Bootcamp Larener suit lieb entroligenite in Switz analytic larener suit lieb entroligenite in Switz analytic inder Switz and Switz and Switz and Switz Market and Switz and Switz and Switz Accessed and Switz and Switz and Switz Market and Switz and Switz and Switz Switz and Switz and Switz and Switz and Switz Switz and Switz and Switz and Switz and Switz and Switz Switz and Switz and Switz and Switz and Switz and Switz Switz and Switz and S		Yes Yes
Sole position: Servicy positions and spays at nuclicred spaporch and sparsation methods to prostem solelity Pure view Name Accurany Literates size in the competency of ESD at holds to prostem solelity of ES	Sole potencie Sole potencie<	Undersken mediad for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plan work to be understaten including the appropriate resource.		Entry level competency	W2 Diagram Reading	decidy Inter: It der appreticien), Ackelorient Apprechending im portice NMS Sententen Apprechending im portice NMS Sententen Busie Competency, aller their developing met Aller in all der training inside through their qualifications and amplyment. Salits Bootcamp Lareners will be introllegation. This will analytic lareners will be introllegation. This will analytic interformation of the statistication of the Interformation of the statistication of the Inte		Yes
Ind dispose. In the spectra of the s	and suppose. In the concert utilisation and storage of tools, materials and equipment, and the lifting and storage of tools, materials and equipment, and the lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment, and the lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment, and the lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment, and the lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and and storage of tools, materials and equipment. The lifting and storage of tools, materials and equipment. The lifting and storage of tools and storage	Undersken mediad for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plan work to be understaten including the appropriate resource.		Entry level competency	W2 Diagram Reading	decidy Inter: It der appreticien), Ackelorient Apprechending im portice NMS Sententen Apprechending im portice NMS Sententen Busie Competency, aller their developing met Aller in all der training inside through their qualifications and amplyment. Salits Bootcamp Lareners will be introllegation. This will analytic lareners will be introllegation. This will analytic interformation of the statistication of the Interformation of the statistication of the Inte	Minimum 12 months duration	Yes Yes
Image records is under a support Image records is under a support is under	Manage resources including the correct utilisation and storage of tools, materials and equipment, and the information Reserve allow the developed encode in the superficiency of the source product in the source and the source product in the source product	Introduce needs for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plans work to be understaten including the spongostate resources.	Not covered	competency Entry level		decido Junter in their appretication). Addecidantes Appreciation). Mari Standards and Appreciation (Section 2014). Mari Standard Laward Constraints (Section 2014). Mari Standard Laward Constraints (Section 2014). Sillin Bootcamp Lamens will be introduced to diagrams and how to interpret them a Signature. This will easily in the Advectered Appreciation (Section 2014). Interpret Marine Signature (Section 2014). These Asilia will be developed once in employment.	Minimum 12 months duration	Yes Yes
Indextry	Name Description Descripion <thdescription< th=""> <thdes< td=""><td>Introduce needs for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plans work to be understaten including the spongostate resources.</td><td>Not covered</td><td>competency Entry level</td><td></td><td>decidy Inter is their appretication), decide protein Appretication is to sortical contentier Appreciation is to sortical NMS Sententier Appreciation is an advertised into the equilibrium and the equilibrium and the approximation of the equilibrium and the interpretion of the equilibrium and the equilibrium is an experiment. Sality Bostoneous Interpretion of the equilibrium and the interpretion of the equilibrium and the equilibrium and the equilibrium and the contexpect of the equilibrium and the solution between the equilibrium and the interpretion of the equilibrium and the solution between the equilibrium and the solution of the equilibrium and the equilibrium and the solution between the equilibrium and the equilibrium and solution between the equilibrium and the equilibrium and the solution and the equilibrium and the equilibrium and the solution a</td><td>Minimum 12 months duration</td><td>Yes</td></thdes<></thdescription<>	Introduce needs for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plans work to be understaten including the spongostate resources.	Not covered	competency Entry level		decidy Inter is their appretication), decide protein Appretication is to sortical contentier Appreciation is to sortical NMS Sententier Appreciation is an advertised into the equilibrium and the equilibrium and the approximation of the equilibrium and the interpretion of the equilibrium and the equilibrium is an experiment. Sality Bostoneous Interpretion of the equilibrium and the interpretion of the equilibrium and the equilibrium and the equilibrium and the contexpect of the equilibrium and the solution between the equilibrium and the interpretion of the equilibrium and the solution between the equilibrium and the solution of the equilibrium and the equilibrium and the solution between the equilibrium and the equilibrium and solution between the equilibrium and the equilibrium and the solution and the equilibrium and the equilibrium and the solution a	Minimum 12 months duration	Yes
Manage resources including the correct utilisation and storage of tool, materials and equipment, and the lifting Not correct and specification in the storage of tool in the storage of tool in the storage of the storage of tool in the s	Manage resources including the correct utilisation and storage of tools, materials and equipments, will be determined. Manage resources including will be determined will be determine	Introduce needs for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plans work to be understaten including the spongostate resources.	Not covered	competency Entry level		decidy Inter is their appretication. Advancement Appreciation and an advancement of the advancement of periodication and approximately and a second and a second advancement of the adv	Minimum 12 months duration	Yes
Manage resources industing the correct utilisation and torong and specificity and in the locality and the provide the specific and have the specific and h	Manage resources to dutilisation and storage of tools, materials and equipment, and the lifting and moining of materials, composition and storage of tools, materials and equipment, and the lifting and moining of materials, composition and storage of tools, materials and equipment, and the lifting and moining of materials, composition and the appendixed materials and equipment. Notes earlies will be developed once in the supervision of the supervision and the supervision of the supervision and the supervision of the sup	Introduce needs for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plans work to be understaten including the spongostate resources.	Not covered	competency Entry level		decidy Inter in their appretication. Addressing Appreciation (appreciation) as particular NMS Senderstand Appreciation (appreciation) as particular their developing test with an is other training inter through their applifications and analogic test applications and analogic test applications SNBs Bootcamp Licenses will be introduced to diagrams and how to interpret them is applications (applications) Licenses will be introduced to diagrams and how to interpret them is applications (applications) interstanding the introduced to diagrams and how to interpret them is applications (applications) Licenses will particulate the diagram reading with interstanding the interpret of the standing with them to full their role as a schedular. These skills will be developed once in employment.	Minimum 12 months duration	Yes Yes
Advange resources low-tigte per control Perse available with be developed once in horizontal components and requipament, and the time and moving of materials, components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament. Perse available with be developed once in horizontal components and requipament components and requipament component to horizontal bills doctame are to expenditional on the horizontal be developed on the horizontal be devel	Advange resources invising the courses strikturger and toolsy, materials and equipment, and the lifting and moving of materials, components and equipment. For an allow will be developed once in mightingwent. For an allow will be developed once in mightingwent. For an allow will be developed once in mightingwent. For an allow will be developed once in selected periods. For an allow will be developed once in selected periods. For an allow will be developed once in selected periods. For an allow will be developed once in selected periods. For an allow will be developed once in selected periods. For an allow will be developed once in selected periods. For an allow will be developed once in all the appendix the course of the allow once in the allow o	Introduce needs for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plans work to be understaten including the spongostate resources.	Not covered	competency Entry level		decido Inter in their appretication). Advancement Appreciation) as positive Mini Sentenera Appreciation and a sentenciation and a Mini Sentenera table and a sentenciation and a manipument. Salah Bootcamp Lancers and positive qualifications and a manipument. Salah Bootcamp Lancers and positive transmit ministrature. Advancement Appreciation This will be advantage to the Advancement Appreciation and and and a sentenciation and a sentenciation and a sentenciation of the Ministrature.	Minimum 12 months duration	Yes Yes
Mage resources industing the correct utilisation and storage of tools, materials and equipment, and the lifting and moving of materials, components and equipment.	In the expected amount of reduction on the supervised set of the expected amount of the expected am	Introduce needs for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plans work to be understaten including the spongostate resources.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as possible MMS Senderstand Appreciation (a) as possible MMS Sender basic convertince, single sith function developed metals in all other training analyzinnes. SNBs Bootcamp Learners will be introduced to diagrams and how phone to available that when working on the indication of the sendence of the sendence of the indications of the sendence of the sendence of the conversion of the sendence of the sendence of the indications of the sendence of the sendence of the sendence of the sendence of the sendence of the learners of the sendence of the learner of the sendence of the learners of the sendence of the learner of the sendence of the learners of the sendence of the learner of the sendence of the learners of the sendence of the learners of	Minimum 12 months duration	Yes
Mage resources industing the correct utilisation and storage of tools, materials and equipment, and the lifting and moving of materials, components and equipment.	In the expected amount of reduction on the supervised set of the expected amount of the expected am	Introduce needs for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plans work to be understaten including the spongostate resources.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as possible MMS Senderstand Appreciation (a) as possible MMS Sender basic convertince, single sith function developed metals in all other training analyzinnes. SNBs Bootcamp Learners will be introduced to diagrams and how phone to available that when working on the indication of the sendence of the sendence of the indications of the sendence of the sendence of the conversion of the sendence of the sendence of the indications of the sendence of the sendence of the sendence of the sendence of the sendence of the learners of the sendence of the learner of the sendence of the learners of the sendence of the learner of the sendence of the learners of the sendence of the learner of the sendence of the learners of the sendence of the learners of	Minimum 12 months duration	Yes
Velicited provides the apportunity for a featurent to progress in their ratio, becoming more employable from the SUB Bocch mean building them raises and expension structures the expension of the sub-structure and the sub-structure and the sub- statis Bocchamp and the appointed through the sub-structure structure and the sub-structure and the sub-structure and the sub- statis Bocchamp and the appointed structure and the sub-structure and the sub-structure and the sub-structure and the sub- structure and the appointed structure and the sub-structure and the sub-structure and the sub-structure and the sub- structure and the sub-structure and the sub-structure and the sub-structure and the sub-structure and the sub- structure and the sub-structure and the sub-structure and the sub-structure and the sub-structure and the sub- structure and the sub-structure and the sub- structure and the sub-structure and the sub-structure and the sub-structure and the sub-structure and the sub- structure and the sub-structure and the sub- structure and the sub-structure and the sub-	selected provides the exportantly for a facturent to progress in the inter the According more encourses from the S when the Mining there inter less the approprint generations and the approximation of the approximation o	Introduce needs for the development of rail engineering activity planning. Detailed impection and performance & control on analysis of assets, Plans work to be understaten including the spongostate resources.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as possible MMS Senderstand Appreciation (a) as possible MMS Sender basic convertince, single sith function developed metals in all other training analyzinnes. SNBs Bootcamp Learners will be introduced to diagrams and how phone to available that when working on the indication of the sendence of the sendence of the indications of the sendence of the sendence of the conversion of the sendence of the sendence of the indications of the sendence of the sendence of the sendence of the sendence of the sendence of the learners of the sendence of the learner of the sendence of the learners of the sendence of the learner of the sendence of the learners of the sendence of the learner of the sendence of the learners of the sendence of the learners of	Minimum 12 months duration Any requirements specific to the accupational standard Prease equilations your rationale for why you have chosen to cover	Yes Yes
Velicited provides the apportunity for a featurent to progress in their ratio, becoming more employable from the SUB Bocch mean building them raises and expension structures the expension of the sub-structure and the sub-structure and the sub- statis Bocchamp and the appointed through the sub-structure structure and the sub-structure and the sub-structure and the sub- statis Bocchamp and the appointed structure and the sub-structure and the sub-structure and the sub-structure and the sub- structure and the appointed structure and the sub-structure and the sub-structure and the sub-structure and the sub- structure and the sub-structure and the sub-structure and the sub-structure and the sub-structure and the sub- structure and the sub-structure and the sub-structure and the sub-structure and the sub-structure and the sub- structure and the sub-structure and the sub- structure and the sub-structure and the sub-structure and the sub-structure and the sub-structure and the sub- structure and the sub-structure and the sub- structure and the sub-structure and the sub-	selected provides the exportantly for a facturent to progress in the inter the According more encourses from the S when the Mining there inter less the approprint generations and the approximation of the approximation o	Introduce mediate for the development of rail engineering activity planning. Detailed impection and performance & control manipulated and exects, Plans work to be understated including the sponsoriate resources.	Not covered	competency Entry level		decidy Inter in their appreticition). Advancement Appreciation is to protect NMT Senderstand Appreciation is to protect NMT Senderstand Appreciation is an advancement inter through their application and methyle developing end is an advancement inter through their applications and methyle applications and a methyle applications and a methyle application and a sender applications and methyle applications and a sender applications and methyle applications and a sender applications and applications and a sender applications and a methyle application applications and a sender to interpret them in Signality. This will enable the the transmission applications applications and a Advancements and particular data applications and a methyle applications and applications and applications and applications applications and applications and applications and applications applications and applications protein single and advancements and applications induced applications applications applications applications applications and applications applications applications induced applications applications applications applications applications and applications applications applications applications applications applications applications applications applications applications applications applicati	Minimum 12 mosts duration Any requirements specific to the accupational standard Please explain your reliance for only you have chosen to over in the expected association of reduction on to	Yes Yes Yes The spacified ansate of KBs and have this overage receives resources for the fill of the the second processor of t
Salls Bostamp and the apprenticeship is aimed at KSB's, which makes the reduction of the apprenticeship justifiable	Suits Bostcamp and the apprenticeably is almed at KBS v, which makes the reduction of the apprenticeably	Indexide media for the devicement of rail engineering activity planning. Detailed impection and performance & condition analysis of assets, Para work to be understate including the spin-pointer resource.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as positive NMS Sententier Appreciation (b) as positive that the sentence of the sentence of the sentence of the manipument. Skills Bootcamp Lancence will be introduced in diagrams and has any sentence of the sentence of the sentence of the interactive of the sentence of the sentence of the sentence of the sentence of the sentence of the hashes sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the the sentence of the sentence	Minimum 12 months duration Any requirements specific to the accupational standard Please explain your rationale for why you have chosen to cover In the reported anount of reduction on to The stript of the table bab to you but the industry.	Yes Yes the specified amount of KSBs and how this coverage rese agencefording (100 word analymus).
		Indexide media for the devicement of rail engineering activity planning. Detailed impection and performance & condition analysis of assets, Para work to be understate including the spin-pointer resource.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as positive NMS Sententier Appreciation (b) as positive that the sentence of the sentence of the sentence of the manipument. Skills Bootcamp Lancence will be introduced in diagrams and has any sentence of the sentence of the sentence of the interactive of the sentence of the sentence of the sentence of the sentence of the sentence of the hashes sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the the sentence of the sentence	Minimum 12 months duration Any requirements specific to the accupational standard Pease explains your relinant for why you have chosen to cover in the expected amount of reduction on th The KBR covered in the Skills Bootsmap have been selected to the form the Skills Bootsmap have been selected.	Yes Yes Yes Yes the specified amount of KSRs and how this coverage result experience/big (158 word machinus). They are non-pilo (coverage to the fit coverage product to an arrow of the state of the st
Skills Bootcamp and the apprenticeship is armed at KSIPs, which makes the reduction of the apprenticeship	Skills Bootcamp and the apprenticeship is aimed at KSB's, which makes the reduction of the appr	Understanding methods for the development of rail engineering activity planning. Detailed inspection and performance & condition analysis of exercits. Plans work to be understate including the spanporter resources.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as positive NMS Sententier Appreciation (b) as positive that the sentence of the sentence of the sentence of the manipument. Skills Bootcamp Lancence will be introduced in diagrams and has any sentence of the sentence of the sentence of the interactive of the sentence of the sentence of the sentence of the sentence of the sentence of the hashes sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the the sentence of the sentence	Minimum 12 months duration Any requirements specific to the accupational standard Press explain your reliance for only you have ducates to open in the expected associat of creditions and The KIT is corred in the Site Boccamp have been indicated agreement in order for them to be also to work in the adulty to the Ministry been associated and agreement and and any different to the site of adulty to the site of a displayer for the site of adulty to the site of a displayer for the site of adulty to the site of adulty the site of adulty to the site of adulty the site of adulty the site of adulty the site of adulty to the site of adulty to the site of adulty the site of a	Yes Yes Yes Yes the specified amount of KSIh and how this coverage result argeneration by the coverage result argeneration of the coverage result of the coverage result for the specified amount of the forwards are been by the coverage result of the cov
anna sociality into the approximation by most the realistic of the approximation of the appro	arm Bookumy and une apprendecting in a single at Line 3, which makes the reduction of the spon	Understanding methods for the development of rail engineering activity planning. Detailed inspection and performance & condition analysis of exercits. Plans work to be understate including the spanporter resources.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as positive NMS Sententier Appreciation (b) as positive that the sentence of the sentence of the sentence of the manipument. Skills Bootcamp Lancence will be introduced in diagrams and has any sentence of the sentence of the sentence of the interactive of the sentence of the sentence of the sentence of the sentence of the sentence of the hashes sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the the sentence of the sentence	Minimum 12 months duration Any requirements specific to the accupational standard Press explain your reliance for only you have ducates to open in the expected associat of creditions and The KIT is corred in the Site Boccamp have been indicated agreement in order for them to be also to work in the adulty to the Ministry been associated and agreement and and any different to the site of adulty to the site of a displayer for the site of adulty to the site of a displayer for the site of adulty to the site of adulty the site of adulty to the site of adulty the site of adulty the site of adulty the site of adulty to the site of adulty to the site of adulty the site of a	Yes Yes Yes Yes the specified amount of KSIh and how this coverage result argeneration by the coverage result argeneration of the coverage result of the coverage result for the specified amount of the forwards are been by the coverage result of the cov
		Understanding methods for the development of rail engineering activity planning. Detailed inspection and performance & condition analysis of exercits. Plans work to be understate including the spanporter resources.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as positive NMS Sententier Appreciation (b) as positive that the sentence of the sentence of the sentence of the manipument. Skills Bootcamp Lancence will be introduced in diagrams and has any sentence of the sentence of the sentence of the interactive of the sentence of the sentence of the sentence of the sentence of the sentence of the hashes sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the the sentence of the sentence	Minimum 12 mosts duration Any requirements specific to the accupational standard Rease regular year influence for only you have shorted to only a specific cover of the Section Section and the Section and the Section and the Section and the Section Section and the Section and th	Yes
		Idendue readed for the development of rail engineering activity planning. Detailed impection and performance & condition analysis of assets, Plan work to be understate including the appropriate resource.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as positive NMS Sententier Appreciation (b) as positive that the sentence of the sentence of the sentence of the manipument. Skills Bootcamp Lancence will be introduced in diagrams and has any sentence of the sentence of the sentence of the interactive of the sentence of the sentence of the sentence of the sentence of the sentence of the hashes sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the the sentence of the sentence	Minimum 12 mosts duration Any requirements specific to the accupational standard Rease regular year influence for only you have shorted to only a specific cover of the Section Section and the Section and the Section and the Section and the Section Section and the Section and th	Yes
		Idendue readed for the development of rail engineering activity planning. Detailed impection and performance & condition analysis of assets, Plan work to be understate including the appropriate resource.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as positive NMS Sententier Appreciation (b) as positive that the sentence of the sentence of the sentence of the manipument. Skills Bootcamp Lancence will be introduced in diagrams and has any sentence of the sentence of the sentence of the interactive of the sentence of the sentence of the sentence of the sentence of the sentence of the hashes sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the the sentence of the sentence	Minimum 12 mosts duration Any requirements specific to the accupational standard Rease regular year influence for only you have shorted to only a specific cover of the Section Section and the Section and the Section and the Section and the Section Section and the Section and th	Yes
		Unbedder eredef for the development of rail engineering activity planning. Detailed impaction and performance & condition analysis of east, Piers work to be undertaken inducing the appropriate resource. Outliver a high standard of technical work. Undertake engineering activities in relation to maintenance, contraction, installation and or relevant of sectorical work. Undertake engineering activities in relation to maintenance, contraction, installation and or relevant of sectorical work. Undertake engineering activities in relation to maintenance, contraction, installation and or relevant of sectorical work. Even and the sectorical segments and approach and approach.	Not covered	competency Entry level		decision function in their appretication). Advancement Appreciation (a) as positive NMS Sententier Appreciation (b) as positive that the sentence of the sentence of the sentence of the manipument. Skills Bootcamp Lancence will be introduced in diagrams and has any sentence of the sentence of the sentence of the interactive of the sentence of the sentence of the sentence of the sentence of the sentence of the hashes sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the sentence of the the sentence of the sentence	Minimum 12 mosts duration Any requirements specific to the accupational standard Rease regular year influence for only you have shorted to only a specific cover of the Section Section and the Section and the Section and the Section and the Section Section and the Section and th	Yes

Communicate (Recruity, Use on), written, electronic and IT based methods and systems for the accurate communication, reporting & recording of technical and other information, using correct terms, standards, templates and certifications.	Partially covered	Entry level competency	WE Employability Skills, Employer Engagement	Suite sociame Learners will karn in thereives kills and communicate with employes. Accelerated Appendicus physics Learners will be able to develop the the communication of the social to the social physics and providing briefings on technical information.	Explain how you will support learners to ensure Prior Learning for Initial Assessment. This should also address where the Salit Roc mainter and a support of the support of the support and of to ensure that support of the support and the appendick needs. The prior learning that summe we completing their appendicular	tcamp provider is not the apprenticeship provider (100 word um). nt knowledge skills and behaviours against the apprenticeship d, and a tailored training plan is designed to suit the employer's licalic conclusions will allow apprentices
	Partially covered	Entry level competency	W1-W4 Basic Signalling 1 (B51), SPWEF, Dagram Reading, Electronic Principles ((P1)	Skills footcamp Lances will gain levers: Dayon making Incorrect models and the second second Incorrect models and a signaling problem scients of electratic and a signaling problem scients of advectratic and a signaling models applied generally more to head out the lange of the second to head out the specific for all filters of the lance of the full competency will be adde to further develop their skills.		
Behaviours						
Ac profectionally demonstrating dependebility, determination, howesty and integrity, Respect others, act ethically and contribute to sustainable development.	Partially covered	Entry level competency	W18WS Behaviours (FTBS / Struttonal & Incidents / Awareness & Change)	Suits Bordnerge, lacatenes will carry suit Behavioural training which includes free to be stafe culture. Statutional & includes I Avareness & Change, This training will be added value to employees as use more safety controls. Accelerates & Apprentications. Accelerates Apprentications. Accelerates Apprentications. Accelerates and the statute of the statute of the statute embedded throughout the industry.		
Se risk aware as as to reduce risks through potential monitoring and checking of information, concentration on the task, and awareness of changing circumstances on activity.	Partially covered	Full level competency	WLAWE BANJOONT (PTES / Structure) & Indontry / Avarence & Changel, Ruis & Hazards in the Workplace, PTS AC, OATCL, BSJ, SPWEF, EP1	Selie Society of the WWS Society of Losse Competences in PTAC and DELC, memory have enter regarder standing of safety in Industry Learners carry on Ethaborical transmit (industry competences) and the safety of the safety standing employees at increases a culture of tablety/safety commons of competences and the safety/safety commons of the safety of the safety/safety commons of the safety of the safety safety of the Accelerated Appendicable Managements and the safety of the safety safety employment and the safety of the safety safety employment and through their transmit employment and through tr		
Dipplay a self-disciplined, self-motivated, proactive approach to work, able to make independent decisions whild Nonview grant's limitations and when to ask for help or to encable.	Partially covered	Entry level competency	WLEWS Behaviours (PTIS / Stuational & Incidents / Ausrements & Chalippi	salis accessing of the second training which includes the to be slife culture, builden the second builden second second second that training will be added value to employers as it can be added to added the second second Appendixed Appendixed by the second second and the second second second and the second second second employment and through their training a this is embedded throughout the industry.		
Work reliably and safely, often without close supervision, to approved industry standards and safe working practices	Partially covered	Full level	WLEWS Rohoviours (FTR) (Structional & Insidents / Awareness & Changel, PTS AC	suite accessment of the second tracking which includes reverse to be site culture, the second second second second second second second tracking with a second second second second are not settly controls. Accelerated second se		

Work effectively and efficiently, includeually and as part of a team, maintaining effective relationships with colleagues, clients, suppliers and the public.	Partially covered	Entry level competency	WIAWS lehaviours (TIS / Studiou & Incidents / Awareness & Change)	Skills Bootcamp Learners will carry out Behavioural training which includes trace to be Sele Could to the Selection Doublewide & Bootcamp / Aventma & & Change Doublewide & Could Selection (Council Selection) Research as culture of Lafety and employees as Research as culture of Lafety and employees. Apprenticeship Learners will develop their behaviours through employment and through their training as this is embedded throughout the industry.	
Receptor to foedbady, willing to beam new skills and sajust to change. Monthlying, carrying out and recording CPD in necessary to maintain and enhance competence.			WLAWS Shenvioury (FTRS / Skutona I & Indonts / Avarences & Change). Employability Skils	Skills Bootcamp The Skills Bootcamp offers employability skills in CV writing interview skills and CPQ, which will be industry, luciment will carry out beinviscue that industry, luciment will carry out beinviscue Structures at large carry of skills of the skills into training will be added value to employers as the training will be added value to employers as the creates a culture of skills and employees who are more sitely conscious. Accelerated Appendicantly employment and the clashing as this is embedded throughout the industry.	
Prepared to make a personal commitment to their employer, the industry and its professional standards.	Partially covered		Incidents / Awareness & Change),	Still Sociaring Offens learness opportunity to meet with industry employers, providing a chance of securing employment/discourse opportunities of employment and pathways, along with employment and pathways, along with employment and pathways, along with an industry, learners carry out Behavioral Stafe catures, situational & incidents/ Awarness Stafe catures, situational & incidents/ Awarness Stafe catures, situational & incidents/ Awarness and Changes. The training will is added value to an employment and threads the incident employees who are more safety conscious. Accessed approaches the industry beams will develop their behaviours through employment and threads their range as its is embedded throughout the industry.	

			2: Pathway to Accelerated All sections (A, B and C) mu	Apprenticeships		
A) CONTENT MAPPING					B) ADDITIONAL CONTENT	
In this section place copy and pasts the KSN's for the algord accountional standard.	In this section you will need to select the level of coverage the Skills Bootcamp will provide against each KSB (not covered, partially covered or fully covered or fully covered).	In this section you will need to select the level of competency. This is whether the Skills Bootzamp will provide the learner with full or entry level occupational competence for the linked KSB.	Where in the Skills Bootcamp is this statement being covered? Indicate the module/unit titles and the wee of delivery.	Provide 3 Mort explanation on what type vocal open more inverse to the able to Do Or NOW by the end of the Skills Boccare to Provide the Inviets CRE streamment and New the will be Author developed in the apprenticeship to resch Aut competency.	As well as specific content that signs to an ecoparison at same elevant content with the set This may insule wrighten account of the set	med of value to employers.
п она зестем решае сору вна ране оне као а ток оне веднео оссорносни языковно.					The maj mouse in sparouna support, non-resumes and source	an concern char, papper of green a pointy specific on project measure
	Please select one option from the	Please select one option from the	You only need to provide evidence of delivery for those statements where you	You only need to provide expected outcomes information for those statements where you	Please list any additional content beyond the KSBs being	Please provide a rationale for the inclusion of this content and
Mapping to: A.R., INGREETING TICHNICAN LVEL3- https://www.instituteforapprenticeships.org/apprenticeship-standardu/nai ergineering-technicas-v1-1	drep down list.	drop down list.	have indicated coverage.	have indicated coverage.	delivered .	how it will compliment the KSBs.
KSB Statement	KSB Coverage	KSB Competency	50 words maximum per entry Evidence of Delivery	100 words maximum per entry Expected Outcomes	Additional Content	100 words maximum per entry Rationale
Conversion det and Professional working practices including legislation, regulation, industry procedures and safety requirements.	Partially covered	Full level competency	& Prep Day, PTS AC, W4 Health & Safety	With increase this Sub Boomset with a full WM Setting that and any elements of PTS Sub Boomset (1998) Setting the Setting Setting Setting Setting Setting Setting Setting Setting Setting Setting And Langerson (1998) and any elements in Compares (1998) and any Work, Working Reitleght America Taining, Langer Mark Setting Reitleght America Taining, Langer Mark Setting Setting Setting Setting Setting Accelerated Ageretischeh Accelerated Ageretischeh Acte Setting Heiting Mehr America Setting Accelerated Ageretischeh Anter America Setting Mehr America Setting Anter America Setting Mehr America Setting Anter America Setting America Setting Mehr America Setting America Setting Mehr America Setting America Setting Am	Soci: Signaling & O.Z. Awanness	The sector will provide an introduction to software provide incompany who must be alwarent to software to the incompany who must be advected and and the the nativey operates, the dangers of other demonst, and will complianent the divergence of various KBY.
The advantify, too history, expressing, mathematical and odego principles that are required to support the mentmance, interval and controlston of The Relative;	Partially covered	Entry level competency	W1-W5 Implementing Base Plate Design	Reactamp The lossicamp will provide an introduction to base parter designs in which their approximation. In strandender and their approximation is Accelerated Apprenticently Larrent ordering their involvedge on base parter competencies, slowy with industry segeritors until regards to scalefile, technical, equipeeing, mathematical and design principles.	Hand Trolley	The compensation will evaluate a tensor to understand the way search of all one priority that and the priority of the selfsy tisses that can also and the operation of a doos and appendix. This comparison that RSVs required as part of the apprenticeship standard.
Now is work effectively and combuct to regimening solutions and smoothin including understanding and paying problem solving technologies on diagonatic, project paramig and management, then management and painting menagement and source on some	Not covered			These skills will be developed once in employment.	Emergency Response	Emergency response is a vital element in understanding what action to take in the event of injuny, damage or ioss of the All of which from part of the safety culture within the industry and provides the learner the opportunity to develop their behaviours.
The reportance of bid party requirements and client confidentiality and the read to understand and adhere to corporate policies on effect, equality and diversity.	Partially covered	Entry level competency	WE Commercial Operations & Corporate Policies	Naite Beckenne The Salitis Bockenne will provide an introduction to commercial operations and comprehe policies when it comes to completing their knowledge when it comes to completing their knowledge when it comes to completing their knowledge schematic states and the source of the completing engineering and apply this to their learning and assessments in the knowledge qualification.	Enronment	The section will camplianted the bioxologie regularity data in the section of the section of the section of the section of the and plot have on the environment. This will also provide an introduction to an element of their incovidege qualification.
How the railway works commercially including contractual principles and implications.	Partially covered	Entry level competency	W4 Commercial Operations & Corporate Policies	Bostang the bencary will provide an introduction to commencial operations and comparts policies, when it comes to competent pelite knowledge when it comes to competent pelite knowledge autification as part of the apprenticehip. Accelerated Apprenticehip knowledge gained and apply this to their harming and assessments in the knowledge gualification.		
					C) ACCELERATED APPRENTICESHIP QUESTIONS Provide the typical duration and price of the app	mationhis at as MATE's accustional man-
					 Typical duration to gateway (full apprenticeship)	30 months
					Typical price of the apprenticeship (full apprenticeship)	E24,000.00
					State the impact Prior Learning from the Skills Bootcam	a will have an the second cable duration and price
						27 months
					 Estimated duration to gateway (accelerated apprenticeship)	£21,600.00
Skills					Estimated price of the apprenticeship (acclerated apprenticeship) Can you confirm that the Skills Bootcamp will provide a route	to an accelerated apprenticeship, whilst ensuring the below
regulations and organizational safety requirements, licituding safe access to work locations.			B. Prop. Dip., 173 AC, W4 Health A. Sufery Ligations and Reparations, Nias & Heardson in the Waterstein, Reporting & Recording Working et Height and Manness Training. Behaviours, Safety	SNIIs Booksamp (The Skiis) Socrame will provide the learner with a full WM Sentin basid competence for FIDS 4. Suit-basid sends there to suit out the rela- shit of the sentimetry of the sentimetry of the full competence in Onegreen (Find & Work, Working in Height A full meets Taining, Lanner and a loss provided with the receivary alls in a construct Agreembergh Accelerated Agreembergh MM Beneficial Sand competency, sing with MM Beneficial Sand competency, sing with the dimensional sector and sectors and employment.	requirements of the ap	ennoceng ar mer
Pan - high studead of stochical work (achtering and integrating information including particular); same, lackbalais needed for development of all angieven grating particular and angestion and performance. B candition analosis of assets, Pan work to be undertaken including the appropriate resources.		Entry level competency	W3 Work Reclage Plans, Task Briefs, CPP, Uit Planner Awareness	Learners will learn new skills in elements of planning a job in regards to completing work package plans, task briefs and CPPs. Accelerated Agereafteshilp Learners will be able to build on the bootcamp skills gained by applying this in their role as a technician, as these are used for every job.	Minimum 12 months duration	Yes
belier a high standard of schroll and with Understate engineering activities in relation to maintenance, communicals/initiating/initiation due to research of standard Comprete Integrity Comprete Integrit				These skills will be developed once in employment.	Any requirements specific to the occupational standard	Yes
Solve problem: dentify problem and apply a thruchwel approach and appropriate methods to problem solving and diagnosis.	Partially covered	Full level competency	WD Track induction County (Tot, WI TST) Interpropring of Hermate and its Weinpack (Wi County in the Weinpack Weinpack)	Suits Bookcamp Lanamus still gan Ab Notits General Insel Lanamus still gan Ab Notits General Insel Insel Table State State State State State Market State State State State State State Landthird Research (problems on the Landthird Research (problems) on the Landthird Research (problems) on the Accelerator Agerenticateship Table State State State State State President Research (problems) on the Insel Institution (problems) on the Insel	Peeze estale your referate for very you have chosen to cove	The specified amount of K06, and how this coverage results
Manase recourse including the correct utilization and choses of tools: materials and environment and the lifting	Partially covered	Full laval	W2 Track Induction Course (TIC)	Skills Rootramo	in the expected amount of reduction on the The KSR's muered in the Rontramo have been celected as they are	e apprenticeship (100 word maximum). e naturally covered by the cources provided to an apprentice in

					Hease registing your attitude for why you have chosen to crear: the specified assumed of 2016 and have this energy results in the expected assumed of induction on the approximately (2019 word maximum).
Nange resource is shafting the control children and storage of tools, materials and explanent, and the Hing and moving an indexist, components and explanent.	Partially covered	Full level competency	W2 Track induction Course (TIC)	Alth Bocksong Lawrenz will gen full XVM Servinel kased competency for first induction Gouvae (IC) enabling them to early with material and components on the inductories. And the service of the service of the service of the law of the service of the service of the service of the law of the service of the service of the service of the law of the service of the service of the service of the service on service of the service of the service of the law of the service of the service of the service of the service on service of the se	The GPU several in the Boatspan has been instead at intery or naturally covered by the covera provided to an approximation in order for them the last severals in the index particularly that been provided in the Boatspan and Bo
Communicate effectively ton one writes, electronic and IT based methods and systems for the accurate communication, reporting & recording of bachward and active information, using connect terms, transfords, templates and conflications.	Partially covered	Full level competency	913-99 97 24 42, 7 343 kolectatio docume (Tric), Taia Merki, Angelopality Sells, Engloye Eggenerest	Alth Assessment the Stath Monitorian will provide the learner with a Kell MMN strategies and provide the learner with a Monitorian strategies with the and Monitorian. Where the shall be more than the Monitorian strategies with the strategies and the Kell monitorian strategies and the strategies and the strategies and the strategies and the strategies with and commutation will be not the strategies with and commutation with and physics.	nalar bar ya yil ugant baran ta man Pari jamin fan he bili batam ia manind i har jamintaka
lywolfe troovinge \$ 500					Replace have yes will support learner to ensure Protor Learning from the Suffix Instances in suspension of the apportation of the apportance of the apportan
Net. Sectorsanding of total generative procluding total gauge, alignment, devoting, constants and total surface and the impact of hour when the programment, methods and tabelogues for the matilation and nationates of the total surface and the impact of the programment, methods and tabelogues for the matilation and nationates of the total surface and the impact of the programment of the surface of the match of the programment of the surface of the match of the surface of the surface of the surface of the surface of the surface of the	Partially covered	Full level competency	NR ARE The Mandeline Convert (TCL Stud- temport, TCL Studency and Permaneter Lines in subdividue, registraneting law and stud- cessing studied in the studency of the studency Tradews, TKBL Stores Restoration	Nihi Boosenan This Shih Boosenan Han Shih Boosenan will provide the latent with a full AVM Enterthal based competencies for PTSAC, and the AVM Enterthal Boosenan and the PTSAC Provide the AVM And AVM AND AVM AND AVM AND AVM Restoration with metalate them is used as the full and Restoration and provide the Internet Program and Avm Restoration and Provide the Internet Program and Program and Provide the Internet Provid	
Relations					
An graphonematic discovering dependential discovering data and the second s	Partially covered	Entry level competency	WS Behavious (PTIS / Sharihout & Incoletts / Awareness & Chargo)	Shik Bootsamp Lanners etti array on di Behasiouri tuoining white incluster nere bis Safé culture, Boltzminal & Riccoler, Marrenes B. Dange. This barrane array and the American Safe Safe Safe series a subties of safety and employees who are more safety constant. <i>Clearised Appendicable</i> Lanners will directly the behaviour the bits ambedded throughout the industry.	
Cold searce as as to solve risk through symmetry monthing and the bing of information, consectation on the task, and meanwests of changing channelsness on withing.	Partially covered	Full level completincy	W4 49 Bohoson (JTES / Stoatone & Indones / Awareness & Courge), Roka & Rauenho in the Workplane, PTS AC, OLCI	Shik Bostcamy Process Insures of MA 2005 Section based importances on PTMA Can of OSEE, some right of moders is assessed as a section of the section of the Moders is assessed as a section of the section of the Moders is assessed as a section of the section of the Moders and assessed as a more a setting and and the amplytoms as it creates a cliner of the PTMA CODECs will be profile and profile and of the Moders and assessed as a section of the sec- Moders and assessed as a section of the sec- Moders and Assessed as a section of the sec- ments and the section of the section of the sec- ments and the section of the section of the sec- ments and the inductory. Section of the sec- ments and the inductory is and section of the section of the section of the sec- ments and the inductory.	
Dayles and disclobed, with methods, presenter approach to avera, allo to make independent decision while beaving and institution and when to add for fiving or to academic.	Partially covered	Entry level competency	W4-W2 Behaviours (*785 / Stratfond & Incident / Awareness & Chango)	Sink because inclusions of sum of a bibacistured training when inclusions from the bib of a chance. Structured is the bible chances, and increase is a bible of a chance of the constant is a bible of a bible of a set of a bible constant is a bible of a bible of a bible memory will be and a bible of a bible constant is a bible of a bible of a bible memory will be and a bible of a bible memory will be an a bible of a bible memory will be a bible of	

Work reliably and safely, often without close supervision, to approved industry standards and safe working practices	Partially covered	Full level competency	W1-W5 Behaviours (FTBS / Situational & Incidents / Awareness & Change), PTS AC,	Skills Bootcamp Learners will carry out Behavioural training which		
		compiltency	Industris / Awweness & Chango, PISAC, Track Induction Course (TIC)	Lamon will carry of Behaviout lamon who behaviour as the star of the star of the star behaviour as the star of the star of the star training will be added value to employeen as it restricts a culture of the part of the star more star of the star of the star of the star more star of the star of the star of the star employees and the star of the star of the star employees and the star of the star of the star employees and the star of the star of the star employees and the star of the star of the star employees and the star of the star of the star employees and the star of the star of the star employees and the star of the star of the star of the star employees and the star of the star of the star of the star employees and the star of the star of the star of the star employees and the star of the star of the star of the star employees and the star of the star of the star of the star of the star employees and the star of the star of the star of the star of the star employees and the star of		
Nork effectively and efficiently, included usily and an part of a stame, maintaining effective relationships with calculated, clience, Replanes and the public.	Partially covered	Full level compilency	W1-W5 Shihanicom (1785 / Shustionil & Incidents / Awareness & Champo), Tack Induction Course (TIC)	Such Brokenne Mark Broken wir son Britanisaut straning windle Includen frem to the Safe Calhurs, Bruntmund & Enclosed, Navenense & Callange, This Bruntmund & Enclosed, Navenense & Callange, This Bruntmund & Enclosed, Navenense & Callange, This Bruntmund & Safe Callange, Navenense & Callange, This Bruntmund & Safe Callange, Navenense & Callange, This Bruntmund & Safe Callange, Navenense & Callange, This Internet and Brundle Bruntmund and Brundle Brundle and Brundle Brundle Brundle Brundle Brundle Brundle Market Brundle Brun		
Nanghes to bedieck, will by to loar new sith and algorit to charge. Herefying, carrying out and recording CP secondary to markelin and onlinear computation.	Partially covered	Entry level competency	We MS Alwadowa (J THS / Standarda & Indeatin / Anamara & Cango), Employedny (ski)	Such account that is a bound on the second		
Programed to make a generated convolutioned to their simplicity, the industry and its productioned databases.	Partially covered	competency	MAAND BANKINGON (TITES / Shaudhood & Indianter, / Anaemanika Engloyebility Skills, Engloyer Engagement	shift horsesti (file) the supporting a bisness of source) implicing, providing a bisness of source) implicing and a particular sources and a source of the source of sources and in the order, management of the source of the sources of the sources of the source of the implicing and a source of the source of the source of company. This toring will be ideal sources the implicing and in contents a solution of undersources bisness of the sources of the sources of the sources of the sources of the sources of the sources bisness of the sources of the sources of the sources of the sources of the sources of the bisness of the sources of the sources of the bisness of the source		



DfE Skills Bootcamps Dynamic Purchasing System

Document 4b. Contract Order Form Engineering and Manufacturing ITT_2394 Engineering & Manufacturing Category

DPA Schedule 4 (Contract Order Form Template)

Order Form

THE CUSTOMER: The Department for Education

CUSTOMER ADDRESS: Sanctuary Buildings Great Smith Street Westminster London, SW1P 3BT

 THE SUPPLIER:
 Intertrain UK Limited

 SUPPLIER ADDRESS:
 Balby Court Business Campus, Balby Carr Bank, Balby, Doncaster, DN4 8DE

 REGISTRATION NUMBER:
 4696164

 DUNS NUMBER:
 734440899

DPA SUPPLIER REGISTRATION SERVICE ID:

APPLICABLE DPA

This Order Form is for the provision of the Service Requirements and dated 29th June 2023 It is issued under the DPA with the Jaggaer Category Assessment reference number **Category** for the provision of Skills Bootcamps.

DPA SERVICE CATEGORY(IES):

As set out in Annex 1

DPA SUB-CATERGORY(IES)

As set out in Annex 1

GEOGRAPHICAL FILTER(S)

As set out in Annex 1

PATHWAY TO ACCELERATED APPRENTICESHIP Yes

GREEN SKILLS

Yes

ORDER INCORPORATED TERMS

The following documents are incorporated into this Call Off Contract. If the documents conflict, the following order of precedence applies:

- 1. This Order Form [including the Order Special Terms and Order Special Schedules].
- 2. The Call-Off Contract conditions and schedules
- 3. This will incorporate all of the DPS Special Terms into the Order Contract.
- 4. Order Schedule 4 (Order Tender) as long as any parts of the Order Tender that offer a better commercial position for the Customer (as decided by the Customer) take precedence over the documents above.

No other Supplier terms are part of the Call Off Contract. That includes any terms written on the back of, added to this Order Form, or presented at the time of delivery.

ORDER SPECIAL TERMS

CALL OFF CONTRACT COMMENCEMENT DATE:01/02/2024

CALL OFF CONTRACT EXPIRY DATE: 31/01/2026

CALL OFF CONTRACT INITIAL PERIOD: 2 Years

CALL OFF CONTRACT EXTENSION PERIOD(S): The contract contains extension provisions which enable the Authority to exercise discretion and extend for up to 24 months on a number of occasions up to a maximum total term (including the initial term) to align with the DPS lifespan of 6 years. Any extension will be subject to funding approval and government priorities for skills.

SERVICE REQUIREMENTS:

As set out in DPA and Annex 1 below.

SUPPLEMENTARY SERVICE REQUIREMENTS (See Schedule 2 Part A Section D) Yes

– As set out in Annex 2

DELIVERY PLAN TRAJECTORY POINTS: See Schedule 2 Parts 2 (performance Management) and 3 (Contract Management and Management Information).

MAXIMUM LIABILITY

The limitation of liability for this Order Contract is stated in Clause 25.3 of the Call Off Contract. The Contract Charges used to calculate liability under Clause 25.3 of the Call Off Contract

CONTRACT CHARGES

See details in Call Off Contract Schedule 3 (Contract Charges, Payment, and Invoicing)

PAYMENT METHOD

Through the ILR and as set out in Clause 14 and Paragraphs 5.2 – 5.4 of Schedule 3 of the Call Off Contract.

CUSTOMER'S AUTHORISED REPRESENTATIVE:

Deputy Director Skills & Strategy Commercial Directorate

Sanctuary Buildings Great Smith Street Westminster London, SW1P 3BT

CUSTOMER'S SECURITY POLICY

As set out at Schedule 4 (Security) of the Call Off Contract

SUPPLIER'S AUTHORISED REPRESENTATIVE

Managing Director Balby Court Business Campus, Balby Carr Bank, Doncaster, DN4 8DE

PROGRESS REPORT FREQUENCY

As set out in Part 3 (Contract Management and Management Information) of Schedule 2 (Service Requirements) in the Call Off Contract.

PROGRESS MEETING FREQUENCY

As set out in Part 3 (Contract Management and Management Information) of Schedule 2 (Service Requirements) in the Call Off Contract.

KEY SUBCONTRACTOR(S)

As set out in Schedule 14 (Sub-Contracting) of the Call Off Contract

COMMERCIALLY SENSITIVE INFORMATION

Not applicable

GUARANTEE

Not applicable

SOCIAL VALUE COMMITMENT

The Supplier agrees, in providing the Service Requirements and performing its obligations under the Call Off Contract, that it will comply with the social value commitments in Schedule 13 (Contract Tender) of the Call-Off Contract.

For and on beh	nalf of the Supplier:	For and on beh	alf of the Customer
Signature:		Signature:	
Name:		Name:	
Role:		Role:	
Date:		Date:	

Annex 1: Service Requirements

Para SR	a(s) in	Requirement							
PART A									

1.	1.2, 9.1.1, (Part B, para 1.1)	The Level of which the SBC will be an equivalent as follows	Level 3 to 5 or equivalent for subcategories: Engineering & Manufacturing (general) Pathway to Accelerated Apprenticeships in Engineering & Manufacturing Level 2-5 for sub-categories: Pathway to Accelerated Apprenticeships in Engineering & Manufacturing (if also Green Skills) Green Skills in Engineering & Manufacturing
2.	2.3	The requirements for a relevant Competition as set out in this Contract Order Form	N/A
3.	8, 9, 10, 11, 12, 13,	The Category of the SBC	The Category is Engineering and Manufacturing
4.	(Part B, paras 3, 4,	The Sub-category (level 1) of the SBC	The Sub-categories (level 1) are:
5.	. 5, 6), (Part C, para 2.2)	The Sub-category (level 2) of the SBC	Engineering and Manufacturing (general) Pathway to Accelerated
6.		The Geographical Filter of the SBC	Apprenticeships in Engineering and Manufacturing Green Skills in Engineering and Manufacturing The Sub-categories (level 2) for
			Engineering and Manufacturing (general) and Green Skills in Engineering and Manufacturing are: Engineering, design & development

			Engineering, manufacturing, process & control Maintenance, installation & repair Emerging The Sub-categories (level 2) for Pathway to Accelerated Apprenticeships in Engineering and Manufacturing are: Engineering, design & development Engineering, manufacturing, process & control Maintenance, installation & repair All competitions are National.
7.	8.3	The minimum percentage of SME Employers that a Supplier must engage as part of its employer engagement	N/A
8.	8.5	For Employers using Skills Bootcamps to train their own employees only: - The amount of Employer Financial Contribution which a Supplier must secure the Employer's commitment to contribute	30% generally; 10% for small and medium enterprises (SMEs)
9.	9.1.3, 12.1.4	The duration (in weeks) of the SBC	Maximum 16 weeks.
10.	9.1.4(c)	The minimum number of Guided Learning Hours that the SBC will comprise	60 GLH
11.	11.1, 11.1.6(c)	The period of time after a Learner finishes the training of a Skills Bootcamp during which the Supplier must track whether the Learner has achieved a positive outcome (See para 11.1.6. of the	Suppliers will be expected to achieve positive outcomes with learners within six months of the learner completing the training of the Skills Bootcamps unless the Customer states otherwise. Suppliers should track learners within this 6-month period until the learner achieves a positive outcome.
		Service Requirements)	

12.	14	The milestone payments to be paid to a Supplier on achievement of Outcomes.	Commencement: 40% Completion: 30% Positive outcome: 30%
12.1		The dates at which payment milestones must be achieved, in line with delivery plans	20% of starts by 31 st March 2024 Completions by 31 st July 2025 Outcomes by 31 st January 2026
13.	14.1.1(d)	In relation to an Independent Positive Outcome, the minimum number of <i>hours per week</i> which a job (that a Learner secures following an SBC) comprises	Not applicable.
14.		In relation to an Independent Positive Outcome, the minimum gross annual salary which a job (that a Learner secures following an SBC) pays	
15.	14.2.1(d)	In relation to a Co-Funded Positive Outcome , the minimum number of <i>hours</i> <i>per week</i> which a job (that a Learner secures following an SBC) comprises	
16.		In relation to a Co-Funded Positive Outcome , the minimum <i>gross annual</i> <i>salary</i> which a <u>job</u> (that a Learner secures following an SBC) pays	
17.	14.2.2(d)	In relation to a Co-Funded Positive Outcome , the minimum number of <i>hours</i> <i>per week</i> which <u>new</u> <u>responsibilities</u> (that a Learner secures with their Employer following an SBC) comprises	
18.		In relation to a Co-Funded Positive Outcome , the minimum gross annual salary which <u>new</u> responsibilities (that a Learner secures with their Employer following an SBC) pays	

19.	14.2.3	In relation to a Co-Funded Positive Outcome , the minimum number of <i>hours</i> <i>per week</i> which a job (that a Learner secures following an SBC) comprises				
20.		In relation to a Co-Funded Positive Outcome , the minimum <i>gross annual</i> <i>salary</i> which a job (that a Learner secures following an SBC) pays				
		PART B				
21.	2.1	The additional requirements that shall apply to each Contract for a Skills Bootcamps for which Pathway to Accelerated Apprenticeship is specified	No additional requirement – requirements in DPA still apply			
22.	3.1, 3.2, 3.3	The equivalent Level at which the Supplier will provide apprenticeship training	Level 3 to 5 or equivalent where Engineering and Manufacturing (general) and Pathway to Accelerated Apprenticeships in Engineering and Manufacturing. Level 2-5 or equivalent where Green Skills in Engineering and Manufacturing.			
23.	7	The milestone payments that the Supplier will receive when the Learner achieves Outcomes	Commencement: 40% Completion: 30% Positive outcome: 30%			
23.1		The dates at which payment milestones must be achieved, in line with delivery plans	20% of starts by 31 st March 2024 Completions by 31 st July 2025 Outcomes by 31 st January 2026			
PART C						
24.	2.1	The additional requirements that shall apply to each Contract for a SBC for which Green Skills is specified	No additional requirement – requirements in DPA still apply			
PART D						
25.	1	The additional requirements that shall apply to each Contract for a Skills Bootcamp for which	N/A			

		Supplementary Requirements are specified	
26.	2	Any supplementary service requirements	Where Engineering Skills Bootcamp delivery requires you to be registered with an accredited body e.g., RTAS (Rail Training Assurance Scheme), provide your registration number and where applicable, grade.

1. As part of the information advice and guidance process ("IAG"), with every potential Learner, suppliers must provide the learner with information about:

- (a) the level of the Skills Bootcamp
- (b) length of the course;
- (c) course outline, including modules taught;
- (d) expected time commitment (including for Self-Guided Learning);
- (e) the offer of an interview on completion of the course and;

(f) potential employment outcomes, career progression routes, potential geographical location of the roles and salary expectations upon receiving a positive outcome completion of the Bootcamp;

(g) any other information useful for the learner to make an informed decision about enrolment

and check that the potential Learner understands the possible employment outcomes and intends to use the Skills Bootcamp to achieve an employment outcome which is related to the Relevant Skills

- 1. Suppliers should provide learners with wraparound support set out in the service requirements. In addition to this, suppliers shall provide throughout the duration of the Skills Bootcamp:
 - (i) personalised wraparound career and personal development support to the Learner, taking into consideration the Learner's employment history and existing skills
 - (ii) Sector-specific Behavioural Skills Support (as defined in the service agreements) for employment
 - (iii) Assistance with the drafting of industry standard, sector-specific CVs and preparation for interviews personalised to the individual learner need

2.1 Such support may include but is but not limited to:

- (a) giving mock interviews from industry professionals or Employers, with experience in the sector
- (b) Tailored career coaching from industry professionals with experience in the Skills Bootcamps sector)

2.2 As part of the support provided to a Learner that is either unsuccessful at Job Interview or does not attend an interview, as defined in the DPS service requirements, the supplier shall:

Based on review, signpost appropriate next steps for the Learner as set out in the DPS service requirements and to include further career and personal development support to pass

See Schedule 2 Parts 2 (performance Management) and 3 (Contract Management and Management Information).

Document 3. Skills Bootcamps Call Off Agreement